

FRANCHISE DISCLOSURE DOCUMENT



HFB FenceCo Franchising, LLC

a North Carolina limited liability company 107 Parr Drive

Huntersville, North Carolina 28078

Phone: (980) 441-1121

Email: development@toprailfence.com Website: www.toprailfence.com

We offer qualified individuals and entities the right to operate a business which sells, furnishes and installs wood, steel, aluminum, vinyl and other types of fencing and related garden products for residential and commercial customers under our then-current proprietary mark, which is currently "TOP RAIL (the "Franchised Business(es)").

The total investment necessary to begin operation of a Franchised Business is between \$103,447,112,944 and \$219,197,213,444. This includes between \$6064,900 and \$747,8,900 that must be paid to franchisor or its affiliate(s).

We also offer qualified parties the right to enter into multiple franchise agreements simultaneously. A multi-unit addendum must be signed to operate multiple Franchised Businesses. The total investment necessary to operate two (2) to 5 Franchised Businesses is between \$146,447.152.944 and \$276,197404.444. This includes between \$97104,900 and \$125269,900 that must be paid to the franchisor or its affiliate(s). The total investment necessary to operate three (3) Franchised Businesses is between \$184,447 and \$328,197. This includes between \$135,900 and \$177,900 that must be The amount paid to franchisorus or its affiliate(s). The total investment necessary to operate four (4) Franchised Businesses is between \$222,447 and \$380,197. This includes between \$173,900 and \$229,900 that must be paid to franchisor or its affiliate(s). The total investment necessary to operate five (5) Franchised Businesses is between \$250,447 and \$422,197. This includes between \$201,900 and \$271,900 that must be paid to franchisor or its affiliate(s).our affiliates includes the Initial Franchise Fee (including additional qualified households) and the Opening Assistance Fee.

This disclosure document summarizes certain provisions of your franchise agreement and other information in plain English. Read this disclosure document and all accompanying agreements carefully. You must receive this disclosure document at least 14 calendar days before you sign a binding agreement with, or make any payment to, us or an affiliate of ours in connection with the proposed franchise sale. We must also identify your territory and any other material terms not set forth in this franchisee disclosure document at least 7 calendar days before you sign a binding agreement with, or make any payment to, us or an affiliate of ours in connection with the proposed franchise sale. **Note, however, that no governmental agency has verified the information contained in this document.**

You may wish to receive your disclosure document in another format that is more convenient to you. To discuss the availability of disclosures in different formats, contact HFB FenceCo Franchising, LLC at 107 Parr Drive, Huntersville, North Carolina 28078 or call (980) 441-1121.

The terms of your contract will govern your franchise relationship. Don't rely on this disclosure document alone to understand your contract. Read your entire contract carefully. Show your contract and this disclosure document to an advisor, like a lawyer or an accountant.

Buying a franchise is a complex investment. The information in this disclosure document can help you make up your mind. More information on franchising, such as "A Consumer's Guide to Buying a Franchise," which can help you understand how to use this disclosure document is available from the Federal Trade Commission. You can contact the FTC at 1-877-FTC-HELP or by writing to the FTC at 600 Pennsylvania Avenue, NW, Washington, DC 20580. You can also visit the FTC's home page at www.ftc.gov for additional information. Call your state agency or visit your public library for other sources of information on franchising.

There may also be laws on franchising in your state. Ask your state agencies about them.

Issuance Date: April 29, 2023 <u>19, 2024</u>

How to Use This Franchise Disclosure Document

Here are some questions you may be asking about buying a franchise and tips on how to find more information:

QUESTION	WHERE TO FIND INFORMATION		
How much can I earn?	Item 19 may give you information about sales, costs, profits, or losses. You should also try to obtain this information from others, like current and former franchisees. You can find their names and contact information in Item 20 or Exhibit HK.		
How much will I need to invest?	Items 5 and 6 list fees you will be paying to the franchisor or at the franchisor's direction. Item 7 lists the initial investment to open. Item 8 describes the suppliers you must use.		
Does the franchisor have the financial ability to provide support to my business?	Item 21 or Exhibit El includes financial statements. Review these statements carefully.		
Is the franchise system stable, growing, or shrinking?	Item 20 summarizes the recent history of the number of company-owned and franchised outlets.		
Will my business be the only Top Rail in my area?	Item 12 and the "territory" provisions in the franchise agreement describe whether the franchisor and other franchisees can compete with you.		
Does the franchisor have a troubled legal history?	Items 3 and 4 tell you whether the franchisor or its management have been involved in material litigation or bankruptcy proceedings.		
What's it like to be a Top Rail franchisee?	Item 20 or Exhibit HK lists current and former franchisees. You can contact them to ask about their experiences.		
What else should I know?	These questions are only a few things you should look for. Review all 23 Items and all Exhibits in his disclosure document to better understand this franchise opportunity. See the table of contents.		

What You Need To Know About Franchising Generally

<u>Continuing responsibility to pay fees</u>. You may have to pay royalties and other fees even if you are losing money.

<u>Business model can change</u>. The franchise agreement may allow the franchisor to change its manuals and business model without your consent. These changes may require you to make additional investments in your franchise business or may harm your franchise business.

<u>Supplier restrictions</u>. You may have to buy or lease items from the franchisor or a limited group of suppliers the franchisor designates. These items may be more expensive than similar items you could buy on your own.

<u>Operating restrictions</u>. The franchise agreement may prohibit you from operating a similar business during the term of the franchise. There are usually other restrictions. Some examples may include controlling your location, your access to customers, what you sell, how you market, and your hours of operation.

<u>Competition from franchisor</u>. Even if the franchise agreement grants you a territory, the franchisor may have the right to compete with you in your territory.

Renewal. Your franchise agreement may not permit you to renew. Even if it does, you may have to sign a new agreement with different terms and conditions in order to continue to operate your franchise business.

When your franchise ends. The franchise agreement may prohibit you from operating a similar business after your franchise ends even if you still have obligations to your landlord or other creditors.

Some States Require Registration

Your state may have a franchise law, or other law, that requires franchisors to register before offering or selling franchises in the state. Registration does not mean that the state recommends the franchise or has verified the information in this document. To find out if your state has a registration requirement, or to contact your state, use the agency information in Exhibit A.

Your state also may have laws that require special disclosures or amendments be made to your franchise agreement. If so, you should check the State Specific Addenda. See the Table of Contents for the location of the State Specific Addenda.

Special Risks to Consider About This Franchise

Certain states require that the following risk(s) be highlighted:

- 1. Out-of-State Dispute Resolution. The franchise agreement requires you to resolve disputes with the franchisor by mediation, arbitration, and/or litigation only in North Carolina. Out-of-state mediation, arbitration, or litigation may force you to accept a less favorable settlement for disputes. It may also cost more to mediate, arbitrate, or litigate with the franchisor in North Carolina than in your own state.
- 2. <u>New Franchise Concept</u>. We do not have a proven concept for the business model we are seeking to franchise and our name and trademarks have no name recognition.
- 3.2. **Financial Condition.** The franchisor's financial condition, as reflected in its financial statements (see Item 21), calls into question the franchisor's financial ability to provide services and support to you.
- 4.3. **Short Operating History**. This Franchisor is at an early stage of development and has a limited operating history. This franchise is likely to be a riskier investment than a franchise in a system with a longer operating history.
- 5.4. **Spousal Liability**. Your spouse must sign a document that makes your spouse liable for all financial obligations under the franchise agreement even though your spouse has no ownership interest in the franchise. This guarantee will place both your and your spouse's marital and personal assets, perhaps including your house, at risk if your franchise fails.
- 6. <u>Unregistered Trademark</u>. The primary trademark that you will use in your business is not federally registered. If the franchisor's right to use this trademark in your area is challenged, you may have to identify your business and its products or services with a name that differs from that used by other franchisees or the franchisor. This change can be expensive and may reduce brand recognition of the products or services you offer.
- 7.5. Mandatory Minimum Payments. You must make minimum monthly royalty fee or advertising fund payments, regardless of your sales levels. Your inability to make the payments may result in termination of your franchise and loss of your investment.

Certain states may require other risks to be highlighted. Check the "State Specific Addenda" (if any) to see whether your state requires other risks to be highlighted.

NOTICE REQUIRED BY STATE OF MICHIGAN

(THE FOLLOWING APPLIES TO TRANSACTIONS GOVERNED BY THE MICHIGAN FRANCHISE INVESTMENT LAW ONLY)

THE STATE OF MICHIGAN PROHIBITS CERTAIN UNFAIR PROVISIONS THAT ARE SOMETIMES IN FRANCHISE DOCUMENTS. IF ANY OF THE FOLLOWING PROVISIONS ARE IN THESE FRANCHISE DOCUMENTS, THE PROVISIONS ARE VOID AND CANNOT BE ENFORCED AGAINST YOU.

- A. A prohibition on the right of a franchisee to join an association of franchisees.
- B. A requirement that a franchisee assent to a release, assignment, novation, waiver, or estoppel which deprives a franchisee of rights and protections provided in this Act. This does not preclude a franchisee, after entering into a franchise agreement, from settling any and all claims.
- C. A provision that permits a franchisor to terminate a franchise prior to the expiration of its term except for good cause. Good cause includes the failure of the franchisee to comply with any lawful provision of the franchise agreement and to cure such failure after being given written notice thereof and a reasonable opportunity, which in no event need be more than 30 days, to cure such failure.
- D. A provision that permits a franchisor to refuse to renew a franchise without fairly compensating the franchisee by repurchase or other means for the fair market value at the time of expiration of the franchisee's inventory, supplies, equipment, fixtures, and furnishings. Personalized materials which have no value to the franchisor and inventory, supplies, equipment, fixtures, and furnishings not reasonably required in the conduct of the franchise business are not subject to compensation. This subsection applies only if: (i) the term of the franchise is less than 5 years; and (ii) the franchisee is prohibited by the franchise or other agreement from continuing to conduct substantially the same business under another trademark, service mark, trade name, logotype, advertising, or other commercial symbol in the same area subsequent to the expiration of the franchise or the franchisee does not receive at least 6 months advance notice of franchisor's intent not to renew the franchise.
- E. A provision that permits the franchisor to refuse to renew a franchise on terms generally available to other franchisees of the same class or type under similar circumstances. This section does not require a renewal provision.
- F. A provision requiring that arbitration or litigation be conducted outside this state. This does not preclude the franchisee from entering into an agreement, at the time of arbitration, to conduct arbitration at a location outside this state.
- G. A provision which permits a franchisor to refuse to permit a transfer of ownership of a franchise, except for good cause. This subdivision does not prevent a franchisor from exercising a right of first refusal to purchase the franchise. Good cause includes, but is not limited to:
 - i. The failure of the proposed transferee to meet the franchisor's then-current reasonable qualifications or standards.
 - ii. The fact that the proposed transferee is a competitor of the franchisor or subfranchisor.

- iii. The unwillingness of the proposed transferee to agree in writing to comply with all lawful obligations.
- iv. The failure of the franchisee or proposed transferee to pay any sums owing to the franchisor or to cure any default in the franchise agreement existing at the time of the proposed transfer
- H. A provision that requires the franchisee to resell to the franchisor items that are not uniquely identified with the franchisor. This subdivision does not prohibit a provision that grants to a franchisor a right of first refusal to purchase the assets of a franchise on the same terms and conditions as a bona fide third party willing and able to purchase those assets, nor does this subdivision prohibit a provision that grants the franchisor the right to acquire the assets of a franchise for the market or appraised value of such assets if the franchisee has breached the lawful provisions of the franchise agreement and has failed to cure the breach in the manner provided in subdivision (c).
- I. A provision which permits the franchisor to directly or indirectly convey, assign, or otherwise transfer its obligations to fulfill contractual obligations to the franchisee unless provision has been made for providing the required contractual services.
- J. If the franchisor's most recent financial statements are unaudited and show a net worth of less than \$100,000, the franchisor will, at the request of a franchisee, arrange for the escrow of initial investment and other funds paid by the franchisee until the obligations to provide real estate, improvements, equipment, inventory, certification, or other items included in the franchise offering are fulfilled. At the option of the franchisor, a surety bond may be provided in place of escrow.

THE FACT THERE IS A NOTICE OF THIS OFFERING ON FILE WITH THE ATTORNEY GENERAL DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION, OR ENDORSEMENT BY THE ATTORNEY GENERAL.

Any questions regarding this notice should be directed to:

State of Michigan Consumer Protection Division Attn: Franchise 670 G. Mennen Williams Building 525 West Ottawa Lansing, MI 48909 (517) 335-7567

Note: Despite paragraph F above, we intend, and we and you agree, to enforce fully the arbitration provisions of our Franchise Agreement. We believe that paragraph F is unconstitutional and cannot preclude us from enforcing these arbitration provisions.

THE MICHIGAN NOTICE APPLIES ONLY TO FRANCHISES WHO ARE RESIDENTS OF MICHIGAN OR LOCATE THEIR FRANCHISES IN MICHIGAN.

TABLE OF CONTENTS

ITEM		PAGE
ITEM 1	THE FRANCHISOR AND ANY PARENTS, PREDECESSORS, AND AFFILIATES	1
ITEM 2	BUSINESS EXPERIENCE	4
ITEM 3	LITIGATION	6
ITEM 4	BANKRUPTCY	6
ITEM 5	INITIAL FEES	6
ITEM 6	OTHER FEES	9
ITEM 7	ESTIMATED INITIAL INVESTMENT	19
ITEM 8	RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES	25
ITEM 9	FRANCHISEE'S OBLIGATIONS	29
ITEM 10	-FINANCING	31
ITEM 11	FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEM AND TRAINING	31
ITEM 12	TERRITORY	43
ITEM 13	TRADEMARKS	47
ITEM 14	PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION	49
ITEM 15	OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISED BUSINESS	50
ITEM 16	RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL	51
ITEM 17	RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION	51
ITEM 18	PUBLIC FIGURES.	61
ITEM 19	FINANCIAL PERFORMANCE REPRESENTATIONS	61
ITEM 20	OUTLETS AND FRANCHISEE INFORMATION	63
ITEM 21	FINANCIAL STATEMENTS.	67
ITEM 22	CONTRACTS	67
ITEM 23	RECEIPTS	67
ITEM 1	THE FRANCHISOR AND ANY PARENTS, PREDECESSORS, AND AFFILIATES	1
ITEM 2	BUSINESS EXPERIENCE	<u></u> 5
ITEM 3	LITIGATION	<u></u> 6
ITEM 4	BANKRUPTCY	<u></u> 7
ITEM 5	INITIAL FEES	7
ITEM 6	OTHER FEES	<u></u> 11
ITEM 7	ESTIMATED INITIAL INVESTMENT	25
ITEM 8	RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES	<u></u> 33
ITEM 9	FRANCHISEE'S OBLIGATIONS	<u></u> 38
ITEM 10	FINANCING	<u></u> 40

	RANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEM AND RAINING	40
	ERRITORY	
ITEM 13 T	RADEMARKS	<u></u> 57
ITEM 14 P	ATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION	<u></u> 59
	BLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE RANCHISED BUSINESS	<u></u> 61
ITEM 16 R	ESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL	62
ITEM 17 R	ENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION	62
ITEM 18 P	UBLIC FIGURES	<u></u> 73
ITEM 19 F	INANCIAL PERFORMANCE REPRESENTATIONS	<u></u> 73
ITEM 20 C	OUTLETS AND FRANCHISEE INFORMATION	<u></u> 81
ITEM 21 F	INANCIAL STATEMENTS	97
ITEM 22 C	CONTRACTS	<u></u> 97
ITEM 23 R	ECEIPTS	97
EXHIBITS:		
Exhibit A	List of State Administrators and Agents for Service of Process	
Exhibit B	Franchise Agreement	
Exhibit C Exhibit D	Statement of Ownership Confidentiality and Restrictive Covenant Agreement	
Exhibit E	Electronic Funds Withdrawal Authorization	
Exhibit E F	Multi-Unit Addendum	
Exhibit G	Sample Assignment and Assumption to a Business Entity	
Exhibit D H	State Specific Addenda	
Exhibit EI	Financial Statements	
Exhibit F	Sample Assignment and Assumption Agreement	
Exhibit G J	Operations Manual Table of Contents	
Exhibit <u>HK</u>	List of Franchisees and Franchisees That Have Franchisee that have Left the Syste	m in
	the Past Fiscal Year	
Exhibit IL	Diligence Review of Franchisee Affirmations and Acknowledgments-Sale Process	SS
Exhibit JM	State Effective Dates	
Exhibit K N	Item 23 Receipts	

ii

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]	
© 20232024 HFB FenceCo Franchising, LLC i 20232024 FDD 4-v1F-(MN)	

ITEM 1 THE FRANCHISOR AND ANY PARENTS, PREDECESSORS, AND AFFILIATES

The Franchisor is referred to in this Franchise Disclosure Document as "we," "us" or "our." "You" or "your" means the person who buys the franchise and includes your owners and principals if you are a corporation, limited liability company, or other business entity.

The Franchisor

HFB FenceCo Franchising, LLC (the franchisor) is a limited liability company organized under the laws of the State of North Carolina formed on May 25, 2022. Our principal business address is 107 Parr Drive, Huntersville, North Carolina 28078. We conduct business under our company name and our thencurrent proprietary marks ("Proprietary Marks"), which currently includes our primary mark: TOP RAIL M. Our agents for service of process are listed in Exhibit A to this Disclosure Document.

We offer qualified individuals and entities (based on level of business acumen and financial strength) the right to operate a business which sells, furnishes and installs wood, steel, aluminum, vinyl and other types of fencing for residential and commercial customers (collectively, the "Approved Services"). You must also offer and sell products to your customers (the "Approved Products") associated with the Approved Services, as we designate them.

We have offered Franchised Businesses since October 26, 2022. We do not operate any Top Rail franchised businesses. However, aone of our minority owner of Franchisorowners operates businesses of the type offered in this disclosure document and has done so since 2015. As of December 29, 2022, these businesses are franchisees. These businesses are referred to as "Affiliate Owned Locations" for purposes of Item 19 of this disclosure document and are listed in Item 20 as franchisees. Other than the aforementioned bove-mentioned services, we do not engage in any other business activities and have not offered franchises in any other line of business.

The Franchised Business

Your Franchised Business will offer the Approved Products<u>and Approved Services</u>, as we designate them, and Approved Services within a defined designated territory (the "Designated Territory"). You do not need any specific prior experience in order to operate a Franchised Business.

Each Franchised Business operates according to our proprietary business system which includes our valuable know-how, information, trade secrets, methods, operations manual, standards, designs, methods of trademark usage, copyrights, sources and specifications, confidential communications, methods of Internet usage, marketing programs, and research and development (collectively, the "System"). We reserve the right to add, modify or change the System, Approved Products and Approved Services at any time.

Upon execution of your Franchise Agreement and for a period of twelve (12) months after the Effective Date of your Franchise Agreement, you may operate the Franchised Business from a home office or storage facility that is approved by us and otherwise meets our standards and specifications with our approval from the Franchisor.

After this initial twelve (12)—month period, you must secure a leased/owned office/warehouse or storage facility within the Designated Territory that meets our current standards and specifications and is approved by us (the "Approved Location"). The Approved Location must include an office and storage

space for equipment and inventory. The Approved Location must typically be between 1,000 and 1,500 square feet.

We identify the System by means of certain trade names, service marks, trademarks, logos, emblems, distinctive trade dress, and indicia of origin, including, but not limited to, the mark TOP RAIL and such other trade names, trademarks, and service marks as we now designate or may in the future designate in writing for use in connection with the System. We continue to develop, use and control the use of the Proprietary Marks in order to identify for the public the source of services marketed under the Proprietary Marks and System to the public, and to represent the System's high standards of quality, appearance and service.

You must enter into our current form of franchise agreement (each, a "Franchise Agreement"), which is attached to this Franchise Disclosure Document as Exhibit B, for each Franchised Business we grant you the right to open and operate.

Multi-Unit Offering

We also offer qualified parties (based on level of business acumen and financial strength) the right to enter into a Multi-Unit Addendum, attached to this Franchise Disclosure Document as Exhibit GF ("Multi-Unit Addendum").") which amends the standard franchise agreement. Subject to the terms of the Multi-Unit Addendum, you will be granted the right to execute up to five (5) Franchise Agreements to open and operate up to five (5) Franchised Businesses at once, each with their own Designated Territory that are typically contiguous to one another ("Multi-Unit Offering"). All agreements must be executed as part of the same transaction at the same time and on the same form of Franchise Agreement. Each Franchised Business will be opened and operated pursuant tounder its respective Franchise Agreement.

You will also need to comply with the terms and conditions set forth in a mutualour form of Multi-Unit Addendum, including, but not limited to, minimum monthly royalty fees, and requirements for the vehicles, equipment and tools (which will apply to your operation of all Franchised Businesses granted as part of your Multi-Unit Offering). As part of the Multi-Unit Offering, you will be required to open at least two (2)1 Approved Locations Location, which may serve as one or morea central Approved Location(s) for all contiguous Designated Territories. You will still be required to commence soliciting customers, marketing and providing the Approved Products and Approved Services, in each of the Designated Territories that are granted under the multiple Franchise Agreements you have entered into.

You will be required to sign all Franchise Agreements associated with your Multi-Unit Offering, as well as your Multi-Unit Addendum, at the same time. As of the Issuance Date of this Franchise Disclosure DocumentAs of April 19, 2024, we do not intend or expect to offer a new prospect the right to participate in a Multi-Unit offering for more than five (5) Franchised Businesses (but we reserve the right to do so). The disclosures in this Franchise Disclosure Document, however, assume that the Multi-Unit Offering will involve the purchase of up to five (5) Franchised Businesses.

Parents, Predecessors and Affiliates

Franchisor does not have a predecessor. Franchisor is owned by DSG Five, LLC, a North Carolina limited liability company with a business address at 107 Parr Drive, Huntersville, North Carolina 28078.

Our parent, HFB HoldCo, LLC is a North Carolina limited liability company with a business address at 107 Parr Drive, Huntersville, NC 28078. HFB HoldCo, LLC is the majority owner of Franchisor. HFB HoldCo, LLC does not provide products or services to our franchisees. HFB HoldCo, LLC does not, and has not, offered or sold franchises in any line of business.

Our affiliate, DDL Investments, Inc. (d/b/a Homefront Brands or "HFB"), is a North Carolina corporation with a business address at 107 Parr Drive, Huntersville, NC 28078. DDL Investments, Inc. provides certain shared services including legal, finance, marketing, and other administrative services to us but does not provide products or services to our franchisees. DDL Investments, Inc. does not, and has not, offered or sold franchises in any line of business.

Our affiliate, A1 Kitchen & Bath Franchising, LLC ("The Designery"), is a North Carolina limited liability company with a business address at 107 Parr Drive, Huntersville, North Carolina 28078. A1 Kitchen & Bath Franchising, LLC franchises businesses under the mark "THE DESIGNERYSM" that provide home improvement and new construction products, as well as design, installation and demolition services for kitchen and bath projects. The Designery began offering franchises on May 15, 2021. As of December 31, 20222023, there were 421 The Designery franchises operating in the United States, and 7 The Designery franchises sold, but not yet operating in the United States.

Our affiliate, LP Franchising, LLC, a North Carolina limited liability company with a business address at 107 Parr Drive, Huntersville, North Carolina 28078 franchises businesses under the mark "WINDOW HERO®" that provide window cleaning, pressure washing, house washing, gutter cleaning, gutter guard installation, window film installation, deck cleaning, and other services. LP Franchising, LLC began offering franchises on April 29, 2022, and its predecessor Labor Panes Franchising, Inc. began offering franchises under our proprietary mark "Labor Panes" in January 2013 in the United States. As of December 31, 20232022, there were 2336 Window Hero franchises operating in the United States, and 52 Window Hero franchises sold, but not yet operating in the United States.

Our affiliate, Temp Walls Franchise Management, LLC, a North Carolina limited liability company with a business address at 107 Parr Drive, Huntersville, North Carolina 28078 franchises businesses under the mark "TEMPORARY WALL SYSTEMS®" that provide rental, installation and servicing of modular containment systems inside existing commercial buildings. Temp Wallwalls Franchise Management, LLC began offering franchises on May 1, 2022. As of December 31, 20232022, there were \$110 Temporary Walls franchises operating in the United States, and 1036 Temporary Walls franchises sold, but not yet operating in the United States.

Our affiliate, BiltRite Franchising, LLC, a North Carolina limited liability company with a business address at 107 Parr Drive, Huntersville, North Carolina 28028. BiltRite Franchising, LLC franchises businesses under the mark "BILTRITE®" that provide inspection services, insurance reports, meter readings, infrared scanning, and other services, including both residential and light commercial applications. BiltRite Franchising, LLC began offering franchises on April 19, 2022. As of December 31, 2022, there were 4 BiltRite affiliate owned locations and no BiltRite franchises operating in the United States.

Our affiliate, HFB MosquitoCo Franchising, LLC, a North Carolina limited liability company with a business address at 107 Parr Drive, Huntersville, North Carolina 28028 franchises businesses under the mark "MOZZIE DOMESM"YARD PATROL PROSSM" (and previously franchised businesses under the "Mozzie Domesm" mark) that sells, designs, installs and services outdoor misting systems and other repetitive application for mosquito, tick and fly control and elimination and control systems, and lawn care services including application of fertilizer, weed control, aeration, pest control, plant care, and tree care, for both residential and commercial use. HFB MosquitoCo Franchising, LLC began offering franchises on December 6, 2022. As of December 31, 20232022, there were no11 Mozzie Dome franchises operating in the United States, and 52 Mozzie Dome franchises sold, but not yet operating in the United States.

Our affiliate National Account Services Company, LLC is a North Carolina limited liability

company with a business address at 107 Parr Drive, Huntersville, North Carolina 28028. National Account Services Company, LLC provides national account marketing and support services for affiliated franchisor companies. The support services provided to affiliate franchisor companies include interaction with and support services to franchisees to coordinate lead flows and compliance with National Account service level requirements required of service providers for those accounts. National Account Services Company, LLC does not, and has not, offered or sold franchises in any line of business.

Our affiliate Bold Future Payment Services, LLC is a North Carolina limited liability company with a business address at 107 Parr Drive, Huntersville, North Carolina 28028. Bold Future Payment Services Company, LLC provides payment card processing services to franchisees of affiliate franchisors. Bold Future Payment Services Company, LLC does not, and has not, offered or sold franchises in any line of business.

Except as stated above, we do not have any parents, predecessors or affiliates that: (a) offer or operate franchises in any line of business; or (b) are otherwise involved in any other business activity.

Market and Competition

Your Franchised Business will offer our Approved Services and Approved Products, as implemented and designated, to the general public, which will include residential and commercial properties. The market for the Approved Products and Approved Services is well developed, and there will be competition from other national and regional chains and local businesses that offer the same or similar services within the Designated Territory you are granted. Certain of our Approved Products and Approved Services are seasonal and may be affected by climate, weather or other environmental conditions.

Your competitive advantage in the marketplace will be based on your adherence to our System standards and guidelines, as well as your entrepreneurial and managerial abilities and focus on customer service in the operation of your Franchised Business.

Industry Specific Regulations

Your Franchised Business will be subject to laws and regulations in your state, county or municipality regarding the operation of an establishment that offers and sells products and services which are the same or similar to our Approved Products and/or Approved Services. The Franchised Business will be subject to federal, state and local regulations, and you must strictly comply with such regulations. There are specific regulations pertaining to this industry and you must comply with all local, state and federal codes and regulations and all Environmental Protection Agency ("EPA") and other environmental regulations pertaining to the Franchised Business.

Some states and municipalities may have strict licensing or permitting requirements which may delay your ability to open your Franchised Business. You may be required, for example, to obtain specific education, training, apprenticeship, licensing, and/or certifications, and comply with certain workplace requirements and/or environmental requirements, related to the operation of your Franchised Business in the Designated Territory or the offer and sale of our Approved Products and/or Approved Services therein.

ITEM 2 BUSINESS EXPERIENCE

President of Top Rail: Todd Bingham

Mr. Bingham has served as our President in Huntersville, North Carolina since our inception. Mr. Bingham has also held the following positions: (i) VP of Development of Restoration 1 Franchise in Waco, Texas from October 2017 to July 2022; (ii) VP of Development for The Driveway Company Franchise in Waco, Texas from February 2021 to July 2022; (iii) Chief Operating Officer of the Driveway Company Franchise in Waco, Texas from February 2021 to July 2022; (iv) VP of Development for Softroc Franchise in Waco, Texas from April 2021 to July 2022; and (v) VP of Development for the Bluefrog Plumbing and Drain Franchise in Waco, Texas from October 2017 to July 2022; (vi) President of Pinnacle Franchise Development in Louisville, Kentucky from January 2015 to August 2017; and (vii) President of FranNet in Louisville, Kentucky from January 2015 to August 2017.

Chief Executive Officer-of HFB: Jeffrey Dudan

Mr. Dudan has served as our Chief Executive Officer in Huntersville, North Carolina since our inception. Mr. Dudan has also held the following positions: (i) Chief Executive Officer of Homefront Brands in Huntersville, North Carolina since March 2022; (ii) Advisor to Beem Light Sauna in Huntersville, North Carolina since January 2022; (iii) Advisor to RockBox Fitness in Huntersville, North Carolina sincefrom January 2019_to December 2023; (iv) Chief Development Officer of Mowbot, Inc. in Huntersville, North Carolina from August 2017 to April 2021; (v) Advisor to FranDevCo in Huntersville, North Carolina since March 2019; and (vi) Advisor to Franchise Genesis in Huntersville, North Carolina since January 2021; and (vii) Chief Executive Officer of AdvantaClean Systems, Inc. in Huntersville, North Carolina from February 1994 to January 2019.

Chief Operating Officer and President of Franchising-of-HFB: Michael O'Driscoll

Mr. O'Driscoll has served as our Chief Operating Officer and the Chief Operating Officer of Homefront Brands in Huntersville, North Carolina since March 2022. Mr. O'Driscoll has also served as our President of Franchising since April 2023. Mr. O'Driscoll has also held the following positions: (i) Global Business Strategist of EMF Advisory, Inc. in New York, New York since April 2021; (ii) Non-Executive Director & Advisor in Brisbane, Australia from March 2019 to April 2021; and (iii) Non-Executive Board Director of Gutter-Vac Pty Ltd Australia from March 2019 to August 2020; and (iv) Chief Executive Officer of Nous House Australia in Brisbane, Australia from November 2016 to March 2019.

Chief Financial Officer of HFB: Michael Dudan

Mr. Dudan has served as our Chief Financial Officer and the Chief Financial Officer of Homefront Brands in Huntersville, North Carolina since March 2022. Mr. Dudan has also held the following positions: (i) Advisor to Beem Light Sauna in Huntersville, North Carolina since March 2022; (ii) Advisor to RockBox Fitness Franchising in Huntersville, North Carolina from March 2019 to March 2022; and (iii) various roles, including Chief Financial Officer of the Carolina Panthers in Charlotte, North Carolina from March 2003 to March 2019.

Chief Marketing Officer of HFB: Carie Beeman

Mrs. Beeman has served as our Chief Marketing Officer and Chief Marketing Officer of Homefront Brands in Huntersville, North Carolina since March 2023. Mrs. Beeman has also held the following positions: (i) Senior Vice President of Marketing, City Facilities Management, Jacksonville, FL from

May 2022 to December 2022; (ii) Vice President, <u>Brand of Brands</u> for PODS Moving and Storage in <u>Clearwater, FL</u> from November 2019 to May 2022; and (iii) Senior Director, <u>Brand of Brands</u> for Hertz Enterprise Holdings in <u>Estero, FL</u> from February 2015 to November 2019.

Vice President of Operations of HFB: Chris Stefanco

Mr. Stefanco has served as the Vice President of Operations of Homefront Brands in Huntersville, North Carolina since March 2022. Mr. Stefanco served as the Vice President of Operations for DDL Investments, Inc. from August 2020 to March 2022. Prior to that time, Mr. Stefanco held various operations roles, including Vice President of Operations, with AdvantaClean Systems, Inc. in Huntersville, North-Carolina from December 1999 to June 2020.

Chief DevelopmentGrowth Officer of HFB: Zack Dudan

Mr. Dudan has served as our Chief <u>DevelopmentGrowth</u> Officer and director and interim Chief Development Officer of Homefront Brands in Huntersville, North Carolina since January 2023. Mr. Dudan has also served as <u>Chief</u> Growth Officer of Homefront Brands since January 2018.

Chief Legal Officer-of HFB: John Haraldson

Mr. Haraldson has served as our Chief Legal Officer and the Chief Legal Officer of Homefront Brands in Huntersville, North Carolina since November 2022. Mr. Haraldson has also held the following positions: (i) General Counsel of Lynx Franchising and its affiliates in Atlanta, Georgia, including the following brands: JAN-PRO, The Intelligent Office, FRSTeam, Archadeck, Conserva, Outdoor Lighting Perspectives, and Superior Fence & Rail from November 2019 to September 2022; and (ii) Vice President and Division General Counsel for the ServiceMaster Franchise Services Group, including the ServiceMaster Restore, ServiceMaster Clean, ServiceMaster Plumbing, Merry Maids, Furniture Medic, and AmeriSpec franchised brands in Memphis, Tennessee from June 2013 to January Private Practice and Consulting from February 2019 to October 2019.

VP of Learning & Development of HFB: Greg Smith

Mr. Smith has served as our Vice President of Learning and Development and the Vice President of Learning and Development of Homefront Brands in Huntersville, North Carolina since November 2022. Mr. Smith has also held the following positions: -(i) Director of Training and Development at Armstrong Transport Group in Charlotte, North Carolina from May 2021 to November 2022; (ii) Senior Instructional Designer at Saddle Creek Logistics in Lakeland, Florida from November 2020 to May 2021; and (iii) Learning Consultant at Delray Learning Solutions in Delray Beach, Florida from February 2019 to June 2021; (iv) Vice President of Centralized Programs at Festiva Vacation Ownership in Orlando, Florida from March 2018 to February 2019; and (v) Director of Training and Development at Bluegreen Vacation Ownership in Boca Raton, Florida from August 2014 to March 2018.

Jeff Dudan and Michael Dudan are brothers and Zack Dudan is Jeff Dudan's son.

ITEM 3 LITIGATION

No litigation is required to be disclosed in this Item.

ITEM 4 BANKRUPTCY

No bankruptcy information is required to be disclosed in this Item.

ITEM 5 INITIAL FEES

Franchise Agreement

Initial Franchise Fee

The initial franchise fee you must pay to us is \$5459,900 (the "Initial Franchise Fee"), which is a fee for a territory of up to 80,000 ownedowner-occupied households ("Qualified Households") and is due upon execution of the Franchise Agreement. If your territory has more than 80,000 Qualified Households, your Initial Franchise Fee will increase from the baseline amount of \$5459,900 by \$0.70 for each additional Qualified Household, up to an additional 20,000 Qualified Households pending our approval (maximum of \$6873,900). The Initial Franchise Fee is deemed fully earned and non-refundable upon payment. During our last fiscal year ended December 31, 2023,2022, we sold 26 Franchises and the Initial Franchise Fee ranged from \$0 to \$5468,900. The founder that operates the Affiliate Owned Locations paid an initial fee of \$0 based on an agreement in the Asset Purchase Agreement to convert its existing operations to franchised businesses. (includes maximum Qualified Household overage charge). We may in the future offer-additional discounts or incentives for a limited time and for specific circumstances that lower the minimum Initial Franchise Fee based on territory size, location and whether the territory is sold to an existing franchisee.

Franchisor Directed Advertising Fee Deposit

Simultaneously with the payment of the Initial Franchise Fee, you are required to pay us an amount equal to \$6,000 representing the first two (2) months of Franchisor Directed Advertising Fee ("Local Advertising Fee Deposit"), as described below in Item 6. The Local Advertising Fee Deposit will be applied towards your first two (2) months of Franchisor Directed Advertising Fees and is deemed a pro-rated credit towards your total required Local Advertising Requirement (defined in Item 6). The Directed Advertising Fee Deposit is refundable on a pro-rated basis if the Franchise Agreement is terminated prior to completing two months of operation. The Franchisor Directed Advertising Fee is a portion of your required Local Advertising Requirement that we direct on your behalf and is not a separate fee.

We may periodically increase or decrease the Franchisor Directed Advertising Fee in an amount we deem appropriate in our Business Judgment, but in no event will that increase cause the Franchisor Directed Advertising Fee to be more than the Local Advertising Requirement. Further, we reserve the right to change the Franchisor Directed Advertising Fee from a dollar amount to a percentage of sales or to any other formula we designate. We will give you thirty (30) days' advance notice by email or other means if we intend to change the Franchisor Directed Advertising Fee. For purposes of the Franchise Agreement and this disclosure document, the term "Business Judgment" means the exercise of reasonable business judgment in making Franchisor's decision or exercising Franchisor's rights. Franchisor's decisions or actions will be deemed to be the result of reasonable business judgment, even if other reasonable or even arguably preferable alternatives are available, if Franchisor's decision or action is intended, in whole or significant part, to promote or benefit the franchise System generally, even if the decision or action also promotes Franchisor's financial or other individual interest. Examples of items that will promote or benefit the franchise System include, without limitation, enhancing the value of the Proprietary Marks, improving

customer service and satisfaction, improving product quality, improving uniformity, enhancing or encouraging modernization, and improving the competitive position of the franchise System. This is a defined term for the purposes of the Franchise Agreement and this disclosure document and is not intended to incorporate principles related to the application of the business judgment rule in a corporate law context.

Opening Assistance Fee

The opening assistance fee you must pay to us is \$5,000 (the "Opening Assistance Fee"), which is a fee for opening assistance services that include: the cost of attending the initial training program, access to the learning management paths, and certain meals during the Initial Training Program, any on-site visits that the operational and support team will make, the initial cost of marketing setup, which includes, but is not limited to, the initial creation of collateral, website, and other marketing setup items (but excludes the ongoing cost that may be paid to us, vendors, or affiliates of additional collateral and other marketing channels), the initial cost of setup with miscellaneous software and technology utilized through the life of the business but does not cover the ongoing cost of software and technology that you may use in your business, and our administrative cost of assisting with of initial legal and compliance guidance that we may provide in our sole discretion. This fee is a single fee for one Franchise Agreement or multiple franchise agreements under a single Multi-Unit Addendum. Each Multi-Unit Addendum will have a separate Opening Assistance Fee.

Multi-Unit Offering

Multi-Unit Fee

If we grant you the right to participate in our Multi-Unit Offering, then you must pay us a multi-unit fee based on the number of Franchised Businesses we grant you the right to develop. Our multi-unit fee is set forth below ("Multi-Unit Fee"):

Number Of Franchised Businesses	<u>Multi-Unit</u> <u>Fee</u>	Maximum Additional Qualified Household Fee	Opening Assistance Fee (assumes one Multi-Unit Addendum)	Total (Multi-Unit Fee + Opening Assistance Fee + Additional Qualified Households)
1	<u>\$59,900</u>	<u>\$14,000</u>	\$5,000	\$ 5 4 <u>64,900 - \$78,</u> 900
2	\$99,900	\$28,000	<u>\$5,000</u>	\$ 94 104,900 - \$132,900
3	<u>\$134,900</u>	<u>\$42,000</u>	<u>\$5,000</u>	\$ 129 139,900 - \$181,900
4	<u>\$169,900</u>	<u>\$56,000</u>	<u>\$5,000</u>	\$ 164 <u>174,900 - \$230,</u> 900
5	<u>\$194,900</u>	<u>\$70,000</u>	<u>\$5,000</u>	\$ 189 <u>199,900 - \$269</u> ,900

The Multi-Unit Offering is open to franchisees that have at least 50% common ownership with an existing franchisee (or a franchisee signed at the same time) and this discount applies to future purchases and purchases of affiliated brands. If we grant you the right to develop more than five (5) Franchised Businesses, then you must pay us a Multi-Unit Fee equal to: (i) \$189194,900 for the first five (5) Franchised Businesses, plus (ii) \$20,000 for each additional Franchised Business you are granted the right to develop. In addition, if any territory in a Multi-Unit Offering has more than 80,000 households Qualified Households, the Initial Franchise Fee allocated to that territory will increase from the applicable baseline amount

(\$5459,900 for unit 1 and the incremental amount for additional units) by \$0.70 for each additional householdQualified Household, up to an additional 20,000 householdsQualified Households. The Multi-Unit Fee must be paid upon execution of the Multi-Unit Addendum and is deemed fully earned and non-refundable upon payment. We may in the future offer additional discounts or incentives for a limited time and for specific circumstances that lower the minimum Initial Franchise Fee based on territory size, location and whether the territory is sold to an existing franchisee.

Discount Programs

Veteran Discount

If you are a United States honorably discharged veteran who meets our qualifications for purchasing a Franchised Business, you will receive a discount of \$5,000 off of the Initial Franchise Fee for your first Franchised Business. Additional Franchised Businesses under a Multi-Unit Addendum will incur the Multi-Unit Fees set forth in the table above. You may use the Veteran's discount only once and only in accordance with the restrictions set forth in this Item 5. To qualify, a prospective franchisee must: (i) request the Top Rail Veteran's discount at the time of application; (ii) meet our then-current qualifications for new franchisees; (iii) be at least 51% legally and beneficially owned by persons meeting our qualifying veteran status; and (iv) not have previously received a Veterans incentive from us. You may meet our qualifying veteran status if you are a veteran who has received an honorable discharge from the U.S. Military, and you must give us a copy of your Form DD 214 showing your status as an honorably discharged veteran. We may discontinue the Veteran's discount program at any time.

Elevate to Owner Program Discount

Under our "Elevate to Owner Program," we offer a discounted Initial Franchise Fee for your first Franchised Business to qualified employees of our franchisees who: (i) have been recommended by a System franchisee; (ii) have been employed by a System franchisee for at least two (2) years; and (iii) otherwise qualify to be our franchisee. The discount is based upon years of service with one of our franchisees and is calculated as follows:

Discount On Initial Franchise Fee	Years Of Consecutive Employment With Franchisee
10%	2
15%	3
20%	4
25%	5+

Discounts under our Elevate to Owner Program will be applied to the Initial Franchise Fee for your first Franchised Business only, and the discount shall be calculated after any third—party broker fees are deducted, if any. Additional Franchised Businesses under a Multi-Unit Addendum will incur the Multi-Unit Fees set forth in the table above. You may use the Elevate to Owner Program discount only once and only in accordance with our standards and specifications.

Conversion Roll In Discount for Existing Businesses

If you have an existing business that is the same or similar to a Franchised Business with annual gross revenue of at least \$500,000 and you agree to convert the existing business into a Franchised

Business, we will discount your royalty fee during your first year of operation. Your Royalty Fee (defined in Item 6)-during the first year of operation will be as follows:

Months of Operations	Royalty Fee
Months 1-4	25% of then current Royalty Fee
Months 5 8	50% of then current Royalty Fee
Months 9-12	75% of then-current Royalty Fee
Months 12+	The then-current Royalty Fee

Combination and Application of Discounts

You may not combine any of the discounts set forth above with any other discount. If you have received a discount for your first purchase of a Franchised Business or multiple Franchised Businesses under a Multi-Unit Addendum, you will not qualify for any of these discounts on subsequent purchases of Franchised Businesses unless we agree in writing. You must provide us with reasonable documentation that we request in order to assess your eligibility for a given discount program and any questions regarding your eligibility for discounts, or for combination of any discounts, will be resolved by us in our sole discretion.

In some circumstances we have deferred payment some or all of the Initial Franchise Fee for a set period of time because of state registration requirements or delays of a franchisee's funding source(s). In addition, in some circumstances we have added more than 20,000 households (up to a maximum of 37,921) to a Designated Territory, and the per household rate for additional households varied from \$0.52 per household to \$0.70 per household and/or did not charge overage charges for additional households. Except as described above, all of the fees described in this Item 5 are paid in a lump sum and are uniformly calculated and imposed.

Initial Training Fee for Replacement or Additional Trainees

We will provide the Initial Training Program (as defined in Item 11) for the first three (3) trainees at no charge. We reserve the right to charge a training fee of \$500, which we may increase upon sixty (60) days' written notice to you, for the following training that occurs prior to opening the business: (i) each person in excess of three (3) trainees, even if they attend the same training session; (ii) each person who is repeating the course or replacing a person who did not pass; and (iii) each subsequent operating principal, designated manager or employee who attends the course at a scheduled training class at our headquarters or other location designated by us. You are responsible for any travel and living expenses, wages, and other expenses incurred by your trainees during the Initial Training Program or any other training program.

ITEM 6 OTHER FEES

Type Of Fee ⁽¹⁾	Amount	Due Date	Remarks
Royalty Fee ⁽²⁾	The greater of: (i) a blended 7% - 5% of your Gross Revenue; or (ii) the Minimum Royalty Fee (defined below in Item 6) each month (collectively, the "Royalty Fee"). The Royalty Fee for newly Approved Products or Approved Services may be different (either higher or lower) but will not be increased by more than 3% of gross revenue above the 7% to 5% Royalty Fee (and this 3% may be allocated between the Royalty Fee and the Brand Fund Contribution). The Royalty Fee is discussed in greater detail in the notes Note 2 below this Item 6 Table.	The Royalty Fee is currently due monthly within ten (10) business days after the end of each calendar month.	See Notes 1 and 2 for the definition of Gross Revenue. Your first Royalty Fee payment is due in the first full month after the date the Franchised Business opens and shall be paid based on all Gross Revenue accrued prior to the opening date and during your first full or partial month of operations.

Type Of Fee ⁽¹⁾	Amount	Due Date	Remarks
Brand Fund Contribution ⁽²⁾	Currently, one percent (14%)2% of your Gross Revenue ("Brand Fund Contribution"). We reserve the right to increase your Brand Fund Contribution to three percent (3%) of your Gross Revenue. 3% of your Gross Revenue. The Brand Fund Contribution for newly Approved Products or Approved Services may be different (either higher or lower) but will not be increased by more than 3% of gross revenue (and this 3% may be allocated between the Royalty Fee and the Brand Fund Contribution).	Same as Royalty Fee.	See Notes 1 and 2 below. This contribution will be used for a system-wide "Brand Fund" for our use in promoting and building the Top Rail brand. The Brand Fund is discussed in Item 11.
Local Advertising Requirement ⁽³⁾	Currently, the minimum required spend is the greater of: (i) three percent (3%)% of Gross Revenue per month; andor (ii) \$3,000 (the "Local Advertising Requirement"). We may increase the Local Advertising Requirement up to a maximum of four percent (4%) of Gross Revenue per month upon 90 days' prior written notice to you. An amount up to the Local Advertising Requirement, currently the greater of \$3,000 per month or three percent (3%) of Gross Revenue per month must be paid to Franchisor ("Franchisor Directed Advertising Fee"), which amount will be applied as a credit to your Local Advertising Requirement amount. The Franchisor Directed Advertising Fee is	Payable by the end of each month, as incurred, except that the Franchisor Directed Advertising Fee must be paid by the 10th of the month.	If you fail to meet your Local Advertising Requirement in any month, we may require you to pay us the shortfall as an additional advertising fee or to pay us the shortfall for us to spend on local advertising in your Designated Territory (defined in Item 12) You are required to sign the Franchisor Directed Advertising Services Agreement attached to the Franchise Agreement as Exhibit H. Under this agreement, you will pay a Franchisor Directed Advertising Fee that is an amount up to the Local Advertising Requirement, currently the greater of \$3,000 per month or three percent (3%) of Gross Revenue per month. This fee is subject to periodic adjustment but will not be higher than the Local Advertising Requirement (and is a credit towards the Local Advertising Requirement).

Type Of Fee ⁽¹⁾	Amount	Due Date	Remarks
	a-portion of your required Local Advertising Requirement that we direct on your behalf and is not a separate fee.4% of Gross Revenue per month upon 90 days' prior written notice to you.		If local advertising directed by us is paid with a franchisee credit card (or other method of direct payment), all such amounts spent that are directed by us will be credited to the monthly Franchisor Directed Advertising Fee.
			Provided, however, that where a group of Designated Territories are under a single Multi-Unit Addendum, the Franchisor DirectedLocal Advertising Fee will be an amount up toRequirement is the greater of: (i) 3% of Gross Revenue per month; or (ii) \$3,000 for the first Designated Territory plus an amount up to \$1,500 for each additional Designated Territory under that one Multi-Unit Addendum. If you have more than one Multi- Unit Addendum, you will pay for each group under a Multi- Unit Addendum an amount up tothe greater of: (i) 3% of Gross Revenue per month; or (ii) \$3,000 for the first Designated Territory plus an amount up to \$1,500 for each additional Designated Territory.
Marketing Management Fee	\$500/month	Same as Royalty Fee.	The Marketing Management Fee currently includes the following localized marketing support: email marketing, social media, promotions and offers, creative support, public relations management, tradeshow support, and paid media support.

Type Of Fee ⁽¹⁾	Amount	Due Date	Remarks
Regional Advertising and Promotional Cooperative	If established, determined by the cooperative. The contribution amount is set by the cooperative, and may only exceed the Local Advertising Requirement of 4% of Gross Revenue if agreed to by 2/3 of the cooperative. If companyowned Franchised Businesses have controlling voting power (defined as 2/3 of the cooperative votes), then the contribution amount is limited to 2% above the maximum Local Advertising Requirement, unless 2/3 of the cooperative vote, excluding company owned Franchised Business votes, to approve a higher amount.	If established, determined by the cooperative.	We currently do not have a cooperative, but reserve the right to require one to be established in the future. Item 11 contains more information about advertising cooperatives.
Technology Fee	Our then-current technology fee, which is currently \$599 per month (the "Technology Fee"). We reserve the right to increase or decrease the Technology Fee at any time upon 30 days' prior written notice to you.	Same as Royalty Fee.	The Technology Fee currently includes fees related to your access to and usage of our reservation system, our intranet, any mobile applications we develop, and the System website. We may add, delete, or otherwise modify the products and services that are included in the Technology Fee. We do not presently expect the Technology Fee to exceed \$1,000 per month, however, there is no cap on the amount the Technology Fee may be increased. An increase will be based on the following: cost of inflation, cost of underlying products and services, including development costs of software, mobile applications, websites, costs of software,

Type Of Fee ⁽¹⁾	Amount	Due Date	Remarks
			subscriptions, and related services, and any other products or services added to the technology stack. The first month will be assessed pro rata from the date on which you begin offering Approved Products and/or Approved Services. Provided, however, that where a group of Designated Territories are under a single Multi-Unit Addendum, you will be required to pay one Technology Fee for all Designated Territories under that one Multi-Unit Addendum.
			If you have more than one Multi-Unit Addendum, you will pay one Technology Fee for each such Multi-Unit Addendum.
Managed Services Desk Fee	Then-current fee (currently, \$55 per month for franchisee and \$20 per month for staff). We reserve the right to increase or decrease the Managed Services Desk Fee at any time upon 30 days' prior written notice to you.	Same as Royalty Fee.	The Managed Services Desk Fee includes management of franchisee emails and software support. This fee is currently included in the Technology Fee.
Online Local Presence Fee	\$249/month	Same as Royalty Fee.	The Online Local Presence Fee currently includes the following: website management, development, updating, maintenance, enhancement, and brand reputation management.

Type Of Fee ⁽¹⁾	Amount	Due Date	Remarks
Call Center Fee	We do not currently charge a call center fee; however, we reserve the right to establish a call center in the future and require your participation in the call center. If we do implement a call center, you will be required to pay up to four percent (4%)2% of your Gross Revenue each month (the "Call Center Fee"). We may increase the Call Center Fee in an amount up to 4% of your Gross Revenue each month.	If implemented, due monthly within ten (10) business days after the end of each ealendar month. Same as Royalty Fee.	If implemented, the The Call Center Fee will include fees related to your account setup and utilization of a call center for routing customers to your Franchised Business. We reserve the right to require you to use our preferred service provider to receive call center services, in which case we may charge the Call Center Fee.
Renewal Fee	\$5,000 (the "Renewal Fee").	Upon execution of a successor franchise agreement.	Payable if you qualify to renew your Franchise Agreement(s) and choose to enter into a new successor franchise agreement(s) You will be obligated to pay a Renewal Fee for each Franchised Business you wish to renew under the Multi-Unit Addendum. Provided, however, that where a group of territories are under a single Multi-Unit Addendum, the Renewal Fee will be \$5,000 for the first Franchise Agreement and \$1,000 for each additional Franchise Agreement renewed at the same time.
Transfer Fee	\$10,000 (the "Transfer Fee") plus the cost of additional training where applicable. Transfer Fees are subject to state law.	A non-refundable \$3,000 deposit of the Transfer Fee is due with your written notice of proposed transfer and the balance of \$7,000 is due upon our approval at closing.	No Transfer Fee is due for transfers upon death or incapacity which occurs prior to the date of transfer.

Type Of Fee ⁽¹⁾	Amount	Due Date	Remarks
Remarketing Fee	\$2,500 listing fee; a success fee equal to 8% of the gross sales price of the Franchised Business; if brokers are used, they may have a separate fee that ranges from 2% to 10%. These fees are in addition to any applicable Transfer Fees.	The listing fee is due upon signing the resale program agreement. The success fee and broker fees, if any, are due upon sale of the Franchised Business.	The resale marketing program is optional and franchisees may use it or choose to sell their business on their own (or with their business broker), and may only be available in certain states.
Late Fee and Interest	The lesser of: (i) one and one half percent (1½%)½% per month; 18% per annum); or (ii) the maximum interest rate allowed by law from the due date of payment. You must also pay us \$100 for each week that a payment is paid after the due date which is in addition to the interest above.	When amount owed becomes past due.	Required whenever a payment to us or our affiliate(s) is made after its due date. Interest accrues from the original due date until payment is received in full.
Nonsufficient Funds Fee	The greater of \$100 per occurrence, or the highest amount allowed by law.	As incurred.	Payable if any check or EFT payment is not successful due to insufficient funds, stop payment, or any similar event.
Payment Service Fee	Up to four percent (4%)% of total charge.	As incurred.	If payment is made to us or our affiliate(s) by credit card for any fee required, we may charge a service charge of up to four percent (4%)% of the total charge.
Initial Training Fee for Replacement or Additional Trainees	Currently, \$0 per trainee for up to three (3) trainees who attend a scheduled training class at our headquarters or other location designated by us, provided that all of your trainees are trained during the same training session. Currently, \$500 per day (plus our travel and living expenses) for additional on-site training in your	Within fourteen (14) days of receipt of an invoice. Depending on the circumstances, this fee may be payable before or after the business is opened.	We will provide the Initial Training Program (as defined in Item 11) for the first three (3) trainees at no charge. We reserve the right to charge a training fee of \$500, per person which we may increase upon sixty (60) days' written notice to you, for training: (i) each person in excess of three (3) trainees, even if they attend the same training session; (ii) each person who is repeating the course or replacing a person who did not pass; and (iii) each

Type Of Fee ⁽¹⁾	Amount	Due Date	Remarks
	Designated Territory. This amount may be modified upon notice by update to the Operations Manual.		subsequent operating principal, designated manager Designated Manager or employee who attends the course at a scheduled training class at our headquarters or other location designated by us. You are responsible for any travel and living expenses, wages, and other expenses incurred by your trainees during the Initial Training Program or any other training program.
Additional Training Programs	The then-current fee, which is currently \$500 per employee or agent for each full or partial day for a scheduled training class at our headquarters or other location designated by us, plus their travel and living expenses, and an additional amount based on the program administered.	Within fourteen (14) days of receipt of an invoice.	We may charge you our then- current fee (currently, \$500 per employee or agent, plus our travel and living expenses) for optional or required training programs or consulting services that we may provide in-person or otherwise. We may change this fee without limitation from time to time upon written notice to you.
In-Person Consulting Services	The then-current fee, which is currently \$500 per employee or agent for each full or partial day, plus their travel and living expenses.	Within 14 days of receipt of an invoice.	Payable if we provide requested consulting services in person at a location other than our offices. We may change this fee without limitation from time to time upon written notice to you.
Temporary Designated Manager	The then-current fee, which is currently \$500 per employee or agent for each full or partial day, plus their travel and living expenses.	Within fourteen (14) days of receipt of an invoice.	Payable if we provide a Designated Manager (defined in Item 15) to work at your Franchised Business after the departure of your previous manager, until a new Designated Manager is hired and trained.

Type Of Fee ⁽¹⁾	Amount	Due Date	Remarks
Temporary Management	The then-current fee charged by us during the period of management, plus any direct out of pocket costs and expenses. We reserve the right to charge up to ten percent (10%)% of the Gross Revenue generated during the time period that we provide temporary management.	Within fourteen (14) days of receipt of an invoice.	Payable in the event of your death or if we exercise our right to manage your Franchised Business after a default.
Annual Conference Fee	We reserve the right to host an annual conference (the "Annual Conference") and charge you the then-current Annual Conference fee. The Annual Conference fee is currently \$1,000 for up to two (2) people to attend. We may charge our then-current fee for additional attendees (currently, \$500 per additional attendee). We reserve the right to host up to two (2) Annual Conferences per year.	Prior to attending the event.	Payable for you and your employee(s) who attend the Annual Conference that we host. You are responsible for the travel and living expenses of you and your employees. If you do not attend the Annual Conference(s), you must still pay us the Annual Conference Fee, regardless of the cause for non-attendance, unless you receive our advance written approval for such absence. This fee is charged on a per franchisee basis.
Product, Service, Supplier, and Service Provider Review	A \$1,000 alternative supplier fee/new product review fee plus our actual costs of the inspection and our actual cost of testing the proposed product or evaluating the proposed service or service provider, including personnel and travel costs.	Upon demand.	Payable if you wish to offer products or use any supplies, equipment, or services that we have not approved or wish to purchase from a supplier or service provider that we have not approved, whether or not we approve the item, service, product, or service provider.
Insurance	Cost of premium plus a \$100 administrative fee for our services in procuring the insurance.	Upon demand.	Payable only if you fail to maintain the minimum insurance we require and we choose to procure the required insurance for you.

Type Of Fee ⁽¹⁾	Amount	Due Date	Remarks
Remodel/ Re-equip/ Refurbish Requirement	\$25,000 Actual costs and expenses which are estimated to be no more than \$25,000	Upon remodel/ re- equip/ refurbish.	Payable if we require you to remodel, re-equip, and otherwise refurbish your Franchised Business to bring it into conformity with our thencurrent brand image. A remodel/re-equip/refurbish will not be required in the last twelve (12) months of a term but will be a condition of renewal.
Relocation Fee	Actual costs and expenses of providing relocation assistance, plus a \$250 administrative fee.	Upon relocation.	You must reimburse us for our actual costs and expenses associated with evaluating your relocation request and/or any locations proposed by you for relocation, plus a \$250 administrative fee. We will provide you with copies of our invoices for our expenses from any third-party providers upon request.
Taxes	Varies.	As Incurred.	You must reimburse us for any taxes that we must pay to any taxing authority for your Franchised Business.
Audit Costs	Our costs and expenses, including costs for an independent accountant and attorneys' fees, any understated amounts, and related travel and living expenses.	Within fourteen (14) days of receipt of an invoice.	Payable if an audit or review shows an understatement of Gross Revenue for the audited or reviewed period of two percent (2%)% or more, or if you fail to submit required reports.

Type Of Fee ⁽¹⁾	Amount	Due Date	Remarks
Indemnification	Actual amount of our liabilities, fines, losses, damages, costs and expenses (including attorneys' fees, except to the extent such fees are determined by an arbitrator or court of competent jurisdiction to be unreasonable, in which case the fees shall be adjusted to be reasonable by such arbitrator or court).	Upon demand.	Payable if we incur losses due to your breach of the Franchise Agreement or any other action or inaction by you or any other person relating to your Franchised Business.
Reimbursement of Professional Fees and Expenses	Varies.	As Incurred.	You must reimburse us for any legal, accounting, or other professional fees that we incur as a result of any breach or termination of your Franchise Agreement. You must reimburse us if we are required to incur any expenses in enforcing our rights against you under the Franchise Agreement. In some circumstances in some states professional fees and expenses are only recoverable where franchisor is the substantially prevailing party.
Enforcement Expenses	Our actual cost of de- identifying your Franchised Business or complying with other post-term obligations on your behalf.	Upon demand.	Payable if your Franchise Agreement expires or is terminated and you fail to de- identify (or comply with other post-term obligations) for your Franchised Business and we take steps to do so.

Explanatory Notes:

1. General. The table above provides recurring or isolated fees or payments that you must pay to us or our affiliates or that we or our affiliates impose or collect in whole or in part on behalf of a third party or that you are required to spend under the Franchise Agreement. All fees and expenses described in this Item 6 are nonrefundable. ExceptWe occasionally may modify or waive certain fees. Otherwise, except as otherwise indicated provided in the above chart above, we uniformly impose all fees and expenses listed on a per franchise agreement basis and they are payable to us and are fully earned upon receipt by us.

We have the right to periodically specify (in the confidential operations manual or otherwise in writing) different payees and/or payment methods, such as, but not limited to, weekly/biweekly/monthly payment, payment by auto-draft, credit card, and payment by check.

All fees are current as of the Issuance Date of this Franchise Disclosure Document. April 19, 2024. Certain fees that we have indicated may increase over the term of the Franchise Agreement.

Manner of Payment. With the exception of the Initial Franchise Fee and the Local Advertising FeeRequirement Deposit, you must pay all fees and other amounts owed to us and/or our affiliates via electronic funds transfer ("EFT") through an electronic funds transfer program (the "EFT Program"), under which we automatically deduct all payments owed to us and/or our affiliates from the bank account you provide to us for use in connection with EFT Program (the "EFT Account").

2. Royalty Fee. You must pay us a monthly Royalty Fee deducted on the tenth (10th) business day at the end of each calendar month in an amount equal to the greater of: (i) the Royalty Fee (as delineated in the chart below); or (ii) the minimum monthly Royalty Fee ("Minimum Royalty Fee") (as defined in the chart below). We reserve the right to change the interval when you pay the Royalty Fee to us upon reasonable notice to you.

Royalty Fee. Your Royalty Fee is calculated as follows:

Monthly Royalty Fee	Gross Revenue for Prior Month	
7.0%	\$0.00 - \$100,000.00	
6.0%	\$100,000.01 - \$300,000.00	
5.0%	\$300,000.01+	

For clarity purposes, if your Gross Revenue for the prior month is \$125,000, you will pay 7.0% for the first \$100,000 in Gross Revenue Generated and 6.0% for the remaining \$25,000 in Gross Revenue generated. The graduated Royalty Fee rate applies on a per franchise agreement basis, and multiple territories may not be aggregated together to qualify for a lower Royalty Fee rate. The Royalty Fee for newly Approved Products or Approved Services may be different (either higher or lower) but will not be increased by more than 3% of gross revenue (and this 3% may be allocated between the Royalty Fee and the Brand Fund Contribution).

Minimum Monthly Royalty Fee. Your Minimum Monthly Royalty Fee is based upon: (i) the number of months the Franchised Business has been open and operating; and (ii) the number of Designated Territories that the Franchised Business operates in. The Franchised Business shall be considered open and operating on the day that you complete the Initial Training Program (defined in Item 11). Your Minimum Monthly Royalty Fee of \$250300 (if you operate a single Franchised Business in one Designated Territory) begins four (in month 4) months after the date you complete the Initial Training Program. The Minimum Monthly Royalty Fee is calculated as follows:

Time of Operation	1 Franchised Business	2 Franchised Businesses	3 Franchised Businesses	4 Franchised Businesses	5 Franchised Businesses
First 3 Months	7% - 5% of Gross Revenue	7% - 5% of Gross Gross Revenue Revenue		7% - 5% of Gross Revenue	7% - 5% of Gross Revenue
4 to 12 Months	\$250	\$500	\$750	\$1,000	\$1,250
Second Year	\$500	\$1,000	\$1,500	\$2,000	\$2,500
Third Year	\$750	\$1,500	\$2,250	\$3,000	\$3,750
Fourth Year +	\$1,000	\$2,000	\$3,000	\$4,000	\$5,000
4 to 12 Months	\$300	<u>\$600</u>	<u>\$900</u>	\$1,200	\$1,500
Second Year	\$1,500	\$3,000	<u>\$4,500</u>	<u>\$6,000</u>	<u>\$7,500</u>
Third Year +	<u>\$2,000</u>	<u>\$4,000</u>	<u>\$6,000</u>	\$8,000	\$10,000

However, you will not be subject to a Minimum Monthly Royalty Fee during the first three (3) months of operationsyou are open and operating on the condition that you strictly comply with all of your obligations under the Franchise Agreement during the first three (3) months of operationsyou are open and operating. Notwithstanding the foregoing, you must still pay the Royalty Fee equal to five to seven percent (7% - 5%)% of Gross Revenue generated by your Franchised Business for sales made during the first three (3)3 months you are open and operating.

For each additional Franchised Business above 5 units:

- the First 3 months-is 7-5% of Gross Revenue
- 4-12 months is an additional \$300 per Franchised Business
- Second Year is an additional \$1,500 per Franchised Business
- Third Year + is an additional \$2,000 per Franchised Business

Conversion Royalty Fee

If you have an existing business that is the same or similar to a Franchised Business with annual gross revenue of at least \$500,000 and you agree to convert the existing business into a Franchised Business, we will discount your royalty fee during your first year of operation. Your Royalty Fee during the first year of operation will be as follows:

Months of Operations	Royalty Fee
Months 1-4	25% of then-current Royalty Fee
Months 5-8	50% of then-current Royalty Fee
<u>Months 9-12</u>	75% of then-current Royalty Fee
Months 12+	The then-current Royalty Fee

Gross Revenue Definition and Reports. For purposes of this Agreement, "Gross Revenue" is defined to include all income of any type or nature and from any source that Franchisee derives or receives directly or indirectly from, through, by or on account of the operation of the Franchised Business (whether or not the products/services are approved by the Franchisor) at any time after

the signing of the Franchise Agreement, in whatever form and from whatever source, including, but not limited to, cash, services in kind from barter and/or exchange, on credit or otherwise, and convenience, credit card user, or other fees charged to a customer, as well as business interruption insurance proceeds, all without deduction for expenses, including marketing expenses and taxes. Exclusions do not include reductions for financing program fees, credit card convenience fees or user fees, or any other fees charged to a customer). However, the definition of Gross Revenue does not include sales tax that is collected from customers and actually transmitted to the appropriate taxing authorities or any bona fide refunds you make to customers in the ordinary course of business. We reserve the right to modify our policies and practices regarding revenue recognition, revenue reporting, and the inclusion or exclusion of certain revenue from Gross Revenue as circumstances, business practices, and technology change.

FranchiseeYou must send Franchisorus a signed report ("Gross Revenue Report(s)") on or before the tenth (10th) day of each month for the immediately preceding calendar month, in the manner and form specified by Franchisorus. Each Gross Revenue Report must set forth: (i) Franchisee'syour Gross Revenue generated during the previous calendar month; (ii) Franchisee'syour calculation of the Royalty Fee and Brand Fund Contribution (defined in Section 3.5); and (iii) any other information Franchisorwe may require. FranchisorIn lieu of the Gross Revenue Report, we currently collect Gross Revenue data through our digital platform system but we reserve the right to require a Gross Revenue Report be submitted to us as described above in the future upon notice. We may change the form, content, and/or interval of the Gross Revenue Report from time to time and/or require Franchiseeyou to submit Gross Revenue Reports upon notice to Franchiseeyou.

3. Local Advertising Requirement. We reserve the right to require that all or a portion of the Local Advertising Requirement be paid to us or an Approved Supplier for products or services that we or a third party provide or for use in marketing channels that we require you to use.

For purposes of the Franchise Agreement and this disclosure document, the term "Business Judgment" means the exercise of reasonable business judgment in making Franchisor's decision or exercising Franchisor's rights. Franchisor's decisions or actions will be deemed to be the result of reasonable business judgment, even if other reasonable or even arguably preferable alternatives are available, if Franchisor's decision or action is intended, in whole or significant part, to promote or benefit the franchise System generally, even if the decision or action also promotes Franchisor's financial or other individual interest. Examples of items that will promote or benefit the franchise System include, without limitation, enhancing the value of the Proprietary Marks, improving customer service and satisfaction, improving product quality, improving uniformity, enhancing or encouraging modernization, and improving the competitive position of the franchise System. This is a defined term for the purposes of the Franchise Agreement and this disclosure document and is not intended to incorporate principles related to the application of the business judgment rule in a corporate law context.

If you default on the payment of your Franchisor Directed Advertising Fee, we reserve the right, among other remedies, to: (a) suspend or refuse production of local advertising services ("Local Advertising Services") until such payment has been made in full; (b) require that you enter into a payment plan for any past due amounts; and/or (c) require that you prepay a minimum of one (1) month of Franchisor Directed Advertising Fees for a period of time that we determine. We may periodically increase or decrease the Franchisor Directed Advertising Fee in an amount we deem appropriate in our Business Judgment (defined in Item 5), but in no event will that increase cause the Franchisor Directed Advertising Fee to be more than the Local Advertising Requirement. We reserve the right to change the Franchisor Directed Advertising Fee from a dollar amount to a

percentage of sales or to any other formula we designate upon thirty (30) days' notice by email or other means.

ITEM 7 ESTIMATED INITIAL INVESTMENT

A. YOUR ESTIMATED INITIAL INVESTMENT FOR A NEW FRANCHISED BUSINESS

Type of	Am	ount	Method of	When Due	To Whom
Expenditure ⁽¹⁾	Low	High	Payment	when Due	Payment is to be Paid
Initial Franchise Fee ⁽²⁾	\$54 <u>59,</u> 90 0	\$ <mark>68<u>73</u>,90</mark>	Lump Sum	Upon execution of the Franchise Agreement	Us
Franchisor Directed AdvertisingOpeni ng Assistance Fee Deposit ⁽³⁾	\$6 <u>5</u> ,000	\$ <u>65</u> ,000	Lump Sum	Upon execution of the Franchise Agreement	Us
Travel Expenses to Training ⁽⁴⁾	\$250 <u>1,00</u> <u>0</u>	\$3,000	As Incurred	As Incurred	Airlines, hotels and restaurants
Security Deposits and Rent ⁽⁵⁾	\$0	\$ <mark>123</mark> ,000	As Incurred	As Incurred	Landlord, utility company
Office Furniture ⁽⁶⁾	\$0	\$ <mark>82</mark> ,000	As Incurred	As Incurred	Third-party vendors
Vehicle and Vehicle Enhancements ⁽⁷⁾	\$3,000	\$15,000	As Incurred	As Incurred	Third-party vendors, us and other third parties
Tools and Equipment ⁽⁸⁾	\$5,000	\$30,000	As Incurred	As Incurred	Third-party vendors, and other third parties
Business Management and Technology System ⁽⁹⁾	\$1,000	\$3,000	As Incurred	As Incurred	Third-party vendors, us and other third parties
Initial Inventory ⁽¹⁰⁾	\$ <u>5</u> 1,000	\$ <u>155,000</u>	As Incurred	Prior to Opening	Third-party vendors, us and other third parties

Type of	Am	ount	Method	WILLIAM	To Whom
Expenditure ⁽¹⁾	Low	High	of Payment	When Due	Payment is to be Paid
Technology Fee (3 Months) ⁽¹¹⁾	\$1,797	\$1,797	As Incurred	As Incurred	Third-party vendors, us and other third parties
Online Local Presence Fee (12)	<u>\$747</u>	<u>\$747</u>	As Incurred	As Incurred	Us and third- party vendors
Marketing Management Fee (13)	<u>\$1,500</u>	<u>\$1,500</u>	As Incurred	As Incurred	Us and third- party vendors
Industry Certifications, Licenses, and Trainings(1214)	\$0	\$ <u>\$10</u> ,000	As Incurred	Prior to Opening	Third-party vendors, us, government agencies, and other third parties
Professional Fees ^(13]5)	\$500	\$2,000	As Incurred	As Incurred	Attorneys, bankers, accountants and other professionals
Insurance Deposit and Initial Premiums(4416)	\$1,000 <u>2,5</u> <u>00</u>	\$4 <u>5</u> ,500	As Incurred	Prior to Opening	Insurance agent or carrier
Grand Opening Marketing ⁽⁺⁵¹⁷⁾	\$ 10 15,00	\$ 15 <u>22</u> ,00	As Incurred	As Incurred	Third-party vendors, us or and other third parties
Additional Funds (3 Months)(16.18)	\$15,000	\$30,000	As Incurred	As Incurred	Employees, utilities, suppliers, us and other third parties
TOTAL ESTIMATED INITIAL INVESTMENT ⁽	\$103,447 112,944	\$219,197 <u>2</u> 13,444			

Explanatory Notes:

- 1. Generally. Although you are not obligated to obtain an Approved Location until twelve (12) months after you execute your Franchise Agreement, the chart above assumes that you operate your Franchised Business from either a leased or owned commercial office/warehouse, which is approximately 1,000 to 1,500 square feet or operate from a home office with a storage space. All fees and payments described above are non-refundable, unless otherwise stated or permitted by the payee. Actual costs will vary for each franchise location depending on a number of factors, including market conditions and the geographic location of your Franchised Business.
- 2. <u>Initial Franchise Fee.</u> Upon execution of your Franchise Agreement, you must pay to us an Initial Franchise Fee of \$5459,900 which you must pay in a lump sum. If you purchase a territory that includes more than 80,000 Qualified Households, then you will be required to pay us \$0.70 for each additional Qualified Household, up to an additional 20,000 Qualified Households. The additional household fee also applies to multi-unit franchise purchases. The Initial Franchise Fee is non-refundable and deemed fully earned upon execution of your Franchise Agreement.
- 3. Franchisor Directed Advertising Opening Assistance Fee Deposit. Simultaneously with execution of your Franchise Agreement and the payment of your Initial Franchise Fee, you must pay us a two month Local Advertisingan Opening Assistance Fee Deposit in an amount equal to \$65,000. The Local AdvertisingOpening Assistance Fee Depositrelates to opening assistance services that include: the cost of attending the initial training program, access to the learning management paths, and certain meals during the Initial Training Program, any on-site visits that the operational and support team will make, the initial cost of marketing setup, which includes, but is not limited to, the initial creation of collateral, website, and other marketing setup items (but excludes the ongoing cost that may be applied towards paid to us, vendors, or affiliates of additional collateral and other marketing channels), the initial cost of setup with miscellaneous software and technology utilized through the life of the business but does not cover the ongoing cost of software and technology that you may use in your first two (2) months of Franchisor Directed Advertising Fees and is deemed a pro-rated eredit towards your total required Local Advertising Requirement. The Directed Advertising Fee Depositbusiness, and our administrative cost of assisting with of initial legal and compliance guidance that we may provide in our sole discretion. This fee is refundable on a prorated basis if the a single fee for one Franchise Agreement is terminated prior to completing two months of operationor multiple franchise agreements under a single Multi-Unit Addendum. Each Multi-Unit Addendum will have a separate Opening Assistance Fee. The Opening Assistance Fee is not refundable.
- 4. Travel Expenses to Training. We estimate that your travel expenses for initial training will be between \$2501,000 and \$3,000. While we do not charge you and up to two (2) additional persons to attend the phases of our Initial Training Program (defined in Item 11), you are required to pay for transportation to and from our Initial Training Program and pay all expenses associated with lodging, meals and other miscellaneous expenses during the time of training (including any employee wages). The range assumes, on the low end, that no additional people other than you will attend the Initial Training Program, and, on the high end, two (2) additional people will attend our Initial Training Program with you. Please see Item 11 of this Franchise Disclosure Document for additional information. Any amounts that may be collected by the Franchisor/affiliate in the future are amounts payable to third party vendors.
- 5. <u>Security Deposit and Rent</u>. Although you are not obligated to obtain a brick-and-mortar location within the first twelve (12) months of operation (in which case you would have \$0 expense at opening and until you sign a lease), if you choose to lease an office or warehouse or storage space

location to start, you will need an office/warehouse location with sufficient space, initially between 1,000 and 1,500 square feet. The amount of your security deposit and rent (a storage space at the start may be as small as 100 square feet). We have assumed that you will rent a storage space or warehouse space during the first year. If you rent an office/warehouse, your costs may be as high as \$12,000 for the first three months. The amount of your security deposit will vary depending on your landlord, occupancy rate, length of lease, personal vs. corporate signature, and your personal financial history. The amount of your rent will vary depending on the landlord, the location, square footage, length of the lease, and general market conditions.

- 6. Office Furniture. In addition to your vehicle, you will have an office even if you are operating from a home office during the first 12 months. You are required to establish a functional office with a desk, file cabinet, chairs, miscellaneous office supplies, etc. We estimate that your office equipment, furniture and supplies will cost up to \$2,000, depending on what items you already have that are available for use. If you already have sufficient office furniture, your opening expense for these items would be \$0.
- 7. <u>Vehicle and Vehicle Enhancements</u>. You must purchase a vehicle wrap or other branded vehicle signage which displays our then-current marks from our approved supplier and have the vehicle used by you or your technician in operating the Franchised Business branded according to our specifications. We reserve the right to require you to lease or purchase a-½ ton truck (with a branded vehicle wrap) or a ¾ ton truck (with branded signage). Although it is not currently required, we strongly encourage you to lease or purchase one of the vehicles that we currently recommend. <u>We have assumed that you own or will lease a vehicle</u>.

An estimate has been included for a vehicle to be used in connection with the Franchised Business in the event that you do not already own a personal vehicle that you can use for business purposes. The actual expense may vary depending on factors such as the type of vehicle chosen, whether you purchase or lease, and if purchased, how much is financed. We recommend, and our standard offering assumes, that you do not purchase a vehicle and pay for it in full. If you decide to purchase a vehicle, the high-end cost could be an additional \$40,000 or more depending on the type of vehicle you purchase.

- 8. Tools and Equipment. You are required to purchase certain types of equipment, including tools for performing Approved Services and also purchasing such work equipment from our approved suppliers, manufacturers, and/or subject to our specifications. We reserve the right for us or our affiliates to become an approved supplier or manufacturer. A list of tools is available upon request.
- Business Management and Technology System. You must purchase the Business Management and Technology System (Item 11.F below) comprised of the computer hardware and software we designate for use in connection with your Franchised Business prior to commencing operations. You must purchase the components of the Business Management and Technology System from third-parties or approved suppliers. Approved Suppliers. The Business Management and Technology System may include, but is not limited to: (i) a business-class laptop computer with performance capabilities sufficient to run all of the software required to operate your Franchised Business; (ii) a laser printer meeting our standards and specifications; (iii) updated versions of QuickBooks, Microsoft Office, and antivirus software; (iv) a tablet with Internet access and a smartphone; and (iv) fees related to the managed service desk required to operate the Franchised Business. This includes setup fees and licensing fees for the first three months of operation. The Business Management and Technology System includes software and related services that include fees that are paid through the Technology Fee (i.e., ServiceMinderour digital platform system, Managed Service Desk Fee, among others). The low end of this estimate assumes that you already

- have some or all of the required hardware or software. The high end assumes that you will need to purchase all of the components. The estimated amount for this section does not include the Technology Fee, which is set forth separately in the Item 7 table.
- 10. <u>Initial Inventory. Generally, materials for projects are purchased on a per project basis rather than keeping materials in stock for use on a project.</u> You are required to obtain an initial inventory of materials to meet the demand in your Designated Territory. The low end of this estimate represents the minimum amount of inventory that you are required to purchase while the high end of the estimate represents the largest amount of inventory that we recommend you purchase. <u>Initial inventory consists of post caps, gate hardware, bulk nails, and similar items.</u>
- 11. <u>Technology Fee (3 Months)</u>. The Technology Fee currently includes fees related to your access to and usage of our reservation system, intranet, any mobile applications we develop, and the System website. We may add, delete, or otherwise modify the products and services that are included in the Technology Fee. There is no cap on the amount the Technology Fee may be increased. The first month will be assessed pro rata from the date on which you begin offering Approved Products and/or Approved Services and collected on the 10th of the following month.
- Online Local Presence Fee. The Online Local Presence Fee currently includes the following: website management, development, updating, maintenance, enhancement, and brand reputation management.
- Marketing Management Fee. The Marketing Management Fee currently includes the following localized marketing support: email marketing, social media, promotions and offers, creative support, public relations management, tradeshow support, and paid media support.
- 12.14. Industry Certifications, Licenses and Training. You must acquire any third-party certifications required by us. You must acquire a general business license, any specialty licenses required by your state or federal agency. If your jurisdiction has additional licensing laws or other laws or regulations that require additional instruction or supervised inspections or other criteria, such as examinations or peer review, that require individualized effort in order to comply with the laws or regulations and we or our affiliate cannot provide such instruction/services to you, you must retain a third-party licensing compliance service provider to assist you in navigating the required criteria. In some jurisdictions, you are required to apprentice for one or more years in order to get a license to perform services. You may incur costs to pay a person who has such a license in order to perform services. These costs can range from \$1,000 to \$3,000 per month or may be based on a percentage of each project's cost. In some jurisdictions, industry certifications, licensing, and/or training may not be required, in which case your cost would be \$0.
- 13-15. Professional Fees. This estimate is based on the fees necessary to create a franchisee entity and retainingretain legal counsel and accountants to do initial corporate filings, and set up bookkeeping and bank accounts.
- 14.16. Insurance Deposit and Initial Premiums. We estimate that your initial insurance deposit and initial premiums will be approximately \$1,000 to \$4,500.2,500 to \$5,500. We assume that you will pay monthly or quarterly. The average annual cost is approximately \$9,000-\$12,500. You should check with your local agent for actual premium quotes and costs, as well as the actual cost of the deposit. The cost of coverage will vary based upon the area in which your Franchised Business will be located, your experience with the insurance carrier, your loss experience, your level of sales, and other factors beyond our control. In some states the cost could be higher depending on availability, competition and broker coverage. You should also check with your insurance agent or broker

regarding any additional insurance that you may want to carry beyond what we require you to obtain.

- 45.17. Grand Opening Marketing. In connection with the opening of the Franchised Business, you must spend a minimum of \$10.00 for grand opening advertising and promotion in the 30 to 60 days prior to opening the Franchised Business and the 300 days after opening the Franchised Business in accordance with a plan that you must submit to us. We have the right to modify your grand opening plan, in our sole discretion, and may require you to use a public relations or other advertising firm to assist with your grand opening. No amount paid by you for your grand opening will be credited toward your Local Advertising Requirement. You must provide us with supporting documentation evidencing these expenditures upon request. If you enter into Multi-Unit Addendum with us, you will only be required to spend \$10.500 in total for Grand Opening Advertising for all Franchised Businesses so long as they are located in contiguous Designated Territories. We reserve the right to increase the required spend to \$1522,000 as we deem necessary.
- 46.18. Additional Funds (3 Months). The estimate of additional funds between \$15,000 and \$30,000 is for a period of at least three (3) months, and is based on an owner-operated business and does not include any allowance for an owner's draw or salary. This estimate includes ongoing fees that you will be required to pay throughout your operation of the Franchised Business such as Royalty Fees, local advertising expenses, Call Center Fees, Brand Fund Contributions, etc. This estimate does not include any fees paid for debt services. These figures are estimates and we cannot guarantee that you will not have additional expenses in starting the business. If you are converting your business to a Franchised Business, you may already have equipment, insurance and other items and this amount may be lower.
- 47.19. Total. This is an estimate of your initial start-up expenses for one (1) Franchised Business. These estimates are based on our experience in offering and selling franchises, as well as the experience of our existing franchisees and estimates we have received from third-party vendors.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

B. YOUR ESTIMATED INITIAL INVESTMENT FOR A MULTI-UNIT OFFERING

Type of Expenditure ⁽¹⁾	Offering	Amo	ount	Method of Payment	When Due	To Whom Payment is to be Paid
		Low	High			10 00 1 414
	2-Pack	\$ 9 4 <u>99</u> ,900	\$ 122 <u>127</u> ,900	Lump	Upon	Us
Multi-Unit	3-Pack	\$ 129 <u>134</u> ,900	\$ 171 <u>176</u> ,900	Sum	execution	US
Fee ⁽²⁾	4-Pack	\$ 164 <u>169</u> ,900	\$ 220 225,900		of the Multi-Unit	
	5-Pack	\$ 189 <u>194</u> ,900	\$ 259 264,900		Addendum	
Initial Investment to Open Initial Franchised Business ⁽³⁾		\$4 8,5 47 <u>53,044</u>	\$ 150,297 <u>139,544</u>	See Item 7, Chart A-Lump Sum or As Incurred	Upon Execution of Franchise Agreement, Prior to Opening, or As Incurred	Us; Approved Suppliers; Airlines, third-party suppliers; and employees
	2 Pack	\$ 3,000	\$ 3,000	Lump Sum	Upon execution of the Multi Unit Addendum	Us -
Additional Initial Franchisor	3-Pack	\$ 6,000	\$6,000	Lump Sum	Upon execution of the Multi- Unit Addendum	Us-
Directed Advertising Fee ⁽⁴⁾	4-Pack	\$ 9,000	\$ 9,000	Lump Sum	Upon execution of the Multi Unit Addendum	Us
	5 Pack	\$12,000	\$ 12,000	Lump Sum	Upon execution of the Multi Unit Addendum	Us

Type of Expenditure ⁽¹⁾	Offering	Amo	ount	Method of Payment	When Due	To Whom Payment is to be Paid
		Low High				
	2-Pack	\$ 146,447 <u>152,944</u>	\$ 276,197 <u>267,444</u>	This is the total estimated initial investment to enter into a Multi-Unit		
	3-Pack \$184,447		\$328,197 <u>316,444</u>	Addendum for the right to develop each of the multi-unit offerings we		
Total ⁽⁵⁴⁾	4-Pack	\$222, 447<u>944</u>	\$380,197 <u>366,444</u>	typically offer under a Addendum, as well as		
	5-Pack	\$ 250,447 <u>247,944</u>	\$ 422,197 <u>404,444</u>	open and commence operating your initial Franchised Business for the first three (3) months.		0.5

Explanatory Notes:

- Generally. All fees and payments are non-refundable, unless otherwise stated or permitted by the
 payee. This Chart details the estimated initial investment associated with executing a Multi-Unit
 Addendum for the right to own and operate between two (2) and five (to 5) Franchised Businesses,
 as well as the initial investment to open your initial Franchised Business.
- 2. Multi-Unit Fee. You will be required to execute a Franchise Agreement for each of the Franchised Businesses you are granted as part of your Multi-Unit Offering, as well as our prescribed form of Multi-Unit Addendum, all at the same time. Upon the execution of these agreements, you will be required to pay us a Multi-Unit Fee in the amount of \$9499,900 for the right to operate two (2) Franchised Businesses, \$129134,900 for the right to operate three (3) Franchised Businesses, \$164169,900 for the right to operate four (4) Franchised Businesses, and \$189194,900 for the right to operate five (5) Franchised Businesses, which must be paid in a lump-sum at the time you sign the Multi-Unit Addendum. This includes the Multi-Unit Fee if you elect not to increase the number of households within your Designated Territory (Low) and if you increase to the maximum number of additional households (High). It does not include the Opening Assistance Fee, which is included in the Initial Investment to Open Initial Franchised Business. If you wish to increase the number of Qualified Households in your Designated Territory, then you will be required to pay us an additional \$0.70 per Qualified Household up to an additional 20,000 Qualified Households per Designated Territory. The Multi-Unit Fee is deemed fully earned and non-refundable upon payment.
- 3. <u>Initial Investment to Open Initial Franchised Business.</u> This figure represents the total estimated initial investment required to open the initial Franchised Business under the Multi-Unit Addendum. You will be required to enter into our current form of Franchise Agreement to govern each Franchised Business you are granted the right to operate at the same time you sign your Multi-Unit Addendum. The range includes all the items outlined in Chart 7(A) of this Item, except for the \$5459,900 Initial Franchise Fee because, upon full payment of the Multi-Unit Fee, you will not be required to pay any Initial Franchise Fee in connection with your Franchised Businesses.
- 4. Additional Initial Franchisor Directed Advertising Fee. You will be required to execute a Franchisor Directed Advertising Agreement. The Initial Franchisor Directed Advertising Fee for the first Franchise Agreement is included in the "Initial Investment to Open Initial Franchised Business" section. The incremental Initial Franchisor Directed Advertising Fee for each Franchise

Agreement is \$1,500 per month times 2 months. A second Franchise Agreement adds \$1,500 x 2 months = \$3,000. Each additional franchise agreement adds \$3,000 to the Initial Franchisor Directed Advertising Fee deposit.

5.4. Total. Please note that this row does not include the initial investment you will need to undertake in connection with opening your second and any other additional Franchised Businesses you are granted the right to open under your Multi-Unit Addendum. The amounts listed in this Section include the initial investment to open your Initialinitial Franchised Business set forth in Note 2 immediately above, as well as the applicable Multi-Unit Fee (which is substituted for the Initial Franchise Fee that would otherwise be due with each franchise agreement). The amounts listed assume that all territories are contiguous. If the territories are not contiguous, then for each additional noncontiguous territory or group of noncontiguous territories, the grand opening marketing fee will increase from a low of \$1015,000 to a high of \$1522,000.

ITEM 8 RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

You must operate your Franchised Business in conformance with our methods, standards and specifications which we prescribe in our confidential and proprietary operations manual (the "Operations Manual") and various other confidential manuals, writings, and other information prepared by us for your use in operating a Franchised Business (together with the Operations Manual, the "Manuals"). We may periodically change our standards and specifications at our sole discretion, and you must comply with all changes.

Approved Products and Approved Services

All Approved Products that we may designate, Approved Services, Approved Location/Storage Facility, vehicles, supplies, equipment, tools, uniforms, forms, advertising materials, computer hardware and software, and inventory used by you in connection with the Franchised Business must meet our thencurrent System standards and specifications, including, but not limited to, branding requirements (including color and label requirements), which we will establish and modify at our discretion. You may incur an increased cost to comply with such changes at your own expense.

We reserve the right to require you to purchase any of the items or services necessary to establish and operate your Franchised Business in accordance with our standards and specifications and/or from an Approved Supplier, from us, our affiliate(s), or our designated vendors and suppliers.

You must offer for sale all Approved Products and Approved Services that we may later designate and Approved Services which we prescribe, and only those products and services which we prescribe. You may not offer any other products or services for sale without having received our prior written authorization. You must at all times maintain sufficient levels of inventory as specified in the Operations Manual to adequately satisfy consumer demand. You must offer, use and sell all private label products which we may now or in the future designate for sale by System franchisees. Private label products, to the extent required, are included in the Initial Inventory expense set forth in the Item 7 table.

Some suppliers may provide us with test equipment for use in our training center, advertise in our newsletters, and may also sponsor events and/or rent booth space at our Annual Conference or regional meetings.

Approved Suppliers

If establishedCurrently, we will beare the Approved Supplier for the Call Center_and marketing services and reserve the right to be the Approved Supplier for other products and services in the future. You must purchase your Initial Inventory, Business Management and Technology System—(defined below in Item 11), branded items, signage and marketing materials, computer equipment and software, insurance, grand opening marketing, and other ongoing marketing from one of our Approved Suppliers (defined in the following paragraph) or in accordance with our standards and specifications. Currently, our affiliates are not Approved Suppliers, but we reserve the right to require you to purchase additional items or services from our affiliates in the future.

Recognizing that preservation of the System depends upon product and service uniformity and the maintenance of our trade dress, you must purchase product samples and other supplies, services, computer hardware and software, and other equipment from us or from approved or designated suppliers that we will specify from time to time in the Operations Manual and otherwise in writing (each an "Approved Supplier"). We, our affiliate, or a designated third party may be one of several, or the only, Approved Supplier of any item. We are not currently the only Approved Supplier of any item. We reserve the right to require you to purchase any products and services, including equipment, supplies, computer hardware and software directly from us or our affiliate. We and our affiliates have the right to realize a profit or otherwise derive revenue on any products or services that we, our affiliates, or our Approved Suppliers supply and/or provide to you.

We may establish business relationships, from time to time, with suppliers who may produce and/or provide certain products or services that you are required to purchase from only that supplier (each a "System Supplier"). These System Suppliers may provide, among other things, supplies, fixtures, technology, software, and equipment, all in accordance with our proprietary standards and specifications or private label goods that we have authorized and prescribed for sale by System Franchised Businesses.

Ownership Interest in a Supplier / Revenue Derived from Regional Franchise Purchases and Leases

Except where we are the approved supplier for items you must purchase, none of our officers <a href="https://www.except.com/ww

We may derive income in the form of rebates or marketing allowances paid to us by Approved Suppliers that we require you to use. During our last fiscal year ended December 31, 2023, Franchisor received 2022, we did not receive any rebates of \$12,982 and did not derive any Franchisor derived revenue from franchisees' required purchases; of Call Center Services and Marketing Fees of \$29,285. During the fiscal year ended December 31, 20232022, our affiliates did not derive any revenue from our franchisees' required purchases. The total revenue of \$42,267 represents .9% of Franchisor's total revenue.

Your obligations to purchase certain products or services from us or our Approved Suppliers and to purchase products, services, supplies, fixtures, equipment, computer hardware and software, training and real estate that meet our specifications are considered "Required Purchases." We estimate that your Required Purchases will account for approximately 70% to 95% of your total purchases and leases incurred in establishing your Franchised Business, and approximately 70% to 95% of your total ongoing purchases and leases to operate the Franchised Business after the initial start-up phase.

Alternative Product or Supplier Approval

If you wish to purchase any unapproved item, including inventory, and/or acquire approved items from an unapproved supplier, you must provide us the name, address and telephone number of the proposed supplier, a description of the item you wish to purchase, and the purchase price of the item, if known. At our request, you must provide us, for testing purposes, a sample of the item you wish to purchase. We are not required to approve any particular product or supplier. We may base our approval of any such proposed item or supplier on considerations relating not only directly to the item or supplier itself, but also indirectly to the uniformity, efficiency and quality of operation we consider necessary or desirable in our System as a whole, as well as the maintenance of our Confidential Information (defined below in Item 14). We reserve the right to receive payments from suppliers on account of their dealings with you and other Franchised Businesses and to use all amounts we receive without restriction (unless instructed otherwise by the supplier) for any purposes we deem appropriate. We are not required to approve an unreasonable number of suppliers for a given item if we believe that such approval may result in higher costs or prevent the effective or economical supervision of approved suppliers.

You will be required to pay us a \$1,000 alternative supplier fee/new product review fee plus our actual cost of the inspection and our actual cost of testing the proposed product or evaluating the proposed service or service provider, including personnel and travel costs to review any alternate supplier. We will use reasonable efforts to notify you in writing if your request is approved or denied within 30 days of: (i) our receipt of all supporting information from you regarding your request; and (ii) our completion of any necessary inspection or testing associated with your request. If we do not provide written approval within this time period, then your request will be deemed denied.

We may, but are not required to, provide your proposed supplier or provider with its specifications for the item that you wish the third party to supply, provided that the third party executes our required non-disclosure agreement form. Each supplier that we approve of must comply with our requirements regarding insurance, indemnification and non-disclosure. If we approve any supplier, you may enter into supply contracts with that third party, but under no circumstances will we guarantee your performance of any supply contract.

We may reinspect and revoke our approval of particular products or suppliers if we determine that such products or suppliers no longer meet our standards. Once you receive written notice from us that we have revoked our approval, you must immediately cease purchasing products from that supplier.

We do not provide any material benefit to you if you buy from sources we approve, but we may default you under (or terminate) your Franchise Agreement, or otherwise deny your request to renew these agreements, based upon your failure to make required purchases from our Approved Suppliers or otherwise in accordance with our standards and specifications.

Approved Location

Upon execution of your Franchise Agreement and for a period of twelve (12) months after the Effective Date of your Franchise Agreement, you may operate the Franchised Business from a home office or storage facility that is approved by us and otherwise meets our standards and specifications.

After this initial twelve (Within 12)-month period months after the Effective Date, you must secure a leased office/warehouse or storage facility within the Designated Territory that meets our standards and specifications and is approved by us. The Approved Location must include an office anda storage space for equipment and inventory and an office unless otherwise approved in writing by us. The Approved Location must typically be between 1,000 and 1,500 square feet.

We must review and approve any proposed location, as well as any lease associated with the proposed location, prior to your entering into any lease for the proposed location. You must secure the Approved Location within thirty (30) days 12 months of executing the Franchise Agreement in the event that you have not already obtained our approval prior to executing the Franchise Agreement. We may provide you with standards and specifications for the design and layout of the premises of the Approved Location.

Advertising and Promotional Materials

We must approve all self-generated advertising materials (<u>advertising materials created by you)</u> prior to publication or use.

Insurance

You must purchase and maintain, at your own expense, insurance covering the operation and location of your Franchised Business as we may require. You must purchase the required insurance at least thirty (30) days before opening your Franchised Business or upon signing a lease for the Approved Location. The limits described in the paragraph below are the minimum amounts that you are required to purchase. If you sign a lease or contract that requires a higher amount than provided below, then you must obtain the higher level of coverage under the terms of the lease or contract. If you sign a lease or contract that does not require as much coverage, you must still purchase enough insurance to meet our requirements.

You must obtain the insurance coverage that we require from time to time as presently disclosed in the Manuals and as we may modify it. All insurance policies required under your Franchise Agreement and as set forth in the Franchise Agreement (excluding workers' compensation) must name us, our officers, directors, shareholders, partners, agents, and representatives as additional insureds. The insurance policies must include a provision that the insurance carrier must provide us with no less than thirty (30) days' prior written notice in the event of a material alteration to, or cancellation of, any insurance policy. A certificate of insurance must be furnished by you to us at the earlier of ninety (90) days after the Effective Date of the Franchise Agreement or prior to the commencement of our Initial Training Program (defined in Item 11). Insurance coverage must be at least as comprehensive as the minimum requirements set forth in the chartparagraphs below and in the Franchise Agreement (Article 9 of the Franchise Agreement). You must consult your carrier representative to determine the level of coverage necessary for the Franchised Business. Higher exposures may require higher limits. Presently, our insurance requirements are as follows:

- A. Comprehensive general liability insurance extended to include contractual liability, products and completed operations, and personal and advertising injury, with a combined bodily injury and property damage limit of at least \$1,000,000 per occurrence, \$2,000,000 in aggregate;
- B. Statutory workers' compensation insurance and employers' liability insurance as required by the law of the state in which the Franchised Business is located, including statutory workers' compensation limits and employers' liability limits of at least \$1,000,000; (include coverage of independent contractors who do not have worker's compensation);
- C. Commercial vehicle liability insurance with limits of not less than \$1,000,000 combined single limit for all owned, hired, and non-owned vehicles you, your employees, or any other person performing work on your behalf, operate;

 D. Business interruption insurance equal to twelve (12) months of your net income and continuing expenses, including Royalty Fees;

- E.D. Commercial umbrella liability insurance with a total liability limit of at least \$1,000,000;
- F.E. Employment practices liability insurance with a limit of at least \$500,000,250,000 (increasing to \$500,000 if you have 5 or more employees), including actions of a third party, and a minimum of \$25,000 for wage and hour disputes;
- G.F. Cyber liability insurance for financial losses arising from unauthorized access, loss or corruption of data, including, but not limited to, privacy and data security breaches, misdirected funds, virus transmission, denial of service, and loss of income from network security failures, with a minimum limit of \$500250,000 per claim and in the aggregate; and
- H.G. All other insurance that we require in the Operations Manual or that is required by law or by the lease or sublease for the Franchised Business.
- I.H. For all subcontractors and independent contractors that you use, we recommend the following insurance requirements:
- i. The subcontractor(s) and/or independent contractor(s) shall carry the same limits and coverages that are carried by you;
- ii. You shall be named as an additional insured under the subcontractor(s) and/or independent contractor(s) policy(ies);
- iii. The subcontractor(s) and/or independent contractor(s) policy(ies) shall carry a waiver of subrogation endorsement as well as a primary and non-contributory endorsement; and
 - iv. Any policy shall provide a thirty (30) days' notice of cancellation.

We must approve all insurance carriers in advance and in writing. All insurance policies must be issued by insurance companies with a rating of A-VII or better, as reported in the most recent edition of A.M. Best's Insurance Reports. Our acceptance of an insurance carrier does not constitute our representation or guarantee that the insurance carrier will be capable of meeting claims during the term of the insurance policy.

Computer Hardware and Software

You must purchase any computer hardware, software and peripherals that meet our System standards and specifications. Please see Item 11 of this Franchise Disclosure Document for additional information regarding our computer hardware and software purchasing requirements. You must obtain, maintain, and use the business management and technology system that we specify periodically in the Manuals to: (i) enter and track purchase orders, and receipts and customer information; (ii) update inventory; (iii) enter and manage your customer's contact information; (iv) generate sales reports and analysis relating to the Franchised Business, and (iv) provide other services relating to the operation of the Franchised Business ("Business Management and Technology System").

The Business Management and Technology System currently includes a laptop computer, a tablet computer, a printer and a smartphone device. Specifications for the brand, operating capabilities, and functionality of these hardware components will be set forth in the Manuals and is subject to change. At a minimum, the components of the Business Management and Technology System must be connected to the Internet via a high-speed Internet connection, and must be able to run our designated software programs

and general business software such as email, word processing, and similar programs.

The Business Management and Technology System will use third-party software from our approved vendors, including our digital platform system, for point-of-sale functions, email marketing, and all customer management functions. For any proprietary software or third-party software that we require you to use, you must execute and be responsible for the fees associated with any software license agreements or any related software maintenance agreements that we or the licensor of the software require.

Purchasing and Distribution Cooperatives; Rebate Programs

We currently do not have any purchasing or distribution cooperatives; however, we reserve the right to establish these types of cooperatives in the future. We may negotiate certain purchase arrangements (including price terms) for the purchase of certain items with suppliers for the benefit of franchisees.

ITEM 9 FRANCHISEE'S OBLIGATIONS

The table lists your principal obligations under the Franchise Agreement and other agreements. It will help you find more detailed information about your obligations in these agreements and in other items of this disclosure document.

	Obligation	Section in Franchise Agreement	Section in Multi-Unit Addendum	Item of Disclosure Document
a.	Site selection and acquisition / lease	Section 7.1	Not Applicable	Items 7, 11 and 12
b.	Pre-opening purchases / leases	Sections 7.1, 7.3, 7.4, 7.6, 7.8 and 7.11	Section 4	Items 7 and 8
c.	Site development and other pre-opening requirements	Sections 7.1 and 9	Not Applicable	Items 6, 7, 8 and 11
d.	Initial and ongoing training	Sections 7.2 and 8	Section 6	Item 11
e.	Opening	Section 7.3	Not Applicable	Item 11
f.	Fees	Sections 2.2.9, 3, 6, 7.4.3.1, 7.6.5.1, 8.1.2, 8.1.3, 8.2, 8.3, 9.5, 11.1, 12.3, 12.5.3,14.3.2.8 22.8, and the Data Sheet, and Exhibit H	Sections 2, 3, 7, 8 and 9	Items 5, 6, 7 and 11
g.	Compliance with standards and policies / operations manual	Sections 6, and 7	Not Applicable	Items 8 and 11

	Obligation	Section in Franchise Agreement	Section in Multi-Unit Addendum	Item of Disclosure Document
h.	Trademarks and proprietary information	Sections 4, 5, 7.8 and 7.14	Not Applicable	Items 13 and 14
i.	Restrictions on products / services offered	Sections 1.2 through 1.8, 7.4 and 7.5	Not Applicable	Items 8, 12 and 16
j	Warranty and customer service requirements	Section 7.18	Not Applicable	Item 15
k.	Territorial development and sales quotas	Not Applicable	Not Applicable	Items 12 and 17
1.	Ongoing product / service purchases	Sections 7.4 and 7.5	Not Applicable	Items 8 and 11
m.	Maintenance, appearance and remodeling requirements	Sections 7.1.2, 7.15 and 7.17	Not Applicable	Items 6, 8, 11 and 12
n.	Insurance	Section 9	Not Applicable	Items 7 and 8
0.	Advertising	Section 12	Not Applicable	Items 6 and 11
p.	Indemnification	Section 13.2	Not Applicable	Item 6
q.	Owners' participation / management / staffing	Sections 7.6.3 through 7.6.5 and 7.10	Not Applicable	Items 11 and 15
r.	Records and reports	Sections 10 and 11	Not Applicable	Item 6
s.	Inspections and audits	Sections 7.7 and 11	Not Applicable	Items 6 and 11
t.	Transfer	Section 14	Section 10	Item 17
u.	Renewal	Section 2.2	Section 11	Item 17
v.	Post-term obligations	Sections 16.1 and 17.2	Not Applicable	Item 17
w.	Noncompetition covenants	Section 17	Not Applicable	Item 17
х.	Dispute Resolution	Section 18	Not Applicable	Item 17

ITEM 10 FINANCING

We do not offer direct or indirect financing to you. We do not guarantee your note, lease or other obligations.

ITEM 11 FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEM AND TRAINING

Except as listed below, we are not obligated to provide you with any assistance.

A. Pre-Opening Obligations

Before you open your Franchised Business, we (or our designee(s)) will provide you with the following assistance:

- 1. We will define your Designated Territory (Franchise Agreement, Section 1.2). We do not generally own a premises which is then leased to a franchisee.
- 2. We will provide you with site selection guidelines and general specifications and standards, including any specific requirements (Franchise Agreement, Exhibit E). We consider some or all of the following factors when determining whether or not a site meets our approval: general location and neighborhood, size, physical characteristics of existing buildings, and lease terms. Upon execution of your Franchise Agreement and for a period of 12 months after the Effective Date of your Franchise Agreement, you may operate the Franchised Business from a home office or storage facility that is approved by us and otherwise meets our standards and specifications. By the end of the initial 12-month period, you must secure a leased office/warehouse or storage facility within the Designated Territory that meets our standards and specifications and is approved by us. We may extend this deadline, in our sole discretion, which we may condition on you executing a general release in our favor (Franchise Agreement, Section 7.3). The Approved Location must include an office and storage space for equipment and inventory. The Approved Location must include an office and storage space for equipment and inventory. The Approved Location must typically be between 500 and 1,500 square feet. Failure to secure a leased office/warehouse or storage space is a default under the Franchise Agreement and may result in termination of the franchise agreement.
- 3. We will provide initial training in the System and our policies and procedures to your required trainees (Franchise Agreement, Section 8).
- 4. We will provide you with either a written copy or electronic access to our Operations Manual and other manuals, on loan as long as the Franchise Agreement or successor franchise agreement remains in effect. The table of contents for the Operations Manual is attached as Exhibit GJ to this Franchise Disclosure Document. The Operations Manual is currently 16961 pages. (Franchise Agreement, Section 6.1).
- 5. We will provide you with a list of approved and recommended suppliers and a list of equipment that you will need to operate your Franchised Business. We do not deliver or install any of these tools or pieces of equipment (Franchise Agreement, Section 6.2).
- 6. We will provide advice and guidance, as we deem necessary in our sole discretion, in preparing you to open your Franchised Business, including standards and procedures for obtaining inventory and supplies, providing approved services, advertising and promoting the business, and otherwise operating the Franchised Business during the start-up phase (Franchise Agreement, Section 6).

 We will initiate the Local Advertising Services described in Item 8, subject to your payment of the applicable fees (Franchise Agreement, Section 12.5.3).

We will approve your Franchised Business opening, provided that you have met all of our requirements for opening, including completion of initial training, obtaining a vehicle that meets our requirements, and purchase of the Initial Inventory and all necessary tools and equipment. We estimate that the typical length of time between signing a Franchise Agreement and opening your Franchised Business is approximately thirty (30) to ninety (90) days. Factors affecting this length of time include, among others: hiring of the requisite employees; successful completion of initial training; local ordinances or community requirements; obtaining a vehicle that meets our requirements; issuance of all necessary licenses, permits and approvals; and procuring required insurance. You must select and secure your Approved Location no later than thirty (30) days 12 months after the effective date of the Franchise Agreement, and must open the Franchised Business no later than ninety (90) days after the effective date of the Franchise Agreement. WeThe Approved Location may extend this deadline, in our sole discretion, which we may condition on you executing be a general release in our favor (Franchise Agreement, Section 7.3), home office or storage space for the first 12 months. You must successfully comply with and complete all of your pre-opening obligations (including identifying and agreeing on an Approved Location) and open your Franchised Business within 90 days of executing your Franchise Agreement or we may terminate your Franchise Agreement upon notice to you (Section 7.3 of the Franchise Agreement), and retain the Initial Franchise

98. We will provide you with a dedicated phone number which you must use in connection with your Franchised Business and in all marketing items (Franchise Agreement, Section 7.11).

We will not provide general business or operations training to your employees or independent contractors; however, we may provide limited training on the System and brand standards to your key employees. You will be responsible for training your employees and independent contractors, including any training on the day-to-day operations of the Franchised Business. You will be responsible for hiring, training, directing, scheduling, and supervising your employees and independent contractors in the day-to-day operations of the Franchised Business. We do not provide the above services to renewal franchisees and may not provide all of the above services to franchisees that purchase existing Franchised Businesses.

B. <u>Training</u>

Initial Training

You (or your operating principal if you are an entity) and your Designated Manager (defined below in Item 15) (the "Required Trainees") must attend and satisfactorily complete to our satisfaction the initial training program before you open your Franchised Business. Initial training currently consists of approximately five (5) days of training to be held at our Home Office located at 107 Parr Drive,in Huntersville, North Carolina, and other location(s) designated by us near Charlotte, North Carolina at least thirty (30) days prior to opening your Franchised Business ("Initial Training Program"). We may provide on-site follow-up training, as we, in our sole discretion, deem necessary to be conducted four (4) to six (6) weeks after your Franchised Business opens for business. We will conduct initial training at least six (6) times per year (approximately every other month), but may conduct training sessions more frequently in our discretion. We reserve the right to modify the length, location and timing of initial training. We may waive a portion of the Initial Training Program or alter the training schedule if we determine that your trainese have sufficient prior experience or training. The Initial Training Program will be provided as soon as practicable after you sign your Franchise Agreement(s) and must be completed within 90 days after the effective date of the Franchise Agreement (Franchise Agreement, Section 8).

We will provide instructors, facilities and materials for initial training (currently \$0) for up to three (3) of your representatives (including your Required Trainees) for the training fee, provided that all of your trainees are trained during the same training session. If space is available, you may bring more than three (3) representatives to initial training. We reserve the right to charge a training fee of \$500, which we may increase upon sixty (60) days' written notice to you, for training: (i) each person in excess of three (3) trainees, even if they attend the same training session; (ii) each person who is repeating the course or replacing a person who did not pass; and (iii) each subsequent operating principal, Designated Manager (defined in Item 15), or employee who attends the course at a scheduled training class at our headquarters or other location designated by us. You are responsible for any travel and living expenses, wages, and other expenses incurred by your trainees during the Initial Training Program or any other training program.

Our Initial Training Program currently consists of the following. We reserve the right to vary the length and content of the Initial Training Program based upon the experience and skill level of the individual attending the Initial Training Program. Some or all of the classroom training may now or in the future be provided virtually.

TRAINING PROGRAM

TRAINING PROGRAM						
Subject	Hours Of Classroom Training	Hours Of On-The-Job Training ⁽¹⁾	Location			
History and Culture 90 Day Sprint	.75 2_	0 1	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Structuring Your OrganizationBusiness Planning & Discussions	<u>0</u> .5	θ <u>1.5</u>	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Compliance & RoyaltiesFundamentals of Marketing	.5 0	θ <u>2</u>	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Fencing Overview	4	θ	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Intro to ToolsCompliance and GearRoyalties	<u>-51.25</u>	0	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Fence Types & Accessories Estimating Overview	4 <u>3</u>	θ <u>1.5</u>	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Hand on Training / Fence Types & AccessoriesFencecloud Software Exercises and Discussions	1	θ <u>3</u>	Our Home Office (Huntersville, NC), virtually, or other location we designate			

TRAINING PROGRAM						
Subject	Hours Of Classroom Training	Hours Of On-The-Job Training ⁽¹⁾	Location			
TechnologyFencing Overview	<u>2.253</u>	0	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Safety Consideration / Hazards	4	θ	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Installation Procedures Techniques and Exercises	÷ <u>3.25</u>	<u>14</u>	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Estimating	2	8	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Sketching PrinciplesOrdering and Delivery	<u>0</u> .5	<u>80</u>	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Ordering Supplies Safety	<u>.51.25</u>	0	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Product KnowledgeService Minder Software and Exercises	<u>24.8</u>	<u>01</u>	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Sales Techniques and Discussions and Role-Play	7. <u>\$25</u>	0 3.2	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Marketing	3.75	θ	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Common Issues on the Job TR Sub-Contractors	1	4 <u>0</u>	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Hand-on Training / Fence Types & Accessories	2	θ	Our Home Office (Huntersville, NC), virtually, or other location we designate			
90 DAY Plan Preparation	2	0	Our Home Office (Huntersville, NC), virtually, or other location we designate			
Leadership and Next Steps/Building Your Team	1.5 <u>0</u>	θ <u>2</u>	Our Home Office (Huntersville, NC), virtually, or other location we designate or other location we designate			

TRAINING PROGRAM					
Subject	Hours Of Classroom Training	Hours Of On-The-Job Training ⁽¹⁾	Location		
Totals	31.75 <u>29</u> Hours	18 19 Hours			

^{.(1)} At our sole discretion, we may provide up to 20 hours of additional "On-The-Job" training or consulting at no cost to you after your completion of the Initial Training Program.

Explanatory Notes:

- The training subjects may vary, and the training may be less than the times indicated above, depending on the number and experience of the attendees. We use Manuals, PowerPoint presentations, online videos and tutorials, a learning management system, and other materials as instructional materials in our Initial Training Program and other training programs. In addition to the above training, we may, in our discretion offer or require additional On-The-Job Training through "ride-a-long" training sessions with experienced franchisees in or around Huntersville, NC or other locations we designate. We expect there to be approximately 8 hours (one day) allocated to this training.
- 2. The instructors for our Initial Training Program and other training programs all have experience working with us or similar businesses. The following individuals will lead our training programs: (i) Todd Bingham, President of Top Rail, who has been with us since our inception and has over tenll years of experience in franchise operations; (ii) Chris Herman, who has been with us since our inception and has over five6 years of experience in our industry, (iii) Tucker Berry, who has been with us since our inception and has over five6 years of experience in the industry; (iv) Greg Smith, Vice President of Learning and (ivDevelopment since November 2022, who has over 30 years' experience in sales and training; and (v) Chris Stefanco, Vice President of Operations of HFB since March 2022, who has over 2021 years of experience in franchise operations, training and business coaching andin the services business industry and who has been with us since our inception. We may also use certain vendors and current franchisees to assist in the training program.
- 3. Your Required Trainees must successfully complete the Initial Training Program at least thirty (30) days before opening your Franchised Business. We will determine, in our discretion, what constitutes successful completion of the Initial Training Program. If your Required Trainees are unable to successfully complete, in our sole discretion, the Initial Training Program for any reason, your Required Trainees must repeat the Initial Training Program, or you must send replacement Required Trainees to complete the Initial Training Program. If your Required Trainees have not, in our sole discretion, successfully completed the Initial Training Program thirty (30) days before the opening of your Franchised Business, we may terminate the Franchise Agreement, in which case we will not refund any initial fees paid by you.

Additional Training

We may periodically conduct mandatory or optional training programs for your Required Trainees and/or your employees at our office or another location that we designate. There will be no charge for training programs that we require you or your employees to attend, but we may charge you our then-current fee for optional training programs. We may provide additional training in person or via recorded media, teleconference, videoconference, the Internet, webinar, or any other means, as we determine. We may require your Required Trainees or employees to satisfactorily complete any additional training programs

that we specify. We may require your Required Trainees to participate in refresher or advanced training each year.

If, in our sole judgment, you fail to maintain the quality and service standards set forth in the Manuals, we may, in addition to all of our other rights and remedies, assign trainers to the Franchised Business to retrain Franchised Business employees and restore service levels and/or require you or your employees to repeat initial training or attend additional training programs at a location that we designate. We may charge our then-current fee for each trainer assigned to your Franchised Business and any remedial training.

If your Designated Manager (defined in Item 15) ceases to be employed by you at the Franchised Business and you are unable to immediately appoint and train a new manager, we may, in our sole discretion, and for our then-current fee, provide a Designated Manager to work at your Franchised Business temporarily until a new Designated Manager is appointed and trained.

Training by You

You and/or your operating principal and your Designated Manager(s) are responsible for training all of your other employees (and subsequent Designated Managers) in accordance with our standards and training programs. If, in our sole judgment, you fail to properly train your employees or other personnel in accordance with our standards, we may prohibit you from training additional employees and either require them to attend training at our headquarters or pay for our costs and expenses to send one of our representatives to train them at your Franchised Business.

Delegation

We may delegate the performance of any or all of our obligations under the Franchise Agreement to an area representative, affiliate, agent, independent contractor, or other third party. If we appoint an area representative in the area that includes your Franchised Business, the area representative will provide training, support, marketing, and other services to you on our behalf and will have the authority to exercise many of our rights and perform many of our obligations under the Franchise Agreement.

In addition to participating in ongoing training, you will be required to attend an Annual Conference of all franchisees at a location we designate and pay a conference fee if we hold an annual meeting of all franchisees (See Item 6). We reserve the right to host an Annual Conference and charge you the thencurrent Annual Conference fee. The Annual Conference fee is currently \$1,000 for up to two (2) people to attend. We may charge our then-current fee (currently, \$500 per additional attendee), for additional attendees. We reserve the right to host up to 2 Annual Conferences per year. You are responsible for all travel and expenses for your attendees. If you do not attend the Annual Conference(s), you must still pay us the Annual Conference Registration Fee, regardless of the cause for non-attendance, unless you receive our advance written approval for such absence. This fee is charged on a per franchisee basis.

C. Post-Opening Obligations

After you open your Franchised Business, we (or our designee(s)) will provide you with the following assistance:

1. <u>Review Advertising</u>. We will review any advertising or promotional programs or materials that you develop (Franchise Agreement, Section 12.1).

- Administer Local Advertising. We will administer the Local Advertising Services
 described in Item 8 for the Franchisor Directed Advertising Fee set forth in Item 6 (Franchise Agreement,
 Section 12.5.3 and Franchise Agreement Exhibit H).
- 3-2. Brand Fund Management. We will manage the Brand Fund as described below in this Item 11. We will prepare an unaudited statement of contributions and expenditures for the Brand Fund and make it available within 120 days after the close of our fiscal year for the prior fiscal year only to franchisees who make a written request for a copy (Franchise Agreement, Section 12.3).
- 4.3. Requested Consulting Services. We will provide you additional consulting services with respect to the operation of the Franchised Business upon your reasonable request and subject to the availability of our personnel at a mutually convenient time. We will make available to you information about new developments, techniques and improvements in the areas of advertising, management, operations, and Franchised Business design. We may provide such additional consulting services through the distribution of printed or filmed material, an Intranet or other electronic forum, meetings or seminars, teleconferences, webinars, or in person. If such services are rendered in person other than at our offices, you must pay us our then current training fee and our expenses (Franchise Agreement, Section 6.3).
- 5.4. Additional Training. We may schedule, and require you, your Designated Manager (if applicable), and other employees, to attend additional or refresher training courses. We may charge you our then-current tuition fee for you and any other persons that attend such additional or refresher training, and you will be solely responsible for all expenses associated with such training (including travel, lodging, meals, and employee wages incurred). We will provide this training to you and your employees at our corporate headquarters, at any other training facility we designate, or virtually. We may also provide you with remedial training if we determine, in our sole discretion after conducting an audit or inspection of your Franchised Business, that you are not complying with our System standards and specifications. The purpose of remedial training is to get you back on track and in compliance with our standards and specifications (Franchise Agreement, Section 6.5). We may periodically conduct optional training programs for your Required Trainees and/or your employees at our office or another location that we designate. We may charge you our then current fee for optional training programs. We may provide additional training in person or via recorded media, teleconference, video conference, the Internet, webinar, or any other means, as we determine.
- 6.5. System Updates. We may, as we deem necessary in our sole discretion, modify and update the System and Manuals, including any standards and specifications, and provide you with updated lists of: (i) Approved Products and Approved Services; (ii) Approved Suppliers; and (iii) items you must purchase in accordance with our System standards and specifications (i.e., equipment, fixtures, inventory and supplies) (Franchise Agreement, Section 1.1 and Section 6.1).
- 7-6. Call Center- and Business Management and Technology System Updates. We reserve the right to establish and administer our Business Management and Technology System and a Call Center, as we deem advisable, to manage prospective and existing customers and route/assign work orders/inquiries as we deem necessary in our sole discretion (Franchise Agreement, Section 6.6), and Section 7.8.1).
- <u>8-7.</u> Evaluate Alternate Suppliers. We will review any alternate supplier or non-approved item you propose for use in connection with the Franchised Business, and subsequently approve or deny these proposals as disclosed more fully in Item 8 of this Franchise Disclosure Document (Franchise Agreement, Section 7.4.3.+).
- <u>9-8. Relocation Review.</u> We will evaluate sites to which you propose to relocate your Franchised Business in accordance with our then-current System Standards for Approved Locations. You must reimburse us for our reasonable costs and expenses associated with evaluating your relocation request

and/or any locations proposed by you for relocation. We will provide you with copies of our invoices for our expenses from any third-party providers upon request (Franchise Agreement, Section 7.1.1).

10.9. Ongoing Advice. We may provide periodic advice and guidance regarding the ongoing operation of your Franchised Business and/or the use of the Proprietary Marks and System in general, as we deem necessary or advisable in our sole discretion. Our advice and assistance may be provided through meetings, printed materials and/or other media that we may make available to you in the System from time to time, or otherwise by telephone, email or other manner of communication. In certain circumstances, we reserve the right to charge our then-current training fee in connection with providing such assistance and/or be reimbursed for our expenses in providing any on-site assistance (Franchise Agreement, Section 6.3).

11.10. Pricing. If we determine that we may lawfully require you to charge certain prices for products or services, certain minimum prices for products or services, or certain maximum prices for products or services, you must adhere to our pricing policies as set forth in the Manuals or otherwise in writing from time to time. Otherwise, you are solely responsible for determining the prices that you charge customers. You must provide us with your current price list upon our request. We reserve the right to negotiate arrangements with customers who have regional or nationwide facilities ("National Accounts"), including pricing, which will bind all Franchised Businesses providing services to such National Accounts. If you wish to perform services for a National Account and we permit you to do so, you may be required to honor pre-negotiated pricing and terms with these National Accounts. We may offer preferred customer plans that offer customers discount prices under certain terms and conditions. You are not required to offer these plans to customers; however, if you do elect to participate in our preferred customer plans, you must offer the discount prices set by the plans in accordance with the terms of the plan (Franchise Agreement, Section 1.4).

D. <u>Advertising and Marketing</u>

Local Advertising Services

You and we agree that as Franchisor, we have the right, in light of our marketing expertise and research regarding effective advertising of the System: (a) to formulate and design the Local Advertising Services (as defined in Item 6), local advertising services, including the service options and vendors made available to franchisees from time to time; (b) to develop and control all content for the Local Advertising Services (the "Advertising"); Requirement; and (c) to determine the uses and distribution of all such Advertisingadvertising. We have the right to administer each service, as well as other services we elect to offer in the future, which may include appointment generation platforms, billboards, radio ads, social media ads, internet advertising, etc. We can change or end a service at any time and add new services at any time. As noted in Item 8, you must sign the Franchisor Directed Advertising Services Agreement and purchase all Local Advertising Services from us unless we provide written permission otherwise. This advertising is generally directed to your Designated Territory. These funds are segregated by franchisee and used for advertising in each franchisee's local marketing area that includes its Designated Territory. The Franchisor Directed Advertising Fee that you pay is allocated to your franchised business and is not part of a general fund for advertising. We will provide unaudited financial reports relating to the Franchisor Directed Advertising Fees to you upon your request at least annually. This advertising is generally directed to your Designated Territory. We may require that all or a portion of the Local Advertising Requirement be paid to us or an Approved supplier for these products and/or services.

The Franchisor Directed Advertising Services Agreement (that you are required to sign and specifies that we will develop and control all content for the Local Advertising Services (the "Advertising")) will remain in effect indefinitely and will automatically terminate upon the expiration or termination of the Franchise Agreement. If you are in default of your obligations under the Franchisor Directed Advertising

Services Agreement or the Franchise Agreement, we may suspend some or all of the Local Advertising Services without terminating either agreement. However, we can also terminate the Franchisor Directed Advertising Services Agreement prior to the end of the term by written notice if: (i) you are in default under the Franchise Agreement or any other agreement with us or one of our affiliates; (ii) you are in default under the Franchisor Directed Advertising Services Agreement and the default continues for a period of ten (10) days after we deliver written notice of default; or (iii) we discontinue the Local Advertising Services. We have the right, in our sole Business Judgment (defined in Section 3.14 of the Franchise Agreement), to discontinue the Local Advertising Services in whole or in part and to require you to pay all or a portion of the amount of the Franchisor Directed Advertising Fee to one or more other services that we establish to supplement or take the place of the Local Advertising Services.

Brand Fund

We have established a creative Brand Fund (the "Brand Fund") for the common benefit of the System. You are currently required to participate in and contribute one percent (1%)2% of your Franchised Business's Gross Revenue; however, we reserve the right to increase the Brand Fund Contribution to three percent (3%)3% of your Gross Revenue (Franchise Agreement, Section 12.3).

The Brand Fund Contribution is presently payable on a monthly basis directly to the Brand Fund via EFT on the fifth (5th)10th day of each month for the sales made during the immediately preceding month. We reserve the right to use the Brand Fund Contributions, in our sole discretion, to develop, produce and distribute national, regional and/or local marketing materials and to create advertising materials and public relations materials which promote, in our sole judgment, the services offered by System franchisees (Franchise Agreement, Section 12.3). We may use the Brand Fund Contributions to satisfy any and all costs offor maintaining, administering, directing, preparing, and producing advertising, including the cost of preparing and producing Internet, television, radio, magazine, and newspaper advertising campaigns; the cost of direct mail and outdoor billboard advertising; the cost of soliciting National Accounts; the cost of public relations activities and advertising agencies; the cost of developing and maintaining an Internet website; personnel and other departmental costs for advertising that we internally administer or prepare; and building partnerships with national and regional brands. Not all System franchisees will benefit directly or on a pro rata basis from such expenditures (Franchise Agreement, Section 12.3). While we do not anticipate that any part of Brand Fund Contributions will be used for advertising that is principally a solicitation for franchisees, we reserve the right to use the Brand Fund Contributions for public relations or recognition of our brand, for the creation and maintenance of a website, a portion of which can be used to explain the franchise offering and solicit potential franchisees, and to include a notation in any advertisement indicating "Franchises Available" (Franchise Agreement, Section 12.3).

If we do not spend all Brand Fund Contributions by the end of each fiscal year, the funds will be carried forward into the next fiscal year. You must contribute to the Brand Fund regardless of amounts due from other System franchisees.

We will prepare on an annual basis; and will have available for you within 120 days of the end of the fiscal year, an unaudited statement of contributions and expenditures for the Brand Fund. Upon your written request, we will provide you with the statement, for the previous fiscal year only. There is no requirement that the Brand Fund be audited (Franchise Agreement, Section 12.3).

We reserve the right to incorporate the Brand Fund as a separate business entity. The Brand Fund is not a trust or our asset and we are not a fiduciary to you with respect to, or a trustee of, the Brand Fund or the monies therein, and we assume no obligation or liability to you with respect to the maintenance, direction or administration of the Brand Fund (Franchise Agreement, Section 12.3).

We may periodically assist franchisees to maintain high quality standards through customer surveys, customer interviews, and other similar initiatives ("<u>Surveys</u>"). The cost of such programs will be borne by the Brand Fund. The cost of these programs may be charged directly to you if your results from a Survey fall below System-established minimum standards for such Surveys, which will be determined at the time we conduct a Survey (Franchise Agreement, Section 12.3).

We have the sole right to determine how to spend contributions to the Brand Fund or any funds from any other advertising program, and the sole authority to determine the selection of the advertising materials and programs; provided, however, that we will make a good faith effort to expend such funds in the general best interests of the System on a national or regional basis. We are not required under the Franchise Agreement to spend any amount on advertising in your Designated Territory. Not all System franchisees will benefit directly or on a pro rata basis from our expenditures (Franchise Agreement, Section 12.3).

We reserve the right to be reimbursed from the Brand Fund Contributions for actual costs and overhead, if any, that we may incur in activities which are related to directing and implementing the Brand Fund (Franchise Agreement, Section 12.3).

Company or affiliate-owned Franchised Businesses may contribute to the Brand Fund, but they are not required to do so (Franchise Agreement, Section 12.3).

We reserve the right to suspend or terminate the Brand Fund at any time, and any surplus funds may only be used for marketing and advertising purposes until fully expended (Franchise Agreement, Section 12.3).

During our last fiscal year ended December 31, 20232022, we did not collect anycollected Brand Fund Contributions or make any expenditures from the. The Brand Fund—was used for production of advertising and promotional materials (70%), media placement (0%), administrative expenses (23%), and other items, including publicity efforts (7%).

Advisory Council

We currently do not have an advisory council, but reserve the right to establish an advisory council for the purpose of exchanging ideas and problem-solving methods, to advise us on expenditures for System-wide advertising, and to coordinate franchisee efforts (an "Advisory Council"). In the event you are elected and accept, you must participate actively in the Advisory Council and participate in all Advisory Council meetings as we require. We reserve the right to prepare and amend the governing documents for the Advisory Council from time to time as we deem necessary, and we will determine the topic areas to be considered by the Advisory Council. The Advisory Council shall act in an advisory capacity only, and we shall have the right to form, change or dissolve an Advisory Council at any time, as we deem necessary in our sole discretion (Franchise Agreement, Section 12.6).

Local Advertising Requirement

In addition to the Brand Fund Contributions, each month you are required to spend the greater of (i) three percent (3%)3% of your Gross Revenue generated during the immediately preceding calendar month; or (ii) \$3,000 per month on advertising and promoting your Franchised Business within the Designated Territory in accordance with our standards and specifications (the Local Advertising Requirement, as defined in Item 6).). We reserve the right to increase your Local Advertising Requirement to four percent (4%)% of Gross Revenue upon 90 days' prior written notice to you. If you wish to use any

advertising or promotional materials other than those currently approved for use by System franchisees, then you must submit the materials you wish to use to us for our prior written approval at least 30 days prior to your intended use or publication. We will use commercially reasonable efforts to notify you of our approval or disapproval of your proposed materials within fifteen (15) days of the date we receive the proposed materials from you. If you do not receive our written approval during that time period, the proposed materials shall be deemed disapproved. Once approved, you may use the materials unless we withdraw or revoke our approval, which we may do at any time with written notice. All advertising must prominently display the Proprietary Marks and must comply with any standards we establish as specified in the Operations Manual or in any other writing. We may require you to discontinue using any advertising or marketing material within a specified time frame, and at your own cost and expense (Franchise Agreement, Section 12.1).

Grand Opening Marketing

In connection with the opening of the Franchised Business, you must spend a minimum of ten thousand dollars (\$10\subseteq\$15,000) for grand opening advertising and promotion in the thirty (30) to sixty (60) days prior to opening the Franchised Business and the thirty (30)\u00e30 days after opening the Franchised Business in accordance with a plan that you must submit to us. We reserve the right to modify your grand opening plan, in our sole discretion, and may require you to use a public relations or other advertising firm to assist with your grand opening. No amount paid by you for your grand opening will be credited toward your Local Advertising Requirement. You must provide us with supporting documentation evidencing these expenditures upon request. If you enter into a Multi-Unit Addendum with us, you will only be required to spend ten thousand dollars (\$10\subseteq\$15,000) in total for Grand Opening Advertising for all Franchised Businesses so long as they are located in contiguous Designated Territories. We reserve the right to increase the required spend to fifteen thousand dollars (\$15\subseteq\$22,000) as we deem necessary (Franchise Agreement, Section 12.5.1).

Advertising Cooperatives

We currently do not have any advertising cooperatives and have no plans to form such cooperatives in the immediate future. However, you must join and actively participate in any organizations or associations of franchisees or advertising cooperatives that we establish or that are established at our direction for the purpose of promoting, coordinating and purchasing advertising in local, regional or national areas where there are multiple Franchised Businesses, and you must abide by the bylaws, rules and regulations duly required by such advertising cooperative, which we reserve the right to mandate or approve if and when we form such cooperative. If we form an advertising cooperative, we will have the right to determine how membership will be defined, whether company-owned or affiliate-owned Franchised Businesses will participate in the cooperative, and whether we, an affiliate, a franchisee or a third party will administer the cooperative. If you join an advertising cooperative, we or the advertising cooperative may require you to spend additional funds on marketing programs conducted by the advertising cooperative, which may be in addition to your Local Advertising Requirement. There is no cap on this potential spending obligation. The contribution amount is set by the cooperative, and may only exceed the Local Advertising Requirement of 4% of Gross Revenue if agreed to by 2/3 of the cooperative. If company-owned Franchised Businesses have controlling voting power (defined as 2/3 of the cooperative votes), then the contribution amount is limited to 2% above the maximum Local Advertising Requirement, unless 2/3 of the cooperative vote, excluding company owned Franchised Business votes, to approve a higher amount. If we form an advertising cooperative, we will make any governing documents available to you for your review.

Promotional Programs

You must participate in all Franchised Business promotional programs that we offer to franchisees. You must follow our guidelines concerning the acceptance and reimbursement of gift certificates, gift cards, coupons, corporate discounts, and other promotional programs as we set forth from time to time in the Manuals or otherwise in writing. You will not allow use of gift certificates, gift cards, or coupons (including Groupon and similar discounts) unless approved or offered by us.

E. Website and Internet Presence

You must have and maintain adequate hardware and software in order to access the Internet at the bit speed we require from time to time. We reserve the right, but not the obligation, to establish and maintain a website that provides information about the System and the products and services offered by Franchised Businesses. If we exercise our right to create such a website, we will have sole discretion and control over it. We also have the right, but not the obligation, to create interior pages on our website(s) that contain information about your Franchised Business and other System locations. If we do create such a page, we may require you to prepare all or a portion of the page for the Franchised Business, at your sole expense, and may require you to use a template that we provide (Franchise Agreement, Sections 12.2.1 and 12.2.2).

Unless you obtain our prior written consent, you are prohibited from establishing or maintaining a separate website, or otherwise maintaining a splash page or other presence on the Internet through any social networking site in connection with the operation of your Franchised Business, including, without limitation, Facebook, Twitter, LinkedIn, YouTube, Pinterest, Instagram, Snapchat, TikTok, or any other social media, referral and/or networking site that uses any variation of the Proprietary Marks or references the System. If you seek and obtain our approval to create a separate website or other web presence, you must: (i) establish and operate the website according to our standards and policies as we describe in the Operations Manual or otherwise in writing from time to time; and (ii) utilize any templates that we provide to you to create and/or modify your site(s). We may require you to update the content of any social media and/or networking site at the times and in the manner we decide (Franchise Agreement, Section 12.2.3).

We reserve the right to modify our policies regarding both our and your use of Internet websites as we deem necessary or appropriate for the best interests of the System (Franchise Agreement, Section 12.2.). We are currently the sole registrant of the domain name www.toprailfence.com and we will be the sole registrant of any other domain names we decide to register in connection with the System in the future. You are prohibited from registering any domain name that contains words used in, or similar to, any trademark or service mark owned or used by us or our affiliate, or any colorable variation thereof (including any abbreviation, acronym, phonetic variation or visual variation) (Franchise Agreement, Section 12.2.5).

F. <u>Computer Hardware and Software</u>

You must obtain, maintain and use the business Management and technology System that we specify periodically in the Manuals to: (i) enter and track purchase orders and receipts and customer information; (ii) update inventory; (iii) enter and manage your customer's contact information; (iv) generate sales reports and analysis relating to the Franchised Business; and (ivy) provide other services relating to the operation of the Franchised Business ("Business Management and Technology System").

The Business Management and Technology System currently includes a laptop computer, a tablet computer, a printer and a smartphone device. Specifications for the brand, operating capabilities, and functionality of these hardware components will be set forth in the Manuals and is subject to change. At a minimum, the components of the Business Management and Technology System must be connected to the

Internet via a high-speed Internet connection, and must be able to run our designated software programs and general business software such as email, word processing, and similar programs.

The Business Management and Technology System will use third-party software from our approved vendors, including our digital platform system for point-of-sale functions, email marketing, and all customer management functions. For any proprietary software or third-party software that we require you to use, you must execute and be responsible for the fees associated with any software license agreements or any related software maintenance agreements that we or the licensor of the software require.

We estimate that the Business Management and Technology System will cost between \$1,000 and \$3,000, which includes the cost of the hardware, software licenses, related equipment, and network connections, including related installation costs. We, or our approved suppliers, will act as vendors or suppliers of some or all of the components of the Business Management and Technology System.

You must maintain the Business Management and Technology System at your expense and must purchase any hardware or software maintenance or technical support programs that we require. You must replace, upgrade, or update the Business Management and Technology System as we may require from time to time. We will establish reasonable deadlines for implementation of any changes to our Business Management and Technology System requirements, but there are no contractual limitations on our right to require changes to the Business Management and Technology System.

We currently do not require you to enter into, or expect that you will need to enter into, any maintenance, updating, upgrading or support contracts related to the Business Management and Technology System. We, our affiliates, and third-party vendors are not obligated to provide you with any ongoing maintenance, repairs, upgrades or updates. Vendors may be able to offer optional maintenance, updating, upgrading, or support contracts to you, which we estimate may cost between \$200 and \$400 per year. You may also be required to use a Managed Services Desk for franchisee emails and software support, which is currently \$55 per month for franchisee and \$20 per month for staff. This fee may be increased or decreased upon thirty (30) days of notice to Franchisee. Franchisor does not have any contractual obligations to maintain, repair, update or upgrade franchisee's computerized system. This fee is currently included in the Technology Fee.

You, at all times, must give us unrestricted and independent electronic access (including user IDs and passwords, if necessary) to the Business Management and Technology System for the purposes of obtaining the information relating to the Franchised Business. You must permit us to download and transfer data via a high-speed Internet connection or such other connection that we specify on a real-time basis. There are no contractual limitations on our right to access data stored in the Business Management and Technology System, and we may program the Business Management and Technology System to automatically transmit data and reports about the operation of your Franchised Business to us. We are the sole owners of all databases, lists, templates, programs, and any other software components that have been created and/or customized by or for us using the Business Management and Technology System, or otherwise generated through your Franchised Business. Franchisor does not have any contractual obligations to maintain, repair, update or upgrade franchisee's computerized system.

ITEM 12 TERRITORY

Designated Territory

You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control.

Upon signing the Franchise Agreement, we will provide you an area in which you will have protected rights (the Designated Territory, as defined in Item 1).). The size of your Designated Territory may vary from other System franchisees based on the location and demographics surrounding your Approved Location. Typically, a Designated Territory will consist of approximately 80,000 Qualified Households, depending upon geography, demographics, and other factors. The demographics, geography, and other factors we use in defining your Designated Territory are based upon information provided to us by third-party sources that we select at our sole discretion. We do not guaranty the accuracy of this information. If your Designated Territory has more than 80,000 householdsQualified Households, your Initial Franchise Fee will increase from the baseline amount of \$5459,900 by \$0.70 for each additional Qualified Household, up to an additional 20,000 Qualified Households. The Initial Franchise Fee is deemed fully earned and non-refundable upon payment.

The boundaries of your Designated Territory may be described in terms of zip codes, streets, landmarks (both natural and man-made) or county lines, or otherwise delineated on a map. The sources we use to determine the population within your Designated Territory will be publicly available population information (such as data published by the U.S. Census Bureau or other governmental agencies and commercial sources).

Relocation of the Franchised Business

If you would like to relocate the Approved Location, you must receive our written consent and our approval, which will not be unreasonably withheld, provided: (i) the new Approved Location is satisfactory to us and within your Designated Territory; (ii) your lease, if any, for the new location complies with our then-current requirements; (iii) you comply with our then-current requirements for constructing, outfitting and furnishing the new Approved Location; (iv) the new Approved Location will not, as determined in our sole discretion, materially and adversely affect the Gross Revenue of any other Franchised Business; (v) you have fully performed and complied with each provision of the Franchise Agreement within the last three (3) years prior to, and as of, the date we consent to such relocation (the "Relocation Request Date"); "); (vi) you are not in default, and no event exists as of the Relocation Request Date; and (vii) you have met all of our then-current training requirements. If you lose your lease, you must secure our approval of another site and enter into a lease for the new approved site within \$\frac{90180}{90180}\$ days after you lose your site lease.

You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control. We will neither operate, nor award to another person a franchise to operate, another Franchised Business in your Designated Territory, nor will we service, or authorize others to service, customers in your Designated Territory, except in limited circumstances described below in this Item 12, and provided you are not in default under the Franchise Agreement.

You may not solicit customers outside of your Designated Territory without our prior written consent (which may be withheld for any reason). If you solicit or sell Approved Products or Approved Services outside of the Designated Territory without our prior written consent, in addition to all other rights and remedies available to us for a breach of the Franchise Agreement, we will have the right to terminate the Franchise Agreement.

Adjacent Territory Policy

You may have customer relationships, or your in-territory advertising may result in customers reaching out to you for Approved Services in an adjacent territory. You may provide Approved Services outside of your Designated Territory in adjacent open territory (i.e., not the Designated Territory of another

franchisee). This policy, so long as it is in effect and not withdrawn or modified, constitutes the required written consent to provide Approved Services in adjacent open territory. You may also provide Approved Services in the Designated Territory of another franchisee whose Designated Territory is adjacent to yours, but only if all of the following conditions are met:

- You must inform the adjacent franchisee of the opportunity before commencing services or agreeing to provide services to the client.
- After you inform the client that there is another franchisee that provides services where the Approved Services are to be performed, the client requests in writing that you provide the services rather than the adjacent franchisee. You must obtain written proof to demonstrate compliance with this requirement.
- 3. You provide a revenue share with the adjacent franchisee as agreed between the parties not to exceed three percent (3%)3% of the Gross Revenue for the work to be performed in the adjacent franchisee's Designated Territory. If there is no agreement as to a revenue share, the revenue share percentage will be set by us in an amount not to exceed three percent (3%):3%.
- 4. You receive prior written permission from us, which permission may be granted for each project or for projects generally in our absolute discretion. We may refuse to grant permission for any reason or no reason.

You and the adjacent franchisee may (but are not required to), by mutual written agreement, enter into a referral fee agreement where one franchisee refers business to the other franchisee for work in the other franchisee's Designated Territory.

Open Territory Transition. If you provide Approved Services to customers in an adjacent territory that becomes a Designated Territory of another franchisee, upon sale of that newly Designated Territory to another franchisee, you will be obligated to transfer customers that are in the newly purchased territory to the franchisee that purchased the newly Designated Territory. This obligation does not include "Work-in-Progress" at the time of the sale of the adjacent open territory. "Work-in-Progress" means any customer with whom you have a contract to perform Approved Services where you have not completed the Approved Services. An Approved Service is deemed completed when final payment is due pursuant tounder the contract. However, any Work-in-Progress is not subject to the revenue share requirement for providing Approved Services in the Designated Territory of another franchisee above. You must use your best, in good faith, the greatest possible efforts to complete any Work-in-Progress within six (6) months after the adjacent open territory is sold. Any leads, in an adjacent territory that becomes a Designated Territory of another franchisee, must be provided to the adjacent franchisee at no cost unless you and the adjacent franchisee have entered into a referral fee agreement. You are obligated to cooperate with the adjacent franchisee to ensure an orderly transition of customers to the adjacent franchisee. You are obligated to transfer a list of all past and present customers in the purchased territory to the adjacent franchisee.

This Adjacent Territory Policy is part of the Operations Manual and is subject to modification or elimination by a change to the Operations Manual.

National Accounts

We have the exclusive right to negotiate and enter into agreements or approve forms of agreements to provide services to any entity that owns or otherwise has responsibility for a building or common services in more than one location whose business is not confined to one particular franchisee's Designated

Territory, regardless of the contract amount of the services to be performed (a National Account). We or any party we may designate shall have the right to perform the services for the National Account within your Designated Territory. Any dispute as to whether a particular customer or account is a National Account will be determined by us, and our determination will be final and binding. You are not entitled to any right to compensation or consideration for work performed by others in your Designated Territory for National Accounts.

You may choose to participate in the National Accounts program ("National Accounts Program") to service National Accounts for which the ultimate client of the National Account is located in the territory consistent with the then-current National Accounts Program Participation Agreement. You acknowledge and agree that weWe or a third party service provider we designate may begin servicing National Accounts in the Designated Territory if: (i) we believe that you are not meeting our standards and specifications applicable to the National Accounts you are servicing or the National Accounts Program generally; (ii) we believe that you are in default of the Franchise Agreement; (iii) the National Account has requested that we or a third party it selects service its clients; (iv) you are not then participating in the National Accounts Program (or are not eligible to participate in the National Accounts Program); (v) you do not offer or are not authorized by us or the applicable licensing authority to offer the service to be provided; or (vi) you refuse to service the National Account.

We have the right to discontinue or terminate a National Account or the entire National Accounts Program. We may impose any restrictions and/or requirements as a condition to your acceptance of each National Account. We reserve the right to charge fees on National Account revenue. These fees, if charged, will be a required condition of participating in our National Accounts Program and these fees are subject to change annually. We may require you to conduct thorough background checks on all employees who will service any National Account and may require proof of compliance with this and other requirements at such intervals as we and/or the National Account dictates.

Rights Reserved By Us

Among other things, we and our affiliates also reserve the exclusive right to: (i) establish and operate, and license third parties the right to establish and operate, other Franchised Businesses using the Proprietary Marks and System at any location outside of the Designated Territory; (ii) acquire, merge with, engage in joint ventures with, or otherwise affiliate with, and thereafter own and operate and franchise others the right to own and operate any business of any kind, including businesses that offer products that are similar to those provided by your Franchised Business, within or outside your Designated Territory; (iii) open and operate, or license third parties the right to open or operate, businesses that offer products and services similar to the Franchised Business under marks other than the Proprietary Marks at any location; (iv) use the Proprietary Marks and System, and license others to use the Proprietary Marks and System, to engage in any other activities not expressly prohibited under your Franchise Agreement; (v) designate and service National Accounts; (vi) service, route and/or assign any and all customer work orders and inquiries received through our Call Center; and (vii) use the Proprietary Marks and System, and license others to use the Proprietary Marks and System, to engage in any other activities not expressly prohibited under your Franchise Agreement.

Alternate Channels of Distribution

We, our affiliates, or third parties may distribute our and our affiliates' products and services in your Designated Territory, including those already developed and those yet to be developed, through alternate channels of distribution that we may choose. These alternate channels of distribution will include, but are not limited to, the sale and distribution of the products and services via the Internet and through joint marketing with partner companies under terms and conditions that we deem appropriate. This does

not give you the right to: (i) distribute such products or services; or (ii) share in any of the proceeds that a party received through these alternate channels. We are not required to pay you for soliciting or accepting orders from inside your Designated Territory.

Restriction on Rights

You do not have the right to open additional Franchised Businesses, nor do you have any right of first refusal on any other location. You do not have the right to use the Proprietary Marks or the System at any location outside of the Designated Territory. You may not offer any Approved Products or Approved Services in wholesale, e-commerce, or other channels of distribution besides the retail operation of the Franchised Business within the Designated Territory without our prior written consent. Any media advertising or direct mail marketing that you conduct must be predominantly focused within your Designated Territory, unless we agree otherwise.

You may service customers outside of your Designated Territory, with our prior written approval, if customers are located in areas geographically contiguous to your Designated Territory and no other franchisee of ours has been awarded that territory, nor is the territory protected as an affiliate-owned business (an "Open Territory"), except for certain limited circumstances, if any, that are more fully set forth in the Operations Manual. You have no rights to use other channels of distribution (Internet, catalog sales, telemarketing, or other direct marketing) except as specifically approved by us in writing or in the Operations Manual.

You may only operate your Franchised Business and service customers within your Designated Territory, unless we have approved you to service customers in Open Territories or Designated Territories of other franchisees adjacent to yours. As long as you are in compliance with the Franchise Agreement, we will not, during the term of your Franchise Agreement, operate or grant others the right to operate any other Franchised Business within the Designated Territory.

If we grant you permission to operate in an Open Territory, we have the right to sell or assign it or any part of it at any time, without notice to you. We may revoke your rights to operate in any Open Territory at any time, even if we permitted you to service customers in that Open Territory in the past. You will not have a right of first refusal or option to buy a territory that was formerly designated as an Open Territory. Once an Open Territory is assigned to another System franchisee or affiliate-owned business, you will have no further rights to service customers in that Open Territory.

Multi-Unit Addendum

If you enter into a Multi-Unit Addendum, you will have the right to simultaneously open and operate a mutually agreed upon number of Franchised Businesses, each with its own Designated Territory. The total number of Franchised Businesses to be opened under your Multi-Unit Addendum and the Designated Territory for each Franchised Business will be dependent upon a number of factors such as: (i) your financialworking capital availability and operational abilities business acumen to operate multiple Franchised Businesses simultaneously; (ii) the location and demographics of the general area where we mutually agree you will be opening these locations; and (iii) other factors that we deem relevant in our sole discretion.

You must simultaneously execute our current Franchise Agreement for each Franchised Business that we grant you the right to open under your Multi-Unit Addendum. Your Designated Territory for each Franchised Business under your Multi-Unit Addendum will typically be contiguous geographic areas and you may operate all of the Franchised Businesses under your Multi-Unit Addendum from one Approved

Location located within one of your Designated Territories. We will use our then-current standards for accepting the Approved Location and designating Designated Territories.

You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control. We will neither operate, nor award to another person a franchise to operate, another Franchised Business in your Designated Territories under the MUA, nor will we service, or authorize others to service, customers in your Designated Territories under the MUA, except in limited circumstances described below in this Item 12, and provided you are not in default under any Franchise Agreement under the MUA.

You must continuously operate each Franchised Business under your Multi-Unit Addendum. If you fail to continuously operate a Franchised Business, we may terminate the Franchise Agreement for that particular Franchised Business and offer the Designated Territory to another System franchisee. We will not terminate any other Franchise Agreement under your Multi-Unit Addendum solely based upon your failure to operate a Franchised Business in a separate Designated Territoryunder one of the Franchise Agreements under your Multi-Unit Addendum. Other events of default which permit us to terminate the Franchise Agreement for one Franchised Business may permit us to terminate any other Franchise Agreement or other agreement between you and us.

ITEM 13 TRADEMARKS

You will have the limited right to use the following Proprietary Mark that is currently pending with the United States Patent and Trademark Office ("USPTO"), as well as any other Proprietary Mark we may now or in the future designate in connection with the System, provided you use these Proprietary Marks in accordance with our System standards and specifications:

Mark	Serial Registration No.	FilingRegistration Date	Status Register
TOP RAIL	97/579,2937,306,587	September 6, February 13, 20242022	Principal Pending

We have also filed to register the following marks on the Principal Register of the USPTO:

<u>Mark</u>	Serial No.	Filing Date	Register
TOP RAIL	98/520789	April 26, 2024	<u>Principal</u>

We do not have a federal registration for the Proprietary Mark listed in the table immediately above (serial number 97/579,293numbers 98/520789). Therefore, this Proprietary Mark does not have the same legal benefits and rights as a federally-registered trademark. If our right to use this Proprietary Mark is challenged, you may have to change to an alternative trademark, which may increase your expenses.

All required affidavits relating to the Proprietary Mark have been filed. There are currently no effective determinations of the USPTO, the Trademark Trial and Appeal Board, the trademark

administrator of any state or any court; no pending interference, opposition or cancellation proceedings; nor any pending material litigation involving the Proprietary MarkMarks. Currently, there is no litigation pending or otherwise that limits our ability to use or license the Marks to you or any other franchisee. There are no other agreements that will affect our right to use, and license you to use, the Proprietary MarkMarks in any manner material to the System and franchises offered in this Franchise Disclosure Document.

Your right to use the Proprietary MarkMarks granted under the Franchise Agreement is non-exclusive, and we retain the right, among others: (i) to use the Proprietary MarkMarks for selling products and services; (ii) to grant others licenses for the Proprietary MarkMarks, in addition to those licenses already granted to existing franchisees; and (iii) to develop and establish other systems using the same or similar Proprietary Marks, or any other proprietary marks, and to grant licenses or franchises in those systems without providing any rights to you.

All your usage of the Proprietary Marks and any goodwill you establish will be for our exclusive benefit and you retain no rights in the Proprietary Marks on the termination or expiration of the Franchise Agreement. You may not use the Proprietary Marks as a part of any corporate or trade name or as part of a domain name, or an electronic address, nor may you use any trade name, trademark, service mark, emblem or logo other than the Proprietary Marks, as we may designate. You must prominently display the Proprietary Marks on the items we designate, including signs and packaging materials. You must obtain fictitious or assumed name registrations we require or under applicable law. You must identify yourself as the owner of the Franchised Business by placing your name on the Franchised Business and on all checks, invoices, receipts, contracts and other documents that bear the Proprietary Mark, and on all printed materials your name must be followed by the phrase "a Top Rail franchisee" or any other phrase as we direct. Upon termination or expiration of the Franchise Agreement, you are required to de identify your Franchised Business, removing all of our Proprietary Marks, logos or other proprietary items that we specify.

You must immediately notify us of any information that you acquire concerning any use by others of names or marks which are the same, or confusingly or deceptively similar to, the Proprietary MarkMarks. At our request, you must assist us to protect and maintain our interest in the Proprietary MarkMarks, and we will pay or reimburse your out-of-pocket costs incurred in rendering such assistance, unless we are required to take action to protect our interests due to your wrongful acts or those of any person under your control.

We are not obligated to protect you from the right to continued use of the Proprietary MarkMarks. Although our right to pursue any third-party infringers of our Proprietary MarkMarks is optional, as a company policy, we may elect to aggressively protect our rights under the Proprietary MarkMarks. If at any time we consider it to be advisable (in our sole discretion) for us and/or you to modify or discontinue the use of any Proprietary MarkMarks and/or use one or more additional or substitute trade or service marks, you must comply with our directions within a reasonable time after receiving notice from us and at your expense. We are not obligated to reimburse you for the costs you incur in complying with our directions or the loss of revenue or expenses caused by any modification or discontinuance of a Proprietary MarkMarks. We are not required to protect you against third-party claims of trademark infringement or unfair competition, however, we reserve the right to assist in the defense of such matters.

You must immediately notify us of any apparent infringement or challenge to your use of any Proprietary MarkMarks, or any claim by any person of any rights in any Proprietary MarkMarks. You agree not to communicate with any person other than us, our attorneys of choice and your attorneys in connection with any such claim or challenge. If we choose to take over or control the defense of any claim or challenge, the cost of such defense will be paid by us, provided that if any claim or challenge is caused by your wrongful acts, we may request that you indemnify us for any claims or damages we incur. This includes paying all of our attorneys', experts' or other professional fees we may incur to defend any claim or

challenge resulting from any of your wrongful acts. In limited instances, if we take over any claim or challenge, we may reimburse you for the reasonable expenses you incur in connection with cooperating with us, as we deem necessary in our sole discretion. We may, but are not obligated to, take over control of litigation or proceedings and Franchisee will control unless and until we have chosen to assert control.

Top Rail Fence Corporation, led by John West, is currently doing business as Top Rail Fence in the following California cities: Elk Grove, Sacramento, Wilton, Folsom, Roseville, Davis, and Galt. This business may have superior common law rights to use "Top Rail" in these markets. We do not intend to sell franchises in these markets at this time.

Top Rail Fencing LLC, led by Dane and Bryce Hooven, is currently doing business as Top Rail Fencing in Northampton, PA 18067. This business may have superior common law rights to use "Top Rail" in this market. We do not intend to sell franchises in this market at this time.

Travis Hayworth operates a "Top Rail Fencing" business in Reeds, MO-Missouri. We believe this use violates our trademark rights. We intend to enforce our rights against this business, but if a court holds that such use is not infringing, we may not be able to use the "Top Rail" trademark in this market.

There is a business located in August, GA that operates a Top Rail Fencing business. We believe this use violates our trademark rights. We intend to enforce our rights against this business, but if a court holds that such use is not infringing, we may not be able to use the "Top Rail" trademark in this market.

There is a business located in August, Georgia that operates a Top Rail Fencing business. We believe this use violates our trademark rights. We intend to enforce our rights against this business, but if a court holds that such use is not infringing, we may not be able to use the "Top Rail" trademark in this market.

There is a business located in Severance, Colorado that operates a Top Rail Fence Company business. We believe this use violates our trademark rights. We intend to enforce our rights against this business, but if a court holds that such use is not infringing, we may not be able to use the "Top Rail" trademark in this market.

There is a business located in Dallas, Texas that operates a Top Rail Fence business. This business may have superior common law rights to use "Top Rail" in this market. We do not intend to sell franchises in this market at this time.

There is a business operated by Jason C. Sarvis, located in Myrtle Beach, South Carolina that operates Top Rail Fence Co., <u>Inc.</u> This business may have superior common law rights to use "Top Rail" in this market. We do not intend to sell franchises in this market at this time.

ITEM 14 PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION

We do not own any registered patents or copyrights that are material to the franchise, however, we claim common law copyright and trade secret protection for several aspects of the System, including our Operations Manual, advertising and business materials. Additionally, we do not have any patent applications that are pending and/or material to this offering.

We own all right, title and interest in and to any Advertising (and related campaign or consumer data) created in or for the Local Advertising Services. You will not acquire any right, title or interest in the Advertising and/or related campaign or consumer data, other than the right to access and use the

Advertising, as expressly granted in the Franchise Agreement, subject to the terms and conditions of the Franchise Agreement and the Franchisor Directed Advertising Services Agreement.

There are no current determinations, proceedings or litigation involving any of our copyrighted materials. Should you become aware that any unauthorized third party is using any of our copyrighted materials, we request that you notify us of this unauthorized use. We may revise any of our copyrighted materials in our discretion, and may require that you cease using any outdated copyrighted material. You will be responsible for printing any revised or new advertising, marketing or other business materials.

During the term of the Franchise Agreement, you will receive information which we consider to be our trade secrets and confidential information, including, but not limited to, information regarding the setup of a Franchised Business; information about proprietary products—and, methods; any proprietary, software we may now or in the future create; our Operations Manual; trade secrets; price marketing mixes related to the sale of products or services offered or authorized for sale by System franchisees; standards and specifications for equipment, design and equipment layout; systems and operations/know-how, financial and pricing information about the franchised business; training—manuals; training systems; compensation systems; and marketing strategies; online marketing systems; sales systems; sales training; location identification and acquisition; general operations; our copyrighted materials; and methods and other techniques and know how concerning the operation of the Franchised Business which may be communicated to you or of which you may be apprised by virtue of your operation of a Franchised Business (collectively, the "Confidential Information").

You must not, during or after the term of the Franchise Agreement, communicate, divulge, or use for the benefit of any other person, partnership, association, corporation or limited liability company, any Confidential Information, except to your employees that must have access to it in order to operate the Franchised Business. Certain additional information, including: (i) current customer and prospective customer names and addresses; (ii) information about credit extensions to customers; (iii) customer service purchasing histories; (iv) rates charged to customers; and (v) sources of suppliers and purchasing arrangements with suppliers, also constitute our trade secrets and Confidential Information. Any and all information, knowledge, know-how, techniques, and other data which we designate as confidential will be deemed Confidential Information for purposes of the Franchise Agreement. We have expended considerable time, effort and money to develop the System, and the Confidential Information is not well known outside of the System. The Confidential Information is of great value to us, and we are implementing this non-disclosure policy in an effort to protect our trade secrets and Confidential Information.

If you, your employees or principals develop any new concept, process or improvement in the operation or promotion of the Franchised Business, you must promptly notify us and provide us with all necessary related information, without compensation. Any such concept, process or improvement will become our sole property and we will be the sole owner of all patents, patent applications, trademarks, copyrights and other related intellectual property rights. You and your principals will assign to us any rights you may have or acquire, including the right to modify the concept, process or improvement, and otherwise must waive and/or release all rights of restraint and moral rights. You and your principals agree to assist us in obtaining and enforcing the intellectual property rights to any such concept, process or improvement in any and all countries and further agree to execute and provide us with all necessary documentation for obtaining and enforcing these rights. You and your principals will irrevocably designate and appoint us as your agent and attorney-in-fact to execute and file any documentation and to do all other lawful acts to further the prosecution and issuance of patents or other intellectual property rights related to any concept, process or improvement. In the event that these provisions are found to be invalid or otherwise unenforceable, you and your principals will grant to us a worldwide, perpetual, non exclusive, fully paid license to use and sublicense the use of the concept, process or improvement if this use or sublicense would otherwise directly or indirectly infringe your rights.

ITEM 15 OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISED BUSINESS

You (or your principals) must devote their personal full-time attention, skill, and best efforts toin good faith, use the greatest possible effort for management and operation of the Franchised Business and to promote and increase the demand for our products and services within the Designated Territory. We may allow you to operate other non-competing businesses, on a case-by-case basis. You agree that you if Franchisee (or at least one of Franchisee's principals, if Franchisee is a corporation or partnership) or the Designated Manager(s) devotes appropriate time, resources and effort to supervising the development and operation of the Franchised Business. The owners of Franchisee must directly oversee the Designated Manager (where one is appointed pursuant to the Franchise Agreements) and such owners may engage in other non-competitive business or activity so long as it does not materially interfere with the operation of the Franchised Businesses. You may not, without our prior written consent, engage in any commercial activity that: (i) may benefit or promote a competing business; and (ii) may be injurious harmful to the Franchised Business or the goodwillgood reputation associated with the Proprietary Marksbrand, trademarks, and the System. Your (generally. If you or your principals') violation principals operate a noncompeting business without our permission or if the operation of these terms will be a material breacha non-competing business materially interferes with the operation of the Franchise Agreement, and Franchised Business, we may terminate the Franchise Agreement with notice and without an opportunity to cure.

Upon your written request, we may permit you to employ a manager to manage the day-to-day operations of the Franchised Business (the "Designated Manager"), provided the Designated Manager: (i) is approved by us in writing prior to hiring; and (ii) successfully completes our Initial Training Program before assuming any managerial responsibility. The Franchised Business must, at all times, be staffed with at least one individual who has successfully completed the Initial Training Program. In the event that a Designated Manager resigns or is otherwise terminated, the replacement must be trained pursuant to our then-current standards. The new Designated Manager must successfully complete training within thirty (30) days of hiring. The Designated Manager is not permitted to seek or maintain other employment or engage in any other business activities during the term of the Franchise Agreement. The Designated Manager is not required to possess any equity interest in the Franchised Business.

You must operate the Franchised Business for at least those days and number of hours we specify in the Operations Manual and operate continuously after opening.

Any Designated Manager and, if you are an entity, an officer that does not own equity in the franchisee entity must sign the Confidentiality and Restrictive Covenant Agreement, the form of which is attached to the Franchise Agreement inas Exhibit CD to this Franchise Disclosure Document. All of your employees, independent contractors, agents or representatives that may have access to our Confidential Information must also sign a Confidentiality and Restrictive Covenant Agreement. If you are a legal entity, each owner (i.e., each person holding an ownership interest in you), including the operating principal, must sign a Personal Guaranty guaranteeing the obligations of the entity in the form of which is attached to the Franchise Agreement inas Exhibit AD to this Franchise Disclosure Document. We also require that the spouses of the Franchise owners sign the Personal Guaranty.

ITEM 16 RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL

You must offer for sale all products and services which we prescribe and only those products and services which we prescribe. You may not offer any other products or services for sale without having received our prior written authorization. You must at all times maintain sufficient levels of inventory, as specified in the Operations Manual, to adequately satisfy consumer demand. You must offer, use and sell all private label products which we may now or in the future designate for sale by System franchisees.

All products and services you use or offer for sale from your Franchised Business must comply with our standards and specifications. Our standards and specifications are set forth in the Operations Manual, which is revised from time to time. You are responsible for ensuring that your Franchised Business meets these standards at all times. We have the right to inspect your Franchised Business or attend a project site for quality control purposes. We have the right to change our System from time to time.

ITEM 17 RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION

THE FRANCHISE RELATIONSHIP

This table lists certain important provisions of the Franchise Agreement and related agreements pertaining to renewal, termination, transfer and dispute resolution. You should read these provisions in the agreements attached to this disclosure document.

	Provision	Section In Franchise Agreement	Summary
a.	Length of the franchise term	2.1	Ten (10) years, which will commence on the date we execute the Franchise Agreement.
b.	Renewal or extension of term	2.2 to 2.3	One (1) successive ten (10)—year term. If the Franchise Agreement expires without you properly exercising your renewal right and you continue to accept the benefits of the Franchise Agreement thereafter, then, at our option, we may treat the Franchise Agreement either as (i) expired as of the date of expiration, with you then illegally operating a franchise in violation of our rights; or (ii) continued on a month-to-month basis (the "Holdover Period") until you and we agree to enter into our then-current form of franchise agreement for a renewal term or until you or we provide the other with written notice of termination, in which case the Holdover Period will terminate thirty (30) days after receipt of the notice of termination. In the latter case, all of your obligations shall remain in full force and effect during the Holdover Period as if the Franchise Agreement had not expired, except that the franchise fee during the Holdover Period will be increased to ten percent (10%)% of Gross Revenue for all types of products/services and without any reductions. All obligations and restrictions imposed on you upon expiration of the Franchise Agreement shall take effect upon termination of the Holdover Period.
c.	Requirements for franchisee to renew or extend	2.2.1 to 2.2.9	You must: (i) provide notice of your renewal no fewer than twelve (12) months and no greater than eighteen (18) months prior to the end of the term; (ii) demonstrate to our satisfaction that you have the

Provision	Section In Franchise Agreement	Summary
		right to operate the Franchised Business at the Approved Location for the duration of the renewal term or, if you are unable to continue operating at the Approved Location, secure a substitute location that is acceptable to us; (iii) complete to our satisfaction, no later ninety (than 90) days prior to expiration of your then-current term, any updates to all required equipment, supplies, inventory, hardware and software, and vehicles and all maintenance, refurbishing, renovating, updating and remodeling of the Franchised Business premises, as well as any update to required hardware and software, as necessary to bring the Franchised Business and all equipment into full compliance with our then-current System standards and specifications for new franchisees; (iv) not be in breach of any provision of the Franchise Agreement, or any other agreement with us, our affiliates, or approved/designated suppliers and vendors, and also have been in substantial compliance with these agreements during their respective terms; (v) satisfy all monetary obligations you have to us, our affiliates, and approved or designated suppliers/vendors and also have timely met these obligations throughout the term; (vi) execute our then-current form of franchise agreement, the terms of which may materially vary from the terms of your current Franchise Agreement at least three (3) months prior to the expiration of your current Franchise Agreement; (vii) satisfy our then-current training requirements and have obtained and maintained all licenses, permits and approvals required by federal and state law applicable to providing the Approved Products and Approved Services at any location within the Designated TerriTerritory; (viii) execute a general release in favor of us and our affiliates in the form we prescribe; and (ix) pay a Renewal Fee equal to five thousand dollars (\$\sum_{2}\$5,000).
		You will be obligated to pay a Renewal Fee for each Franchised Business you wishwant to renew under the Multi-Unit Addendum. Provided, however, that where a group of territories are under a single Multi-Unit Addendum, the Renewal Fee will be \$5,000 for the first Franchise Agreement and \$1,000 for each additional Franchise Agreement renewed at the same time and in a contiguous territory
d. Termination by franchisee	Not Applicable	Not Applicable. However, franchisees may terminate under any grounds permitted by applicable law.
e. Termination by franchisor without cause	Not Applicable	Not Applicable.
f. Termination by franchisor with cause	15.1 to 15.4	We may terminate your Franchise Agreement upon your default and, in some instances, failure to cure. Termination is effective upon delivery of written notice, except as otherwise provided in the Franchise Agreement.

Provision	Section In Franchise Agreement	Summary
g. "Cause" defined – curable defaults	15.3	The following are curable defaults under the Franchise Agreement, provided you cure the default within fifteen (15) days of our notice of: (i) your failure to pay any sums due us, our affiliates or any of our System Suppliers, Approved Suppliers, or vendors; (ii) your failure to immediately endorse and deliver to us any payments due us from any third party that are erroneously made to you; (iii) your failure to maintain sufficient inventory of marketing materials and other supplies; (iv) your failure to maintain the prescribed months, days or hours of operation at the Franchised Business; (v) your failure to personally supervise day-to-day operations or failure to employ a sufficient number of qualified, competent personnel as we required operate the business to the brand standards; (vi) your failure to maintain the strict quality controls reasonably required by the Franchise Agreement and/or the Operations Manual; and (vii) your failure to procure or maintain any licenses, certifications or permits necessary for the operation of the Franchised Business, (provided that additional time may be provided if franchisee has an appropriate substitute license that allows the business to operate and is diligently pursuing licensure).
	15.4	Notwithstanding Sections 15.1, 15.2 and 15.3 of the Franchise Agreement, you will have thirty (30) days from the date of our notice to cure any other default under the Franchise Agreement or any other agreement with us or our affiliates. The cure period will be extended for a period of up to 30 days if: a) the breach cannot reasonably be cured within the original cure period, or b) Franchisee is diligently pursuing a cure of the breach but is unable to complete the cure due to circumstances outside of Franchisee's control.
h. "Cause" defined – non-curable defaults	15.1	The Franchise Agreement will automatically terminate without notice or opportunity to cure upon the occurrence of any of the following: (i) if you make an assignment for the benefit of creditors, file a voluntary petition in bankruptcy, are adjudicated bankrupt or insolvent, file or acquiesce in the filing of a petition seeking reorganization or arrangement under any federal or state bankruptcy or insolvency law, or consent to or acquiesce in the appointment of a trustee or receiver for you or the Franchised Business; (ii) if proceedings are commenced to have you adjudicated bankrupt or seek your reorganization under any state or federal bankruptcy or insolvency law and such proceedings are not dismissed within sixty (60) days, or a trustee or receiver is appointed for you or the Franchised Business without your consent and the appointment is not vacated within sixty (60) days; or (iii) you lose the right to occupy the premises or operate the Franchised Business from the Approved Location and such location is not replaced with an approved location within 6 months after loss of the right to occupy the original site.
	15.2	We have the right to terminate the Franchise Agreement upon notice and without opportunity to cure upon the occurrence of any of the following defaults: (i) you or one of your principal(s) are convicted of, or plead guilty or no contest to, a felony (whether or not related

Provision Franchise Summary Agreement	
to the Franchised business) or other offense related to the	
of the Franchised Business or which we believe, i	
discretion, is likely to have an adverse effect on our	
Marks or the goodwill associated therewith; (ii) yo	-
principal(s) commit any fraud or misrepresentation in the of the Franchised Business, including a misrepresentation	
or otherwise) made in completing your franchise appli	
you or any of your principals, including any shareholder	
guarantor or agent engage in any activity or conduct tha	
impairs the goodwill associated with the System or	
Marks and fail to cease and correct such activities or con	
twenty four (24) hours of being notified of this breach;	
will have only one opportunity to cure under this sec	
subsequent violation after the first violation is subject to	
termination without an opportunity to cure.); (iv) you principal(s) make any misrepresentation or omission in	
with your franchise application, including, but not limit	
financial misrepresentation; (v) you or your Designated N	
to complete the Initial Training Program in the required t	-
(vi) we send you two (2) or more written notices to cure	e any of the
defaults set forth indefault pursuant to Sections 15.3 and	
Franchise Agreement in any twelve (12)_month period	
of whether or not you subsequently cure these defaults;	
your principal(s) materially breach any other agreement our affiliates, or threaten any material breach of these agr	
any lease for the Approved Location, and fail to cure s	
within the prescribed time period set forth in that agree	
you or your principal(s) misuse our Proprietary	
Confidential Information in any manner; (ix) you or your	principal(s)
disclose any contents of the Operations Manual, C	
Information, and/or Trade Secrets; (x) you violate any law	
or regulation, as well as your operation of the Franchised	
a manner that presents a health or safety hazard to custo general public and have not cured the violation within	
has not put in place an action plan to prevent the occurr	
similar violation in the future. The cure period does not a	
repeated occurrence of the same or similar breach in the	future; (xi)
you violate any of the restrictive covenants set forth in the	
Agreement; (xii) a levy or writ of attachment or execu	
other lien is placed against you or any of your principal	-
their assets which are not released or bonded against w (30) days; (xiii) except for adjudicated insolvency which	
to automatic termination, you or your principal(s) become	
or admits insolvency; (xiv) you voluntarily or otherwise a	
Franchised Business; (xv) you make any unauthorized tra	
Franchised Businessany interest in the Franchised	
(including the sale of assets of the Franchised Business)	
void such transfer, or otherwise come into compliance	
transfer obligations within 30 calendar days of n	
Franchisor; (xvi) you offer any unauthorized or unapprov or services at or from the Franchised Business_at	*
immediately cease offering the unauthorized Products	

Provision	Section In Franchise Agreement	Summary
		unless and until you obtain approval from us; (xvii) you order or purchase supplies, signs, or other required items and/or inventory from an unapproved supplier and fails to immediately cease the unapproved purchases unless and until you obtain approval from us; (xviii) you misuse, or make unauthorized use of, any proprietary software that we may develop and does not immediately cease use of such unauthorized use of such software, provided however that immediate termination is permitted even if the unauthorized use is stopped: a) where cure is not possible, b) where such default negatively and materially impacts the rights of the system to use the software, or c) where as a result of such default franchisee is not permitted to continue using software that is a required software for the System; (xix) you fail to maintain the required insurance, and fail to reinstate insurance within 48 hours of notice from us or repay us for insurance we paid for you; (xx) you fail, within fifteen (15) calendar days after notification of non-compliance by federal/state/local government authorities, to comply with any law or regulation applicable to the Franchised Business; (xxi) the government takes any action against you that results in an obligation upon us that we believe is uneconomical, not in our best interest, or which would result in having an unintended relationship or obligation; (xxii) you fail to comply with any anti-terrorism law or provisions; (xxiii) you take any assets or property of the Franchised Business for personal use except for incidental and innocuous sue that does not result in a claim or damage to the goodwill of the brand; (xxiv) there are insufficient funds in your EFT bank account to cover any payment to Franchisor two (2) or more times in any twelve (12)-month period; (xxv) you fail to commence operations of the Franchised Business within the required time period; (xxvi) you operate or conduct business outside of your Designated Territory without our consent; and (xxviii) you or your principal(s) fa
i. Franchisee's obligations on termination / non-renewal	16.1	Upon termination or expiration of the Franchise Agreement, your obligations include: (i) immediately cease all operations of the Franchised Business; (ii) immediately pay all amounts owed to us, our affiliates and our major suppliers and vendors; (iii) immediately discontinue using the Proprietary Marks; (iv) immediately cease using the Business Management and Technology System and Operations Manual, and within ten-(10) days return all proprietary and confidential materials; (v) immediately cease use of all telephone and facsimile numbers and related listings, as well as any permitted domain names and/or social media pages that were used in connection with the Franchised Business (collectively, the "Assigned Property") and take all necessary steps to assign the Assigned Property to us or our designee; (vi) immediately vacate the premises of the Franchised Business and arrange for transfer of the lease, if applicable; (vii) within ten-(10) days, return all stationery, printed matter, signs, advertising materials, supplies and other items containing our Proprietary Marks and trade dress; (viii) cease holding

	Provision	Section In Franchise Agreement	Summary
			yourself or the Franchised Business out as part of our System; (ix) cease all contact with Franchised Business customers; (x) take all actions necessary to amend or cancel any assumed name, business name or equivalent registration that contains any trade name or Proprietary Mark, and furnish evidence to us that you have complied with this obligation within fifteen (15) days; (xi) permit us to make a final inspection of your financial records, books and other accounting records within one (1) month2 years of the termination/expiration or transfer of your Franchise Agreement; (xii) comply with your post-term restrictive covenants set forth in Section 17 of the Franchise Agreement; (xiii) cease using any methods, procedures or techniques associated with us or the System in advertising or any other manner; (xiv) de-identify all vehicles used in connection with the Franchised Business; (xv) execute from time to time any necessary papers, documents, and assurances to effectuate Section 16 of the Franchise Agreement; (xvi) provide us with any social media account information and access thereto; (xvii) reimburse us for any costs that we incur in enforcing your obligations under Section 16 of the Franchise Agreement.
j.	Franchisor's right to transfer	14.5	There are no restrictions on our right to sell, transfer or assign the Franchise Agreement.
k.	"Transfer" by franchisee - defined	14.1	"Transfer" as a verb means to sell, assign, give away, transfer, pledge, mortgage or encumber, voluntarily, involuntarily, or by operation of law (such as through divorce or bankruptcy proceedings), any interest in the Franchise Agreement, the Franchised Business, substantially all the assets of the Franchised Business, or in the ownership of the Franchised Business (if you are an entity). "Transfer" as a noun means any sale, assignment, gift, transfer, pledge, mortgage or encumbrance.
1.	Franchisor approval of transfer by franchisee	14.1	Any transfer requires our prior written consent and a general release. Franchisor will not unreasonably withhold, condition, or delay a request to transfer by Franchisee.
		14.4	If you are an individual and desire to assign your rights under the Franchise Agreement to a corporation or limited liability company, and if certain conditions are met, we will consent to the transfer without assessing a transfer fee, and such assignment will not be subject to our right of first refusal.
m.	Conditions for franchisor approval of transfer	14.3.2	Our approval of a proposed transfer is conditioned upon the satisfaction of the following conditions, as applicable: (i) all of your accrued monetary obligations to us, our affiliates, suppliers and vendors have been paid; (ii) you have cured all existing defaults under the Franchise Agreement and any other agreement with us our affiliates and designated/Approved Suppliers and vendors within the time period permitted for cure, and have substantially complied with these agreements during their respective terms; (iii) you and your principals and the transferee (if it had any prior relationship with us or our affiliates) must execute a general release under seal of any and all claims against us and our affiliates (including our officers,

Provision	Section In Franchise Agreement	Summary
		directors, shareholders and employees, in their corporate and individual capacities) in the form we prescribe; (iv) transferor and transferee must execute a mutual general release, relieving all claims against each other provided, however, the general release shall not be inconsistent with any applicable state statute regulating franchising; (v) you or the transferee must provide us with a copy of the executed purchase agreement for the Franchised Business, as well as all other documents relevant to the transaction; (vi) transferee must satisfactorily demonstrate to us that it meets our educational, managerial and business standards to operate the Franchised Business; (vii) transferee must either execute our then-current form of franchise agreement, which may contain materially different terms than your Franchise Agreement, and assume a full term as set forth in the then-current form of franchise agreement for new franchisees; or, at our option, enter into an assignment and assumption of the Franchised Business and all rights and obligations thereunder; (viii) you must pay us our transfer and training fees; (ix) transferee must satisfactorily complete our Initial Training Program at its own expense within the time frame we designate; (x) you, your principals and members of their respective families must comply with the post-termination provisions of the Franchise Agreement; (xi) transferee must obtain and maintain all permits and licenses required for the operation of the Franchised Business within the time limits we set; (xii) if you are operating from a leased location, the lessor of that location must approve the assignment of the lease to the transferee; (xiii) the transfer must comply with any state and federal laws that apply to the transfer; (xiv) you must ensure that all projects in progress at the time of the transfer will be continued without interruption, and the transferee must promptly notify all current customers of the change in ownership; (xv) the purchase price and terms of transfer must not be
	14.4	You do not need to pay a transfer fee if you are an individual and you wish to transfer the Franchise Agreement to a corporation or limited liability company, provided the following conditions are met: (i) the business entity is newly organized and its activities are confined to operating the Franchised Business; (ii) you remain, at all times, the owner of at least fifty one percent (51%)51% of the outstanding shares of the entity; (iii) the entity agrees to assume all of your obligations under the Franchise Agreement; (iv) all stockholders of the corporation, or members and managers of the limited liability company must execute our prescribed form of personal guaranty; (v) at our request, you must provide true and correct copies of all documents and contracts governing the rights, obligations and powers of the owners. If you are a partnership, corporation or limited liability company, you and your principals, and the transferee (if it

Provision	Section In Franchise Agreement	Summary
		has had any previous relationship with us or our affiliates), must execute a general release. If we approve a Transfer to allow ownership by an ESOP, trust or private equity firm, there may be additional requirements.
n. Franchisor's right of first refusal to acquire franchisee's business	14.3.1	We have the right to match any bona fide third-party offer to buy your franchise rights, assets or controlling interest that is the subject of a proposed Transfer (other than a Transfer from an individual franchisee to a business entity as described in Section 14.4 of the Franchise Agreement). We may exercise this right of first refusal within thirty (30) days of the date you provide us with a copy of the third-party offer and any other information that we request. If we do not exercise this option, you must complete the Transfer to the third-party within sixty (60)120 days, subject to the conditions set forth in Section 14.3.2 of the Franchise Agreement. Otherwise, we will once again have our right of first refusal.
o. Franchisor's option to purchase franchisee's business	16.2 and 16.3	Upon your termination, we or our designee may purchase personal property used in connection with the operation of the Franchised Business by: (i) providing you with notice of our election to do so within sixty (60) calendar days of the expiration/termination of your Franchise Agreement; and (ii) paying you the bookfair market value for such personal property within sixty (60) days of providing you with this notice. The assumption of any debt or lease obligations on personal property is deemed payment of the fulla credit towards purchase priceat fair market value.
p. Death or disability of franchisee	14.2.1	UponIn the event of Franchisee's death, disability, physical or mental incapacity of any person with an interest in the Franchise Agreement, the franchisee, incapacitation (or in allthe death, disability or substantially allincapacitation of the assets of the Franchised Business, the Franchisee's partners or personal guarantors), Franchisee's legal representative—of such person shall—or Franchisee's partner's or guarantor's respective legal representative, as applicable, will have the right to continue the operation of the Franchised Business as franchisee under this Agreement if: (i) within 180 days from the date of death/, disability/or incapacity,—(the representative—180-Day—Period—), such—person: (a) meets ourFranchisor's then-current standards to own—a—Franchised Business, become a franchisee, as described in Section 14.3.2.6; and (b) has obtained ourFranchisor's prior written approval, and has executed ourFranchisor's then-current form of franchise agreement for the unexpired termsterm of the franchise, or has otherwise furnished a personal guaranty of any business entitypartnership, corporation or limited liability company guarantying franchisee's franchisee's obligations to usFranchisor and ourFranchisor's affiliates; and (ii) thissuch person successfully completes our then current Initial Training Program, which will be provided at our then current training tuition rate. In the Franchisor's then-current tuition rate. Such assignment by operation of law will
	14.2.2	not be deemed in violation of this Agreement, provided such heirs or legatees accept the conditions imposed by the Franchise Agreement and are acceptable to Franchisor. However, in the case of a

Provision	Section In Franchise Agreement	Summary
		Transfertransfer by demise or inheritance, if the heirs or beneficiaries of any deceased person are unable to meet the aforementioned conditions contained in this Agreement, the personal representative of the deceased franchiseeFranchisee will have a reasonable time, in ourFranchisor's sole discretion and, which shall not to exceed 180 days from the date or Transfertransfer by demise or inheritance, to dispose of the deceased's interest in the Franchised Business, and such disposition will be subject to all the terms of the Franchiseand conditions for transfer contained in this Agreement. If the interest is not disposed of in the manner and time frame provided in the immediately preceding sentence, then weFranchisor may terminate the Franchisethis Agreement.
		We may, but are not obligated to, operate the Franchised Business during the 180-day period following the death/incapacity/disability, and we may pay ourselves our then-current fee to reimburse us for providing management services and our other costs.
q. Non-competition covenants during the term of the franchise	17.1	During the term of the Franchise Agreement, neither you, nor your owners, officers, directors, principals, nor any member of the their immediate families may directly or indirectly: (a) own, maintain, engage in, be employed as an officer, director or principal of, lend money to, extend credit to, or have any interest in any other business that, directly or indirectly, by itself or through licensees or franchisees, offers any products and/or services authorized or offered for sale by System franchisees (a "Competitive Business") within the Designated Territory or the Designated Territory of any other System franchisee, provided that Section 17.1.1 of the Franchise Agreement does not apply to: (i) such person's ownership of a Franchised Business under a Franchise Agreement with us; or (ii) such person's ownership of a less than five percent (5%)% legal or beneficial interest in any publicly-traded company providing such services; (b) employ or seek to employ any person who is at that time employed by us or our affiliates, or otherwise directly or indirectly induce or try to induce an employee to leave his or her employment with us, our affiliates or other System franchisees; (cor (b) solicit any current, former or prospective customer solicited by your Franchised Business or any other customer that you become aware of as a result of access to our System or other franchisees for any competitive purpose.

	Provision	Section In Franchise Agreement	Summary
r.	Non-competition covenants after the franchise is terminated or expires	17.2	For a period of two—(2) years after the expiration, transfer or termination of the Franchise Agreement, regardless of the cause, neither you, your owners, officers, directors, or principals, nor any member of the immediate family of you or your owners, principals, officers, or directors may, directly or indirectly, for themselves or through, on behalf of, or in conjunction with any other person, partnership or corporation: (1) own, maintain, engage in, be employed as an officer, director or principal of, lend money to, extend credit to, or have any interest in any Competitive Business: (a) within the Designated Territory; (b) within a twenty five (25)—mile radius of the Designated Territory of any other Franchised Business; or (c) within a twenty five (25)—mile radius of any other System business operated by us or our affiliate. This covenant does not apply to: (i) your ownership of a Franchised Business under a Franchise Agreement with us; or (ii) your ownership of a less than five percent (5%)5% legal or beneficial interest in any publicly traded company providing services the same as or similar to a Competitive Business; (2) employ or seek to employ any person who is at that time employed by us or our affiliates, or otherwise directly or indirectly induce or try to induce an employee to leave his or her employment with us, our affiliates, or other System franchisees; and (3or (2) solicit any current, former or prospective customer solicited by your Franchised Business, or any other customer that you have become aware of as a result of access to our System and other franchisees for any competitive purpose.
s.	Modification of the agreement	22.1	The Franchise Agreement may not be modified except by a written agreement that both of us sign.
t.	Integration/ merger clause	22.1	Only the terms of the Franchise Agreement and the Franchise Disclosure Document are binding (subject to state law). Any representations or promises made outside of the Franchise Disclosure Document and Franchise Agreement may not be enforceable. Nothing in the Franchise Agreement or any related agreement is intended to disclaim the representations we made in this Franchise Disclosure Document.
u.	Dispute resolution by arbitration or mediation	18.2	Any disagreement, litigation, claim, or dispute, suit, action, controversy or proceeding of any type whatsoever, including any claim for equitable relief and/or where you are acting as a "private attorney general" suing pursuant to a statutory claim or otherwise, between or involving you and us on whatever theory and/or facts based, and, whether or not arising out of the Franchise Agreement (including any dispute or disagreement relating to arbitration, including the arbitrability of the Franchise Agreement or any of its provisions), our offer, sale or negotiation of a Top Rail Franchise, or the relationship of the parties arising from the Franchised Business or from entering into the Franchise Agreement, or any claim that the Franchise Agreement or any part of the Franchise Agreement is invalid, illegal or otherwise voidable, void, or unenforceable ("Dispute") will be processed in the following manner, you and we each expressly waiving all rights to any court proceeding, except as expressly provided in Section 18.3.1 of the Franchise Agreement.

Provision	Section In Franchise Agreement	Summary
		You must first bring any claim or dispute between you and us to our management, after providing notice as set forth in Section 18.6 of the Franchise Agreement. You must exhaust this internal dispute resolution procedure before you may bring your Dispute before a third party. This agreement to first attempt resolution of disputes internally shall survive termination or expiration of the Franchise Agreement. You and we agree that the franchise relationship is unique and that as a result, it is important that anyone who serves as a mediator or arbitrator in a Dispute must have a minimum of seven (7) that has a minimum of 7 years' substantive experience in franchise law.
	18.3	
		arbitration in accordance with such rules. Except as provided in Section 18.3.1 of the Franchise Agreement, no party may file litigation involving any Dispute unless it has complied with the mediation and arbitration provisions in Article 18 of the Franchise Agreement regarding that Dispute. "Filing litigation" includes asserting a counterclaim, third party claim or other claim relating to a Dispute (irrespective of whether the party asserting the claim filed)

Provision	Section In Franchise Agreement	Summary
	18.3.1	the legal action in which the claim is asserted). Our rights to mediation and arbitration, as set forth herein, may be specifically enforced by us. Each party shall bear its own cost of mediation and arbitration and we and you shall share mediation costs equally. This agreement to mediate and arbitrate shall survive any termination or expiration of the Franchise Agreement.
		The parties shall not be required to first attempt to mediate or arbitrate a controversy, Dispute or claim through mediation if such controversy, Dispute or claim relates to an allegation that a party has violated (or threatens to violate, or poses an imminent risk of violating): (i) any federally protected intellectual property rights in the Proprietary Marks, the System, or in any Confidential Information; (ii) any claims pertaining to or arising out of any warranty issue; (iii) any of the restrictive covenants contained in the Franchise Agreement; (iv) any of your payment obligations that are more than forty-five (45) days past due; (v) any claims arising out of or related to fraud or misrepresentation by you, or your insolvency; (vi) any claims relating to your obligations on termination or expiration of the Franchise Agreement; (vii) any claims relating to any Transfer of an interest in you, the Franchised Business or your assets; or (viii) any matters involving danger, health or safety. Dispute resolution requirements are subject to applicable state law.
v. Choice of forum	18.4	All claims not subject to mediation and/or arbitration must only be brought in a competent court of general jurisdiction located in Mecklenburg County, North Carolina or, if appropriate, the United States District Court for the Western District of North Carolina (subject to applicable state law).
w. Choice of law	18.1	North Carolina law governs all claims arising out of the Franchise Agreement, without reference to its conflict of laws provision (subject to applicable state law).

ITEM 18 PUBLIC FIGURES

We do not currently use any public figure to promote our System.

ITEM 19 FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC's Franchise Rule permits a franchisor to provide information about the actual or potential financial performance of its franchised and/or franchisor-owned outlets, and/or affiliate-owned outlets, if there is a reasonable basis for the information, and if the information is included in the disclosure document. Financial performance information that differs from that included in Item 19 may be given only if: (1) a franchisor provides the actual records of an existing outlet you are considering buying; or (2) franchisor supplements the information provided in this Item 19, for example, by providing information about possible performance at a particular location or under particular circumstances.

As of December 31, 2022, there were nine franchised operating locations (each, a "<u>Franchised Location</u>") that operate 19 Franchised Businesses under 4 multi unit addendums and two locations operated individually, and zero (0) affiliate owned locations (each, an "<u>Affiliate Owned Location</u>") in operation. In June 2022, we reached an agreement to convert nine (9)10 locations offering substantially similar services to the services that we offer to Affiliate Owned Franchised Locations offer, and at that time, each Affiliate-Owned Location began converting their businesses to the "TOP RAIL SM" brand.

This Item discloses As of December 31, 2023, the Gross Revenue generated by the 10 Affiliate-Owned Locations from January 1, operate 21 Franchised Businesses under 4 multi-unit addendums and two Franchised Locations operated individually (each, an "Affiliate-Owned Location") in operation. 2022 through December 31, 2022 (the "2022 Measurement Period").

The Affiliate-Owned Locations have been in operation since 2015 using a system of operations substantially similar to the Top Rail System and upon which the Top Rail System is based. The Affiliate-Owned Locations converted its branding to Top Rail when it became a franchisee on December 29, 2022. The information in this Item has not been audited, is based on historical financial data provided to us by the Affiliate-Owned businesses. Prior to converting to Franchised Businesses the Affiliate-Owned Locations were not subject to the franchisee fees relating to royalties and advertising (including mandatory minimum advertising spend requirements), call center fees, or the annual conventionconference. More than one of the Affiliate-Owned Locations may be managed together by the same personnel (see below for a breakdown of how the locations were managed by group). Other than as described above there are no other material financial or operational differences between the Affiliate-Owned Locations and operational franchise outlets.

Only Part I of this Item discloses the Ticket Size for Franchised Businesses with at least 6 months of operations in 2023. This data is for the period January 1, 2023, through December 31, 2023. This data was collected from our Business Management and Technology System.

Part II of this Item discloses the aggregate Proposal Conversion Ratio of the 8 franchisees with at least 6 months of operations in 2023. This data is for the period January 1, 2023, through December 31, 2023.

Part III of this Item discloses the Gross Revenue generated by the Affiliate-Owned locations from January 1, 2023, through December 31, 2023 (the "Measurement Period").

Some units have soldearned this muchamount. Your individual results may differ. There is no assurance that you will sellyou'll earn as much.

Written substantiation offor the data used in preparing this Itemfinancial performance representation will be made available to the prospective franchisee upon reasonable request.

PART I TICKET SIZE FOR FRANCHISED BUSINESSES WITH 6 OR MORE MONTHS OF OPERATION IN 2023

Ticket Siz	ze - Summary Statistics
<u>Highest</u>	<u>\$309,320</u>

<u>Average</u>	<u>\$6,468</u>
Median	<u>\$5,000</u>
Lowest	<u>\$150</u>

Notes to Part I:

- 1. This data includes 26 Designated Territories operated by 8 Franchised Businesses that operated for 6 or more months in 2023 and does not include Affiliate-Owned Locations.
- 2. Tickets for purposes of Part I is a completed job. The amount of each ticket is the total Gross Sales from a ticket. Change orders are not counted separately as a ticket.

[THE REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

PROPOSAL CONVERSION RATE FOR FRANCHISED BUSINESSES WITH 6 OR MORE MONTHS OF OPERATION IN 2023

Lead to Close Rate - Summary Statistics		
<u>Total Proposals</u>	<u>1,284</u>	
Proposal Conversion Rate Average	35%	

Notes to Part II:

- This data includes 26 Designated Territories operated by 8 Franchised Businesses that operated for 6 or more months in 2023 and does not include Affiliate-Owned Locations.
- 2. The median conversion rate is 34%, with a high of 58% and a low of 11%.
- 3. Proposal Conversion Rate means the percentage of total proposals that were accepted, completed, invoiced, and scheduled.
- 4. Lost Proposals include proposals that were declined, expired, or are still open.

GROSS REVENUE GENERATED BY THE AFFILIATE-OWNED LOCATIONS⁽¹⁾ OVER THE 20232022 MEASUREMENT PERIOD

Top Rail Fence (Group A)		
2023 Measurement Period		
	Total	Percent

Gross Revenue ⁽²⁾	\$5,255,067 <u>\$</u>	
Group ASales(1)	4,024,958	100.0%
Cost of Goods Sold	_	=
Subcontractors ⁽²⁾	\$ 566,271	<u>14.1%</u>
Direct Materials & Repair ⁽³⁾	\$ 1,510,531	<u>37.5%</u>
Equipment Rental ⁽⁴⁾	<u>\$ 1,963</u>	<u>0.0%</u>
Total Cost of Goods Sold(5)	\$ 2,078,765	<u>51.6%</u>
Gross Revenue	\$ <u>\$</u>	
Group BProfit ⁽⁶⁾	1, 554, 946 <u>,193</u>	48.4%
Disclosed Expenses		_
Management Compensation, Burden, & Benefits ⁽⁷⁾	\$ 553,526	13.8%
Miscellaneous G&A ⁽⁸⁾	\$ 6,268	0.2%
Marketing, Advertising, & Promotion ⁽⁹⁾	\$ 213,224	<u>5.3%</u>
Building Maintenance & Utilities (10)	\$ 10,010	<u>0.2%</u>
Vehicle Expense - Fuel, Repair, & Maintenance ⁽¹¹⁾	\$ 4,981	<u>0.1%</u>
Travel & Entertainment ⁽¹²⁾	\$ 13,892	<u>0.3%</u>
Insurance ⁽¹³⁾	\$ 52,582	<u>1.3%</u>
Professional Fees ⁽¹⁴⁾	\$ 54,204	<u>1.3%</u>
Franchisor Fees	_	_
Royalty (Effective Assumed Rate of 6.29%) ⁽¹⁵⁾	\$ 253,170	<u>6.3%</u>
Brand Fund (2%) ⁽¹⁶⁾	\$ 80,499	<u>2.0%</u>
Call Center Fee (2%) ⁽¹⁷⁾	\$ 80,499	2.0%
Technology Fee (\$599/mo) (18)	\$ 7,188	0.2%
Local Advertising Requirement (\$3,000/month) (19)	\$ 0	0.0%
Annual Conference Fee (\$1,000/year)	\$ 1,000	<u>0.02%</u>
Marketing Management Fee (\$500/mo)	\$ 6,000	<u>0.1%</u>
Online Local Presence Fee (\$249/mo)	\$ 2,988	<u>0.1%</u>
Total Disclosed Expenses ⁽²⁰⁾	\$ 1,340,034	33.3%
Adjusted Profit ⁽²¹⁾	\$ 606,159	<u>15.1%</u>

2023 Measurement Period (Group B)			
	Total	Percent	
Gross Sales ⁽¹⁾	\$ 5,412,868	<u>100.0%</u>	
Cost of Goods Sold	_	_	
Subcontractors ⁽²⁾	\$ 1,197,942	22.1%	
Direct Materials & Repair ⁽³⁾	\$ 2,224,802	<u>41.1%</u>	
Equipment Rental ⁽⁴⁾	\$ -	0.0%	
Total Cost of Goods Sold(5)	\$ 3,422,744	<u>63.2%</u>	
Gross Profit ⁽⁶⁾	\$ 1,990,124	<u>36.8%</u>	
Disclosed Expenses	_	_	
Management Compensation, Burden, & Benefits ⁽⁷⁾	\$ 366,428	<u>6.8%</u>	
Miscellaneous G&A ⁽⁸⁾	\$ 65,074	<u>1.2%</u>	
Marketing, Advertising, & Promotion ⁽⁹⁾	\$ 164,909	<u>3.0%</u>	
Building Maintenance & Utilities ⁽¹⁰⁾	\$ 19,157	0.4%	
Vehicle Expense - Fuel, Repair, & Maintenance(11)	\$ 34,501	<u>0.6%</u>	
Travel & Entertainment ⁽¹²⁾	\$ 35,574	<u>0.7%</u>	
Insurance ⁽¹³⁾	\$ 96,515	<u>1.8%</u>	
Professional Fees ⁽¹⁴⁾	\$ 1,063	<u>0.0%</u>	
Franchisor Fees	_	=	
Royalty (Effective Assumed Rate of 5.89%) ⁽¹⁵⁾	\$ 318,643	<u>5.9%</u>	
Brand Fund (2%) ⁽¹⁶⁾	\$ 108,257	2.0%	
Call Center Fee (2%) ⁽¹⁷⁾	\$ 108,257	2.0%	
<u>Technology Fee (\$599/mo) (18)</u>	\$ 7,188	<u>0.1%</u>	
Local Advertising Requirement (\$3,000/month) (19)	\$ 0	<u>0.0%</u>	
Annual Conference Fee (\$1,000/year)	\$ 1,000	0.02%	
Marketing Management Fee (\$500/mo)	\$ 6,000	<u>0.1%</u>	
Online Local Presence Fee (\$249/mo)	\$ 2,988	<u>0.1%</u>	
Total Disclosed Expenses (20)	\$ 1,335,554	<u>24.7%</u>	
Adjusted Profit ⁽²¹⁾	\$ 654,570	12.1%	

Notes to **Item 19**Part III:

- 1. "Gross Sales" is defined to include all income of any type or nature and from any source that you derive or receive directly or indirectly from, through, by or on account of the operation of the Franchised Business at any time after the signing of your Franchise Agreement, in whatever form and from whatever source, including, but not limited to, cash, services in kind from barter and/or exchange, on credit or otherwise, as well as business interruption insurance proceeds, all without deduction for expenses, including marketing expenses and taxes. However, the definition of Gross Revenue does not include sales tax that is collected from customers and actually transmitted to the appropriate taxing authorities.
- "Subcontractors" includes multiple independent businesses that are contracted to perform installation services for customers.

- 3. "Direct Materials and Repair" includes fence material and hardware and other job materials that are used in the performance of services such as fence paint or stain.
- "Equipment Rental" includes equipment or other miscellaneous onetime use equipment that is a direct cost of completing a specific job.
- "Total Cost of Goods Sold" is calculated by adding Subcontracts, Direct Materials and Repair; and Equipment Rental.
- 6. "Gross Profit" is calculated by subtracting Total Costs of Goods Sold from Gross Sales.
- 7. "Management Compensation, Burden, & Benefits" includes all wages and salaries paid to the owners, union and non-union employees, payroll taxes and employee benefits. The Owner of the Affiliate_Owned Franchise Outlet took \$200,000 in W-2 income that has been removed from Management Compensation, Burden, and Benefits. There is a full management team, operations team, and sales team in the Management Compensation, Burden, and Benefits Section of the P&L.
- 8. "Miscellaneous G&A" includes bank charges, dues & subscriptions, insurance, license & permits, and telephone.
- 9. "Marketing, Advertising, & Promotion" includes advertising & promotion, new business development; plus the following office and warehouse expenses: rent, utilities, & repairs;
- 10. "Building Maintenance & Utilities" includes fees for upkeep of office/warehouse space and utilities for the Approved Locations."
- 11. "Vehicle Expense" includes automobile fuel, repairs, maintenance, insurance, parking and tolls, and other travel expenses.
- 12. "Travel & Entertainment" includes expenses incurred for travel, meals, lodging related to support of the business.
- 13. "Insurance" includes cost of required and optional insurance.
- 14. "Professional Fees" includes fees for the following professional services: accounting services, legal services, outside consulting services.
- 15. "Royalty Fee." The Term "Royalty Fee" means the Royalty Fee that the Affiliate Location would have had to pay us over the 2023 Measurement Period if the Affiliate Location were owned by a System franchisee and governed by our current form of Franchise Agreement. We calculated the Royalty Fee by multiplying Gross Sales by 0.0629 (6.29% for Group A) and 0.0589 (5.89% for Group B) to account for the sliding scale Royalty Fee of 5%-7% set forth and required under our current form of Franchise Agreement.
- 16. "Brand Fund Fee." The term "Brand Fund Contribution" means the Brand Fund Contribution that the Affiliate Location would have had to pay us over the 2023 Measurement Period if the Affiliate Location were owned by a System franchisee and governed by our current form of Franchise Agreement. We calculated the Brand Fund Contribution by multiplying Gross Sales by .02 to account for the Brand Fund Contribution of 2% set forth and required under our current form of Franchise Agreement.

- 17. "Technology Fee" is calculated by taking our current Technology Fee of \$599 and multiplying that number by 12. We do not presently expect the Technology Fee to exceed \$1,000 per month, however, there is no cap on the amount the Technology Fee may be increased. An increase will be based on the following: cost of inflation, cost of underlying products and services, including development costs of software, mobile applications, websites, costs of software, subscriptions, and related services, and any other products or services added to the technology stack.
- 18. "Call Center Fee" is calculated by multiplying Gross Revenue by the current Call Center Fee of 2%. We have the right to increase the Call Center fee to an amount equal to 4% of Gross Revenue.
- 19. "Local Advertising Requirement" for Group A is calculated by taking our current Local Advertising Requirement of \$3,000 per month for the Asheville location and the first location under the South Carolina Multi-Unit Addendum plus \$1,500 per month each for the other three locations (total of \$10,500 per month) and multiplying that number by 12 (\$126,000). The Local Advertising Requirement can be increased to \$4,000 per month upon a 90-day notice. We have adjusted the number in the table to reflect advertising included in the Marketing, Advertising, & Promotion figure under Disclosed Expenses for 2023 (\$213,224).

The Local Advertising Requirement for Group B is calculated as follows:

Columbia group: \$3,000 for the first location and \$1,500 each for the other 4 Designated Territories. (\$9,000)

Wilmington/Sunset Beach: \$3,000

Knoxville: \$3,000 for the first location and \$1,500 each for the other 4 Designated Territories. (\$9,000)

Suburban North and East Atlanta: \$3,000 for the first location and \$1,500 each for the other 4 Designated Territories. (\$9,000)

For a total of \$30,000/month multiplied by 12 months (\$360,000)

The Local Advertising Requirement can be increased to \$4,000 per month upon a 90-day notice. We have adjusted the number in the table to reflect advertising included in the Marketing, Advertising, & Promotion figure under Disclosed Expenses as well as salaries for employees who are dedicated to Sales (See Franchise Agreement Section 12.5.2 "salaries/commissions for one or more sales people with approval from Franchisor") for 2023 (\$164,909 + \$197,116 = \$362,025).

- 20. "Total Disclosed Expenses" is calculated by adding: (a) Management Compensation, Burden, & Benefits; (b) Miscellaneous G&A; (c) Marketing, Advertising, & Promotion; (d) Building Maintenance & Utilities; (e) Vehicle Expense; (f) Travel & Entertainment; (g) Insurance; (h) Professional Fees; (i) Royalty Fee; (j) Brand Fund Contribution; (k) Technology Fee; (l) Local Advertising Requirement; (m) Call Center Fee; (n) Annual Conference Fee; (o) Marketing Management Fee; (p) Online Local Presence Fee. The following expenses are excluded from Total Disclosed Expenses: interest, taxes, depreciation, amortization; and credit card rewards.
- 21. "Adjusted Profit" is calculated by taking the Gross Profit and subtracting that number by the Total Disclosed Expenses. Adjusted Profit is not equal to net profit or income and, except as to Cost of Goods Sold and Services Sold and Disclosed Expenses, does not include the deduction of all other expenses incurred by a Franchised Business including, but not limited to, other operating expenses, management wages, interest, taxes, depreciation and amortization.
- 22. There were no franchisees that were open and operating for 12 months in 2023.

+23. Affiliate-Owned Locations. "Affiliate-Owned Locations" means the nineten original Top Rail locations that operated a businessFranchised Businesses operating in 21 designated territories offering substantially similar services to those offered under the Top Rail brand and System. These businesses are operated as two Multi-Unit locationsdistinct businesses which we have designated as Group A and Group B.

Below is a chart detailing the city and state of the nine(9)10 Affiliate-Owned Locations as well as the date each of those began providing substantially similar services to the Approved Services that you will offer as a franchisee and the number of Designated Territory equivalents based on the size of territories currently be offered.

 Gross Revenue. "Gross Revenue" means the total revenue derived by the Affiliate Owned Locations by providing the Approved Services, less sales tax, discounts, allowances, and returns.

Location/ Designated Territory description	Group	Opening Date	Franchise Agreements/Designated Territories
Spartanburg, SC	A	June 2015	1
Greenville, SC (2 Territories)	A	June 2015	2
Anderson, SC	A	June 2015	1
Ashville Asheville, NC	A	June 2015	1
Suburban North and East Atlanta, GA	В	July 2022	5
Wilmington – Sunset Beach, NC	В	June 2018	1
Myrtle Beach, SC	В	October 2018	1
Charleston, SC	В	January 2020	2
Knoxville, TN/Johnson City, TNColumbia, SC	В	March 2020July 2021	<u>25</u>
Columbia, SCKnoxville, TN/Johnson City, TN	В	July 2021 March 2020	<u>52</u>

Spartanburg, Greenville, and Anderson are operated under one multi-unit addendum—(MUA #1)... Ashville is operated individually.—Suburban North and East Atlanta are operated under one multi-unit addendum (MUA #2). Wilmington—Sunset Beach is operated individually. Columbia, Charleston, and Myrtle Beach are operated under one multi-unit addendum—(MA#3). Wilmington—Sunset Beach is operated individually. Knoxville and Johnson City are operated under one multi-unit addendum—(MUA#4).

Suburban North and East Atlanta are operated under one multi-unit addendum.

Other than the preceding financial performance representation, we do not make any financial performance representations. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an existing Franchised Business, however,

we may provide you with the actual records of that Franchised Business. If you receive any other financial performance information or projections of your future income, you should report it to our management by contacting our Chief Development Officer, Zack Dudan, at 107 Parr Drive, Huntersville, NC 28078, (980) 441-1121, the Federal Trade Commission, and the appropriate state regulatory agencies.

ITEM 20 OUTLETS AND FRANCHISEE INFORMATION

Table No. 1 System-wide Outlet Summary For Years 20202021 to 20232022

Systemwide Outlet Summary

Systemwide Outlet Summary				
Outlet Type	Year	Outlets At The at the Start Of Theof the Year	Outlets At The at the End Of Theof the Year	Net Change
	<u>2021</u> 2020	0	0	0
Franchised	2021	0	0	0
Francinsed	2022	0	21	+21
	<u>2023</u>	<u>21</u>	<u>93</u>	<u>+72</u>
Company-	<u>2021</u> 2020	0	0	0
Owned and Affiliate	2021 <u>2022</u>	0	0	0
Owned*	2022 <u>2023</u>	0	0	0
	<u>2021</u> 2020	0	0	0
Total	2021	0	0	0
Outlets	2022	0	21	+21
	<u>2023</u>	<u>21</u>	<u>93</u>	<u>+72</u>

Table No. 2

<u>Transfers of Outlets from Franchisees to New Owners (Other than the Franchisor)</u>

<u>For Years 20202021 to 20232022</u>

State	_Year	Number of Transfers
	<u>2021</u> 2020	0
<u>Alabama</u> Total	2022 <mark>2021</mark>	0
	2022 <u>2023</u>	0
	<u>2021</u>	<u>0</u>
<u>Alaska</u>	<u>2022</u>	<u>0</u>
	<u>2023</u>	<u>0</u>
<u>Arizona</u>	<u>2021</u>	<u>0</u>
	<u>2022</u>	<u>0</u>

	2023	0
	<u>2023</u> 2021	0
Arkansas	2022	0
THRUISES	2023	0
	2021	0
California	2022	0
<u>Cumorina</u>	2023	0
	2021	0
Colorado	2022	0
00101440	<u>2023</u>	0
	2021	0
Connecticut	2022	0
	2023	0
	<u>2021</u>	0
Delaware	2022	0
	2023	0
	2021	0
District of Columbia	2022	0
	2023	0
	2021	0
Florida	2022	0
	2023	<u>0</u>
	<u>2021</u>	<u>0</u>
Georgia	<u>2022</u>	<u>0</u>
	<u>2023</u>	<u>0</u>
	<u>2021</u>	<u>0</u>
<u>Hawaii</u>	2022	<u>0</u>
	<u>2023</u>	<u>0</u>
	<u>2021</u>	<u>0</u>
<u>Idaho</u>	<u>2022</u>	<u>0</u>
	<u>2023</u>	<u>0</u>
	<u>2021</u>	<u>0</u>
<u>Illinois</u>	<u>2022</u>	<u>0</u>
	<u>2023</u>	<u>0</u>
	<u>2021</u>	<u>0</u>
<u>Indiana</u>	<u>2022</u>	<u>0</u>
	<u>2023</u>	<u>0</u>
	<u>2021</u>	<u>0</u>
<u>Iowa</u>	<u>2022</u>	<u>0</u>
	<u>2023</u>	<u>0</u>
Kansas	<u>2021</u>	<u>0</u>
<u>IXansas</u>	<u>2022</u>	<u>0</u>

	<u>2023</u>	0
Kentucky	2021	0
	2022	0
	2023	0
	2021	0
Louisiana	2022	0
	2023	0
	2021	0
Maine	<u>2022</u>	<u> </u>
	2023	0
	2021	<u>0</u>
Maryland	2022	0
	2023	0
	2021	<u>0</u>
Massachusetts	2022	0
1111051101111051115	2023	0
	2021	0
Michigan	2022	0
- Internation	2023	0
	2021	0
Minnesota	2022	0
	2023	<u>=</u> 0
	<u>2021</u>	0
Mississippi	<u>2022</u>	0
	2023	<u> </u>
	2021	0
Missouri	2022	0
	2023	0
	2021	0
Montana	2022	0
	<u>2023</u>	<u> </u>
	2021	0
<u>Nebraska</u>	2022	<u>0</u>
	2023	0
	<u>2021</u>	<u>0</u>
<u>Nevada</u>	2022	0
110111111	2023	0
	2021	0
New Hampshire	<u>2022</u>	<u> </u>
	2023	0
	2021	0
New Jersey	2022	0

	2023	0
	2021	0
New Mexico	2022	0
	2023	0
	<u>2021</u>	0
New York	2022	0
INCW TOTA	2023	0
	2021	0
North Carolina	2022	0
	2023	0
	<u>2021</u>	0
North Dakota	2022	<u>0</u>
	2023	0
	2021	0
Ohio	2022	0
	2023	0
	2021	0
Oklahoma	2022	0
	2023	0
	2021	0
Oregon	<u>2022</u>	<u>0</u>
	<u>2023</u>	0
	<u>2021</u>	<u>0</u>
Pennsylvania	<u>2022</u>	<u>0</u>
	<u>2023</u>	<u>0</u>
	<u>2021</u>	<u>0</u>
Puerto Rico	<u>2022</u>	<u>0</u>
	<u>2023</u>	<u>0</u>
	<u>2021</u>	<u>0</u>
Rhode Island	<u>2022</u>	<u>0</u>
	<u>2023</u>	<u>0</u>
	<u>2021</u>	<u>0</u>
South Carolina	<u>2022</u>	<u>0</u>
	<u>2023</u>	<u>0</u>
	<u>2021</u>	<u>0</u>
South Dakota	<u>2022</u>	<u>0</u>
	<u>2023</u>	<u>0</u>
	<u>2021</u>	<u>0</u>
Tennessee	<u>2022</u>	<u>0</u>
	<u>2023</u>	<u>0</u>
Texas	<u>2021</u>	<u>0</u>
ICAAS	<u>2022</u>	<u>0</u>

		T
	<u>2023</u>	<u>1</u>
	<u>2021</u>	<u>0</u>
<u>Utah</u>	<u>2022</u>	<u>0</u>
	<u>2023</u>	0
	<u>2021</u>	<u>0</u>
Vermont	<u>2022</u>	<u>0</u>
	<u>2023</u>	<u>0</u>
	2021	<u>0</u>
Virginia	2022	<u>0</u>
	2023	0
	<u>2021</u>	0
Washington	<u>2022</u>	<u>0</u>
	2023	0
	<u>2021</u>	<u>0</u>
West Virginia	2022	0
	2023	<u>0</u>
	2021	<u>0</u>
Wisconsin	<u>2022</u>	0
	2023	0
	<u>2021</u>	<u>0</u>
Wyoming	2022	<u>0</u>
	<u>2023</u>	0
	<u>2021</u>	<u>0</u>
TOTAL	2022	0
	2023	1

Table No. 3 Status of Franchised Outlets For Years 20202021 to 20232022

State	Year	Outlets At at Start Of Theof the Year	Outlets _Opened	Terminations Terminations	Non- Renewals	Reacquired By Re- acquired by Franchisor	Ceased Operations - Other Reasons	Outlets At at End Of Theof the Year
	2020 <u>2</u> 021	0	0	0	0	0	0	0
Georgia Alab ama	2021 <u>2</u> 022	0	0	0	0	0	0	0
	2022 <u>2</u> 023	0	<u>50</u>	0	0	0	0	<u>50</u>
	2020 <u>2</u> 021	0	0	0	0	0	0	0

North Carolina Alas	<u>2022</u> ⊋ 021	0	0	0	0	0	0	0
<u>ka</u>	2022 <u>2</u> 023	0	<u>20</u>	0	0	0	0	<u>20</u>
South	2020 <u>2</u> 021	0	0	0	0	0	0	0
Carolina Ariz	2022 ₂ 021	0	0	0	0	0	0	0
<u>ona</u>	2022 <u>2</u> 023	0	<u>90</u>	0	0	0	0	<u>90</u>
	2020 2 021	0	0	0	0	0	0	0
Tennessee Ar kansas	<u>2022</u> ⊋ 021	0	0	0	0	0	0	0
	2022 <u>2</u> 023	0	5 <u>1</u>	0	0	0	0	5 <u>1</u>
	2020 <u>2</u> 021	0	0	0	0	0	0	0
California _T	2022 ² 021	0	0	0	0	0	0	0
	2022 <u>2</u> 023	0	<u>210</u>	0	0	0	0	21 0
<u>Colorado</u>	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2023	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Connecticut	2022	0	0	0	0	0	<u>0</u>	0
	2023	0	0	0	0	0	<u>0</u>	0
	2021	0	0	0	<u>0</u>	<u>0</u>	<u>0</u>	0
Delaware	2022	0	0	0	<u>0</u>	<u>0</u>	<u>0</u>	0
	2023	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
District of Columbia	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Columbia	2023	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Florida</u>	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2023	<u>0</u>	<u>5</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>5</u>
	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Georgia	2022	<u>0</u>	<u>5</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>5</u>
	2023	<u>5</u>	<u>5</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>10</u>
	2021	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Hawaii	2022	0	0	0	0	0	0	0
	2023	0	0	0	0	0	0	0
Idaho	2021	0	0	0	0	0	0	0

	2022	0	0	<u>0</u>	0	0	0	0
	2023	0	0	0	0	0	0	0
	2021	0	0	0	0	0	0	0
Illinois	2022	0	0	0	0	0	0	0
	2023	0	4	0	0	0	0	4
	2021	0	0	0	0	0	0	0
Indiana	2022	0	0	0	0	0	0	0
	2023	0	<u>3</u>	<u> </u>	0	0	0	<u>3</u>
	2021	0	0	0	0	<u>0</u>	0	0
Iowa	2022	0	0	<u>0</u>	0	0	0	0
	2023	0	2	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	2
	<u>2021</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Kansas	2022	<u>0</u>						
	<u>2023</u>	<u>0</u>	<u>4</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>4</u>
	<u>2021</u>	<u>0</u>						
<u>Kentucky</u>	<u>2022</u>	<u>0</u>						
	<u>2023</u>	<u>0</u>						
	<u>2021</u>	0	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
<u>Louisiana</u>	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	<u>2023</u>	<u>0</u>	<u>3</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>3</u>
	<u>2021</u>	<u>0</u>						
Maine	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	<u>2023</u>	<u>0</u>						
	<u>2021</u>	<u>0</u>						
Maryland	<u>2022</u>	<u>0</u>						
	<u>2023</u>	<u>0</u>						
Massa	<u>2021</u>	<u>0</u>						
Massa- chusetts	<u>2022</u>	<u>0</u>						
	<u>2023</u>	<u>0</u>						
	<u>2021</u>	<u>0</u>						
Michigan	<u>2022</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2023</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2021</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Minnesota	<u>2022</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2023	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2021	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Mississippi	<u>2022</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	<u>2023</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0	0
Missouri	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0

	2022	0	0	<u>0</u>	0	0	<u>0</u>	0
	2023	0	1	0	0	0	0	1
	2021	0	0	0	0	0	0	0
Montana	2022	0	0	0	0	0	0	0
	2023	0	0	0	0	0	0	0
	2021	0	0	0	0	0	0	0
Nebraska	2022	0	0	0	0	0	0	0
	2023	0	0	<u> </u>	0	0	<u>0</u>	0
	2021	0	0	0	0	0	0	0
Nevada	2022	0	0	<u>0</u>	0	0	0	0
	2023	0	0	0	0	0	0	0
	2021	0	0	0	0	0	0	0
New 1	2022	0	0	0	0	<u>0</u>	0	0
<u>Hampshire</u>	2023	0	0	0	0	<u>0</u>	0	0
	2021	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
New Jersey	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2023	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2021	0	0	<u>0</u>	0	0	<u>0</u>	0
New Marriag	2022	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Mexico</u>	2023	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
New York	<u>2022</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2023	0	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	2021	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>North</u> Carolina	2022	0	<u>2</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>2</u>
Caronna	<u>2023</u>	<u>2</u>	<u>12</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>14</u>
NT d	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>North</u> Dakota	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Dukotu	<u>2023</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Ohio</u>	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2023</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Oklahoma</u>	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2023</u>	<u>0</u>	1	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	1
	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Oregon	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2023</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>

ъ .	2022	0	0	0	0	0	0	0
Pennsyl- vania	2022 2023	0	0	0	0	0	0	<u>0</u> 0
vania	2023	0	0	0	0	0	0	0
Duranta Dina			0	_	0	0	_	
Puerto Rico	2022	0	_	0	_	_	0	0
	2023	0	0	0	0	0	0	0
Rhode	2021	0	0	0	0	0	0	0
Island	2022	0	0	0	0	0	0	0
	2023	0	0	<u>0</u>	0	0	0	0
South	2021	0	0	0	0	0	0	0
Carolina	2022	0	9	0	0	0	<u>0</u>	9
	2023	9	1	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>10</u>
South	2021	0	0	0	0	0	0	0
Dakota	2022	0	0	<u>0</u>	0	0	0	0
	2023	0	0	0	0	0	<u>0</u>	0
	2021	0	<u>0</u>	<u>0</u>	0	0	0	0
<u>Tennessee</u>	2022	<u>0</u>	<u>5</u>	<u>0</u>	0	0	0	0
	<u>2023</u>	<u>5</u>	9	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>14</u>
_	<u>2021</u>	0	<u>0</u>	<u>0</u>	0	<u>0</u>	0	0
Texas	<u>2022</u>	0	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	<u>2023</u>	0	<u>13</u>	0(1)	<u>0</u>	<u>0</u>	0	<u>13</u>
	2021	0	0	<u>0</u>	0	0	0	0
<u>Utah</u>	2022	0	0	<u>0</u>	0	<u>0</u>	0	0
	<u>2023</u>	0	<u>3</u>	<u>0</u>	<u>0</u>	<u>0</u>	0	<u>3</u>
	<u>2021</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Vermont</u>	<u>2022</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	<u>2023</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	<u>2021</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
<u>Virginia</u>	<u>2022</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	<u>2023</u>	0	2	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	2
Washing-	<u>2021</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
ton	2022	0	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	<u>2023</u>	0	<u>3</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>3</u>
West	<u>2021</u>	0	<u>0</u>	<u>0</u>	0	<u>0</u>	0	0
Virginia	<u>2022</u>	0	0	<u>0</u>	0	0	0	0
	<u>2023</u>	0	0	<u>0</u>	0	0	0	0
	<u>2021</u>	0	0	<u>0</u>	<u>0</u>	0	<u>0</u>	<u>0</u>
Wisconsin	2022	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2023</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Wyoming	<u>2021</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>

	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2023	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
TOTAL	2022	<u>0</u>	<u>21</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>21</u>
	2023	<u>21</u>	<u>72</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>93</u>

⁽¹⁾ A Franchisee purchased 2 territories in Texas and did not pay the initial fee or attend training and stopped communicating. The business was never opened by the franchisee. The 2 franchise agreements were terminated. Because the franchises were never opened, these two franchises are not included in the terminations number for Texas.

Table No. 4
Status of Company-Owned Outlets
For Years 20202021 to 20232022

State	Year	Outlets At at Start Of Theof Year	Outlets Opened	Outlets Reacquired Re- acquired From Franchisee Franchisees	Outlets Closed	Outlets Sold To to Franchisee	Outlets At at End Of Theof Year
	2020 <u>2021</u>	0	0	0	0	0	0
Total Alabama	<u>2022</u> 2021	0	0	0	0	0	0
	2022 <u>2023</u>	0	0	0	0	0	0
	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Alaska</u>	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2023</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
Arizona	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	2023	<u>0</u>	0	<u>0</u>	0	<u>0</u>	<u>0</u>
	2021	<u>0</u>	0	<u>0</u>	0	<u>0</u>	<u>0</u>
Arkansas	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2023	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
California	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2023	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Colorado</u>	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2023	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Connecticut	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2023	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Delaware	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
	<u>2021</u>	<u>0</u>	0	<u>0</u>	0	<u>0</u>	<u>0</u>
District of Columbia	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Columbia	2023	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Florida</u>	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2023	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Georgia	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>

I				Ī		I	
	<u>2022</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	<u>2023</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Hawaii</u>	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2023</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Idaho</u>	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2023</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Illinois</u>	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2023</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Indiana</u>	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2023	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Iowa</u>	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2023	0	0	0	0	<u>0</u>	0
Kansas	2021	0	0	0	0	<u>0</u>	0
	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2023	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Kentucky	2022	0	0	0	0	<u>0</u>	0
	2023	0	0	0	0	<u>0</u>	0
	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
Louisiana	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	2023	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
<u>Maine</u>	2022	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	2023	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
	2021	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
Maryland	2022	0	0	0	0	<u>0</u>	<u>0</u>
	2023	0	0	0	0	0	0
	2021	0	0	<u>0</u>	0	<u>0</u>	0
Massachusetts	2022	0	0	<u>0</u>	0	0	0
	2023	0	0	0	0	0	0
	2021	0	0	0	0	0	0
Michigan	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
Minnesota	2021	0	0	0	0	0	0

	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
	2021	0	0	0	0	0	0
Mississippi	2022	0	0	0	0	0	0
<u> </u>	2023	0	0	0	0	0	0
	2021	0	0	0	0	0	0
Missouri	2022	0	0	0	0	0	0
1/11555411	2023	0	0	0	0	0	0
	2021	0	0	0	0	0	0
Montana	2022	0	0	0	0	0	0
<u>iviolitalia</u>	2023	0	0	0	0	0	0
	<u>2023</u> <u>2021</u>	0	0	0	0	0	0
Nebraska	2022	0	0	0	0	0	0
INCOTASKA	2023	0	0	0	0	0	0
	<u>2023</u>	0	0	0	0	0	0
<u>Nevada</u>	2022	0	0	0	0	0	0
<u>140 vada</u>	2023	0	0	0	0	0	0
	2021	0	0	0	0	0	0
<u>New</u> <u>Hampshire</u>	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
	2021	0	0	0	0	0	0
New Jersey	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
	2021	0	0	0	0	0	0
New Mexico	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
	2021	0	0	0	0	0	0
New York	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
	2021	0	0	0	0	0	0
North	2022	0	0	0	0	0	0
<u>Carolina</u>	2023	0	0	0	0	0	0
	2021	0	0	0	0	0	0
North Dakota	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
	2021	0	0	0	0	0	0
Ohio	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
Oklahoma	2021	0	0	<u>0</u>	0	0	0

2022 Q Q Q Q Q Q Q Q Q
Oregon 2021 0
Oregon 2022 0
Pennsylvania 2021 0 0 0 0 0 0 0
Pennsylvania 2021 0
Pennsylvania 2022 0 0 0 0 0 0 2023 0 0 0 0 0 0 0 Puerto Rico 2021 0 0 0 0 0 0 0 2022 0 0 0 0 0 0 0 0 2023 0 0 0 0 0 0 0 0
2023 0 0 0 0 0 0 2021 0 0 0 0 0 0 0 Puerto Rico 2022 0 0 0 0 0 0 0 0 2023 0 0 0 0 0 0 0 0 0
Puerto Rico 2021 0 0 0 0 0 0 2022 0 0 0 0 0 0 0 2023 0 0 0 0 0 0 0
Puerto Rico 2022 0 0 0 0 0 0 2023 0 0 0 0 0 0 0
<u>2023</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
$\begin{array}{c c c c c c c c c c c c c c c c c c c $
<u>Rhode Island</u> <u>2022</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
$egin{array}{ c c c c c c c c c c c c c c c c c c c$
$\underline{2021}$ $\underline{0}$ $\underline{0}$ $\underline{0}$ $\underline{0}$ $\underline{0}$ $\underline{0}$
South Carolina 2022 0 0 0 0 0
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
<u>2021</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
<u>South Dakota</u> <u>2022</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
<u>2021</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
<u>Tennessee</u> <u>2022</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$
<u>2021</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
<u>Texas</u> <u>2022</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
<u>2021</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
<u>Utah</u> <u>2022</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$
<u>2021</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
<u>Vermont</u> <u>2022</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
<u>2023</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
<u>2021</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
<u>Virginia</u> <u>2022</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
<u>2023</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
<u>2021</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
<u>Washington</u> <u>2022</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
<u>2023</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>
West Virginia 2021 0 0 0 0 0

	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2023</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
Wisconsin	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2022</u>	0	<u>0</u>	<u>0</u>	0	<u>0</u>	<u>0</u>
	<u>2023</u>	0	0	0	0	0	0
	<u>2021</u>	0	0	<u>0</u>	0	<u>0</u>	<u>0</u>
Wyoming	2022	0	0	0	0	0	0
	<u>2023</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2021</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
TOTAL	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2023</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>

<u>Table No. 5</u>
Projected Openings As Of
December 31, 20232022

State	Franchise Agreements Signed But Outlet Not Opened	Projected New Franchised Outlets In The in the Next Fiscal Year	Projected New Affiliate Company Owned Outlets In The Next in the Current Fiscal Year
Alabama	<u>01</u>	<u>23</u>	0
Arizona	0	<u>40</u>	0
Arkansas	0	<u> 40</u>	0
<u>California</u>	<u>0</u>	<u>4</u>	<u>0</u>
Colorado	<u>010</u>	<u>210</u>	0
Connecticut	1	<u>4</u>	<u>0</u>
<u>Delaware</u>	<u>0</u>	<u>0</u>	<u>0</u>
District of Columbia	<u>0</u>	<u>0</u>	<u>0</u>
Florida	<u>010</u>	10 <u>18</u>	0
Georgia	0	<u>67</u>	0
<u>Hawaii</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Idaho</u>	<u>2</u>	<u>2</u>	<u>0</u>
Illinois	0	<u>24</u>	0
Indiana	0	<u> 20</u>	0
<u>Iowa</u>	<u>0</u>	<u>0</u>	<u>0</u>
Kansas	<u>0</u>	1	<u>0</u>
<u>Kentucky</u>	<u>0</u>	<u>3</u>	<u>0</u>
Louisiana	0	4 <u>1</u>	0
<u>Maine</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Maryland</u>	<u>0</u>	<u>0</u>	<u>0</u>

Massachusetts	0	<u>0</u>	0
Michigan	4	4	0
Minnesota	4(1)	7	0
	<u>-</u>	_	+
Mississippi	1	1	0
Missouri	<u>0</u>	0	<u>0</u>
<u>Montana</u>	0	0	<u>0</u>
<u>Nebraska</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>Nevada</u>	<u>0</u>	<u>0</u>	<u>0</u>
New Hampshire	<u>0</u>	<u>0</u>	<u>0</u>
New Jersey	0	2 <u>5</u>	0
New Mexico	<u>0</u>	<u>0</u>	<u>0</u>
New York	<u>0</u>	<u>2</u>	<u>0</u>
North Carolina	3 0	4 <u>2</u>	0
North Dakota	<u>0</u>	<u>0</u>	<u>0</u>
<u>Ohio</u>	<u>3</u>	<u>8</u>	<u>0</u>
<u>Oklahoma</u>	<u>0</u>	<u>4</u>	<u>0</u>
Oregon	<u>0</u>	<u>0</u>	<u>0</u>
Pennsylvania	0	<u>34</u>	0
Puerto Rico	<u>0</u>	<u>0</u>	<u>0</u>
Rhode Island	<u>0</u>	<u>0</u>	<u>0</u>
South Carolina	0	4 <u>2</u>	0
South Dakota	<u>0</u>	<u>0</u>	<u>0</u>
Tennessee	0	4 <u>0</u>	0
Texas	0 11 ⁽²⁾	10 11	0
Utah	0	<u>21</u>	0
Vermont	<u>0</u>	<u>0</u>	<u>0</u>
Virginia	0	<u>23</u>	0
Washington	<u>32</u>	<u>04</u>	0
Totals West Virginia	<u>60</u>	64 0	0
Wisconsin	<u>0</u>	<u>2</u>	<u>0</u>
Wyoming	<u>0</u>	<u>0</u>	<u>0</u>
TOTAL	49	117	0
(1) T1 4 4 4 4		1. T1 C 1:	

⁽¹⁾ These 4 territories were originally sold in Nevada. The franchisees subsequently requested we amend the Franchise Agreement to change the Designated Territory to Minnesota.

In some instances, current and former franchisees will sign provisions restricting their ability to speak openly about their experience with the System. <u>Some franchisees have signed confidentiality clauses</u>

⁽²⁾ A Franchisee purchased 2 territories in Texas and did not pay the initial fee or attend training and stopped communicating. The business was never opened by the franchisee. The 2 franchise agreements were terminated. Because of the termination, these two franchises are not included in the sold but not open number for Texas.

in the last three years. You may wish to speak to current and former franchisees, but be aware that not all such franchisees will be able to communicate with you. There are no trademark-specific organizations formed by our franchisees that are associated with our System. If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

As of the Issuance Date of this Franchise Disclosure Document, there There are no franchise organizations sponsored or endorsed by us, and no independent franchisee organizations have askedrequired to be included in this Franchise Disclosure Document.

There are no trademark-specific organizations formed by our franchisees that are associated with our System. If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

ITEM 21 FINANCIAL STATEMENTS

Attached as Exhibit EI to this Franchise Disclosure Document are our audited financial statements as of December 31, 2022, and December 31, 2023. We have not been in business for three years or more, and therefore cannot include the same financial statements as a franchisor that has been in business for three or more years. Our fiscal year end is December 31st.

ITEM 22 CONTRACTS

Attached to this Franchise Disclosure Document are the following contracts and their attachments:

Exhibit B	Franchise Agreement and Schedules
Exhibit C	Statement of Ownership
Exhibit D	Confidentiality and Restrictive Covenant
Exhibit E	Electronic Funds Withdrawal Authorization
Exhibit EF	Multi-Unit Addendum
Exhibit D	State Specific Addenda
Exhibit FG	Sample Assignment and Assumption Agreement
Exhibit H	State Specific Addenda
Exhibit L	Diligence Review of Franchisee Sales Process

ITEM 23 RECEIPTS

Exhibit <u>KN</u> of this Franchise Disclosure Document contains a detachable document, in duplicate, acknowledging receipt of this Franchise Disclosure Document by a prospective franchisee. You should sign both copies of the Receipt. You should retain one signed copy for your records and return the other signed copy to HFB FenceCo Franchising, LLC, 107 Parr Drive, Huntersville, North Carolina 28078.

EXHIBIT Exhibit A

TO

to FDD

HFB FENCECO FRANCHISINGFenceCo Franchising, LLC FRANCHISE DISCLOSURE DOCUMENT

$\underline{\textbf{LIST OF STATE ADMINISTRATORS AND AGENTS FOR SERVICE OF PROCESS}}$

LIST OF STATE ADMINISTRATORS AND AGENTS FOR SERVICE OF PROCESS

Listed here are the names, addresses and telephone numbers of the state agencies having responsibility for franchising disclosure/registration laws and for service of process. We may not yet be registered to sell franchises in any or all of these states.

LIST OF STATE ADMINISTRATORS

CALIFORNIA

(state administrators)

Department of Financial Protection and Innovation 320 West 4th Street, Suite 750 Los Angeles, CA 90013 (213) 576-7500 Toll Free (866) 275-2677

2101 Arena Blvd. Sacramento, CA 95834 (866) 275-2677

1350 Front Street San Diego, CA 92101 (619) 525-4233

One Sansome St., #600 San Francisco, California 94104 (415) 972-8559

(agents for service of process)
California Commissioner of the Department of Financial Protection and Innovation
320 West 4th Street, Suite 750
Los Angeles, CA 90013-2344

Commissioner of Department of Financial Protection and Innovation
One Sansome Street #600

San Francisco, California 94104

Commissioner of Department of Financial Protection

and Innovation 2101 Arena Blvd. Sacramento, CA 95834

CONNECTICUT

(state administrator) State of Connecticut Department of Banking Securities & Business Investments Division 260 Constitution Plaza Hartford, CT 06103-1800 (860) 240-8230

(agent for service of process) Banking Commissioner

LIST OF STATE ADMINISTRATORS

HAWAII

(state administrator)

Business Registration Division
Department of Commerce and Consumer Affairs
Business Registration Division
Securities Compliance Branch
335 Merchant Street, Room 203
Honolulu, Hawaii 96813
(808) 586-2722

(agent for service of process)
Commissioner of Securities
State of Hawaii
Department of Commerce and Consumer Affairs
Business Registration Division
Securities Compliance Branch
335 Merchant Street, Room 203
Honolulu, Hawaii 96813

ILLINOIS

Franchise Bureau
Office of the Illinois
500 South Second Street
Springfield, Illinois 62706
(217) 782-4465

<u>INDIANA</u>

(808) 586-2722

(state administrator)
Indiana Secretary of State Securities Commissioner
Securities Division, E-111
302 Washington Street, Room E-111
Indianapolis, Indiana 46204
(317) 232-6681

(agent for service of process)
Indiana Secretary of State
201 State House
200 West302 Washington Street, Room E-018
Indianapolis, Indiana 46204
(317) 232-6531

MARYLAND

(state administrator)
Office of the Attorney General
Division of Securities Division
200 St. Paul Place
Baltimore, Maryland 21202-2021
(410) 576-6360

(agent for service of process) Maryland Securities Commissioner 200 St. Paul Place Baltimore, Maryland 21202-2021 (410) 576-6360

LIST OF STATE ADMINISTRATORS

MICHIGAN

(state administrator)

Consumer Protection Division

Antitrust and Franchise Unit

Michigan Department of Attorney General

Consumer Protection Division

525 W. Ottawa Street

G. Mennen Williams Building, 1st Floor

-Lansing, Michigan 48933

(517) 373-7117

(agent for service of process)

Corporations Division

Bureau of Commercial Services

Michigan Department of Labor and Economic

GrowthAttorney General

P.O. Box 30054

Consumer Protection Division

525 W. Ottawa Street

G. Mennen Williams Building, 1st Floor

Lansing, Michigan 4890948933

(517) 373-7117

MINNESOTA

(state administrator) Minnesota Department of Commerce 85 7th Place East, Suite 280 St. Paul, Minnesota 55101 (651) 539-1500

(agent for service of process)
Minnesota Commissioner of Commerce
Minnesota Department of Commerce
85 7th Place East, Suite 280
St. Paul, Minnesota 55101
(651) 539-1500

NEW YORK

(state administrator)
Office of the New York State Attorney General Attention: Barbara Lasoff
NYS Department of Law
Investor Protection Bureau
Franchise Section
120 Broadway, 23rd Floor
28 Liberty St. 21st FL
New York, NY 10271 033210005
(212) _416-8236

New York Department of State Attention: UCC
One Commerce Plaza,
99 Washington Avenue, 6th Floor Albany, NY 12231-0001
(518) 473-2492

(agent for service of process)

NORTH DAKOTA

Securities Commissioner North Dakota Securities Department State Capitol, Fifth Floor, Dept. 414 600 East Boulevard Avenue State Capitol, 14th Floor, Dept. 414 Bismarck, North Dakota 58505 (701) 328-4712

LIST OF STATE ADMINISTRATORS **OREGON** RHODE ISLAND Department of Insurance and Finance Director, Department of Business Regulation Corporate Securities Section Securities Division Labor and Industries Building Department of Business Regulation, Salem, Oregon 97310 Bldg. 69, First Floor (503) 378-4387 1511 Pontiac Avenue John O. Pastore Center - Bldg. 68-2 1511 Pontiac Avenue Cranston, Rhode Island 02920 (401) 462-9582 SOUTH DAKOTA **VIRGINIA** Department of Labor and Regulation (state administrator) Division of Insurance State Corporation Commission Division of Securities and Retail Franchising - Securities Regulation 124 S. Euclid, Suite 104 1300 East Main Street, 9th Floor Pierre, South Dakota 57501 Richmond, Virginia 23219 (804) 371-9051 (605) 773-3563 (for service of process) Clerk of the State Corporation Commission 1300 East Main Street, 1st Floor Richmond, Virginia 23219 (804) 371-9733 WASHINGTON WISCONSIN (state administrator) (state administrator) Department of Financial Institutions Commissioner of Securities Securities Division Division of Securities P.O. Box 9033 Department of Financial Institutions Olympia, Washington 98507-9033 <u>Division of Securities</u> (360) 902-8760 201 W Washington Avenue, 3rd Floor 4822 Madison Yards Way, North Tower (agent for service of process) Madison, Wisconsin 5370353705 Director, Department of Financial Institutions (608) 266-1064 Securities Division 150 Israel Road S.W. (agent for service of process) Tumwater, Washington 98501 Administrator, DivisionCommissioner of Securities Department of Financial Institutions (360) 902-8760 201 W Washington Avenue, 3rd Floor

Division of Securities

4822 Madison Yards Way, North Tower Madison, Wisconsin 5370353705

EXHIBIT Exhibit B TO HIFB FENCECO FRANCHISING, LLC FRANCHISE DISCLOSURE DOCUMENT to FDD

HFB FenceCo Franchising, LLC

FRANCHISE AGREEMENT

EXHIBIT B



HFB FENCECO FRANCHISING, LLC FRANCHISE AGREEMENT

Franchise #:	
Franchisee:	
Date:	
Designated Territory:	

See Map and list of zip codes on the Data Sheet to this Franchise Agreement

TABLE OF CONTENTS

CTIO	<u>N</u>	PAGE
1	GRANT OF FRANCHISE	2
2	TERM AND RENEWAL	5
3	—FEES	6
4 —	PROPRIETARY MARKS	11
5 —	—CONFIDENTIAL INFORMATION	13
6	FRANCHISOR'S OBLIGATIONS	15
7 —	FRANCHISEE'S OBLIGATIONS	17
8	—TRAINING	26
9—	—INSURANCE	27
10	FINANCIAL RECORDS AND REPORTS	28
11-	BOOKS AND RECORDS	29
12 -	-ADVERTISING	31
13	INDEPENDENT BUSINESS; INDEMNIFICATION	35
14 -	—SALE OR TRANSFER	36
15—	BREACH AND TERMINATION	41
16	RIGHTS AND DUTIES UPON TERMINATION OR EXPIRATION	44
17—	-COVENANTS	46
18 —	DISPUTE RESOLUTION	48
19	REPRESENTATIONS	52
20	GUARANTEE OF PRINCIPALS AND THEIR SPOUSES	53
21—	-NOTICES	54
22—	MISCELLANEOUS	55
1	GRANT OF FRANCHISE	2
2	TERM AND RENEWAL	6
3	FEES	7
4	PROPRIETARY MARKS	13
5	CONFIDENTIAL INFORMATION	15
	FRANCHISOR'S OBLIGATIONS	
7	FRANCHISEE'S OBLIGATIONS	19
8	TRAINING	28
9	INSURANCE	
10	FINANCIAL RECORDS AND REPORTS	31
	BOOKS AND RECORDS	
12	ADVERTISING	33

13	INDEPENDENT BUSINESS; INDEMNIFICATION	38
14	SALE OR TRANSFER	39
15	BREACH AND TERMINATION	<u></u> 43
16	RIGHTS AND DUTIES UPON TERMINATION OR EXPIRATION	48
17	COVENANTS	50
18	DISPUTE RESOLUTION	<u></u> 52
19	REPRESENTATIONS	<u>5</u> 6
20	GUARANTEE OF PRINCIPALS AND THEIR SPOUSES	<u></u> 57
21	NOTICES	58
22	MISCELLANEOUS	59

EXHIBITS:

EXHIBIT A	Personal Guaranty and Guaranty of Spouses
Exhibit B	Conditional Assignment of Franchisee's Telephone Numbers, Facsimile Numbers and
	Domain Names
Exhibit C	Confidentiality and Restrictive Covenant Agreement for Employees
Exhibit D	Electronic Funds Withdrawal Authorization
Exhibit E	Site Selection Addendum
Exhibit F	Statement of Ownership
Exhibit G	Collateral Assignment of Lease
The faithful TT	Formation Discount Advantage Construction



DATA SHEET

Franchisee:	
Guarantors:	
Effective Date:	
Approved Location:	
Designated Territory:	See mapMap and list of Zip Codes on the following page-
Telephone Number:	
Email Address:	
Initial Franchise Fee:	\$
Initial Franchisor Directed Advertising Fee: Oper Assistance Fee:	

Multi-Unit Addendum: \longrightarrow Yes \longrightarrow No

Territory Map and Zip Codes – the "Designated Territory"

[insert map and list of zip codes]

In the event of a conflict between the map and zip codes, the zip codes control.

[insert map and list of zip codes]

The terms of this Data Sheet are incorporated into the attached Franchise Agreement.

$\begin{array}{c} \text{HFB FENCECO FRANCHISING, LLC} \\ \underline{FRANCHISE\ AGREEMENT} \end{array}$

This FRANCHISE AGREEM	ENT (the "Agree	reement" or "Franchise Agreement") is entered into
and made effective on	, 20	(the "Effective Date"), by and between: (i) HFE
FenceCo Franchising, LLC, a North Ca	rolina limited lia	liability company, with its principal place of business
at 107 Parr Drive, Huntersville, North O	Carolina 28078 ("	("Franchisor"); and
	, 8	a(n)
with its principal place of business at,		
which is identified more fully in the att	ached Data Shee	eet ("Franchisee").

RECITALS

- A. Through the expenditure of a considerable amount of time, effort and money, Franchisor has developed a system for the operation of franchised businesses under its then-current proprietary marks currently "Top Rail®" (each, a "Franchised Business(es)") that sell, furnish and install wood, steel, aluminum, vinyl, and other types of fencing and related garden products to residential and commercial customers, and any services that Franchisor may later designate (collectively, the "Approved Services").
- B. Franchised Businesses are established and operated using Franchisor's proprietary operating system, the distinguishing characteristics of which include: (i) valuable know-how, information, trade secrets and methods; (ii) Franchisor's proprietary standards and specifications for certain products and services used in connection with providing Franchisor's Approved Products and Approved Services to customers; (iii) certain proprietary products developed by Franchisor (the "Approved Products"); (iv) Franchisor's standards and specifications for sales techniques, marketing and advertising programs, and call center; (v) Franchisor's proprietary, centralized system-wide call center; (vi) proprietary initial and ongoing training programs; and (vivii) standards and specifications for operating the Franchised Business in the manner set forth in this Agreement and Franchisor's proprietary and confidential Operations Manual (defined in Section 6.1) that franchisees have access to, which may be modified from time to time by Franchisor (collectively, the "System").
- C. The System is identified by Franchisor's proprietary trademarks, service marks, trade dress, logos and other indicia of origin (collectively, the "Proprietary Marks"). The rights to all such Proprietary Marks as are now, or shall hereafter be, designated as part of the System will be owned exclusively by Franchisor or its affiliates and be used for the benefit of Franchisor, its affiliates, and Franchisor's franchisees to identify to the public the source of the products and services marketed thereunder. Franchisor may continue to develop, expand, use, control, and add to the Proprietary Marks and System for the benefit of itself, its affiliates, and its franchisees and licensees in order to identify for the public the source of products and services marketed thereunder and to represent the System's high standards of quality and service. For purposes of this Franchise Agreement Franchisor affiliates means DDL Investments, Inc., National Account Services Company, LLC, Bold Future Payment Services, LLC, HFB Holdco, LLC, franchisors and other companies majority owned by HFB HoldCo, LLC, and each of their subsidiaries or parents.
- D. Franchisor offers franchises for the development and operation of Franchised Businesses to be operated and promoted within a designated geographical territory <u>identified on the Data Sheet</u> (the "Designated Territory").

- E. Franchisee desires to establish and operate a Franchised Business within the Designated Territory hereinafter designated, to use in connection therewith, Franchisor's System and the Proprietary Marks and to derive the benefits of Franchisor's information, experience, advice, guidance and customer goodwill.
- F. Franchisor desires to grant Franchisee the right to open and operate a Franchised Business based on Franchisee's representations to Franchisor, including those representations set forth in Franchisee's franchise application, in accordance with the terms and conditions set forth in this Agreement.
- G. Franchisee recognizes the importance to Franchisor, to its other franchisees, and to the public of maintaining the integrity, standards, qualities and attributes of products and services associated with the Proprietary Marks and System, and is willing to adhere to certain uniform standards, procedures and policies to maintain such integrity, standards, qualities and attributes.

NOW, THEREFORE, in consideration of the foregoing recitals and the mutual promises, commitments and understandings contained herein, Franchisor and Franchisee hereby agree as follows:

1. GRANT OF FRANCHISE

Grant and Acceptance. Franchisor hereby grants to Franchisee, upon the express terms and conditions contained in this Agreement, and Franchisee hereby accepts, a non-exclusive license to establish and operate one (1) Franchised Business under the System and Proprietary Marks identified below, and the right to use the System and Proprietary Marks in the operation of the Franchised Business. Except as otherwise provided in this Agreement, Franchisee may offer and sell only Franchisor's Approved Products and Approved Services within the Designated Territory set forth in Section 1.2 of this Agreement and the Data Sheet. Franchisee must request permission to provide additional/new Products and Approved Services from Franchisor, which permission Franchisor may grant, deny, or condition in its sole discretion. In addition, Franchisee may provide Franchisor's Approved Products and Approved Services in adjacent territories (as those are determined by Franchisor) if permitted by and subject to the terms and conditions set forth in the Operations Manual (defined in Section 6.1). Franchisor has the right to supplement, improve or otherwise modify the System from time to time in Franchisor's discretion, and Franchisee agrees to comply with all changes which may include, without limitation, the offer and sale of new or different products or services as Franchisor may specify. Franchisee acknowledges and agrees that this Agreement does not grant Franchisee the option or right to purchase additional Franchised Businesses and/or additional Designated Territories designated territories.

1.1.1 System Updates. Franchisee agrees that it is critical for Franchisee, franchisor and other franchisees that the System is flexible to respond to commercial opportunities and challenges. An inability to change the System could adversely affect all franchisees. Therefore, Franchisee agrees and anticipates that the Operations Manual and the System may be changed by Franchisor, from time to time in our Business Judgment. Franchisee agrees to comply with the specifications and standards set forth in the Operations Manual and that make up the System as they are changed by Franchisor. Therefore, additional services may be added in our Business Judgment upon notice to Franchisee. The Royalty Fee and/or Brand Fund Fee for new services may be higher or lower than the Royalty Fee and/or Brand Fund Fee for existing approved services may be higher or lower than the Royalty Fee/Brand Fund Fee of an additional 3% of Gross Revenue over the then current rates. Any added products or services that Franchisee performed prior to the addition of such products or services to the definition of Approved Products or Approved Services permitted will be subject to the then current Royalty Fees, Brand Fund fees, and Local Advertising Requirement chosen for the newly Approved Products and Approved Services. Franchisor may, but is not required to, provide a ramp-up schedule for application of fees to new service as determined in our Business

Judgment. If Franchisee has commenced operations within 6 months of the date of the notice from Franchisor, then Franchisee will have a period of up to 12 months as determined in our Business Judgment, from the date of the notice from Franchisor to complete the upgrade or add a new service. However, in any case, Franchisee is not obligated to complete such upgrade or add a new service if (a) the notice from Franchisor is delivered to Franchisee within 6 months before the end of the Term of this Agreement, and either (b) Franchisor does not offer an extension or renewal under this Agreement, or (c) Franchisee provides notice of its intent not to renew. Franchisor may remove Approved Products or Approved Services in our Business Judgment upon notice to Franchisee. Franchisee will have up to 6 months to cease providing services that are no longer approved, unless Franchisor specifies otherwise in its sole discretion, in which case the period to cease providing services may be immediate or such longer period as Franchisor deems appropriate in its sole discretion.

- Designated Territory. Except as otherwise provided in this Agreement (and the Operations Manual (defined in Section 6.1)) and for so long as Franchisee is not in default of this Agreement, Franchisor shall not establish and operate, nor license any other third party the right to establish and operate, any Franchised Business under the System and the Proprietary Marks within the protected area identified in the Data Sheet, the terms of which are incorporated herein by reference (the Designated Territory) during the term of this Agreement. Franchisor and its affiliates retain all other rights, including, without limitation, those rights set forth in Sections 1.4 through 1.8 of this Agreement. Franchisee is permitted to conduct the Franchised Business outside of the Designated Territory in adjacent open territory (i.e., not the Designated Territory of another franchisee) if Franchisee receives Franchisor's prior written consent, and in compliance with -our then current Adjacent Territory Policy (described in the Operations Manual). The Adjacent Territory Policy is subject to modification or elimination by a change to the Operations Manual. Franchisee may not operate within another franchisee's Designated Territory, except for certain limited circumstances, if any, that are more fully set forth in the Operations Manual, and other franchisees may operate in your Franchisee's Designated Territory, in limited circumstances, if any, as set forth in the Operations Manual. Other than these operations, Franchisee is not permitted to operate or market the Franchised Business outside of the Designated Territory without Franchisor's prior written consent. All sales and other activities conducted within or outside the Designated Territory must be conducted in accordance with the terms of this Agreement and Franchisor's operating methods, standards and specifications as set forth in the Operations Manual (defined in Section 6.1).
- Approved Location. Franchisee must operate the Franchised Business from a location that is approved by Franchisor and meets Franchisor's standards and specifications (the "Approved Location"). Upon execution of this Agreement and for a period of 12 months after the Effective Date of this Agreement, Franchisee may operate the Franchised Business from an Approved Location that is a home office or storage facility that is approved by Franchisor and meets Franchisor's standards and specifications (the "Approved Location"). Franchisee must secure this initial Approved Location within 90 days of the Effective Date of this Agreement. After this initial 12-month period, Franchisee shall operate the Franchised Business from an Approved Location that is a leased office/warehouse or storge facility within the Protected Designated Territory that is approved by Franchisor and meets Franchisor's standards and specifications. Franchisee must lease approximately 1,000 to 1,500 square feet of commercial real estate for the office/warehouse and to securely store equipment/inventory. The office may be located in the warehouse or a home office where a storage space is approved, provided that it meets Franchisor's standards and specifications. Franchisor may provide Franchisee with standards and specifications for the design and layout of the premises of the Approved Location, and Franchisor must review and approve any proposed location, as well as any lease associated with the proposed location, prior to Franchisee entering into any lease for the proposed location. If Franchisor has not approved a location from which Franchisee must operate the Franchised Business as of the date Franchisee signs this Agreement, the parties will enter into Franchisor's prescribed form of Site Selection Addendum in follow the form attached as Exhibit E to obligations set forth in Section 7.1 of this Franchise Agreement, the terms of which will govern the parties'

site selection obligations. Franchisee may not relocate the Franchised Business without Franchisor's prior written consent.

- 1.4 **National Accounts.** Franchisor will have the exclusive right, on behalf of itself, its affiliate(s), Franchisee, and/or other franchisees utilizing the Proprietary Marks, to negotiate and enter into agreements or approve forms of agreement to provide services to "National Accounts," (defined in Section 1.4.1), including National Accounts that Franchisee has solicited or serviced. Franchisee may not solicit any National Accounts outside of the Designated Territory, or solicit any National Accounts within or outside of the Designated Territory who are already under contract with Franchisor.
- 1.4.1 The term "National Account" means any business or businesses under common control, ownership, or branding which operate locations in, or deliver products and services beyond one designated territory, regardless of the volume of products and/or services to be purchased by the customer. Any dispute as to whether a particular customer is a National Account will be determined by Franchisor, in its sole discretion, and Franchisor's determination will be final and binding.
- 1.4.2 Franchisee may choose to participate in the National Accounts program ("National Accounts Program") to service National Accounts for which the ultimate client of the National Account is located in the Designated Territory consistent with the then-current National Accounts Program Participation Agreement. Franchisee acknowledges and agrees that Franchisor or a third party service provider Franchisor designates may begin servicing National Accounts in the Designated Territory if: (i) Franchisor believes that Franchisee is not meeting Franchisor's standards and specifications applicable to the National Accounts Franchisee is servicing or the National Accounts Program generally; (ii) Franchisor believes that Franchisee is in default of this Agreement; (iii) the National Account has requested that Franchisor or a third party it selects service its clients; (iv) Franchisee is not then participating in the National Accounts Program (or is not eligible to participate in the National Accounts Program); (v) Franchisee does not offer or is not authorized by Franchisor or the applicable licensing authority to offer the service to be provided; or (vi) Franchisee refuses to service the National Account.
- 1.4.3 Franchisee agrees that neither the direct provision by Franchisor or a franchisee, licensee, or designee of Franchisor of services to National Account customers as authorized above, nor Franchisor's contracting with another party to provide such services as authorized above, shall constitute a violation of the grant of license contained in this Agreement or any other provision of this Agreement, even if such services are delivered from a location within the Designated Territory. Franchisee disclaims any right to compensation or consideration for work performed by others in the Designated Territory pursuant to this Section.
- 1.4.4 Franchisor has the right to discontinue or terminate a National Account or the entire National Accounts Program. Franchisor may impose any restrictions and/or requirements as a condition to yourFranchisee's acceptance of each National Account. Franchisor reserves the right to charge fees on National Account revenue. These fees, if charged, will be a required condition of participating in our National Accounts Program and these fees are subject to change annually. Franchisor may require Franchisee to conduct thorough background checks on all employees who will service any National Account and may require proof of compliance with this and other requirements at such intervals as Franchisor and/or the National Account dictates.
- 1.5 **Reservation of Rights**. Franchisee acknowledges that, except as otherwise provided in this Agreement, Franchisee's right to provide the Approved Products and Approved Services, and otherwise use the Proprietary Marks and System within the Designated Territory is non-exclusive, and Franchisor and its affiliates expressly reserve the right to: (i) open and operate, and license others the right to open and operate, Franchised Businesses that offer products and use the Proprietary Marks and System at any location

outside of the Designated Territory; (ii) open and operate, and license others the right to open and operate, businesses that offer similar products and services to those offered by the Franchised Business under any other mark at any location, within or outside the Designated Territory; (iii) acquire, merge, or be acquired by, any company, including a company operating one or more businesses near the Franchised Business that offer products and services that are the same or substantially similar to the Approved Products and Approved Services; (iv) exercise its rights under Section 1.6 of this Agreement in the events of catastrophe(s); (v) designate and service National Accounts; (vi) service, route, and/or assign any and all customer work orders and inquiries received through Franchisor's Call Center (defined in Section 6.6) in accordance with this Agreement; and (vii) use the Proprietary Marks and System, and license others to use the Proprietary Marks and System, to engage in any other activities not expressly prohibited in this Agreement.

- 1.6 **Events of Catastrophe**. Notwithstanding Section 1.4, in the event of a natural disaster or other similar catastrophic situation, as Franchisor determines in its sole discretion, Franchisor, its affiliate(s) and other System franchisees may be permitted to provide support to Franchisee and/or perform work in the Designated Territory, including the provision of labor, materials, equipment, and project management on projects in the Designated Territory, and Franchisee will not be entitled to any proceeds from the provision of these services performed by third parties within the Designated Territory.
- 1.7 Alternate Channels of Distribution. Franchisee acknowledges and agrees that certain of Franchisor's or its affiliates' products and services, whether now existing or developed in the future, may be distributed in Franchisee's Designated Territory by Franchisor, Franchisor's affiliates, or other third parties that Franchisor designates, in such manner and through such channels of distribution as Franchisor, in its sole discretion, shall determine. Such alternate channels of distribution will include, but are not limited to, the sale and distribution of the products and services via the Internet and through joint marketing with partner companies under terms and conditions that Franchisor deems appropriate. Franchisee understands that this Agreement grants Franchisee no rights: (i) to distribute Approved Products and Approved Services as described in this Section; or (ii) to share in any of the proceeds received by any such party therefrom.

1.8 Right to Service Customers in Designated Territory; Use of Call Center.

1.8.1 If Franchisor establishes a Call Center (defined in Section 6.6), then 1.8.1

Franchisee must ensure that all initial calls made to the Franchised Business are forwarded to the Call Center-, which is part of Franchisor's proprietary Call Center. Once a customer's call is routed to the Call Center and assigned to the Franchised Business, Franchisor will route that customer's work to Franchisee if the customer's location (where the work is to be performed) is within the Designated Territory, unless: (i) Franchisor determines that the work is in the nature of an emergency and: (a) Franchisee does not respond to the assignment within a time period Franchisor deems reasonable, in its sole discretion, appropriate under the circumstances; or (b) Franchisee is not able to perform the required services for the customer within a time period Franchisor deems reasonably appropriate, in its sole discretion; (ii) the work is of such a large scope, complexity and/or commercial nature that Franchisor determines, in its sole discretion, that Franchisee is not capable of performing the work requested in accordance with System standards and specifications and/or the prevailing standard of care in the industry for the type of work requested (in which case Franchisor may route the work order to Franchisee and additional franchisees, or other franchisees, or Franchisor's affiliate, for completion); (iii) the work order is mistakenly routed to another franchisee or affiliate-owned business due to either the customer providing incorrect information to a Call Center representative or an inadvertent error on the part of the Call Center representative when taking the information from the customer; (iv) Franchisee is not operating the Franchised Business in compliance with this Agreement; or (v) Franchisor reasonably determines that a portion of the Designated Territory has been subjected to a disaster or catastrophe.

1.8.2 Franchisee acknowledges and agrees that Franchisor's rights under this Section are necessary to: (i) maintain uniformity across the System and ensure that all work performed under the Proprietary Marks meets Franchisor's System standards for customer service; and (ii) account for inadvertent mistakes by Franchisor's customers and Franchisor's Call Center. Franchisee further acknowledges that it does not have any right to share in the Gross Revenue generated from customers that are serviced within the Designated Territory unless Franchisee is assigned, and subsequently provides services to, such customers.

2. TERM AND RENEWAL

- 2.1 **Term.** The initial term of the Franchise is for a period of ten (10) years, which will commence on the date Franchisor executes this Agreement Effective Date.
- 2.2 **Renewal.** Franchisee has the right to renew this Agreement for one (1) successive, additional ten (10)—year period, provided Franchisee has met the following conditions:
- 2.2.1 Franchisee has notified Franchisor of Franchisee's intention to renew this Agreement in writing at least twelve (12) months, and no more than eighteen (18) months, prior to expiration of the current term;
- 2.2.2 Franchisee has demonstrated to Franchisor's satisfaction that Franchisee has the right to operate the Franchised Business at the Approved Location for the duration of the renewal term; or, if Franchisee is unable to operate the Franchised Business at the Approved Location, Franchisee has secured a substitute location within the Designated Territory acceptable to Franchisor;
- 2.2.3 Franchisee has completed to Franchisor's satisfaction, no later than 90 days prior to the expiration of the then-current term, any updates to all required equipment, supplies, inventory, hardware and software, and vehicles to bring the Franchised Business into full compliance with Franchisor's then-current System standards and specifications, and Franchisee has completed all maintenance, refurbishing, renovating, updating and remodeling of the Franchised Business's Approved Location so that it satisfies Franchisor's then-current standards;
- 2.2.4 Franchisee is not in breach of any provision of this Agreement, or any other agreement between Franchisee and Franchisor, Franchisor's affiliates, or Franchisor's approved/designated suppliers and vendors, and Franchisee has substantially complied with all such agreements during their respective terms;
- 2.2.5 Franchisee has satisfied all monetary obligations Franchisee owes Franchisor, Franchisor's affiliates, and Franchisor's approved/designated suppliers and vendors, and has timely met these obligations throughout the term of this Agreement;
- 2.2.6 Franchisee executes Franchisor's then-current form of franchise agreement, the terms of which may vary materially from the terms of this Agreement (which may include, without limitation, increased royalty and other fees and insurance requirements), at least three (3) months prior to the expiration of this Agreement;
- 2.2.7 Franchisee satisfies Franchisor's then-current training requirements for renewing franchisees at Franchisee's expense, if any, as of the date of such renewal, and Franchisee has otherwise obtained and maintained all licenses, permits and approvals required by federal and state law applicable to providing the Approved Products and Approved Services at any location within the Designated Territory;

- 2.2.8 Franchisee and its principals execute a general release in the form Franchisor prescribes; and
- 2.2.9 Franchisor may condition the renewal on resizing the territory into multiple territories consistent with the then current size of territory being sold, which may result in more territories with separate franchise agreements and the requirement for a multi-unit addendum; and
 - 2.2.10 Franchisee pays a renewal fee in the amount of five thousand dollars (\$\sum_5,000).
- 2.3 **Holdover Period.** If this Agreement expires without Franchisee properly exercising its renewal right and Franchisee continues to accept the benefits of this Agreement thereafter, then, at Franchisor's option, Franchisor may treat this Agreement either as: (i) expired as of the date of expiration, with Franchisee then illegally operating a franchise in violation of Franchisor's rights; or (ii) continued on a month-to-month basis (the "Holdover Period") until both parties agree to enter into Franchisor's thencurrent form of franchise agreement for a renewal term or until one party provides the other with written notice of termination, in which case the Holdover Period will terminate 30 days after receipt of the notice of termination. In the latter case, all of Franchisee's obligations shall remain in full force and effect during the Holdover Period as Hif this Agreement had not expired, except that the license fee during the Holdover Period will be increased to 10% of Gross Revenue for all types of products/services and without any reductions. All obligations and restrictions imposed on Franchisee upon expiration of this Agreement shall take effect upon termination of the Holdover Period.

3. FEES AND MANNER OF PAYMENT

3.1 **Initial Franchise Fee.** In consideration of the franchise granted to Franchisee by Franchisor, Franchisee must pay Franchisor a lump sum initial franchise fee set forth on the Data Sheet that is part of this Agreement (the "Initial Franchise Fee"), which is due at the signing of this Agreement and deemed fully earned and non-refundable upon payment, in consideration of administrative and other expenses Franchisor incurs in granting the franchise and for Franchisor's lost or deferred opportunity to franchise to others.

3.1.1 Elevate to Owner Program Discount. Franchisor offers a discounted Initial Franchise Fee for the first Franchised Business to qualified employees of a franchisee who: (i) have been recommended by a System franchisee; (ii) have been employed by a System franchisee for at least two (2) years; and (iii) otherwise qualifies to be Franchisor's franchisee. The discount is based upon years of service with one of Franchisor's franchisees and is calculated as follows:

Discount on Initial Franchise Fee	Years of Consecutive Employment With Franchisee
10%	2
15%	3
20%	4
25%	5 +

Discounts under the Elevate to Owner Program will be applied to the Initial Franchise Fee for Franchisee's first Franchised Business only and the discount shall be calculated after any third party broker fees are deducted, if any.

- 3.2 **Royalty Fee.** Franchisee must pay Franchisor a monthly royalty fee (the "Royalty Fee") deducted on the tenth (10th) business day of the following month for the prior month in an amount equal to the greater of: (i) the Royalty Fee (as delineated in Section 3.2.1 below); or (ii) the minimum monthly royalty fee ("Minimum Royalty Fee"), as described in Section 3.2.2 below.
 - 3.2.1 Royalty Fee Structure. Franchisee's Royalty Fee is calculated as follows:

Monthly Royalty Fee	Gross Revenue for Prior Month
7.0%	\$0.00 - \$100,000.00
6.0%	\$100,000.01 - \$300,000.00
5.0%	\$300,000.01+

3.2.2 *Minimum Monthly Royalty Fee*. Franchisee's Minimum Monthly Royalty Fee for one (1) Designated Territory is based upon the number of months the Franchised Business has been open and operating. Franchisee is considered "open and operating" on the date that Franchisee completes the Initial Training Program (defined in Section 8.1). The Minimum Monthly Royalty Fee is as follows:

Time of Operation	1 Designated Territory
First 3 Months	The Royalty Fee (7%-5% of Gross Revenue)
4 to 12 Months	\$ 250 300
Second Year	\$ <u>1,</u> 500
Third Year +	\$ 750 2,000
Fourth Year+	\$1,000

- 3.2.3 Special Programs. Franchisor reserves the right, but not the obligation, to establish special programs that reward franchisees for meeting certain criteria. In the event Franchisor establishes any special programs, Franchisor will have the right, in Franchisor's sole discretion, to change, modify or dissolve any special programs upon notice to Franchisee.
- 3.2.4 Conversion Roll-In Discount for Existing Businesses. If Franchisee has an existing business that is the same or similar to a Franchised Business with annual gross sales of at least \$500,000 and Franchisee agrees to convert the existing business into a Franchised Business, Franchisor will discount Franchisee's Royalty Fee for the initial period of operation. The Royalty Fee will be calculated as follows:

Months of Operation	Royalty Fee
Months 1-4	25% of then-current Royalty Fee
Months 5-8	50% of then-current Royalty Fee
Months 9-12	75% of then-current Royalty Fee
Month 12+	The then-current Royalty Fee

After the initial twelve (12)_month period, Franchisee shall pay the Royalty Fee as set forth in Section 3.2 of this Agreement.

3.3 Gross Revenue Definition and Reports. For purposes of this Agreement, "Gross Revenue" is defined to include all income received of any type or nature and from any source that Franchisee derives or receives directly or indirectly from, through, by or on account of the operation of the Franchised Business (whether or not the products/services are approved by the Franchisor) at any time after the signing of the Franchise Agreement, in whatever form and from whatever source, including, but not limited to, cash, services in kind from barter and/or exchange, on credit or otherwise, and convenience, credit card user, or other fees charged to a customer, as well as business interruption insurance proceeds, all without deduction for expenses, including marketing expenses and taxes. Exclusions do not include reductions for financing program fees, credit card convenience fees or user fees, or any other fees charged to a customer. However, the definition of Gross Revenue does not include sales tax that is collected from customers and actually transmitted to the appropriate taxing authorities or any bona fide refunds you make Franchisee makes to customers in the ordinary course of business. We reserve Franchisor reserves the right to modify our policies and practices regarding revenue recognition, revenue reporting, and the inclusion or exclusion of certain revenue from Gross Revenue as circumstances, business practices, and technology change.

Franchisee must send Franchisor a signed report ("Gross Revenue Report(s)") on or before the tenth (10th) day of each month for the immediately preceding calendar month, in the manner and form specified by Franchisor. Each Gross Revenue Report must set forth: (i) Franchisee's Gross Revenue generated during the previous calendar month; (ii) Franchisee's calculation of the Royalty Fee and Brand Fund Contribution (defined in Section 3.5); and (iii) any other information Franchisor may require. In lieu of the Gross Revenue Report, weFranchisor currently collectcollects Gross Revenue data through the ServiceMinder Systemour digital platform but we reserveFranchisor reserves the right to require a Gross Revenue Report be submitted to usFranchisor as described above in the future upon notice. Franchisor may change the form, content, and/or interval of the Gross Revenue Report from time to time and/or require Franchisee to submit Gross Revenue Reports upon notice to Franchisee.

Method of Payment. With the exception of the Initial Franchise Fee, Franchisee shall pay all fees and other amounts due to Franchisor and/or its affiliates under this Agreement through an electronic funds transfer program (the "EFT Program"), under which Franchisor automatically deducts all payments owed to Franchisor under this Agreement or any other agreement between Franchisee and Franchisor or its affiliates, from the bank account Franchisee provides to Franchisor for use in connection with the EFT Program (the "EFT Account"). Franchisee shall immediately deposit all revenues from operation of the Franchised Business into this bank account within two (2) days upon receipt, including eash, checks, and eredit eard receipts. At least ten (10) At least 10 days prior to opening the Franchised Business, Franchisee shall provide Franchisor with: (i) Franchisee's bank's name, address and account number; and (ii) a voided check from such bank account. Contemporaneous with the execution of this Agreement, Franchisee shall sign and provide to Franchisor and Franchisee's bank, all documents, including Franchisor's form of EFT Withdrawal Authorization Form in the form attached hereto as Exhibit D to this Agreementas updated from time to time, necessary to effectuate the EFT Program and Franchisor's ability to withdraw funds from such bank account via electronic funds transfer ("EFT"). Franchisee shall immediately notify Franchisor of any change in Franchisee's banking relationship, including any change to the EFT Account. Franchisor reserves the right to require Franchisee to pay any fees due under this Agreement or any other agreement related to the Franchised Business by such other means as Franchisor may specify from time to time, including, but not limited to, a requirement to use a specific payment processor from whom Franchisor or its affiliate may collect fees at the time of such payment. If any Gross Revenue Report has not been received within the required time period, then Franchisor may process an EFT for the subject month based on the most recent Gross Revenue Report provided by Franchisee to Franchisor, provided, that if a Gross Revenue Report for the subject month is subsequently received and reflects: (i) that the actual amount of the fee due was more than the amount of the EFT, then Franchisor may withdraw additional funds through an EFT from Franchisee's designated bank account for the difference; or (ii) that the actual amount of the fee due was

less than the amount of the EFT, then Franchisor shall credit the excess amount to the payment of Franchisee's future obligations. If any check or electronic payment is not successful due to insufficient funds, stop payment or any similar event, any excess amounts that Franchisee owes will be payable upon demand, together with a nonsufficient funds fee of one hundred dollars (\$100) per occurrence or the highest amount allowed by law, plus the lesser of: (i) one and one half percent (1½%) per month ((eighteen percent (18%) per annum); or (ii) the maximum interest rate allowed by law from the due date of payment. Franchisee must also pay Franchisor one hundred dollars (\$100) for each week that a payment is paid after the due date which is in addition to interest.\$100 per occurrence or the highest amount allowed by law. If Franchisee makes any payment to Franchisor or Franchisor's affiliate(s) by credit card for any fee required, Franchisor may charge a payment service fee of up to four percent (4%)4% of the total charge.

- 3.5 **Brand Fund Contribution**. As set forth more fully in Section 12 of this Agreement, Franchisor has established a brand fund for advertising and brand promotion (the "Brand Fund"). Franchisee shallmust make monthly Brand Fund Contributions of one percent (1%)2% of Gross Revenue generated by the Franchised Business for sales made during the immediately preceding calendar month (the "Brand Fund Contribution"). Presently, Brand Fund Contributions are to be paid as directed by Franchisor via the EFT Program—on the tenth—(10th) business day of each month, or as otherwise required by the Operations Manual (defined in Section 6.1) or in writing by Franchisor. Franchisor will have the right to expend the funds accumulated in the Brand Fund in Franchisor's sole discretion. Notwithstanding the foregoing, Franchisor reserves the right to increase the Brand Fund Contribution at any time upon providing notice to Franchisee, provided that the Brand Fund Contribution will not exceed three percent (3%)3% of Gross Revenue during the term of this Agreement.
- Local Advertising Requirement. Franchisee shall expend the greater of: (i) three percent (3%)% of Gross Revenue per month; and (ii) \$3,000 per month on local advertising (the "Local Advertising Requirement"). Franchisor reserves the right to increase the Local Advertising Requirement to four percent (4%)% of Gross Revenue upon ninety (90) days' prior written notice. While the Local Advertising Requirement will typically be paid to third parties that Franchisor designates, Franchisor and its affiliate(s) reserve the right to collect this fee or to designate an Approved Supplier require that will collect this fee. Allall or a portion of the Local Advertising Requirement may be allocatedpaid to the Franchisor Directed Advertising Fee described in Section 12.5.3 or an Approved Supplier for Local Advertising Services products or services that we provide to you. We Franchisor or a third party provides or for use in marketing channels that Franchisor requires Franchisee to use. Franchisor and ourits affiliates may earn revenue and profits on products or services we provide Franchisor provides, and may receive rebates, licensing fees, administrative fees, commissions, or other payments on products and services that third party vendors provide. With respect to all Local Advertising Requirement funds you payFunds Franchisee pays to a third party, you are Franchisee is required to provide us Franchisor with monthly expense statements (including receipts supporting the reported expenditures) evidencing compliance with the Local Advertising Requirement spending obligations. The Initial Franchisor Directed Advertising Fee in the amount set forth in the Data Sheet must be paid at the signing of this Agreement and deemed fully earned and non-refundable upon payment. The initial Franchisor Directed Advertising Fee (defined in Section 12.5.3) is a credit to that fee for the first two (2) months of the term. The Franchisor Directed Advertising Fee must be paid in the month prior to the month in which the fees will be spent, in an amount up to the greater of \$3,000 or three percent (3%) of the prior month's required spend, whichever is greater. Thereafter, in the second month, the Franchisor Directed Advertising Fee must be paid to Franchisor or its designee on or before the tenth (10th) day of the month. If local advertising directed by us is paid with a franchisee credit card (or other method of direct payment), all such amounts spent that are directed by us will be credited to the monthly Franchisor Directed Advertising Fee.
- 3.7 **Technology Fee**. In addition to the fees set forth above, Franchisee must pay Franchisor and/or Franchisor's designated vendors the then-current technology fee as set forth in the Operations

Manual (defined in Section 6.1) ("Technology Fee"), associated with using and maintaining Franchisor's reservation system, intranet, mobile applications, required computer hardware and software, hosting services and solutions, and any other technology used in the operation of the Franchised Business, and such payment shall be made in the manner prescribed by Franchisor or the designated vendor(s), as applicable. The current Technology Fee is \$599.00 per month and is subject to increase upon 30 days' notice to youFranchisee. Franchisee shall pay the Technology Fee inon the same manner as the Royalty Fee and as described in Section 3.410th business day of this Agreementeach month, or as otherwise set forth inrequired by the Operations Manual or in writing by Franchisor. Franchisor reserves the right to change the amount of the fee described in this Section as changes are made to the System's hardware, software and other computer requirements or as required by the third-party service provider(s) or by any regulatory agency upon thirty (30) days' prior written notice to youFranchisee.

- 3.8 Call Center Fee. Franchisor reserves the right to establish a Call Center. If established, Franchisee shall pay Franchisor a call center fee deducted on the tenth (10th) business day after the end of each calendar month for the prior calendar month in an amount up to four percent (4%) of Franchisee's Gross Revenue (the "Call Center Fee").
- 3.93.8 Online Local Presence Fee. In addition to the fees set forth above, Franchisee must pay Franchisor and/or Franchisor's designated vendors the then-current Online Local Presence Fee as set forth in the Operations Manual ("Online Local Presence Fee"), associated with costs of website management, development, updating, maintenance, enhancement, and brand reputation management. The current Online Local Presence Fee is \$249 per month and is subject to increase upon 30 days' notice to Franchisee. Franchisee shall pay the Online Local Presence Fee on the 10th business day of each month, or as otherwise required by the Operations Manual or in writing by Franchisor. Franchisor reserves the right to change the amount of the fee described in this Section to cover costs incurred in the development, management, and maintenance of the brand websites or as required by the third-party service provider(s) or by any regulatory agency upon 30 days' prior written notice to Franchisee.
- 3.9 Marketing Management Fee. In addition to the fees set forth above, Franchisee must pay Franchisor and/or Franchisor's designated vendors the then-current marketing management fee as set forth in the Operations Manual ("Marketing Management Fee"), associated with localized marketing including: email marketing, social media, promotions and offers, creative support, public relations management, tradeshow support, and paid media support. The current Marketing Management Fee is \$500 per month and is subject to increase upon 30 days' notice to Franchisee. Franchisee shall pay the Marketing Management Fee on the 10th business day of each month, or as otherwise required by the Operations Manual or in writing by Franchisor. Franchisor reserves the right to change the amount of the fee described in this Section to cover costs incurred in localized marketing support or as required by the third-party service provider(s) or by any regulatory agency upon 30 days' prior written notice to Franchisee.
- 3.10 Opening Assistance Fee. In consideration of the opening assistance services provided to Franchisee by Franchisor, Franchisee must pay Franchisor a lump sum opening assistance fee set forth on the Data Sheet that is part of this Agreement (the "Opening Assistance Fee"), which is due at the signing of this Agreement and deemed fully earned and non-refundable upon payment in consideration of administrative and other expenses Franchisor incurs in providing pre-opening services, including setting up and preparing its systems for Franchisee to open its business.
- 3.11 Call Center Fee. Franchisee shall pay Franchisor a call center fee equal to 2% of Franchisee's Gross Revenue for the prior calendar month (the "Call Center Fee"). Franchisor reserves the right to increase the Call Center Fee in an amount up to 4% of Franchisee's Gross Revenue upon 30 days'

prior written notice to Franchisee. Franchisee shall pay the Call Center Fee on the 10th business day of each month, or as otherwise required by the Operations Manual or in writing by Franchisor.

- 3.12 Late and/or Underpayments and Interest. All fee payments, amounts due for purchases by Franchisee from Franchisor and/or its affiliated company;(ies), and other amounts which Franchisee owes to Franchisor and/or its affiliated company not received on or before the due date shall be deemed past due. If any payment or contribution is past due, Franchisee shall pay to Franchisor immediately upon demand, in addition to the past due amount, Franchisor's then-current late fee per incident, plus interest on the past due amount from the date it was due until paid at the rate-lesser of one and one half percent (: (i) 1½%)½% per month (eighteen percent (18%)—%per annum);); or (ii) the maximum interest rate permitted allowed by law, whichever from the due date of payment. In addition to interest and fees for insufficient funds, stop payment, or similar events, Franchisee must also pay Franchisor \$100 for each week that a payment is lesspaid after the due date which is in addition to interest. Nothing contained in this Section shall prevent Franchisor from exercising, in Franchisor's sole judgment, any other rights or remedies available to Franchisor under this Agreement. In addition to the interest set forth above, Franchisee shall also pay Franchisor one hundred dollars (\$100) for each week that a payment is paid after the due date.
- 3.1013 No Right to Offset. Franchisee shall not be entitled to set off any payments required to be made under this Section 3 against any monetary claim it may have against Franchisor.
- 3.4114 Non-Exclusive Remedies. Franchisor's right to recover interest and late payment fees under this Section shall not prevent Franchisor from obtaining, or otherwise waiving, any other remedy available to Franchisor for Franchisee's breach of this Section, as set forth in this Agreement or under applicable law.
- 3.1215 **Taxes on Payments**. In the event any taxing authority, wherever located, imposes any future tax, levy or assessment on any payment Franchisee makes to Franchisor, Franchisee must, in addition to all payments due to Franchisor, pay such tax, levy or assessment.
- 3.16 Franchisor's Right of Setoff. As to Franchisee and/or any affiliate of Franchisee, Franchisor can:
 - 3.16.1 apply any payments received to any past due, current, future or other indebtedness of any kind, in Franchisor's Business Judgment (defined in Section 3.4417 below), no matter how payment is designated by Franchisee (including any amount due under any Promissory Note), except that the Brand Fund Contributions may only be credited to the Brand Fund;
 - 3.16.2 set off from any amounts that may be owed by Franchisor, any amount owed to Franchisor or any marketing fund; and
 - 3.16.3 retain any amounts received for Franchisee's account (and/or that of any affiliate of Franchisee), whether rebates from suppliers, payment for National Account work, or otherwise, as a payment against any amounts owed to Franchisor or any marketing fund.

Franchisor can exercise any of the foregoing rights in connection with amounts owed to or from Franchisor and/or any affiliate. For purposes of this Agreement, "affiliate" means any person or entity which controls, is controlled by, or is under common control with another person or entity, including any

person or entity that provides services to Franchisee. In addition, as to the Franchisee, "affiliate" also means any owner of any interest in Franchisee or the Franchised Business, any employee or agent of the Franchisee, and/or any independent contractor performing functions for, or on behalf of, Franchisee, and any entity controlled by any of the foregoing.

3.1417 Business Judgment. For purposes of this Agreement, the term "Business Judgment" means the exercise of reasonable business judgment in making Franchisor's decision or exercising Franchisor's rights. Franchisor's decisions or actions will be deemed to be the result of reasonable business judgment, even if other reasonable or even arguably preferable alternatives are available, if Franchisor's decision or action is intended, in whole or significant part, to promote or benefit the franchise System generally, even if the decision or action also promotes Franchisor's financial or other individual interest. Examples of items that will promote or benefit the franchise System include, without limitation, enhancing the value of the Proprietary Marks, improving customer service and satisfaction, improving product quality, improving uniformity, enhancing or encouraging modernization, and improving the competitive position of the franchise System. This is a defined term for the purposes of this Agreement and is not intended to incorporate principles related to the application of the business judgment rule in a corporate law context.

4. PROPRIETARY MARKS

4.1 Franchisee's Use of the Proprietary Marks and Other Proprietary Material.

- 4.1.1 Franchisee shall use only the Proprietary Marks which Franchisor designates, and shall use them only in the manner Franchisor authorizes and permits.
- 4.1.2 Franchisee shall use the Proprietary Marks only for the operation of the Franchised Business and only in the Designated Territory and in sales and marketing for the Franchised Business.
- 4.1.3 Franchisee shall use all Proprietary Marks without prefix or suffix and in conjunction with the symbols "TM," "SM," "S," or "®," as applicable. Franchisee may not use the Proprietary Marks in connection with the offer or sale of any services or products which Franchisor has not authorized for use in connection with the System. Franchisee may not use the Proprietary Marks as part of Franchisee's corporate or other legal name. Franchisee's corporate name and all fictitious names under which Franchisee proposes to do business must be approved by Franchisor in writing before use. Franchisee must use Franchisee's corporate or limited liability company name either alone or followed by the initials "D/B/A" and a business name approved in advance by Franchisor. Franchisee must promptly register at the office of the county in which Franchisee's Franchised Business is located, or such other public office as provided for by the laws of the state in which Franchisee's Franchised Business is located, as doing business under such assumed business name.
- 4.1.4 Franchisee must identify itself as the owner of the Franchised Business (in the manner Franchisor prescribes) in conjunction with any use of the Proprietary Marks, including, without limitation, on invoices, order forms, receipts, customer forms and questionnaires, business stationery, and advertisements, as well as at such conspicuous locations as Franchisor may designate in writing at the Franchised Business premises.
- 4.1.5 Franchisee's right to use the Proprietary Marks is limited to such uses as are authorized under this Agreement, and any unauthorized use thereof shall constitute an infringement of Franchisor's rights.

- 4.1.6 Franchisee shall not use the Proprietary Marks to incur any obligation or indebtedness on Franchisor's behalf.
- 4.1.7 Franchisee shall execute all documents Franchisor deems necessary to obtain protection for the Proprietary Marks or to maintain their continued validity and enforceability.
- 4.1.8 Franchisee must promptly notify Franchisor of any suspected unauthorized use of the Proprietary Marks that Franchisor may now or hereafter designate for use in connection with the System. any challenge to the validity of the Proprietary Marks, or any challenge to Franchisor's ownership of, Franchisor's right to use and to license others to use, or Franchisee's right to use, the Proprietary Marks and Operations Manual (defined in Section 6.1) (collectively the "Proprietary Material"). Franchisee acknowledges that Franchisor has the sole right to direct and control any administrative proceeding or litigation involving the Proprietary Material, including any settlement thereof. Franchisor has the right, but not the obligation, to take action against uses by others that may constitute infringement of the Franchisor's rights to the Proprietary Material. Franchisor shall defend Franchisee against any third-party claim, suit or demand arising out of Franchisee's use of the Proprietary Material. If Franchisor, in Franchisor's sole discretion, determines that Franchisee has used the Proprietary Material in accordance with this Agreement, Franchisor shall bear the cost of such defense, including the cost of any judgment or settlement. If Franchisor, in Franchisor's sole discretion, determines that Franchisee has not used the Proprietary Material in accordance with this Agreement, Franchisee shall bear the cost of such defense, including the cost of any judgment or settlement. In the event of any litigation relating to Franchisee's use of the Proprietary Material, Franchisee shall execute any and all documents and do such acts as may, in Franchisor's opinion, be necessary to carry out such defense or prosecution, including, without limitation, becoming a nominal party to any legal action. Except to the extent that such litigation is the result of Franchisee's use of the Proprietary Material in a manner inconsistent with the terms of this Agreement, Franchisor agrees to reimburse Franchisee for Franchisee's out-of-pocket costs in performing such acts.
 - 4.1.9 Franchisee expressly understands and acknowledges that:
- 4.1.9.1 Franchisor or its affiliates or licensors own all right, title and interest in and to the Proprietary Marks and the goodwill associated with and symbolized by them, and Franchisor has the right to use, and license others to use, the Proprietary Marks;
- 4.1.9.2 The Proprietary Marks are valid and serve to identify the System and those who are authorized to operate under the System;
- 4.1.9.3 During the term of this Agreement and after its expiration or termination, Franchisee shall not directly or indirectly contest the validity of, or Franchisor's ownership of, or right to use and to license others to use, the Proprietary Marks or any other Proprietary Material;
- 4.1.9.4 Franchisee's use of the Proprietary Material does not give Franchisee any ownership interest or other interest in or to the Proprietary Material;
- 4.1.9.5 Any and all goodwill arising from Franchisee's use of the Proprietary Material shall inure solely and exclusively to Franchisor's benefit, and upon expiration or termination of this Agreement, no monetary amount shall be assigned as attributable to any goodwill associated with Franchisee's use of the System, the Proprietary Marks, or any other Proprietary Material;
- 4.1.9.6 Except as specified in this Agreement, the license of the Proprietary Marks granted to Franchisee hereunder is non-exclusive and Franchisor retains the right, among others, to: (i) use the Proprietary Marks itself in connection with selling products and services; (ii) grant other licenses for

the Proprietary Marks; and (iii) develop and establish other systems using the Proprietary Marks, similar proprietary marks, or any other proprietary marks, and to grant licenses thereto without providing any rights therein to Franchisee; and

- 4.1.9.7 Franchisor reserves the right, in Franchisor's sole discretion, to substitute different proprietary marks for use in identifying the System and the businesses operating thereunder. Franchisee shall discontinue using all Proprietary Marks which Franchisor has notified Franchisee, in writing, have been modified or discontinued within thirty (30) days of receiving written notice and, at Franchisee's sole cost and expense, shall promptly begin using such additional, modified or substituted Proprietary Marks.
- 4.2 Copyrighted Materials. You acknowledge Franchisee acknowledges that Franchisor is the owner of certain copyrighted or copyrightable works (the "Works") and that the copyrights in the Works are valuable property. The Works include, but are not limited to, the Operations Manuals, advertisements, promotional materials, signs, Internet sites, mobile applications, vehicle graphics, and facility designs. We authorize youFranchisor authorizes Franchisee to use the Works on the condition that you complyFranchisee complies with all of the terms and conditions of this Section 4. This Agreement does not confer any interest in the Works on youFranchisee, other than the right to use them in the operation of the Franchised Business in compliance with the terms of this Agreement. If you prepareFranchisee prepares any adaptation, translation or other work derived from the Works, whether or not authorized by us, you agreeFranchisor, Franchisee agrees that the material will be our property and youFranchisee hereby assignassigns all your of its right, title and interest therein to us. You agreeFranchisor. Franchisee agrees to sign any documents we deemFranchisor deems necessary to confirm ourFranchisor's ownership.

5. CONFIDENTIAL INFORMATION

- 5.1 **Nondisclosure**. During the term of this Agreement, Franchisee will receive information which Franchisor considers its trade secret and confidential information. Franchisee may not, during the term of this Agreement or thereafter, communicate, divulge, or use for the benefit of any other person or entity any Confidential Information (defined in Section 5.2). Upon termination or expiration of this Agreement, regardless of the reason, Franchisee shall return all copies of such Confidential Information to Franchisor immediately, and Franchisee may not use the Confidential Information for any purpose other than operating the Franchised Business in accordance with Franchisor's standards and specifications.
- Confidential Information. Confidential Information hereby includes, without limitation, any and all confidential, proprietary, and trade secret information relating to the operation of a Franchised Business, such as: (i) all financial, operational, technical and marketing information; (ii) Operations Manual; (iii) Franchisor's System policies or procedures, and franchising materials, brochures, marketing plans, forecasts, and related information; (iv) cost data; (v) pricing information; (vi) business plans; (vii) financial records and results of Franchisor's operations and other persons or entities operating a Franchised Business; (viii) photographs, devices, samples, models and illustrations; (ix) software developed by or for Franchisor; (x) customer and/or client lists and any information relating to Franchisor's customers or the customers of other System franchisees; (xi) prospect customer/client lists; (xii) patent, trademark, service mark and copyright applications; (xiii) information relating to inventions, discoveries, software, and any other research and development information; (xiv) methods of conducting the business developed by Franchisor or other franchisees, and any forms, memoranda, outlines, protocol, presentations, proposals, software, or other documents or information related to such methods; (xv) any information of a customer not generally known or available to the public; (xvi) any Trade Secrets (defined in Section 5.3), or of a customer of Franchisor, or of any other franchisee; and (xvii) any information about or originating from any Franchisee which, if it was information of Franchisor, is expressly deemed Confidential Information pursuant to the foregoing (collectively, "Confidential Information"). Any and all information, knowledge,

know-how, techniques, and other data which Franchisor designates as confidential will be deemed Confidential Information for purposes of this Agreement.

- 5.3 **Trade Secrets**. Notwithstanding Section 5.2, "trade secret" means information (including, but not limited to, components of the System, product marketing and promotional techniques, confidential business information, technical or non-technical data, formulas, patterns, compilations, programs, devices, methods, techniques, drawings, processes, financial data, financial plans, product plans, and lists of actual or potential customers or suppliers) that: (i) derive economic value, actual or potential, from not being generally known to, and not being readily ascertainable by proper means by, other persons who can obtain economic value from its disclosure or use; and (ii) is the subject of efforts that are reasonable under the circumstances to maintain its secrecy (collectively, "Trade Secrets"). To the extent that applicable law mandates a definition of "trade secret" inconsistent with the foregoing definition, then the foregoing definition shall be construed in such a manner as to be consistent with the mandated definition under applicable law.
- 5.4 **Employees/Personnel**. All of Franchisee's employees or other personnel must execute covenants that they will maintain the confidentiality of information they receive in connection with their employment by Franchisee at the Franchised Business. Such covenants shall be in a form satisfactory to Franchisor and substantially similar to the Confidentiality and Restrictive Covenant Agreement in the form attached to the Franchisor Disclosure Document as Exhibit C to this AgreementD. Employee non-compete and restrictive covenant agreements must include, without limitation, specific identification of Franchisor as a third-party beneficiary of such covenants with independent rights to enforce them.
- New Concepts. If Franchisee, or Franchisee's employees or principals, develop(s) any new concept, process or improvement in the operation or promotion of the Franchised Business, including, but not limited to, any modifications or additions to the Proprietary Material, Franchisee shall promptly notify Franchisor and provide Franchisor with all necessary related information, without compensation. Any such concept, process or improvement shall become Franchisor's sole property, and Franchisor shall be the sole owner of all patents, patent applications, trademarks, copyrights, and other intellectual property rights related thereto. Franchisee and Franchisee's principals and agents hereby assign to Franchisor any rights they may have or acquire therein, including the right to modify such concept, process or improvement, and otherwise waive and/or release all rights of restraint and moral rights therein and thereto. Franchisee and Franchisee's principals and agents agree to assist Franchisor in obtaining and enforcing the intellectual property rights to any such concept, process or improvement in any and all countries, and further agree to execute and provide Franchisor with all necessary documentation for obtaining and enforcing such rights. Franchisee and Franchisee's principals and agents hereby irrevocably designate and appoint Franchisor as their agent and attorney-in-fact to execute and file any such documentation and to do all other lawful acts to further the prosecution and issuance of patents or other intellectual property rights related to any such concept, process or improvement. In the event that the foregoing provisions of this Section 5.5 are found to be invalid or otherwise unenforceable, Franchisee and Franchisee's principals and agents hereby grant to Franchisor a worldwide, perpetual, non-exclusive, fully-paid license to use and sublicense the use of the concept, process or improvement to the extent such use or sublicense would, absent this Agreement, directly or indirectly infringe Franchisee's rights therein.
- 5.6 **Customer Privacy**. Franchisee agrees to adhere to the terms of Franchisor's customer privacy policies Franchisor may now or in the future develop. Franchisee may not divulge personal information regarding any customers, except as absolutely necessary to operate the Franchised Business.

6. FRANCHISOR'S OBLIGATIONS

- Operations Manual. Prior to commencing operation of the Franchised Business, Franchisor will provide Franchisee with secure access to its operations manual, which contains mandatory and suggested specifications, standards and operating procedures for the System, which may be modified and/or supplemented by Franchisor at any time as Franchisor deems advisable in its sole discretion Business Judgement, including Franchisor's proprietary and confidential operations manual for operating a Franchised Business ("Operations Manual"). The Operations Manual may cover such topics as pre-opening procedures; systems and procedures; personnel policies; specifications for vehicles, supplies, equipment and inventory; marketing; accounting and bookkeeping and related matters as may be incorporated from time to time. Any required specifications, standards and operating procedures exist to protect Franchisor's interests in the System and the Proprietary Marks and to create a uniform customer experience, and not for the purpose of establishing any control or duty to take control over those matters that are reserved to Franchisee. The Operations Manual will remain confidential and the property of Franchisor, constituting a trade secret of Franchisor, and may not be shared, loaned out, duplicated, distributed or copied in whole or in part in any manner. The provisions of the Operations Manual constitute provisions of this Agreement as if fully set forth herein. Franchisor will have the right to add to and otherwise modify the contents of the Operations Manual from time to time in writing in any manner, including through the Operations Manual, email, Franchisor's website, or any other means. Franchisee must always follow the required directives in the Operations Manual, as they may be modified by Franchisor from time to time. Such compliance by Franchisee is necessary to protect the integrity and reputation of the System. The Operations Manual may also contain provisions that share recommended best practices, which provisions are not mandatory.
- 6.2 **Initial Supplies.** Franchisor will provide Franchisee with a list of all items and equipment needed to open the Franchised Business, along with the proprietary list of Approved Suppliers for those items (as applicable), with which Franchisee must comply.
- 6.3 **Ongoing Assistance**. Franchisor may provide Franchisee continuing consultation and advice, as Franchisor deems necessary and appropriate in its sole discretion, regarding the management and operation of the Franchised Business. Franchisor will provide such assistance, in Franchisor's discretion, by telephone, facsimile, intranet communication, on-site visits, or other means. If Franchisee requires and requests additional on-site assistance from Franchisor, subject to the availability of Franchisor's personnel, Franchisor may provide Franchisee with such assistance at Franchisor's then-current training fee, plus expenses, including Franchisor's travel and lodging expenses, as Franchisor deems necessary in its sole discretion. Franchisor may also use the Operations Manual to provide self-serve training materials.

6.4 Additional Training.

As set forth more fully in Section 8, Franchisor may, in Franchisor's sole discretion, hold refresher and ongoing training courses, or training courses to provide additional information and/or updates regarding Franchisor's System and/or the operation of the Franchised Business. Except as otherwise provided in this Agreement, Franchisor may require Franchisee and Franchisee's personnel to attend such additional training up to twenty (20) days per year at a location to be selected by Franchisor except that up to twenty (20) days per year may be required where in person training is required due to a default/non-compliance. All expenses, including Franchisee's and Franchisee's employees' transportation, meal, and lodging expenses to attend such training shall be Franchisee's Designated Territory, Franchisee will be responsible for all of Franchisor's employees' expenses to conduct such training, including transportation, meal and lodging expenses. Franchisor will provide Franchisee with thirty (30) days' notice of any upcoming additional or refresher training that Franchisee is required to attend. Franchisor may periodically conduct optional training programs for Franchisee and/or Franchisee's personnel at Franchisor's office or another location that Franchisor designates. Franchisor may charge Franchisee our then current fee for optional

training programs. Franchisor may provide any of the additional training in person or via recorded media, teleconference, video conference, the Internet, webinar, or any other means, as Franchisor determines.

- 6.5 **Remedial Training**. In the event Franchisor determines that Franchisee is not operating the Franchised Business as required under the Franchise Agreement or in compliance with the System standards, Franchisor may require Franchisee to attend up to five (5) days of remedial training a year (in addition to any required training). If established, Franchisor reserves the right to schedule remedial training at its corporate headquarters or other designated training facility, or Franchisor may provide such training on-site at the Franchised Business. In either case, Franchisor may charge Franchisee its then-current tuition training fee (plus our travel and living expenses) to provide such remedial training.
- 6.6 Call Center. Franchisor reserves the right to established and maintainmaintains a centralized call center for the purpose of accepting telephone, Internet and other inquiries from potential customers and forwarding such customer information to the appropriate franchisee (the "Call Center"). If established, Franchisee must comply with Franchisor's procedures for using the Call Center as Franchisor specifies in the Operations Manual or otherwise in writing, including any fees Franchisee must pay in connection with administering and maintaining this service. Franchisor reserveshas the absolute right to receive all customer calls to the Franchised Business, and subsequently service, route and/or assign any work orders or inquiries resulting from such calls as it deems advisable in its sole discretion, regardless of whether the customer is located within Franchisee's Designated Territory. If established, all All System related phone numbers and Internet lead sources are required to be ported to or directed to Franchisor.
- 6.7 **Pricing.** Franchisor may advise Franchisee from time to time concerning suggested retail prices. Franchisor and Franchisee agree that any list or schedule of prices furnished to Franchisee by Franchisor is a recommendation only and is not to be construed as mandatory upon Franchisee. Nothing contained herein shall be deemed a representation or warranty by Franchisor that the use of Franchisor's suggested prices will result in a profit. If we determine Franchisor determines that we Franchisor may lawfully require you Franchisee to charge certain prices for products or services, certain minimum prices for products or services, or certain maximum prices for products or services, you Franchisee must adhere to our pricing policies as set forth in the Manuals or otherwise in writing from time to time. Otherwise, you are Franchisee is solely responsible for determining the prices that you charge Franchisee charges customers. You Franchisee must provide us Franchisor with your that you charge Franchisee charges customers.
- Annual Conference. Franchisor may, in Franchisor's discretion, hold one or multiple annual conferences at a location to be selected by Franchisor (the "Annual Conference"). Franchisor will determine the topics and agenda for such conference to serve the purpose, among other things, of updating franchisees on new developments affecting franchisees, exchanging information between franchisees and Franchisor's personnel regarding operations and programs, and recognizing franchisees for their achievements. Franchisor may require Franchisee to attend the Annual Conference and pay Franchisor's then-current registration fee if Franchiseo chooses to charge a registration fee, in its sole discretion. Attendance at the Annual Conference expenses in the event the Franchisee chooses not to attend. All expenses, including Franchisee's and Franchisee's employees' transportation to and from the Annual Conference, as well as lodging, meals and salaries during the Annual Conference are Franchisee's sole responsibility. Franchisor may use Brand Fund Contributions for purposes related to the Annual Conference, including costs related to production, programs and materials.

7. FRANCHISEE'S OBLIGATIONS

Site Location and Lease Approval. Upon the execution of this Agreement and for a period of twelve (12) months from the Effective Date of this Agreement, Franchisee may operate the Franchised Business from a home office that meets Franchisor's standards and specifications. After this initial twelve (12) month period, Franchisee must operate the Franchised Business from an approved office/warehouse space that meets Franchisor's then current standards and specifications with approval from Franchisor for an the Approved Location. Franchisee will need to lease approximately 1,000 to 1,500 square feet of commercial real estate for Franchisee's office and to securely store all equipment and inventory. Franchisor may provide Franchisee with standards and specifications for the design and layout of the premises of the Approved Location, and Franchisor must review and approve any proposed location, as well as any lease associated with the proposed location, prior to Franchisee entering into any lease for the proposed location. only. If Franchisor has not approved a location for Franchisee to operate the Franchised Business as of the date Franchisee signs this Agreement, the parties shall enter into Franchisor's prescribed form of Site Selection Addendum, the terms of which shall govern the parties' Franchisee must work with Franchisor to identify an Approved Location. Prior to Franchisee's acquisition by lease or purchase of a site selection obligations.for the Franchised Business, Franchisee must submit to Franchisor, in the form Franchisor specifies, such information or materials as Franchisor may reasonably require, and a letter of intent or other evidence satisfactory to Franchisor that confirms Franchisee's favorable prospects for obtaining the proposed site. Recognizing that time is of the essence, Franchisee must submit a proposed site, together with the information and materials required by this Paragraph 3, to Franchisor for Franchisor's approval within 12 months after signing the Franchise Agreement. Franchisor will have 30 days after receipt of such information and materials from Franchisee to approve or disapprove, in Franchisor's sole discretion, the site as a location for the Franchised Business. No proposed site will be deemed approved unless Franchisor has expressly approved it in writing. Franchisor has the right to review, evaluate and approve Franchisee's proposed lease for the Approved Location ("Lease") prior to execution. Franchisor may condition Franchisor's approval of any proposed Lease on, among other things, Franchisee and Franchisee's landlord's execution of a Collateral Assignment of Lease in the form prescribed by Franchisor and attached to this Agreement as Exhibit G.

7.1.1 Relocation. If, for any reason, the Lease term is shorter than the term of this Agreement and the Lease cannot be renewed or extended, or Franchisee cannot continue for any other reason to occupy the Approved Location, Franchisee must relocate Franchisee's Franchised Business to a mutually acceptable site within Franchisee's Designated Territory to complete the unexpired portion of the term of this Agreement. Franchisee must notify Franchisor of Franchisee's intention to relocate, procure a site acceptable to Franchisor within ninety (90) days prior to closing operations at Franchisee's current Approved Location, and open for business at the new Approved Location within thirty (30) days of closing business at Franchisee's existing Approved Location. Franchisor may require Franchisee to reimburse Franchisor for its actual costs and expenses associated with evaluating Franchisee's relocation request and/or any locations proposed by Franchisee for relocation, plus a \$250 administrative fee. We will provide you with copies of our invoices for our expenses from any third party providers upon request.

7.1.2 7.1.2 Third-Party Provider Invoices. Franchisor will provide Franchisee with copies of Franchisor's invoices for its expenses from any third-party providers upon request.

7.1.27.1.3 Franchised Business Appearance and Construction. Franchisee agrees that the Franchised Business must conform to Franchisor's standards and specifications for the appearance, layout and design of a Franchised Business. Franchisee is solely responsible for the preparation of architectural and working drawings necessary to complete construction and/or build-out at the Approved

Location and must ensure that plans meet with applicable ordinances, building codes, permitspermit requirements, and any other applicable local, state or federal law.

- 7.2 **Training.** Franchisee (or if Franchisee is an entity, then Franchisee's principals) must attend and successfully complete Franchisor's Initial Training Program (defined in Section 8.1), as set forth more fully in Section 8 of this Agreement. Franchisor has the right to require up to one (1) individual to attend in addition to Franchisee.
- Opening Requirements. Franchisee shall open and commence operating the Franchised Business within ninety (90) days of executing this Agreement. Franchisee must obtain the Approved Location within thirty (30) days of executing this Agreement, and continuously operate the Franchised Business thereafter. In addition to any other pre-opening obligations set forth in this Agreement, Franchisee is required to complete the following prior to commencing operations: (i) obtain all required licenses, certifications, permits, and other governmental approvals necessary to operate the Franchised Business in the Designated Territory, and provide Franchisor with written proof thereof; (ii) purchase/lease all required vehicles, equipment, supplies and inventory in accordance with Franchisor's standards and specifications and, if appropriate, from Franchisor's Approved Suppliers (defined in Section 7.4.2), that Franchisee is required to purchase prior to opening; (iii) attend and successfully complete Franchisor's Initial Training Program, as defined and described in Section 8.1 of this Agreement, as well as any other pre-opening training Franchisor may prescribe; and (iv) provide Franchisor with any and all documents and information necessary for Franchisor to effectuate the EFT Program to automatically withdraw all payments due and owing Franchisor and its affiliates under the Franchise Agreement. Without limiting the foregoing, if youFranchisee or any of Franchisee's owners is not a U.S. national, you represent that youFranchisee and/or such owners have an immigration status that allows youFranchisee and/or such owners to live and work in the United States, and youFranchisee hereby promises that youFranchisee and/or such owners will maintain such status during the term of this Agreement. Failure to maintain an immigration status that allows Franchisee and/or such owners to live and work in the United States is a material breach of the Franchise Agreement.

7.4 **Purchasing Requirements**.

- 7.4.1 Compliance with Standards. Franchisee acknowledges and agrees that Franchisee's obligations set forth in this Agreement and the Operations Manual are reasonable and necessary for the operation of the Franchised Business and to maintain uniformity throughout the System. Franchisee shall adhere to the standards and specifications set forth in this Agreement and the Operations Manual and any revisions or amendments to same. Franchisee shall use the furnishings, supplies, fixtures, equipment, computer hardware and software, and product samples and promotional materials that comply with Franchisor's then-current standards and specifications, which Franchisor will establish and modify at Franchisor's discretion. Franchisee acknowledges that Franchisee may incur an increased cost to comply with such changes at Franchisee's expense.
- 7.4.2 Designated and Approved Suppliers. The Franchisee must currently use Franchisor's designated suppliers to purchase any items and/or services necessary to operate the Franchised Business. Recognizing that preservation of the System depends upon product and service uniformity and the maintenance of Franchisor's trade dress, Franchisee agrees to purchase product samples and other

supplies, services, furnishings, fixtures, computer hardware and software, and other equipment from Franchisor or from approved or designated suppliers as Franchisor shall specify, from time to time, in the Operations Manual and otherwise in writing (each an "Approved Supplier"). Franchisee hereby acknowledges that Franchisor, Franchisor's affiliate(s) and/or a third party may be one of several, or the only, Approved Supplier of any item. Franchisee further acknowledges and agrees that Franchisor and/or Franchisor's affiliates have the right to realize a profit or otherwise derive revenue from any products or services that Franchisor, Franchisor's affiliates, or Franchisor's Approved Suppliers supply and/or provide to Franchisee. Franchisor has the irrevocable right to modify, supplement, or otherwise change its lists of Approved Suppliers and any items that must be purchased from such Approved Suppliers at any time, as Franchisor deems advisable in its sole discretion. Franchisor may mandate that Franchisee use a specific payment processor in its Franchised Business, in which Franchisor or an affiliate may (or may not) be an owner, and Franchisor or its affiliate may collect fees owed to it or an affiliate at the time of payment by a customer using such processor. Franchisor may provide Franchisee with notice of such modifications to these lists via the Operations Manual or any other manner Franchisor deems appropriate.

7.4.3 Supplier Approval. In the event Franchisee wishes to purchase any unapproved item, including inventory, and/or acquire approved items from an unapproved supplier, Franchisee must provide Franchisor the name, address and telephone number of the proposed supplier, a description of the item Franchisee wishes to purchase, and the purchase price of the item, if known. At Franchisor's request, Franchisee must provide Franchisor, for testing purposes, a sample of the item Franchisee wishes to purchase. Nothing in the foregoing shall be construed to require Franchisor to approve any particular supplier. Franchisor may base Franchisor's approval of any such proposed item or supplier on considerations relating not only directly to the item or supplier itself, but also indirectly to the uniformity, efficiency and quality of operation Franchisor's deems necessary or desirable in Franchisor's System as a whole, as well as the maintenance of Franchisor's Confidential Information. Franchisor has the right to use all amounts Franchisor receives without restriction (unless instructed otherwise by the supplier) for any purposes Franchisor deems appropriate. Nothing herein shall require Franchisor to approve an unreasonable number of suppliers for a given item, which approval might, in Franchisor's reasonable judgment, result in higher costs or prevent the effective or economical supervision of approved suppliers.

7.4.3.1 Franchisee or the proposed supplier must pay Franchisor its then-current alternative supplier or new product review fee, in addition to Franchisor's actual costs that Franchisor estimates it will incur in connection with inspecting the alternate supplier, its facilities, and/or the previously non-approved item(s) proposed by Franchisee. If the costs Franchisor incurs are more than the amount Franchisee or the proposed supplier advanced, then Franchisor may withdraw additional funds through the EFT Program from Franchisee's designated bank account for the difference, or if the actual amount Franchisor incurs is less than the amount of the advancement, then Franchisor shall credit the excess amount to the payment of Franchisee's future obligations.

7.4.3.2 Franchisor will use commercially reasonable efforts to notify Franchisee in writing whether or not Franchisee's request is approved or denied within thirty (30) days of: (i) Franchisor's receipt of all supporting information from Franchisee regarding Franchisee's request under this Section; and (ii) if applicable, Franchisor's completion of any inspection or testing associated with Franchisee's request. If Franchisor does not provide written approval within this time period, then Franchisee's request will be deemed denied.

7.4.3.3 Franchisor may, but is not obligated to, provide Franchisee's proposed supplier or provider with its specifications for the item that Franchisee wishes the third-party to supply, provided that third-party executes Franchisor's prescribed form of non-disclosure agreement.

- 7.4.3.4 Each supplier that Franchisor approves of must comply with Franchisor's requirements regarding insurance, indemnification and non-disclosure. If Franchisor approves any supplier, Franchisee may enter into supply contracts with such third party, but under no circumstances will Franchisor guarantee Franchisee's performance of any supply contract.
- 7.4.3.5 Franchisor may re-inspect and revoke Franchisor's approval of particular products or suppliers when Franchisor determines, in Franchisor's sole discretion, that such products or suppliers no longer meet Franchisor's standards. Upon receipt of written notice of such revocation, Franchisee must cease purchasing products from such supplier.
- 7.4.4 System Suppliers. Franchisor may establish business relationships, from time to time, with suppliers who may produce and/or provide certain products or services that Franchisee is required to purchase from only that supplier (each a "System Supplier"). These System Suppliers may provide, among other things, supplies, fixtures, technology, software and equipment, all in accordance with Franchisor's proprietary standards and specifications, or private label goods that Franchisor has authorized and prescribed for sale by System franchisees. Franchisee recognizes that such products and services are essential to the operation of the Franchised Business and to the System generally. Franchisee further recognizes that Franchisee's failure to pay System Suppliers may interfere with such suppliers' willingness to supply the System and may result in other System franchisees' inability to obtain product or ability to obtain product only on less favorable credit terms. Accordingly, Franchisee agrees to pay System Suppliers as and when due. Franchisee must use products purchased from Approved Suppliers solely in connection with the operation of the Franchised Business and not for any competitive business purpose.
- 7.5 **Approved Products and Services.** Franchisee shall offer for sale all products and services which Franchisor prescribes and only those products and services which Franchisor prescribes. Franchisee may not offer any other products or services for sale without having received Franchisor's prior written authorization. Franchisee shall at all times maintain sufficient levels of inventory, as specified in the Operations Manual, to adequately satisfy consumer demand. Franchisee must offer, use and sell all private label products which Franchisor may now or in the future designate for sale by System franchisees. In the event Franchisee wishes to offer any Approved Products or Approved Services that Franchisor indicates requires additional training or certification from Franchisor or its designee, then Franchisee must complete such training and/or obtain such certification, at Franchisee's sole expense, prior to providing these specialized Approved Products and Approved Services.

7.6 **Operations**.

- 7.6.1 *Hours of Operation.* The Franchisee must operate the Franchised Business for at least those days and number of hours Franchisor specifies in the Operations Manual.
- 7.6.2 Maintenance of Premises and Project Sites. Franchisee must maintain the Franchised Business and all project sites in a clean, safe and attractive manner, and in accordance with all applicable requirements of law, including all federal, state and local health laws, as well as this Agreement and the Operations Manual. Franchisee and Franchisee's employees must give prompt, courteous and efficient service to the public and otherwise operate the Franchised Business so as to preserve, maintain and enhance the reputation and goodwill of the System.
- 7.6.3 *Personnel/Staffing*. Franchisee must employ a sufficient number of qualified, properly licensed, and competent personnel to offer prompt, courteous and efficient service to the public, and otherwise operate the Franchised Business in compliance with the System so as to preserve, maintain and enhance the reputation and goodwill of the System. All employees engaged in the operation of the Franchised Business during working hours shall dress conforming to Franchisor's standards and shall

present a neat and clean appearance in conformance with Franchisor's reasonable standards and shall render competent, efficient service to the customers of the Franchised Business.

- 7.6.4 Compliance with Operations Manual and Training of Employees. Franchisee agrees to conduct the Franchised Business in accordance with the Operations Manual. Franchisee shall immediately train and instruct Franchisee's employees in accordance with the Operations Manual, and shall continue such training and instruction as long as each employee is employed. The Operations Manual shall set forth the practices, procedures and methods to be utilized in the Franchised Business and Franchisor may require Franchisee to conform Franchisee's practices to national programs, which Franchisor has designed as part of Franchisor's System. mandatory portions of the Operations Manual.
- 7.6.5 Management Participation. Franchisee (or at least one of Franchisee's principals if Franchisee is a corporation or partnership) must devote their personal full-time attention and best efforts to the management and operation of the Franchised Business. Upon Franchisee's written request, Franchisor may permit Franchisee to employ a manager to manage the day-to-day operations of the Franchised Business (the "Designated Manager"), provided the Designated Manager: (i) is approved by Franchisor in writing prior to hiring; and (ii) successfully completes Franchisor's Initial Training Program (defined in Section 8.1) before assuming any managerial responsibility. The Franchised Business must, at all times, be staffed with at least one (1) individual who has successfully completed Franchisor's Initial Training Program. In the event that Franchisee operates more than one Franchised Business, Franchisor may require Franchisee to have a properly trained Designated Manager who has been approved by Franchisor at each location. Franchisee will keep Franchisor informed at all times of the identity of any employee acting as Designated Manager of the Franchised Business. In the event that a Designated Manager resigns or is otherwise terminated from the Franchised Business, the replacement must be trained pursuant to Franchisor's then-current standards. Franchisee must replace the Designated Manager within a reasonable amount of time not to exceed 90 days. The new Designated Manager must successfully complete training within thirty (30) days of hiring. Franchisor reserves the right, without the obligation, to train the new Designated Manager directly. Franchisee and any Designated Manager(s) are not permitted to seek or maintain other employment or engage in any other business activities during the term of this Agreement.
- 7.6.5.1 If Franchisee's Designated Manager departs the Franchised Business and Franchisee does not have another Designated Manager that satisfies the requirements in Section 7.6.5 above, then Franchisor may, at its option, provide Franchisee with a temporary replacement Designated Manager and Franchise shall be obligated to pay the replacement Designated Manager the then-current fee for each full or partial day.
- 7.6.5.42 Notwithstanding the foregoing, Franchisee will not be in violation of the obligations of this Section 7.6.5 and Section 7.10 if Franchisee (or at least one of Franchisee's principals if Franchisee is a corporation or partnership) or the Designated Manager(s) devotes appropriate time, resources, and effort to supervising the development and operation of the Franchised Business. The owners of Franchisee must directly oversee the Designated Manager (where one is appointed pursuant to the Franchise Agreements) and such owners may engage in other non-competitive business or activity so long as it does not materially interfere with the operation of the Franchised Businesses.
- 7.6.6 *Working Capital*. Franchisee must at all times maintain such working capital as may be reasonably necessary to enable Franchisee to properly and fully carry out and perform all of Franchisee's duties, obligations and responsibilities hereunder and to operate the business in a businesslike, proper and efficient manner.
- 7.6.7 *Inventory.* Prior to commencement of operations, Franchisee shall adequately supply the Franchised Business with supplies, vehicles, equipment and inventory as prescribed by the

Franchisor in the Operations Manual, and any other items of the type, quantity and quality as specified by the Franchisor in the Operations Manual. Franchisee must, at all times, maintain sufficient levels of inventory, including Franchisor's proprietary products and other equipment and supplies used at project sites, as required by Franchisor in the Operations Manual to adequately meet consumer demand. Vehicles may be leased or rented. Provided however, rented vehicles may not display the trademarks of a Competing Business.

- 7.6.8 *Products with Proprietary Marks*. Franchisee shall, in the operation of its Franchised Business, use and display labels, forms, vehicles, supplies, equipment and inventory imprinted with the Proprietary Marks and colors as prescribed by the Franchisor.
- 7.6.9 *Market Research*. Franchisor may, from time to time, conduct market research and testing to determine the viability of new products and services. Franchisee must cooperate by participating in such programs and by purchasing and promoting the sale of such test products and services, if required by the Franchisor.
- 7.6.10 *Customer Contracts*. In the operation of the Franchised Business, Franchisee is required to use only the customer contracts, waivers, and/or other forms that are designated by Franchisor from time to time. Franchisor is not obligated to designate any such items and Franchisee's use obligations do not apply where Franchisor does not designate such items. Franchisor may provide Franchisee with templates or sample forms of such items, but it is Franchisee's responsibility to have all items which are to be used with prospective and/or actual customers reviewed, at Franchisee's expense, by an attorney licensed to practice law in the state(s) where the Franchised Business is operated, for compliance with all applicable state and local legal requirements. Franchisor makes no warranty or representation that any contracts, waivers and/or other forms and/or materials, whether supplied by Franchisor or otherwise, are in compliance with the laws of any particular state(s) or locality.
- 7.7 **Franchised Business Inspection**. Franchisee agrees that, in order to maintain the high quality and uniform standards associated with the Franchise System and to protect its goodwill and reputation, Franchisee will permit Franchisor, during business hours, to inspect Franchisee's Franchised Business or attend a project site, confer with Franchisee and Franchisee's employees and customers, observe and evaluate Franchisee's sales techniques and operation methods, and perform any other inspection which Franchisee's performance under this Agreement, the Operations Manual, and other standards and specifications required by Franchisor. Franchisee is obligated to make changes to Franchisee's operations based upon any inspections by Franchisor. Franchisor is not required to provide Franchisee with any notice prior to conducting such an inspection.

7.8 Computer Software and Hardware.

7.8.1 Business Management and Technology System. Franchisor's business management and technology system ("Business Management and Technology System") is a proprietary integrated business management system, whichthat includes customized software that facilitates the flow of business-related information between Franchisee, Franchisor, and Franchisee's customers. The Business Management and Technology System will be managed and maintained by either Franchisor, a third-party company selected by Franchisor, or a combination of both. The Business Management and Technology System includes software that manages customer / Franchisee / Franchisor interactions from demand generation and lead intake through invoicing and collections, Franchiser's websites, telematics, and the Call Center (if established). If the Call Center is established, Franchise will beis required to forward any and all calls made to the Franchised Business (and all telephone numbers associated therewith) so that Franchisor can receive/route/assign these customer calls through the Business Management and

Technology System as more fully set forth in this Agreement. Franchisee will have access to the Business Management and Technology System through a unique login.

- 7.8.2 Computer System. Franchisor shall have the right to specify or require that certain brands, types, makes, and/or models of communications, computer systems and hardware be used by Franchisee, including, without limitation: (a) a compatible "back office" computer system that complies with Franchisor's standards and specifications and is capable of operating designated and Required Software (as defined below in Section 7.8.3); (b) Franchisor's Business Management and Technology System; (c) a custom and proprietary point of sale system (the "POS System"), if Franchisor makes such a POS System part of its proprietary operating system in the future; (d) accounting software; Ke software applications and programs; (f) printers and other peripheral hardware or devices; (g) archival back-up systems; (h) Internet access mode and speed; and (i) physical, electronic, and other security systems (collectively, the "Computer System"). Franchisee must dedicate its Computer System for use as the Business Management and Technology System only and use the Business Management and Technology System in accordance with Franchisor's policies and operational procedures. Franchisee's employees must complete any and all training programs Franchisor reasonably requires for the proper operation and use of the Business Management and Technology System. Franchisee may not use any other cash registers or computer systems in its Franchised Business.
- 7.8.3 Required Software. Franchisor shall have the right, but not the obligation, to develop or designate: (i) computer software programs that Franchisee must use in connection with any component of the Computer System, including Franchisor's Business Management and Technology System software (the "Required Software"), which Franchisee shall install at Franchisee's expense; (ii) updates, supplements, modifications, or enhancements to the Required Software, which Franchisee shall install at Franchisee's expense; (iii) the tangible media upon which Franchisee records data; and (iv) the database file structure of the Computer System.
- 7.8.4 Compliance with Requirements. At Franchisor's request, Franchisee shall purchase or lease, and thereafter maintain, the Computer System and, if applicable, the Required Software. Franchisee expressly agrees to strictly comply with Franchisor's standards and specifications for all items associated with Franchisee's Computer System and any Required Software. (including the Business Management and Technology System). Franchisee agrees, at its own expense, to keep its Computer System in good maintenance and repair and. Franchisee and Franchisor agree that maintaining current technological capabilities is necessary and appropriate in order to meet customer preferences and/or requirements as they evolve over time, and that as a result of such changes over time, updates and/or new or additional software, hardware, and systems may be added to the specifications. Franchisee agrees, at its own expense to install such additions, changes, modifications, substitutions, and/or replacements to Franchisee's Computer System or Required Software as Franchisor directs from time to time in writing. Franchisee agrees that its compliance with this Section shall be at Franchisee's sole cost and expense. Franchisee agrees that maintaining current technological capabilities is necessary and appropriate in order to meet customer preferences and/or requirements as they evolve over time, and that as a result of such changes over time, updates and/or new or additional software, hardware, and systems may be added to the specifications.
- 7.8.5 Franchisor's Access. Franchisor may require that Franchisee's Computer System be programmed to automatically transmit data and reports about the operation of the Franchised Business to Franchisor. Franchisor shall also have the right to, at any time without notice, electronically connect with Franchisee's Computer System to monitor or retrieve data stored on the Computer System or for any other purpose Franchisor deems necessary. There are no contractual limitations on Franchisor's right to access the information and data on Franchisee's POS system and Computer System. Franchisee shall deliver to Franchisor all access codes, static Internet protocol ("IP") addresses and other information to facilitate

Franchisor's access to the data described in this Section within thirty (30) days of opening the Franchised Business.

- 7.8.6 *Proprietary Software*. Franchisor has a proprietary interest in all databases, lists, templates, programs, and any other software components that have been created and/or customized by Franchisor using the Computer System and/or Required Software (the "Proprietary Software"). In the future, Franchisor may further customize the Proprietary Software and create programs that conduct, among other things, scheduling, accounting, inventory, and related activities. Franchisee must obtain the computer hardware necessary to implement the Proprietary Software into the Franchised Business; and comply with all specifications and standards prescribed by Franchisor regarding the Proprietary Software as provided in the Operations Manual. This Proprietary Software will be Franchisor's proprietary product and the information collected therefrom will be deemed Franchisor's Confidential Information.
- 7.8.7 Computer Network. Franchisee is required to participate in any System-wide computer network, intranet system, extranet system or community portal that Franchisor implements and that may be required by Franchisor to use such computer network, intranet system, or extranet system to, among other things: (i) submit Franchisee's reports due under this Agreement to Franchisor online; (ii) view and print portions of the Operations Manual, including any updates or modifications thereto; (iii) download approved marketing materials; (iv) communicate with Franchisor and other System franchisees; and (v) complete any initial or ongoing training that Franchisor designates, in the event Franchisor makes such training accessible through this medium. Franchisee agrees to use the facilities of any such computer network, intranet system or extranet system in strict compliance with the standards, protocols and restrictions that Franchisor included in the Operations Manual, including those related to the encryption of confidential information and prohibitions against the transmission of libelous, derogatory or defamatory statements. Franchisee understands and agrees that it is solely responsible for any licensing and/or maintenance fee(s) associated with any intranet or extranet described in this Section.
- 7.9 **Personal Conduct**. Franchisee agrees to refrain from committing any act or pursuing any course of conduct that tends to bring Franchisor's Proprietary Marks into disrepute.
- 7.10 **Best Efforts.** Franchisee (or Franchisee's principals) must devote their personal full-time attention, skill and best efforts to the management and operation of the Franchised Business and to promote and increase the demand for the Franchisor's products and services within the Designated Territory. In consideration of the grant of the Franchised Business, Franchisee (or Franchisee's principals) agrees that it will not own, maintain, engage in, be employed by, or have any interest in any other business other than the Franchised Business without the prior written consent of Franchisor. Franchisee agrees that Franchisee may not, without the prior written consent of Franchisor, engage in any commercial activity that: (i) is not performed for the sole and direct benefit of the Franchised Business; (ii) may benefit or promote any other business; or (iii) may be injurious to the Franchised Business or the goodwill associated with the Proprietary Marks and the System. Franchisee acknowledges that Franchisee's (or Franchisee's principals') violation of the terms in this Section will be a material breach of this Agreement, and Franchisor may terminate this Agreement with notice and without an opportunity to cure. The foregoing remedy shall be in addition to any other legal or equitable remedies that the Franchisor may possess.
- 7.11 **Telephone and Email Access.** Franchisor reserves the right to procure and supply all telephone numbers and email accounts associated with the Franchised Business.
- 7.12 **Payment of Debts**. Franchisee is solely responsible for: (i) selecting, retaining and paying Franchisee's employees; (ii) the payment of all invoices for the purchase of products and services used in connection with operating the Franchised Business; and (iii) determining whether, and on what terms, to

obtain any financing or credit which Franchisee deems advisable or necessary for the conduct of the Franchised Business. Franchisee agrees to pay all current obligations and liabilities to suppliers, lessors, and creditors on a timely basis. Franchisee agrees to indemnify Franchisor in the event that Franchisor is held responsible for debts owed by Franchisee if Franchisor elects to pay any of Franchisee's obligations in order to preserve the relationship between System Suppliers and System franchisees. Franchisee agrees to make prompt payment of all federal, state and local taxes, including individual and corporate taxes, sales and use taxes, franchise taxes, gross receipts taxes, employee withholding taxes, FICA taxes, personal property, and real estate taxes arising from Franchisee's operation of the Franchised Business. Franchisee agrees to indemnify Franchisor in the event that Franchisor is held responsible for these taxes.

7.13 Compliance with Applicable Laws.

__Franchisee must comply with all applicable federal, state and local laws, ordinances and regulations regarding the operation of the Franchised Business (including, without limitation, all government regulations relating to required licensing; occupational hazards and health; trademark and copyright infringement; fair marketing laws; consumer protection; trade regulation; workers' compensation; unemployment insurance; withholding and payment of Federal and State income taxes and social security taxes; sales, use and property taxes; and the applicable provisions of the Americans with Disabilities Act ("ADA") regarding the construction, design and operation of the Franchised Business). Franchisee will have sole authority and control over the day-to-day operations of the Franchised Business and Franchisee's employees and/or independent contractors. Franchisee is solely responsible for all employment decisions and is solely responsible for compliance with all local, state, and federal and local hiring laws and functions of the Franchised Business, including, without limitation, those related to hiring, firing, training (except for brand standards certification training we offer Franchisor offers to you Franchisee and yourits Designated Manager), wage and hour requirements, compensation, promotion, recordkeeping, supervision, and discipline of employees, paid or unpaid, full or part-time. At no time will Franchisee or Franchisee's employees be deemed to be employees of Franchisor or Franchisor's affiliates. Any information we provide Franchisor provides about employment matters, whether voluntarily or in response to your request, and whether directly or by means of any technology tools, is a recommendation only and is not intended to and is not an actual exercise of control over the wages, hours or working conditions of yourFranchisee's employees or the means and manner by which they carry out their duties. You are Franchisee is strongly encouraged to seek independent competent labor law counsel in the state/locality where you do Franchisee does business for all employment matters.

YouFranchisee alone will direct and control all employees of the Franchised Business, subject only to the Brand Standards that we prescribeFranchisor prescribes to protect the goodwill associated with the Marks, which may include the requirement of initial and periodic drug testing and background checks. You areFranchisee is required to clearly inform all workers, before hiring and periodically thereafter, that Franchisee, and not Franchisor, is their employer. You agreeFranchisee agrees to indemnify usFranchisor for any liability, cost, expense, loss or damage, including attorney's fees and costs, arising from—(i) any claim or allegation that Franchisor or any affiliate is the employer, co-employer, or joint employer of Franchisee, its Owners, or any workers in the Franchised Business.

- 7.14 **Trade Secrets and Confidential Information**. Franchisee and all of its employees must maintain the confidentiality of all Confidential Information as set forth in this Agreement.
- 7.15 **Image.** Franchisee acknowledges that Franchisor has developed the System to offer and sell products and services which will distinguish the Franchised Business from other service businesses that offer similar products and services valued at different prices and with less attention paid to product quality and customer service. Franchisee agrees to offer products and services and to conduct the Franchised Business in such a manner which will serve to emulate and enhance the image Franchisor intended for the System. Franchisee further acknowledges and agrees that each aspect of the System is important not only

to Franchisee, but also to Franchisor and to other System franchisees in order to maintain the highest operating standards, achieve System-wide uniformity, and increase the demand for the products sold and services rendered by System franchisees. Franchisee agrees to comply with the standards, specifications and requirements Franchisor set forth in order to uniformly convey the distinctive image of a Franchised Business. Franchisee shall, in the operation of the Franchised Business, use only displays, bags, labels, forms, stationery, and other products Franchisor designates that are imprinted with the Proprietary Marks and colors, as prescribed from time to time by Franchisor.

- 7.16 **Pending Actions.** Franchisee shall notify Franchisor in writing, within five (5) days of the commencement of any action, suit or proceeding and the issuance of any order, suit or proceeding of any court, agency or other government instrumentality, including the receipt of any notice or citation, which may adversely affect the operation or financial condition of Franchisee or the Franchised Business.
- 7.17 **Standard Maintenance and System Conformity**. Franchisee agrees to repair, refinish, replace, and/or otherwise refurbish the Franchised Business's Computer System, POS System, Required Software, vehicle(s), trailer(s) (if applicable), equipment, tools, and the Approved Location's furnishings, fixtures, Idecor, and any other tangible part or property of the Franchised Business at Franchisee's sole expense at such times as Franchisor may reasonably direct. Franchisor has the right to direct Franchisee to remodel, re-equip, and otherwise refurbish the Approved Location in the manner necessary to bring it into conformance with other franchises of the type Franchisor's franchisees are opening at the time of such direction. A remodel/re-equip/refurbish will not be required in the last twelve (12) months of a term, but will be a condition of renewal. If at any time, in Franchisor's judgmentBusiness Judgment, the general state of repair or the appearance of the premises of the Approved Location or its vehicles or equipment does not meet Franchisor's standards, Franchisor shall so notify Franchisee, specifying the action to be taken by Franchisee to correct such deficiency. If Franchisee fails or refuses to initiate within fifteen (15) days after receipt of such notice, Franchisor shall have the right, in addition to all other remedies, to enter upon the premises of the Approved Location and affect such maintenance on behalf of Franchisee, and Franchisee shall pay the entire costs thereof on demand.
- 7.18 **Customer Service**. Franchisee must comply with any standards, specifications or methodologies that Franchisor establishes in the Operations Manual or otherwise in writing regarding customer service requirements, a customer or product warranty, or a satisfaction guarantee on any Approved Products or Approved Services offered or sold by the Franchised Business, refund policies, and other standards and specifications. If specified in the Operations Manual, you are Franchisee is required to provide the warranty or satisfaction guarantee to each customer and comply with the requirements of the warranty/guarantee program, as set forth in the Operations Manual.

8. TRAINING

8.1 **Initial Training Program**. Franchisor will provide Franchisee (or one of Franchisee's principal owners if Franchisee is a business entity) and up to two (2) additional representatives that Franchisee designates with Franchisor's then-current initial training program (the "Initial Training Program") tuition-free, subject to the availability and schedule of Franchisor's training personnel. Franchisee must attend and successfully complete the Initial Training Program to Franchisor's satisfaction at Franchisor's corporate headquarters (the "Home Office"), or other location that Franchisor designates, prior to commencing operations of the Franchised Business. Some or all of the classroom training may now or in the future be provided virtually. If Franchisee is a business entity, each franchise owner must attend and successfully complete the Initial Training Program. The Initial Training Program lasts approximately five (5) days, and Franchisee will be solely responsible for costs and expenses Franchisee and its representative(s) incur in connection with attending the Initial Training Program, including travel, lodging, meals or any wages incurred for Franchisee's employees. In the event Franchisor permits

Franchisee to employ a Designated Manager (defined in Section 7.6), such Designated Manager must attend and complete the Initial Training Program to Franchisor's satisfaction prior to commencing any managerial duties of the Franchised Business.

- 8.1.1 Timing for Completion. Franchisee and its designated trainees must participate in and complete the Initial Training Program to Franchisor's satisfaction thirty (30) days prior to opening the Franchised Business and within sixty (60)90 days from the Effective Date of this Agreement. Upon completion of the Initial Training Program, the Franchised Business shall be deemed "open and operating." In the event Franchisee does not complete the Initial Training Program to Franchisor's satisfaction within this 90-day period, then Franchisor may terminate this Agreement or Franchisor may provide Franchisee with additional time in exchange for a general release.
- 8.1.2 Additional Employees. In the event Franchisee requests that more than two (2) additional people participate in the Initial Training Program (other than Franchisee or Franchisee's partner or principal shareholder), even if they attend the same training session, Franchisor may provide the Initial Training Program to such additional persons, subject to the availability of Franchisor's personnel, and charge Franchisee its then-current training tuition fee. All training related expenses for Franchisee's additional personnel, including transportation to and from the training site, lodging, meals, and salaries during training are Franchisee's sole responsibility.
- 8.1.3 Replacement Personnel. In the event Franchisee or Franchisee's employee(s) fail(s) to complete the Initial Training Program to Franchisor's satisfaction, the respective person may repeat the course, or, in the case of an employee, Franchisee may send a replacement (the "Replacement Personnel") to the next available training session. Franchisor may charge its then-current training tuition fee for such Replacement Personnel to attend the Initial Training Program. Franchisor may also charge its then-current training tuition fee for each subsequent operating principal, designated manager or employee who attends the course at a scheduled training class at our headquarters or other location designated by usFranchisor. Failure by Franchisee, subsequent operating principals, designated managers, an employee, or any Replacement Personnel to complete the Initial Training Program to Franchisor's satisfaction within the time period prescribed in this Agreement shall constitute default of this Agreement and Franchisor may terminate the Agreement.
- 8.1.4 Employee Training. Franchisee must ensure that any and all employees of the Franchised Business that do not attend the Initial Training Program are properly trained to perform their respective duties in connection with the Franchised Business prior to such employee(s) undertaking the duties.
- 8.1.5 Training Materials. Franchisor will provide Franchisee with training materials for Franchisee to use in training Franchisee's personnel. Only Franchisor's providedFranchisee must use Franchisor training materials to train Franchisee's personnel on brand standard topics and may be used by Franchisee insupplement training Franchisee's personnel with its own materials so long as they are not inconsistent with Franchisor training materials. Updated training materials may be available to Franchisee in the Operations Manual or by other means, in Franchisor's sole discretion. All training materials provided to Franchisee by Franchisor shall at all times remain Franchisor's property, and Franchisee agrees not to challenge Franchisor's or Franchisor's affiliates' title or rights in or to the training materials. Franchisee may not make any disclosure, duplication or other unauthorized use of any portion of the training materials.
- 8.2 Additional and Remedial Training. Franchisor may conduct, and require Franchisee, Franchisee's Designated Manager (if applicable), and other employees to attend additional and/or refresher training courses that Franchisor develops for the benefit of the System, as Franchisor deems advisable in its sole discretion. Franchisor may charge Franchisee its then-current training tuition fee for Franchisee and

any other persons that attend such additional or refresher training, and Franchisee will be solely responsible for any and all expenses associated with such training (including travel, lodging, meals and employee wages incurred). Additional and/or refresher training may take place at Franchisor's Home Office or any other location that Franchisor designates. Franchisor will provide Franchisee with thirty (30) days' notice of any upcoming additional or refresher training that Franchisee is required to attend.

8.3 **Reasonable Training and Assistance Requests**. Upon Franchisee's written request, Franchisor may provide Franchisee with additional training and/or assistance, as Franchisor deems necessary in its sole discretion, subject to the availability and schedules of Franchisor's personnel. Franchisor may charge Franchisee its then-current tuition fee for any training or assistance that Franchisor provides at Franchisee's request, and Franchisee is solely responsible for any expenses and costs incurred: (i) by Franchisee and its representatives in connection with attending such additional training; and (ii) by Franchisor in connection with providing such training or assistance, whether at the Home Office or on-site at Franchisee's Approved Location or within the Designated Territory. Additional assistance may be provided by Franchisor over the phone, via email or Franchisor's Operations Manual.

9. INSURANCE

- General. Franchisee must maintain, at Franchisee's expense, in full force and effect throughout the term of this Agreement, the types of insurance and the minimum policy limits specified in the Operations Manual. In determining and modifying such requirements, Franchisor agrees to use reasonable business judgment and only require such insurance and minimum policy limits that are reasonable in its Business Judgment. The insurance policy or policies must be in effect at the earlier of: (i) sixty (60) days after the Effective Date of this Agreement; or (ii) prior to attending the Initial Training Program. The insurance policy or policies must protect Franchisee, Franchisor, and Franchisor's past, present, and future: officers, directors, owners, managers, members, stockholders, affiliates, employees, consultants, attorneys and agents, against any loss, liability, personal injury, death, property damage, or expense whatsoever arising out of or occurring upon or in connection with the condition, operation, use, or occupancy of the Franchised Business. Franchisee shall list Franchisor as an additional insured under each policy, except for policies required by statute in Franchisee's jurisdiction, including, but not limited to, workers' compensation and employer's liability insurance policies. Franchisor reserves the right to amend, modify, and/or supplement additional types of coverage and/or increase the required minimum amount of coverage upon providing Franchisee reasonable notice through the Operations Manual or otherwise in writing by Franchisor. Franchisee's obligation to obtain coverage is not limited in any way by insurance that Franchisor maintains. Upon Franchisor's request or as specified in the Operations Manual, Franchisee shall provide Franchisor with certificates of insurance evidencing the required coverage and any other documentation in connection therewith.
- 9.2 **Insurance Rating, Approval, and Certification**. All insurance carriers must be approved by Franchisor in advance and in writing. All insurance policies must be issued by insurance companies with a rating of A-VII or better as reported in the most recent edition of A.M. Best's Insurance Report. Franchisor's acceptance of an insurance carrier does not constitute Franchisor's representation or guarantee that the insurance carrier will be capable of meeting claims during the term of the insurance policy. Franchisee agrees to carry such insurance as may be required by the Lease of the Approved Location or by any of Franchisee's lenders or equipment lessors, and such workers' compensation insurance as may be required by applicable law. Franchisee must deliver a certificate of insurance to Franchisor at least twenty (20) days prior to opening the Franchised Business and ten (10) days prior to any renewal of the required policies, as evidence that all insurance requirements have been met. All insurance policies held by Franchisee will be primary to any policy or policies held by Franchisor or its affiliates.

- 9.3 **Designees**. All liability policies will list Franchisor as an additional insured except the Employment Practices Liability policy where Franchisor will be named as co-defendant. The Commercial General Liability policy shall contain a waiver of subrogation in favor of Franchisor and any parties Franchisor designates, and will be primary and non-contributory to any insurance Franchisor might carry. Franchisor reserves the right to modify required insurance coverage during the course of Franchisee's Agreement based on changes in risk factors, for which Franchisee will comply with upon written notice from Franchisor.
- 9.4 **Claims Cancellation.** Franchisee must provide Franchisor with copies of any insurance claims or insurance cancellations within twenty-four (24) hours of Franchisee's receipt of said claims or cancellations. Franchisee has a twenty four (24) 48 hour opportunity to cure any lapses in insurance coverage. No insurance policy must be subject to cancellation, termination, non-renewal or material modification, except upon at least thirty (30) calendar days' prior written notice from the insurance carrier to Franchiser. Franchisee must submit a certificate of insurance which demonstrates compliance with this Section 9.
- 9.5 **Failure to Maintain Insurance**. If Franchisee fails to comply with the minimum insurance requirements set forth herein, Franchisor has the right to obtain such insurance and keep same in force and effect, and Franchisee shall pay Franchisor, on demand, the premium cost thereof and a \$100 administrative fee for the costs incurred in connection with Franchisor obtaining the insurance.
- 9.6 **Modification of Requirements**. Franchisor has the right to increase or otherwise modify the minimum insurance requirements upon thirty (30) days' prior written notice to Franchisee, and Franchisee shall comply with any such modification within the time specified in said notice.

10. FINANCIAL RECORDS AND REPORTS

- 10.1 **Reporting**. Franchisee must maintain, for at least ten—(10) fiscal years from their preparation, full, complete accurate records of all sales, marketing activities, contracts, estimates, authorizations, receipts, payroll and accounts payable, and any other documents and records used in connection with the Franchised Business, in accordance with the standard accounting system described by the Franchisor in the Operations Manual or otherwise specified in writing. Franchisee must also provide Franchisor with complete financial records for the operation of the Franchised Business as described in this Section in accordance with generally accepted accounting principles.
- 10.1.1 Franchisee will, at its expense, submit to the Franchisor within sixty (60) days of the end of each calendar year of the Franchised Business during the term of this Agreement, a complete financial statement for the said calendar year, including, without limitation, both an income statement and balance sheet, which may be unaudited, together with such other information in such form as the Franchisor may require.
- 10.1.2 Each financial statement shall be signed by Franchisee or by an individual authorized by Franchisee, attesting that the statement is true and correct and prepared in accordance with Franchisor's requirements.
- 10.1.3 Franchisee must provide Franchisor with access to the information generated by any software Franchisor requires Franchisee to use in connection with its bookkeeping and other accounting obligations under this Agreement, including QuickBooks and Franchisor's software provider(s). Franchisee must, upon Franchisor's request, grant temporary administrative access to any software Franchisor requires Franchisee to use in connection with its bookkeeping and other accounting obligations (including, but not limited to, QuickBooks or our then-current accounting software) to Franchisor and/or

its designee for the limited purpose of integration with franchise systems (i.e., ServiceMinderour digital platform or similar systems) or collecting financial reporting data as required under the Franchise Agreement.

- 10.1.4 Franchisee shall provide Franchisor with any other data, information and supporting records that Franchisor designates from time to time, including any and all reports set forth in the Operations Manual.
- 10.2 **Tax Returns.** In addition to the information and materials set forth in Section 10.1, Franchisee agrees to maintain, and furnish to Franchisor within thirty (30) days of filing: (i) complete copies of all federal, state and local tax returns, including those detailing income, sales, value added, use and service taxes, as well as employee withholding, workers' compensation, and similar reports filed by Franchisee reflecting financial activities of the Franchised Business; and (ii) Franchisee's (or Franchisee's principals') personal federal, state and local tax returns; and Franchisee hereby waives, to the extent not prohibited by applicable law, any right to object to disclosure of any tax returns.
- 10.3 **Right to Disclose Information**. Franchisor has the right to use and disclose data derived from the reports Franchisee furnishes to Franchisor. Franchisee hereby consents to usFranchisor obtaining, using and disclosing to third parties (including, without limitation, prospective franchisees, financial institutions, legal and financial advisors), for any purpose whatsoever or as may be required by law, any financial or other information contained in or resulting from information, data, materials, statements and reports received by usFranchisor or ourits affiliates (or disclosed to usFranchisor or ourits affiliates) in accordance with this Agreement. Notwithstanding the foregoing, tax return data will only be disclosed to legal and financial advisors or as may be required by law.

11. BOOKS AND RECORDS

11.1 **Records and Audits.** Franchisee must establish and maintain, at its own expense, accurate business records, reports, accounts, books and data relating to the operation of the Franchised Business that conforms to the requirements and formats (including a specific chart of accounts) Franchisor prescribes from time to time. WeFranchisor may designate an approved bookkeeping vendor (or one of a list of vendors) that youFranchisee must use and youFranchisee hereby eonsents to such vendor providing access to any business records that Franchisor would otherwise be entitled to pursuant to the Franchise Agreement.

Franchisee may be required to use the Computer System to maintain sales data and other information and to generate the reports Franchisor requires. Franchisor may require use of a specific accounting software and version thereof as set forth in the Operations Manual, and Franchisee will be solely responsible for any associated cost.

Franchisor and Franchisor's designees have the right to inspect and/or audit Franchisee's business records at any time during normal business hours to determine whether Franchisee is current with suppliers and otherwise operating in compliance with the terms of this Agreement and the Operations Manual. Moreover, Franchisee must also provide Franchisor with access to the information generated by any software Franchisor requires Franchisee to use in connection with its bookkeeping and other accounting obligations under this Agreement, including QuickBooks (or ourFranchisor's then-current accounting software, if different) and Franchisor's software provider.

If any audit reveals that Franchisee has understated Franchisee's financial information, including, but not limited to, Royalty Fee or Brand Fund Contribution payments by more than two percent (2%),%, or if Franchisee has failed to submit timely reports and/or remittances for any two (2) reporting periods

within any twelve (12)-month period, Franchisee must pay our costs and expenses, including the cost of such audit and/or inspection, and including the cost of outside auditors and attorneys (to the extent Franchisor incurs such costs), together with amounts due for Royalty Fees and other fees as a result of such underreporting and/or failure to submit reports, along with all late fees and interest which may otherwise be due under this Agreement.

- 11.2 **Corporate or Limited Liability Company Franchisee Records**. If Franchisee becomes a corporation, limited liability company or other business entity either prior to executing this Agreement or at any time during the term of this Agreement, the following requirements, when applicable, shall apply:
- 11.2.1 Copies of Franchisee's Articles of Incorporation or Charter, minutes of the annual meeting, by-laws and other governing documents, and any amendments thereto, copies of initial shareholder certificates and Shareholder Agreements, if any, and the Resolutions of the Board of Directors authorizing entry into this Agreement as required by the Franchisor and as set forth in the Operations Manual shall be promptly furnished to Franchisor.
- 11.2.2 Franchisee shall maintain a current list of all owners or members of record and all beneficial owners of any class of stock of Franchisee and shall furnish such list to Franchisor annually.
- 11.2.3 All members with, or shareholders of, Franchisee that directly or beneficially own 5% or more of Franchisee, shall jointly and severally guarantee Franchisee's performance hereunder and shall bind themselves to the terms of a Personal Guaranty in a form approved by Franchisor (see Exhibit A to this Agreement). All members and/or shareholders shall also be individually subject to the non-disclosure and confidentiality provisions set forth in this Agreement, as well as any and all in-term and post-term restrictive covenants. However, the requirements of this subsection shall not apply to any corporation registered under the Securities Exchange Act of 1934 (hereinafter known as a "Publicly Held Corporation").
- 11.2.4 The majority equity owner in the entity that becomes the Franchisee must complete the Initial Training Program and all other training required by the Franchisor and work directly in the day-to-day operations of the Franchised Business and devote his or her personal full-time attention, skill and best efforts to the management and operation of the Franchised Business, unless Franchisor agrees otherwise in writing.
- 11.2.5 All issued and outstanding stock certificates of such corporation shall bear the following legend:

EXAMPLE: "Transfers of these shares is subject to certain restrictions contained in a Franchise Agreement between _______and HFB FenceCo Franchising, LLC dated _____

_____11.2.6 A business entity Franchisee may own multiple Franchised Businesses of the same brand, but may not own more than one brand of Franchised Business, and any such other brands must be owned and operated through a separate business entity.

12. ADVERTISING

Recognizing the value of advertising and promotion, and the importance of the standardization of advertising and promotion programs to further the goodwill and public image of the System, the parties agree as follows:

- 12.1 **Generally**. With regard to advertising generally for the Franchised Business, Franchisee will only use or display the advertising materials Franchisor approves in writing. If Franchisee wishes to use any advertising or promotional materials other than those currently approved for use by System franchisees, then Franchisee must submit Franchisee's proposed materials to Franchisor for approval at least thirty (30) days prior to its intended use. Franchisor will use commercially reasonable efforts to notify Franchisee of Franchisor's approval or disapproval of the proposed materials within fifteen (15) days of the date such materials are received. If Franchisee does not receive Franchisor's written approval within fifteen (15) days, the proposed materials shall be deemed disapproved. All advertising must prominently display the Proprietary Marks and will comply with any standards for use of the Proprietary Marks that Franchisor establishes, as set forth in the Operations Manual or otherwise in writing. Franchisor may require Franchisee to discontinue the use of any advertising or marketing material within time frames prescribed by Franchisor, at Franchisee's sole cost and expense.
- 12.2 **Internet Website**. Franchisee must obtain and maintain adequate hardware and software in order to access the Internet at the bit speed Franchisor requires from time to time. Franchisee is prohibited, however, from establishing any website or other presence on the Internet, except as provided herein. Franchisor may unilaterally modify the provisions of this Section from time to time in its sole discretion.
- 12.2.1 Franchisor may, but is not obligated to, establish an Internet website that provides information about the System and the products and services offered by franchised businesses. Franchised Businesses. In the event Franchisor exercises its right to create such a website, Franchisor shall have sole discretion and control over the website (including timing, design, content and continuation).
- 12.2.2 Franchisor may, but is not obligated to, create interior pages on its website(s) that contain information about Franchisee's Franchised Business and other franchised businesses. If Franchisor does create such pages, Franchisor may require Franchisee to prepare all or a portion of the page for Franchisee's Franchised Business, at Franchisee's expense, using a template that Franchisor provides. All such information will be subject to Franchisor's approval prior to posting.
- 12.2.3 Except as approved in advance in writing by Franchisor, Franchisee must not establish or maintain a separate website, splash page, profile or other presence on the Internet, or otherwise advertise on the Internet or any other public computer network in connection with the Franchised Business, including any profile on Facebook, Twitter, LinkedIn, YouTube, Pinterest, Instagram, Snapchat, or any other social media and/or networking site. If such approval is granted by Franchisor, Franchisee must: (i) establish and operate such site or page in accordance with System standards and any other policies Franchisor designates in the Operations Manual or otherwise in writing from time to time; and (ii) utilize any templates that Franchisor provides to Franchisee to create and/or modify such site(s). Franchisor will retain ownership of any and all social media or referral websites / accounts.
- 12.2.4 Franchisor may use a portion of the Brand Fund Contribution to pay or reimburse itself for the costs incurred in connection with the development, maintenance and update of its website.
- 12.2.5 Franchisee acknowledges that Franchisor and/or Franchisor's affiliates are the lawful, rightful and sole owner of the Internet domain name https://toprailfence.com/, as well as any other Internet domain names registered by Franchisor and/or Franchisor's affiliates, and unconditionally disclaims any ownership interest in such Internet domain names and any Internet domain names similar thereto. Franchisee agrees not to register any Internet domain name in any class or category that contains words used in or similar to any brand name owned by Franchisor or Franchisor's affiliates or any abbreviation, acronym, phonetic variation, or visual variation of those words.

- 12.3 **Brand Fund**. Franchisor has established a Brand Fund for the common benefit of System franchisees. Franchisor requires Franchisee to participate in and contribute on a monthly basis, one percent (1%)2% of the Franchised Business's Gross Revenue to the Brand Fund. Notwithstanding the foregoing, Franchisor reserves the right to increase the Brand Fund Contribution at any time upon providing notice to Franchisee, provided that the Brand Fund Contribution will not exceed three percent (3%)% of Gross Revenue during the term of this Agreement except as provided in Section 1.1.1. Franchisee must pay the Brand Fund Contribution directly to the Brand Fund via EFT on a monthly basis on or before the tenth (10th) business day of each month based on the Gross Revenue generated during the preceding calendar month. Franchisor reserves the right to modify the frequency, manner, or method of payment upon providing reasonable notice to Franchisee.
- 12.3.1 Franchisor will use the Brand Fund, in Franchisor's sole discretion, to develop for the general promotion of the Proprietary Marks and the System, including to develop, produce and distribute national, regional and/or local advertising and to create advertising materials and public relations programs which promote, in Franchisor's sole judgment, the services offered by System franchisees. Franchisor has the sole right to determine contributions and expenditures from the Brand Fund or any other advertising program, and sole authority to determine, without limitation, the selection of the advertising materials and programs, provided, however, that Franchisor will make a good faith effort to expend the Brand Fund in the general best interests of the System on a national or regional basis. Franchisor may use the Brand Fund to satisfy any and all costs of maintaining, administering, directing, preparing and producing advertising, including: (i) the cost of preparing and producing Internet, television, radio, magazine and newspaper advertising campaigns; (ii) the cost of direct mail and outdoor billboard advertising; (iii) the cost of soliciting National Accounts; (iv) the cost of public relations activities and advertising agencies; (v) the cost of developing and maintaining an Internet website; (vi) personnel and other departmental costs for advertising that Franchisor internally administers or prepares; and (vii) the costs of building partnerships with national or regional brands. Nevertheless, Franchisee acknowledges that not all System franchisees will benefit directly or on a pro rata basis from such expenditures. The Brand Fund will not be used for the direct solicitation of franchise sales, but Franchisor reserves the right to include a notation in any advertisement indicating "Franchises Available." Franchisor also reserves the right to use the Brand Fund for public relations or recognition of the brand.
- 12.3.2 Franchisor may periodically assist franchisees to maintain high quality standards through customer surveys, customer interviews, and other similar initiatives ("Surveys"). The cost of such programs will be borne by the Brand Fund. The cost of these programs may be charged directly to Franchisee if Franchisee's results from a Survey fall below System-established minimum standards for such Surveys.
- 12.3.3 Franchisor has the right to reimburse itself from the Brand Fund for such costs and overhead, if any, as Franchisor may incur in activities reasonably related to the direction and implementation of the Brand Fund.
- 12.3.4 In Franchisor's sole discretion, units owned and operated by Franchisor or its affiliates may, but are not obligated to, contribute to the Brand Fund. Franchisee acknowledges that the use of the Brand Fund or other advertising funds or accounts and the expenditures made thereby, may benefit Franchisor and its units, even though the units operated by Franchisor or its affiliates may or may not contribute to the Brand Fund or other advertising funds or accounts.
- 12.3.5 Franchisor will prepare on an annual basis, and will have available for Franchisee within one hundred twenty (120) days of the end of the fiscal year for the prior fiscal year only, a statement of contributions and expenditures for the Brand Fund. The statement will be presented to Franchisee upon Franchisee's written request. The Brand Fund is not required to be independently audited.

- 12.3.6 Franchisor has the right to require that an advertising cooperative and/or franchisee advisory council be formed, changed, dissolved or merged.
- 12.3.7 Franchisor assumes no obligation or liability to Franchisee with respect to the maintenance, direction or administration of the Brand Fund or any other advertising funds or accounts maintained in connection with this Agreement, except as expressly set forth in this Section. Franchisor also has the right to cause the Brand Fund to be incorporated or operated through a separate entity at such time as Franchisor deems appropriate, and such successor entity will have all of the rights and duties specified herein.
- 12.3.8 Franchisee acknowledges that the Brand Fund is not a trust or an asset of Franchisor and that Franchisor is not a fiduciary to Franchisee with respect to, or a trustee of, the Brand Fund or the monies therein.
- 12.3.9 Franchisor may suspend or terminate the Brand Fund at any time, and any surplus funds may only be used for marketing and advertising purposes until fully expended.
- 12.4 **Regional Advertising and Promotional Cooperative**. Franchisor may, in Franchisor's discretion, designate any geographical area for purposes of establishing a regional advertising and promotional cooperative ("Cooperative"), and to determine whether a Cooperative is applicable to the Franchised Business. If a Cooperative is established applicable to the Franchised Business, Franchisee must participate in the Cooperative. Cooperative contributions will be credited towards the Local Advertising Requirement discussed in Section 12.5 below. The following provisions will apply to each Cooperative:
- 12.4.1 Each Cooperative will be organized and governed in a form and manner, and will commence operation on a date, approved in advance by Franchisor;
- 12.4.2 Each Cooperative will be organized for the exclusive purpose of administering regional advertising programs and developing, subject to Franchisor's approval, standardized advertising materials for use by the members in local advertising;
- 12.4.3 No promotional or advertising plans or materials may be used by a Cooperative or furnished to its members without Franchisor's prior approval. All such plans and materials must be submitted to Franchisor in accordance with the procedure set forth in Section 12.1 hereof;
- 12.4.4 Each Cooperative will have the right to require its members to make contributions to the Cooperative in such amounts as determined by the Cooperative; however, the Cooperative may, by a two-thirds majority vote of its members, require a Cooperative contribution in excess of the Local Advertising Requirement;
- 12.4.5 Each member franchisee must submit to the Cooperative its respective contribution, together with such other statements or reports as Franchisor may require or as may be required by the Cooperative with Franchisor's approval;
- 12.4.6 Franchisor may grant to Franchisee, in its sole discretion, an exemption for any length of time from the requirement of membership in a Cooperative, upon written request from Franchisee stating reasons supporting such exemption. Franchisor's decision concerning such request for exemption will be final; and

12.4.7 Franchisor will have the power to require any Cooperative to be formed, changed, dissolved or merged at any time.

12.5 Local Advertising.

12.5.1 Grand Opening Marketing. In connection with the opening of the Franchised Business, Franchisee must spend a minimum amount of ten thousand dollars (\$10\\$15,000) in local advertising, and as otherwise required and directed by Franchisor ("Grand Opening Marketing"). Any marketing must be conducted in accordance with Franchisor's standards and specifications, as described in the Operations Manual or this Agreement. Franchisor reserves the right to collect all or a portion of the Grand Opening Marketing. Franchisor reserves the right to increase the required spend to \$1522,000 as Franchisor deems necessary.

12.5.2 Local Advertising Requirement. In addition to the Brand Fund Contributions described above in Section 12.3, Franchisee will be required to spend the greater of: (i) three percent (3%)3% of Gross Revenue per month; or (ii) \$3,000 per month on local advertising and promotion in accordance with Franchisor's standards and specifications (the Local Advertising Requirement, as defined in Section 3.6). Franchisor reserves the right to increase the Local Advertising Requirement to four percent (an amount up to 4%) of Gross Revenue, upon 90 days' prior written notice. Franchisee must spend the Local Advertising Requirement as Franchisor prescribes in the Operations Manual or otherwise in writing, which may include, without limitation, requirements for placing a certain number and/or type(s) of media advertisements, orpromotions, sponsorships, sales expenses, engaging certain public figures to assist the Franchisee in promoting its Franchised Business, or salaries/commissions for one or more sales people with approval from Franchisor which amounts may be limited to ensure allocation of funds to appropriate local marketing channels. The Local Advertising Requirement must be expended within Franchisee's Designated Territory. Franchisee acknowledges and agrees that Franchisee's Local Advertising Requirement must be expended regardless of the amount(s) spent by other System franchisees on local advertising. Franchisee may spend any additional sums Franchisee determines on local advertising. Franchisee must use only such advertising and promotional materials as have been previously approved by Franchisor. Franchisee must submit to Franchisor proof of Franchisee's expenditures on local advertising. All phone numbers used in local advertising by the Franchised Business of any form must be forwarded to Franchisor's Call Center.

12.5.3 Local Advertising Services. You must sign the Franchisor Directed Advertising Services Agreement and purchase all Local Advertising Services (defined in the Franchisor Directed Advertising Services Agreement attached hereto as Exhibit H) from us unless we provide written permission otherwise. For our Local Advertising Services, you agree to pay us or an affiliate we designate a non-refundable fee (the "Franchisor Directed Advertising Fee") in an amount designated in the Franchisor Directed Advertising Services Agreement, as updated in the Operations Manual periodically. The Franchisor Directed Advertising Fee is credited to your required monthly Local Advertising Requirement. In no event will the Franchisor Directed Advertising Fee be more than the Local Advertising Requirement.

12.6 Advisory Council. Franchisor reserves the right to establish an Advisory Council ("Advisory Council"). Elected franchisee(s) must participate actively in the Advisory Council as Franchisor designates, and participate in all Advisory Council meetings approved by Franchisor. Franchisor reserves the right to prepare and amend the governing documents for the Advisory Council from time to time, in its sole discretion, at any time. Franchisor, in its sole discretion, will determine the topic areas to be considered by the Advisory Council. The purposes of the Advisory Council shall include, but are not limited to: (i) exchanging ideas and problem-solving methods; (ii) advising Franchisor on expenditures for System-wide advertising; and (iii) coordinating franchisee efforts. Amounts and expenditures may vary from time to time due to variations in Advisory Council participation and costs, as determined by the Advisory

Council, and as approved by Franchisor. The Advisory Council shall act in an advisory capacity only. Franchisor will have the right to form, change or dissolve any Advisory Council at any time in its sole discretion.

13. INDEPENDENT BUSINESS; INDEMNIFICATION

- Independent Business Status. Franchisee is an independent business responsible for full control over the internal management and daily operation of the Franchised Business, and neither party to this Agreement is the agent, principal, partner, employee, employer or joint venture partner of the other party. Franchisee may not act or represent itself, directly or by implication, as Franchisor's agent, partner, employee or joint venture partner, and Franchisee may not incur any obligation on Franchisor's behalf or in Franchisor's name. All stationery, business cards and contractual agreements entered into by Franchisee shall contain Franchisee's corporate or fictitious name, and a conspicuously displayed notice, in the place Franchisor designates, that Franchisee operates the Franchised Business as an independently owned and operated business, and that Franchisee independently owns and operates the Franchised Business as a System franchisee. Nothing in this Agreement authorizes Franchisee to make any contract, agreement, warranty, or representation on Franchisor's behalf, or to incur any debt or other obligation in Franchisor's name, and Franchisor shall in no event assume liability for, or be deemed liable hereunder as a result of, any such action, nor shall Franchisor be liable by reason of any of Franchisee's acts or omissions in the operation of the Franchised Business or for any claim or judgment arising therefrom against Franchisee or Franchisor. Neither this Agreement nor Franchisor's course of conduct is intended, nor may anything in this Agreement (nor Franchisor's course of conduct) be construed to state or imply that Franchisor is the employer of Franchisee or the Franchisee's employees and/or independent contractors. It is understood and agreed by the parties hereto that this Agreement does not establish any fiduciary relationship between them.
- 13.2 **Indemnification**. Franchisee and Franchisee's principals agree to indemnify, defend and hold Franchisor, Franchisor's affiliates and their respective shareholders, directors, officers, employees, agents, successors and assignees ("Indemnitees") harmless against, and to reimburse them for the actual costs of all claims, obligations, liabilities, fines, losses, damages, costs and expenses (including attorneys' fees, except to the extent such fees are determined by an arbitrator or court of competent jurisdiction to be unreasonable, in which case the fees shall be adjusted to be reasonable by such arbitrator or court) ("Claims"), including any and all taxes, directly or indirectly arising out of, in whole or in part: (i) the operation of the Franchised Business, including the use, condition, or construction, equipping, maintenance or operation of the Franchised Business and; (ii) Franchisee's promotion and advertising; (iiiii) the unauthorized use of the Proprietary Marks and other Proprietary Material; (iii) the transfer of any interest in this Agreement or the Franchised Business or assets in any manner not in accordance with this Agreement; (ivv) the infringement, alleged infringement, or any other violation or alleged violation by Franchisee or any of Franchisee's principals of any patent, trademark or copyright, or other proprietary right owned or controlled by third parties; or (v (subject to Franchisors obligations in 4.1.8, if any); or (vi) libel, slander, or any other form of defamation of Franchisor, the System, or any franchisee or developer operating under the System, by Franchisee or by any of Franchisee's principals, except for Claims that are determined by an arbitrator or court of competent jurisdiction to have been caused solely and directly by the indemnified party's gross negligence, willful misconduct, strict liability, or fraud. For purposes of this indemnification, "Claims" shall mean and include all obligations, actual, consequential, punitive, and other damages and costs reasonably incurred in the defense of any action, including attorneys', attorney assistants' and expert witness fees, costs of investigation and proof of facts, court costs, other litigation expenses, and travel and living expenses, whether or not such claims exceed the amount of insurance coverage available through Franchisee to Franchisor. Franchisor shall have the right to defend any such claim against it in such manner as Franchisor deems appropriate or desirable, in Franchisor's sole discretion. Such an undertaking by Franchisor shall, in no manner or form, diminish Franchisee's, and each of Franchisee's principals', obligations to indemnify the Indemnities and to hold them harmless. This

indemnity shall continue in full force and effect subsequent to and notwithstanding the expiration or termination of this Agreement.

14. SALE OR TRANSFER

14.1 **Transfer**. Franchisee's rights under this Agreement are personal, and Franchisee shall not sell, transfer, assign or encumber Franchisee's interest in the Franchised Business without Franchisor's prior written consent. Any sale, transfer, assignment or encumbrance made without Franchisor's prior written consent shall be voidable at Franchisor's option and shall subject this Agreement to termination as specified herein. For purposes of this Agreement, "Transfer" as a verb means to sell, assign, give away, transfer, pledge, mortgage or encumber, voluntarily, involuntarily, or by operation of law (such as through divorce or bankruptcy proceedings), any interest in this Agreement, the Franchised Business, substantially all the assets of the Franchised Business, or in the ownership of the Franchised Business (if you are an entity). "Transfer" as a noun means any sale, assignment, gift, transfer, pledge, mortgage or encumbrance. Franchiser will not unreasonably withhold, condition, or delay a request to transfer by Franchisee. Franchisee acknowledges and agrees that the conditions set forth for transfer in this Section 14 are reasonable.

14.2 **Death or Disability**.

14.2.1 Representative's Right to Continue as Franchisee. In the event of Franchisee's death, disability or incapacitation (or the death, disability or incapacitation of Franchisee's partners or personal guarantors), Franchisee's legal representative, or Franchisee's partner's or guarantor's respective legal representative, as applicable, will have the right to continue the operation of the Franchised Business as franchisee under this Agreement if: (i) within one hundred eighty (180) days from the date of death, disability or incapacity (the "180-Day Period"), such person: (a) meets Franchisor's then-current standards to become a franchisee, as described in Section 14.3.2.6; and (b) has obtained Franchisor's prior written approval and has executed Franchisor's then-current franchise agreement for the unexpired term of the franchise, or has furnished a personal guaranty of any partnership, corporatecorporation or limited liability company guarantying franchisee's obligations to Franchisor and Franchisor's affiliates; and (ii) such person successfully completes Franchisor's training program (which Franchisor will provide at Franchisor's thencurrent tuition rate). Such assignment by operation of law will not be deemed in violation of this Agreement, provided such heirs or legatees accept the conditions imposed by the Franchise Agreement and are acceptable to Franchisor. However, in the case of a transfer by demise or inheritance, if the heirs or beneficiaries of any deceased person are unable to meet the conditions contained in this Agreement, the personal representative of the deceased Franchisee will have a reasonable time, in Franchisor's sole discretion, which shall not exceed one hundred eighty (180) days from the date or transfer by demise or inheritance, to dispose of the deceased's interest in the Franchised Business, and such disposition will be subject to all the terms and conditions for transfer contained in this Agreement. If the interest is not disposed of in the manner and time frame provided in the immediately preceding sentence, then Franchisor may terminate this Agreement.

14.2.2 Franchised Business Operation During and After 180-Day Period. Franchisor is under no obligation to operate the Franchised Business or incur any obligation on behalf of any incapacitated franchisee during or after the 180-Day Period. If necessary, Franchisee (or Franchisee's legal representative, as applicable) shall appoint a previously approved acting interim manager to operate the Franchised Business during the 180-Day Period. In the event of Franchisee's death, disability, absence or otherwise, Franchisor may (but is not required to) operate Franchisee's Franchised Business on Franchisee's behalf and at Franchisee's expense for such period of time (and under such terms and conditions) as Franchisor determines, including paying out the assets and/or revenues of the Franchised Business (including

any amounts owed to Franchisor and/or any affiliate) in such priorities as Franchisor determines from time-to-time in Franchisor's sole and absolute discretion. Franchisor may pay itself its then-current fee to reimburse Franchisor for Franchisor's management services and other costs. Franchisor may obtain approval of a court or arbitrator for any such arrangements, the attorneys' fees and other costs incurred in connection with obtaining such approval to be charged against the assets and/or revenues of the Franchised Business. Franchisee (and/or Franchisee's estate) will indemnify Franchisor against any costs and/or liabilities incurred by it in connection with, or related in any way to, the operation (or otherwise) of the Franchised Business.

- 14.3 **Ownership Changes**. A sale, transfer or assignment requiring Franchisor's prior written consent shall be deemed to occur: (i) if Franchisee is a corporation, upon any assignment, sale, pledge or transfer of any fractional portion of Franchisee's voting stock or any increase in the number of outstanding shares of Franchisee's voting stock which results in a change of ownership; (ii) if Franchisee is a partnership, upon the assignment, sale, pledge or transfer of any fractional partnership ownership interest; or (iii) if Franchisee is a limited liability company, upon the assignment, sale, pledge or transfer of any interest in the limited liability company. Any new partner, shareholder, or member or manager will be required to personally guarantee Franchisee's obligations under this Agreement. A transfer pursuant to (i) through (iii) above shall not be subject to Franchisor's right of first refusal as set forth in Section 14.3.1.
- 14.3.1 Right of First Refusal. If Franchisee proposes to transfer either this Agreement or all, or substantially all, of the assets used in connection with the Franchised Business or any interest in Franchisee's lease to any third party (other than a corporation or limited liability company as set forth herein), Franchisee shall first offer to sell such interest to Franchisor on the same terms and conditions as offered by such third party. Franchisee shall obtain from the third party and provide to Franchisor a statement in writing, signed by the third party and Franchisee, of the terms of the offer ("Letter of Intent"). If Franchisor elects not to accept the offer within a thirty (30) day period, Franchisee shall have a period not to exceed sixty (60)120 days to complete the transfer described in the Letter of Intent, subject to the conditions for approval set forth herein. Franchisee shall effect no other sale or transfer as contemplated under the Letter of Intent without first complying with this Section. Any materialsignificant change in the terms of the offer shall be deemed a new proposal, subject to Franchisor's right of first refusal. So long as Franchisee has obtained Franchisor's prior written consent, which shall not be unreasonably withheld, a transfer to an existing partner, shareholder or member, or a transfer as a result of the death, disability or incapacitation of a shareholder or partner in accordance with the provisions set forth below, is not subject to Franchisor's first right of refusal. In the event the consideration, terms and/or conditions offered by a third party are such that Franchisor may not reasonably be required to furnish the same consideration, terms and/or conditions, then Franchisor may purchase the interest proposed to be sold for the reasonable equivalent in cash. If the parties cannot agree within thirty (30) days on the reasonable equivalent in cash of the consideration, terms and/or conditions offered by the third party, an independent appraiser shall be designated by Franchisor at Franchisor's expense, and the appraiser's determination shall be binding.
- 14.3.2 Conditions for Approval. Franchisee shall notify Franchisor in writing of any proposed transfer of this Agreement, any direct or indirect interest in Franchisee, or in all or substantially all of the assets of the Franchised Business, at least thirty (30) days before such transfer is proposed to take place. Franchisor may condition Franchisor's approval of any proposed sale or transfer of the Franchised Business or of Franchisee's interest in this Agreement upon satisfaction of the following occurrences:
- 14.3.2.1 All of Franchisee's accrued monetary obligations to Franchisor, Franchisor's affiliates, and Franchisor's designated/approved suppliers and vendors must be satisfied;
- 14.3.2.2 Franchisee must cure all existing defaults under this Agreement or any other agreement between Franchisee and Franchisor, Franchisor's affiliates, and Franchisor's

designated/approved suppliers and vendors within the period permitted for cure, and have substantially complied with such agreements during their respective terms;

- 14.3.2.3 Franchisee and Franchisee's principals (if Franchisee is a partnership, corporation or limited liability company), and the transferee (if it has had any previous relationship with Franchisor or Franchisor's affiliates), must execute a general release under seal, in a form satisfactory to Franchisor, of any and all claims against Franchisor and Franchisor's affiliates and officers, directors, shareholders and employees, in their corporate and individual capacities; provided, however, the general release shall not be inconsistent with any applicable state statute regulating franchising;
- 14.3.2.4 Transferor and transferee must execute a mutual general release relieving all claims against each other, excluding only such claims relating to any provision or covenant of this Agreement which imposes obligations beyond the expiration of this Agreement;
- 14.3.2.5 Franchisee or transferee must provide Franchisor a copy of the executed purchase agreement relating to the proposed transfer with all supporting documents and schedules, including transferee's assumption of an agreement to faithfully perform all of Franchisee's obligations under this Agreement;
- 14.3.2.6 Transferee must demonstrate to Franchisor's satisfaction that he or she: (i) meets Franchisor's educational, managerial and business standards; (ii) possesses a good moral character, business reputation and credit rating; (iii) has the aptitude and ability to conduct the business to be transferred; and (iv) has adequate financial resources and capital to meet the performance obligations under this Agreement; however, transferee shall not be in the same business as Franchisor either as licensor, franchisor, independent operator or licensee of any other business or chain which is similar in nature or in competition with Franchisor, except that the transferee may be an existing franchisee of ours;
- 14.3.2.7 At Franchisor's option, transferee (and, if the transferee is other than an individual, such owners of a beneficial interest in the transferee as Franchisor may request) either: (i) enters into a written assignment, in a form satisfactory to Franchisor, assuming and agreeing to discharge all of Franchisee's obligations under this Agreement; or (ii) executes execute Franchisor's then-current franchise agreement, and will receive a full then-current initial term for a franchisee, which term will commence on the date the transferee executes the then-current Franchise Agreement;
- 14.3.2.8 Franchisee must pay Franchisor a transfer fee equal to ten thousand dollars (\$(in lieu of an Initial Franchise Fee for a new franchisee) equal to \$10,000) per Designated Territory that is being transferred to transferee. A non-refundable deposit of \$3,000 of the Transfer Fee is due with Franchisee's written notice of proposed transfer and the balance of \$7,000 is due upon Franchisor's approval at closing. In the event Franchisee transfers multiple Designated Territories at once, Franchisor reserves the right, but not the obligation, in Franchisor's sole discretion, to reduce the transfer fee for any of the Designated Territories being transferred by any amount;
- 14.3.2.9 Transferee shall satisfactorily complete Franchisor's training program at the transferee's expense within the time frame Franchisor sets forth;
- 14.3.2.10 Franchisee (and Franchisee's principals if Franchisee is a partnership, corporation or limited liability company) and the members of their respective families must comply with the post-termination provisions of this Agreement;
- 14.3.2.11 Transferee must obtain, within the time limits set by Franchisor, and maintain thereafter, all permits and licenses required for the operation of the Franchised Business;

- 14.3.2.12 To the extent required by the terms of any leases or other agreements, the lessors or other parties must consent to the proposed transfer;
- 14.3.2.13 The transfer must be made in compliance with any laws that apply to the transfer, including state and federal laws governing the offer and sale of franchises;
- 14.3.2.14 Franchisee must ensure that all projects in progress at the time of the transfer will be continued without interruption, and transferee must promptly notify all current customers of the change in ownership;
- 14.3.2.15 The purchase price and terms of the proposed transfer must not be so burdensome to the prospective transferee as to impair or materially—threaten its future operation of the Franchised Business and performance under its franchise agreement;
- 14.3.2.16 Franchisee must request that Franchisor provide the prospective transferee with Franchisor's current form of Franchise Disclosure Document and Franchisor will not be liable for any representations not included in the Franchise Disclosure Document by Franchisee or any of it's agents;
- 14.3.2.17 Franchisor's approval of the transfer will not constitute a waiver of any claims Franchisor may have against the transferring party; and
- 14.3.2.18 Franchisor will have the right to disclose to any prospective transferee such revenue reports and other financial information concerning Franchisee and Franchisee's Franchised Business as Franchisee has supplied Franchisor hereunder—; and
- 14.3.2.19 Franchisor may condition the approval of a transfer on resizing the territory(ies) into a territory or multiple territories consistent with the then current size of territory being sold, which may result in more territories with separate franchise agreements and the requirement for a multi-unit addendum.
- 14.4 **Transfer to a Corporation or Limited Liability Company**. If Franchisee is an individual and desires to assign its rights under this Agreement to a corporation or limited liability company, and if all of the following conditions are met, Franchisor will consent to the transfer without assessing the transfer fees set forth above, and such assignment will not be subject to Franchisor's right of first refusal in Section 14.3.1:
- 14.4.1 The corporation or limited liability company is newly organized and its activities are confined to operating the Franchised Business;
- 14.4.2 Franchisee is, and at all times remains, the owner of at least fifty-one percent (51%) of the outstanding shares of the corporation or a controlling interest in the limited liability company;
- 14.4.3 The corporation or limited liability company agrees in writing to assume all of Franchisee's obligations hereunder;
- 14.4.4 All stockholders of the corporation, or members and managers of the limited liability company, as applicable, must execute Franchisor's prescribed form of personal guaranty;

- 14.4.5 At Franchisor's request, Franchisee will furnish true and correct copies of all documents and contracts governing the rights, obligations and powers of Franchisee's owners and agents (such as articles of incorporation or organization and partnership, operating or shareholder agreements and similar documents);
- 14.4.6 Franchisee and Franchisee's principals (if Franchisee is a partnership, corporation or limited liability company), and transferee (if it has had any previous relationship with Franchisor or Franchisor's affiliates), must execute a general release under seal, in a form satisfactory to Franchisor, of any and all claims against Franchisor and Franchisor's affiliates and officers, directors, shareholders and employees, in their corporate and individual capacities; provided, however, the general release shall not be inconsistent with any applicable state statute regulating franchising; and
- 14.4.7 Non-Traditional Ownership Structures. Franchisor may, but is not obligated to, approve a Transfer to or allow ownership by an ESOP, trust or private equity firm ("Non-traditional Ownership") and any request to Transfer to an ESOP, trust or private equity firm may be denied in our sole discretion. However, if we deFranchisor does approve a Transfer by an ESOP, trust or private equity firm, we may impose additional requirements that include, but are not limited to, the following: (a) the ownership and related requirements set forth in Section 14.4.7.1; (b) additional financial covenants; (c) additional insurance requirements; (d) an enhanced right to purchase the Franchised Business, and (e) such other conditions that Franchisor deems appropriate in Franchisor's Business Judgment.
- 14.4.7.1 Operational Control-Non-Traditional Entities. ESOP or other trust ownership of the Franchised Business is permitted only with our prior written permission, which may be granted or withheld in our sole discretion. At all times, the guarantors directly will: (i) control all aspects of Franchisee and the operation of the Franchised Business; and (ii) serve as trustee of the trust and retain sole control over the voting of the trust's equity interest in the Franchisee. The Trustee must at all times be an individual and must meet the then-current requirements of franchise ownership. Any change in trustee is deemed a Transfer and is subject to approval as set forth in this Agreement. Any change in beneficiaries is deemed a Transfer. Any such Transfer is subject to approval conditions and other rights we haveFranchisor has as set forth in this Agreement. You acknowledge and agree that: (i) if the guarantors do not maintain operational control respecting each of: (a) the Franchised Business; (b) the Franchisee; and (c) the trust, such an event will constitute a Transfer as described in Article 14 of this Agreement and Franchisee must comply with all applicable provisions of such Article; and (ii) if the guarantors desire to turn over operational control respecting the Franchisee, the trust, or the Franchised Business to one or more trust beneficiaries, such beneficiaries must satisfy all conditions to consent as described in Article 20 of this Agreement. The trust must be and remain a revocable trust. If by operation of law or otherwise the trust converts to an irrevocable trust, the trust must divest ownership of the Franchised Business to an approved person or entity pursuant to the terms of Article 14 of this Agreement. The trust must contain a provision equivalent to and granting the trustee the following powers with regard to trust assets: "The trustee may exercise all rights of an absolute owner with respect to stocks, other securities, and closely held business interests. The trustee is hereby authorized to sell, convey, pledge, mortgage, lease or transfer title to any interest in the trust property for a period within or extending beyond the duration of the trust."
- 14.5 **Franchisor's Right to Transfer**. Franchisor has the right to sell, transfer, assign, and/or encumber all or any part of Franchisor's assets and Franchisor's interest in, and rights and obligations under, this Agreement, in Franchisor's sole discretion without consent from or prior notice to Franchisee.

15. BREACH AND TERMINATION

15.1 **Automatic Termination**. This Agreement shall automatically terminate without notice or an opportunity to cure upon the occurrence of any of the following:

- 15.1.1 *Voluntary Bankruptcy*. If Franchisee makes an assignment for the benefit of creditors, files a voluntary petition in bankruptcy, is adjudicated bankrupt or insolvent, files or acquiesces in the filing of a petition seeking reorganization or arrangement under any federal or state bankruptcy or insolvency law, or consents to or acquiesces in the appointment of a trustee or receiver for Franchisee or the Franchised Business.
- 15.1.2 *Involuntary Bankruptcy*. If proceedings are commenced to have Franchisee adjudicated bankrupt or to seek Franchisee's reorganization under any state or federal bankruptcy or insolvency law, and such proceedings are not dismissed within sixty (60) days, or a trustee or receiver is appointed for Franchisee or the Franchised Business without Franchisee's consent, and the appointment is not vacated within 60 days.
- 15.1.3 Loss of Premises. If Franchisee loses the right to occupy the Franchised Business premises or operate the Franchised Business from the Approved Location, and such location is not replaced with an approved location within 180 days after loss of the right to occupy the original site.
- 15.2 **With Notice and Without Opportunity to Cure**. Franchisor has the right to terminate this Agreement upon notice without providing Franchisee an opportunity to cure for any of the following breaches or defaults:
- 15.2.1 *Criminal Acts.* If Franchisee or Franchisee's principals are convicted of or plead guilty or no contest to a felony (whether or not related to the Franchised Business) or other offense related to the operation of the Franchised Business or that Franchisor believes, in its sole discretion, is likely to have an adverse effect on the Proprietary Marks or the goodwill associated therewith.
- 15.2.2 Fraud. If Franchisee or Franchisee's principals commit any fraud or misrepresentation in the operation of the Franchised Business, including, but not limited to, any misrepresentation made in Franchisee's franchise application.
- 15.2.3 Other Actions. If Franchisee or Franchisee's principals, including any shareholder, member, guarantor or agent, engages in activity or conduct whichthat materially impairs thatthe goodwill associated with the System or the Proprietary Marks and fails to cease and correct such activities or conduct within twenty-four (24) hours of Franchisee's receipt of written notice of a breach under this Section. Franchisee will have only one opportunity to cure under this section. Any subsequent violation after the first violation is subject to immediate termination without an opportunity to cure.
- 15.2.4 *Misrepresentation*. If Franchisee or Franchisee's principals make any misrepresentation or omission in connection with Franchisee's franchise application, including, but not limited to, any financial misrepresentation.
- 15.2.5 Failure to Complete Training. If Franchisee (and/or Franchisee's Designated Manager, if applicable) fails to complete the Initial Training Program as provided in Section 8.
- 15.2.6 Repeated Breaches. If Franchisor sends Franchisee two (2) or more written notices to cureof default pursuant to Sections 15.3 or 15.4 hereof in any twelve (12) month period, regardless of whether the defaults set forth in the notices were subsequently cured.
- 15.2.7 Breach of Other Agreements. If Franchisee or Franchisee's principals materially breach any other agreement with Franchisor or any of Franchisor's affiliates, or threaten any material breach of any such agreement, or any lease for the Approved Location, and fails to cure such breach within any

permitted period for cure. This provision does not apply to the breach of agreements related to Franchise Agreements with Franchisor's affiliated Franchisors of a different brand than the franchise to be operated under this Franchise Agreement.

- 15.2.8 *Misuse of the Proprietary Marks or Confidential Information.* If Franchisee or Franchisee's principals violate any provision hereof pertaining to Proprietary Marks or Confidential Information or misuse the Proprietary Marks or Confidential Information.
- 15.2.9 Disclosure of Confidential Information or Trade Secrets. If Franchisee or Franchisee's principals disclose or divulge the contents of the Operations Manual or any other Confidential Information or Trade Secret provided to Franchisee by Franchisor or any of its affiliates to any third party.
- 15.2.10 *Violation of Law.* If Franchisee violates any law, ordinance or regulation, or operates the Franchised Business in a manner that presents a health or safety hazard to customers or the general public that is not cured within 15 days. A cure requires that Franchisee put in place an action plan to prevent the occurrence of any similar violation in the future. The cure period does not apply to any repeated occurrence of the same or similar breach in the future.
- 15.2.11 *Violation of In-term Restrictive Covenant.* If Franchisee violates the in-term restrictive covenant contained in Section 17.1, or any of the other restrictive covenants set forth in this Agreement.
- 15.2.12 *Liens*. If a levy of writ of attachment or execution or any other lien is placed against Franchisee or any of Franchisee's principals or any of their assets which is not released or bonded against within thirty (30) days.
- 15.2.13 Insolvency. <u>IfExcept for adjudicated insolvency which is subject to automatic termination pursuant to 15.1.1, if Franchisee or any of Franchisee's principals become insolvent or admits insolvency other than by adjudication.</u>
- 15.2.14 Abandonment. If Franchisee voluntarily or otherwise abandons the Franchised Business. The term "abandon" includes any objective conduct which indicates a desire or intent to discontinue the Franchised Business in accordance with the terms of this Agreement for seven (for 7) days or more without Franchisor's prior written consent.
- 15.2.15 *Unauthorized Transfer*. If Franchisee purports to sell, transfer or otherwise dispose of any interest in the Franchised Business in violation of Section 14 hereof(including the sale of assets of the Franchised Business) in violation of Section 14 hereof and does not void such transfer, or otherwise come into compliance with its transfer obligations set forth in Section 14 related to such transfer within 30 calendar days of notice from Franchisor.
- 15.2.16 Unauthorized Products or Services. If Franchisee offers any unauthorized and unapproved products or services at or from the Franchised Business- and fails to immediately cease offering the unauthorized Products or Services, unless and until such Products or Services are approved through the alternative product or supplier approval process.
- 15.2.17 *Unapproved Purchases*. If Franchisee orders or purchases supplies, signs, furnishings, fixtures, equipment or inventory from an unapproved supplier or which Franchisor has not approved and fails to immediately cease the unapproved purchases, unless and until such purchases are approved through the alternative product or supplier approval process.

- 15.2.18 *Proprietary Software*. If Franchisee misuses or makes unauthorized use of any Proprietary Software Franchisor may develop for use in connection with the System and does not immediately cease use of such unauthorized use of such software.
- 15.2.19 *Insurance*. If, within 48 hours of notice from Franchisor, Franchisee fails to maintain insurance or to repay Franchisor for insurance paid for by it, or otherwise fails to adhere to the requirements of Section 9.
- 15.2.20 *Government Regulations*. If Franchisee fails, within fifteen (15) calendar days after notification of non-compliance by federal, state or local government authorities, to comply with any law or regulation applicable to the Franchised Business.
- 15.2.21 *Government Actions*. If any government action is taken against Franchisee that results in any obligation upon Franchisor which, in Franchisor's sole judgment, is uneconomical, not in the best interests of Franchisor, or would result in Franchisor having an unintended relationship or obligation.
- 15.2.22 Anti-Terrorist Activities. If Franchisee fails to comply with the provisions of Section 22.8.
- 15.2.23 Personal Use of Franchised Business Property. If Franchisee takes for Franchisee's own personal use any assets or property of the Franchised Business, including employee taxes, FICA, insurance or benefits. Notwithstanding the foregoing, the incidental and innocuous use of Franchised Business assets is not considered a default except to the extent such use causes or results in a claim or liability for Franchisee or Franchisor, or in Franchisor's sole determination may damage to the good will of the system.
- 15.2.24 *Insufficient Funds*. If there are insufficient funds in Franchisee's bank account to cover a check or EFT payment to Franchisor two (2) or more times within any twelve (12)—month period.
- 15.2.25 *Failure to Open.* If Franchisee fails to commence operations of the Franchised Business within the time prescribed in Section 7.3 of this Agreement.
- 15.2.26 Operating Outside of Designated Territory. If Franchisee operates the Franchised Business outside of the Designated Territory without Franchisor's prior written consent, as provided in Section 1.2 of this Agreement.
- 15.2.27 *Violation of Best Efforts.* If Franchisee or Franchisee's principals violate any of the provisions in Section 7.10 of this Agreement.
- 15.2.28 Loss of Immigration Status Necessary to live and work in the United States. If Franchisee or its owners fail to maintain an immigration status that allows Franchisee and/or such owners to live and work in the United States
- 15.3. **Upon 15 Days' Notice to Cure**. Franchisor has the right to terminate this Agreement if any of the following defaults remain uncured after Franchisor provides Franchisee with notice of such default(s) and fifteen (15) days to cure:
- 15.3.1 *Nonpayment*. If Franchisee fails to pay Franchisor as and when due any sums owed to Franchisor, any of Franchisor's affiliates, or any of Franchisor's System Suppliers, Approved Suppliers, or vendors.

- 15.3.2 *Endorsement of Checks.* If Franchisee fails to immediately endorse and deliver to Franchisor any payments due to Franchisor from any third party that is erroneously made to Franchisee.
- 15.3.3 Failure to Maintain Sufficient Marketing Materials and Supplies. If Franchisee fails to maintain sufficient inventory of marketing materials and other supplies necessary to adequately develop the Designated Territory and meet consumer demand.
- 15.3.4 *Interruption of Service*. If Franchisee fails to maintain the prescribed months, days or hours of operation at the Franchised Business- (this provision does not modify the right to termination for abandonment set forth in 15.2.14 without an opportunity to cure).
- 15.3.5 Failure to Personally Supervise Franchised Business Operations or Employ Adequate Personnel. If Franchisee or the Designated Manager fails, in Franchisor's sole discretion, to personally supervise day-to-day operation of the Franchised Business or fails to employ a sufficient number of qualified, competent personnel as Franchisor requires from time to time to operate the business in compliance with brand standards.
- 15.3.6 *Quality Control*. If Franchisee fails to maintain the strict quality controls reasonably required by this Agreement and/or the Operations Manual.
- 15.3.7 *Licenses and Permits.* If Franchisee fails to procure or maintain any licenses, certifications or permits necessary for the operation of the Franchised Business. The cure period for a breach of 15.3.7 will be extended for the amount of time necessary to obtain a required license so long as franchisee is diligently pursuing licensure and is using the license of another person to legally operate the business until Franchisee obtains licensure.
- 15.4 **Upon 30 Days' Notice to Cure**. Franchisor has the right to terminate this Agreement if Franchisee fails to perform or comply with any other term or condition of this Agreement or any ancillary agreements between Franchisee and Franchisor or Franchisor's affiliates, and Franchisee fails to cure such default(s) within thirty (30) days after being provided with notice thereof30 days after being provided with notice thereof. The cure period will be extended for a period of up to 30 days if: a) the breach cannot reasonably be cured within the original cure period, or b) Franchisee is diligently pursuing a cure of the breach but is unable to complete the cure due to circumstances outside of Franchisee's control.
- 15.5 Step In Rights. In addition to Franchisor's right to terminate this Agreement, and not in lieu of such right or any other rights Franchisor may have against Franchisee, upon a failure to cure any default within the applicable time period (if any), Franchisor has the right, but not the obligation, to enter upon the Franchised Business premises and exercise complete authority with respect to the operation of the Franchised Business until such time as Franchisor determines, in Franchisor's sole discretion, that the default has been cured and Franchisee is otherwise in compliance with this Agreement. In the event Franchisor exercises the rights described in this Section, Franchisee must reimburse Franchisor for all costs and overhead, if any, incurred in connection with its operation of the Franchised Business including, without limitation, costs of personnel for supervising and staffing the Franchised Business and their travel and lodging accommodations. If Franchisor undertakes to operate the Franchised Business pursuant to this Section, Franchisee agrees to indemnify and hold Franchisor (and Franchisor's representative(s) and employees) harmless from and against any fines, claims, suits or proceedings which may arise out of Franchisor's operation of the Franchised Business. Franchisor reserves the right to charge Franchisee up to ten percent (10%) of Franchisee's Gross Revenue generated during the period of time that Franchisor provides temporary management of the Franchised Business.

15.6 Nonwaiver. 5 Nonwaiver. Franchisor's delay in exercising or failing to exercise any right or remedy under this Agreement or Franchisor's acceptance of any late or partial payment due hereunder shall not constitute a waiver of any of Franchisor's rights or remedies against Franchisee.

16. RIGHTS AND DUTIES UPON TERMINATION OR EXPIRATION

- 16.1 **Franchisee's Obligations**. Upon termination of this Agreement, or the expiration or termination of any Holdover Period, regardless of the cause, and upon expiration and nonrenewal or transfer of this Agreement, Franchisee must, at Franchisee's cost and expense:
 - 16.1.1 Immediately cease all operations under this Agreement;
- 16.1.2 Immediately pay Franchisor all unpaid fees, and pay Franchisor, Franchisor's affiliates, Franchisor's major suppliers and vendors, all other monies owed;
 - 16.1.3 Immediately discontinue the use of the Proprietary Marks;
- 16.1.4 Immediately cease using the Business Management and Technology System and the Operations Manual, and return all Proprietary Materials and Confidential Information, including, without limitation, all customer lists and data, within ten (10) calendar days and immediately and permanently cease use of such information and materials;
- as any permitted domain names and/or social media pages used in connection with the operation of the Franchised Business (collectively, the "Assigned Property"), and direct the telephone company and/or domain name registrar to transfer all such Assigned Property to Franchisor or Franchisor's designee pursuant to the Conditional Assignment of Telephone Numbers and Domain Name in the form attached as Exhibit B to this Agreement, and transfer all usernames and passwords for all social media pages to Franchisor;
- 16.1.6 Immediately vacate the Franchised Business premises, and if Franchisor exercised Franchisor's rights pursuant to Franchisor's prescribed form of Collateral Assignment of Lease, arrange for transfer of the lease to Franchisor within fifteen (15) calendar days of termination or expiration of this Agreement;
- 16.1.7 Immediately surrender all stationery, printed matter, signs, advertising materials, supplies and other items containing the Proprietary Marks, as Franchisor directs, and all items which are a part of the trade dress of the System no later than $\frac{1}{100}$ calendar days after the termination or expiration of this Agreement;
 - 16.1.8 Immediately cease holding itself out as Franchisor's Franchisee;
 - 16.1.9 Immediately cease to communicate with all customers of the Franchised Business;
- 16.1.10 Take such action as shall be necessary to amend or cancel any assumed name, business name or equivalent registration which contains any trade name or other Proprietary Mark Franchisor licensed to Franchisee, and furnish Franchisor evidence satisfactory to Franchisor of compliance with this obligation within fifteen (15) calendar days after the termination, expiration or transfer of this Agreement;

- 16.1.11 Permit Franchisor to make a final inspection of Franchisee's financial records, books, and other accounting records within one (1) month 2 years of the effective date of termination, expiration, or transfer;
- 16.1.12 Comply with the post-termination covenants set forth in Section 17 hereof, all of which shall survive the transfer, termination or expiration of this Agreement;
- 16.1.13 Cease to use in advertising or in any other manner, any methods, procedures or techniques associated with Franchisor or the System;
- 16.1.14 Immediately remove Franchisor's Proprietary Marks from vehicles used in connection with the Franchised Business, and otherwise de-identify the vehicles from being associated with Franchisor or the System;
- 16.1.15 Execute from time to time any necessary papers, documents, and assurances to effectuate the intent of this Section 16;
- 16.1.16 As applicable, provide Franchisor with access and any account information regarding any social media accounts used in connection with the Franchised Business; and
- 16.1.17 Reimburse Franchisor in connection with any costs Franchisor incurs in connection with enforcing Franchisee's obligations under this Section 16.
- Option to Purchase Personal Property. Upon the termination or expiration of this Agreement, or the expiration or termination of any Holdover Period, Franchisor, or Franchisor's designee shall have the option, but not the obligation, to purchase any personal property used in connection with operation of the Franchised Business by providing Franchisee written notice of Franchisor's election within sixty (60) calendar days after such termination or expiration and paying Franchisee the bookfair market value for such personal property within sixty (60) calendar days of such notice. For purposes of this paragraph, "book The fair market value" means the amount will be determined by an independent appraiser or broker selected by Franchisee actually paid for and Franchisor and subject to the limitations provided herein. The independent appraiser/broker must have 7 years substantive experience valuing the personal property, less depreciation (calculated by using same or similar assets. If the straight line depreciation methodparties cannot agree on a ten (10) year depreciation schedule irrespective of an appraiser/broker, each party will choose one and the depreciation method or seheduletwo chosen appraisers/brokers will choose a third person and the fair market value will be the average of all three valuations. All sales, transfer and/or similar taxes are to be paid by Franchisee uses for accounting purposes).. Notwithstanding the foregoing, to the extent that Franchisor exercises Franchisor's right to purchase any personal property that is subject to a lease or finance agreement, the purchase price of such personal property shall equal the amount of Franchisee's remaining obligations under the lease or finance agreement, as applicable, or the fair market value less the remaining obligations on the finance agreement, as applicable, which Franchisor would assume (such assumption of obligations constituting the full purchase price of such personal property), or pay off. Franchisor shall be entitled to offset the purchase price by the amount of money owed by Franchisee to Franchisor for any payments necessary to acquire clear title to property or for any other debt. If Franchisor exercises Franchisor's option to purchase, pending the closing of such purchase, Franchisor has the right to appoint a manager to maintain operation of the Franchised Business, or Franchisor may require that Franchisee close the Franchised Business during such period without removing any assets. Franchisee is required to maintain in force all insurance policies required under this Agreement until the date of such closing. Franchisor has the unrestricted right to assign this option to purchase the Franchised Business. Franchisor will be entitled to all customary warranties and representations in connection with Franchisor's purchase of Franchisee's property, including, without limitation,

representations and warranties as to ownership and condition of and title to the property; liens and encumbrances on the property; validity of contracts and agreements; and liabilities affecting the property, contingent or otherwise.

- 16.3 **Exclusions**. Franchisor may exclude from the personal property purchased under Section 16.2 cash or its equivalent and any equipment, signs, inventory, materials and supplies that are not reasonably necessary (in function or quality) to the Franchised Business's operation or that Franchisor has not approved as meeting standards for the Franchised Business.
- 16.4 **Damages, Costs, and Expenses**. In the event of termination for any default by Franchisee, Franchisee shall promptly pay to Franchisor all damages, costs and expenses, including reasonable attorneys' fees, incurred by Franchisor as a result of the default, which obligation shall give rise to and remain, until paid in full, a lien in favor of Franchisor against any and all of Franchisee's personal property, furnishings, equipment, signs, fixtures and inventory related to the operation of the Franchised Business.

17. COVENANTS

Franchisee acknowledges that as a participant in Franchisor's System, Franchisee will receive proprietary and Confidential Information and materials, Trade Secrets, and the unique methods, procedures and techniques which Franchisor has developed. Therefore, to protect Franchisor, the Franchisor's System, and Franchiser's other franchisees, Franchisee agrees as follows:

- 17.1 **During the Term of this Agreement**. During the term of this Agreement, and during any Holdover Period, neither Franchisee, Franchisee's owners, officers, directors, principals, nor any member of the immediate family of Franchisee or Franchisee's owners, officers, directors, or principals will, directly or indirectly, for themselves or through, on behalf of, or in conjunction with any other person, partnership or corporation:
- 17.1.1 Own, maintain, engage in, be employed as an officer, director, or principal of, lend money to, extend credit to, or have any interest in any other business that, directly or indirectly, by itself or through licensees or franchisees, offers the Approved Products and/or Approved Services or any other products and/or services authorized or offered for sale by System franchisees (a "Competitive Business") within the Designated Territory or the Designated Territory of any other System franchisee, provided that this Section 17.1.1 does not apply to: (i) such person's ownership of a Franchised Business under a Franchise Agreement with Franchisor; or (ii) such person's ownership of a less than five percent (5%) legal or beneficial interest in any publicly traded company providing such services; or
- 17.1.2—Employ or seek to employ any person who is at that time employed by Franchisor, Franchisor's affiliates, or otherwise directly or indirectly induce or seek to induce such person to leave his or her employment thereat; or
- 47.1.3 Solicit any current, former or prospective customer solicited by Franchisee's Franchised Business or any other customer of whom Franchisee has become aware as a result of access to Franchisor's System, or other franchisees for any competitive purpose.
- 17.2 After the Term of this Agreement. For a period of two (2) years after the expiration, transfer or termination of this Agreement, or after the expiration or termination of any Holdover Period, regardless of the cause, neither Franchisee, Franchisee's owners, officers, directors, principals, nor any member of the immediate family of Franchisee or Franchisee's owners, principals, officers or directors will, directly or indirectly, for themselves or through, on behalf of, or in conjunction with any other person, partnership or corporation:

17.2.1 Own, maintain, engage in, be employed as an officer, director, principal of, lend money to, extend credit to, or have any interest in any Competitive Business: (a) within the Designated Territory; (b) within a twenty five (25)—mile radius of the Designated Territory of any other Franchised Business; or (c) within a twenty five (25)—mile radius of any System business operated by Franchisor or its affiliates, provided that this Section 17.2.1 does not apply to: (i) such person's ownership of a Franchised Business under a Franchise Agreement with Franchisor; or (ii) such person's ownership of a less than five percent (5%) legal or beneficial interest in any publicly traded company providing services the same as or similar to a Competitive Business; or

17.2.2—Employ or seek to employ any person who is at that time employed by Franchisor, Franchisor's affiliates, or otherwise directly or indirectly induce or seek to induce such person to leave his or her employment thereat; or

17.2.3 Solicit any current, former or prospective customer solicited by Franchisee's Franchised Business or any other customer of whom Franchisee has become aware as a result of access to Franchisor's System, or other franchisees for any competitive purpose.

- **Intent and Enforcement**. It is the parties' intent that the provisions of this Section 17 be judicially enforced to the fullest extent permissible under applicable law. Accordingly, the parties agree that any reduction in scope or modification of any part of the noncompetition provisions contained herein shall not render any other part unenforceable. In the event of the actual or threatened breach of this Section 17 by Franchisee, any of Franchisee's principals, or any member of the immediate family of Franchisee or Franchisee's principals, Franchisor shall be entitled to an injunction restraining such person from any such actual or threatened breach. Franchisee acknowledges that the covenants contained herein are necessary to protect the goodwill efassociated with the Franchised Business, other System franchisees, and the System. Franchisee further acknowledges that covenants contained in this Section 17 are necessary to protect Franchisor's Confidential Information, procedures and know-how transmitted made available to Franchisee during the term of this Agreement. Franchisee agrees that in the event of the actual or threatened breach of this Section 17, Franchisor's harm will be irreparable and that Franchisor has no adequate remedy at law to prevent such harm. Franchisee acknowledges and agrees on Franchisee's own behalf, and on behalf of the persons who are liable under this Section 17, that each has previously worked or been gainfully employed in other careers and that the provisions of this Section 17 in no way prevent any such person from earning a living. Franchisee further acknowledges and agrees that the time limitations of this Section 17 shall be tolled during any default under this Section.
- 17.4 **Employees**. Franchisee shall ensure that Franchisee's principals, managers, employees, and anyone else who will have access to Franchisor's Confidential Information, execute a Confidentiality and Restrictive Covenant Agreement; in the form attached to the Franchisor Disclosure Document as Exhibit C to this AgreementD, or as Franchisor, in Franchisor's sole discretion, otherwise prescribes. Franchisee must furnish Franchisor a copy of each executed agreement.
- 17.5 **No Defense.** Franchisee hereby agrees that the existence of any claim Franchisee may have against Franchisor, whether or not arising from this Agreement, shall not constitute a defense to Franchisor's enforcement of the covenants contained in this Section 17. Franchisee agrees to pay all costs and expenses (including reasonable attorneys' fees) that Franchisor incurs in connection with the enforcement of this Section 17.

18. DISPUTE RESOLUTION

- 18.1 **Choice of Law**. This Agreement shall be governed by and construed in accordance with the laws of the State of North Carolina, without reference to its conflict of laws principalsprinciples.
- Internal Dispute Resolution. Franchisee and Franchisor believe that it is important to resolve any disputes amicably, quickly, cost-effectively and professionally and to return to business as soon as possible. Franchisee and Franchisor have agreed that the provisions of this Section 18 support these mutual objectives and, therefore, agree as follows: any disagreement, litigation, claim, dispute, suit, action, controversy or proceeding of any type whatsoever, including any claim for equitable relief and/or where Franchisee is acting as a "private attorney general," suing pursuant to a statutory claim or otherwise between or involving Franchisee and Franchisor on whatever theory and/or facts based, and whether or not arising out of this Agreement (including any dispute or disagreement relating to arbitration, including the arbitrability of this Agreement or any of its provisions) or offer, sale or negotiation of a Top Rail Franchise, or the relationship of the parties arising from the Franchise or from entering into this Agreement, or any claim that this Agreement or any part of this Agreement is invalid, illegal or otherwise voidable, void or unenforceable ("Dispute") will be processed in the following manner, Franchisee and Franchisor each expressly waiving all rights to any court proceeding, except as expressly provided below in Section 18.3.1. Franchisee must first bring any claim or dispute between Franchisee and Franchisor to Franchisor's management after providing notice as set forth in Section 18.6 below. Franchisee must exhaust this internal dispute resolution procedure before Franchisee may bring Franchisee's dispute before a third party. This agreement to first attempt resolution of disputes internally shall survive termination or expiration of this Agreement. Franchisee and Franchisor agree that the franchise relationship is unique and that as a result it is important that anyone who serves as a mediator or arbitrator in a Dispute must have a minimum of seven (7) years' substantive experience in franchise law.
- Mediation/Arbitration. If Franchisee and Franchisor are unable to resolve a Dispute using the internal dispute resolution procedure, then Disputes which are not first resolved through the internal dispute resolution procedure set forth in Section 18.2 above the Dispute must be submitted first to non-binding mediation in Mecklenburg County, North Carolina under the auspices of the American Arbitration Association ("AAA") in accordance with AAA's Commercial Mediation Rules then in effect. A link to the AAA Commercial Arbitration Rules and Mediation Procedures can be found at https://www.adr.org/Rules. The mediation will be conducted by one mediator with at least seven (7) years of substantive experience in franchise law who will be appointed under the AAA's Commercial Mediation Rules and who will conduct the mediation in accordance with such rules. Mediation must be conducted on an individual, not a class-wide, basis; only Franchisor (and/or its affiliates and its and their respective owners, officers, directors, agents and employees) and Franchisee (and/or its Ownersowners, guarantors, affiliates, officers, directors, agents and employees, if applicable) may be the parties to any mediation proceeding described in this Section, and no such mediation proceeding may be consolidated with any other mediation proceeding between Franchisor and any other person, corporation, limited liability company or partnership. Franchisor and Franchisee agree that statements made by Franchisor, Franchisee or any other party in any such mediation proceeding will not be admissible in any arbitration or other legal proceeding. If the mediation proceedings have been terminated either: (i) as the result of a written declaration of the mediator(s) that further mediation efforts are not worthwhile; or (ii) as a result of a if the mediation does not resolve the dispute by written declaration by Franchisoragreement within 14 days after the mediation, then, subject to the exclusions set forth in Section 18.3.1, all Disputes must be submitted to binding arbitration before one arbitrator of the AAA in accordance with its commercial arbitration rules, the AAA Commercial Arbitration Rules. The arbitration will be conducted by one arbitrator with at least seven (7) years of substantive experience in franchise law who will be appointed under the AAA's Commercial Arbitration Rules and who will conduct the arbitration in accordance with such rules. Except as provided in Section 18.3.1, no party may file litigation involving regarding any Dispute unless it has complied with

the mediation and arbitration provisions in this Article regarding that Dispute. ""Filing litigation" includes asserting a counterclaim, third-party claim, or other claim relating to a Dispute (irrespective of whether the party asserting the claim filed the legal action in which the claim is asserted). Franchisor's rights to mediation and arbitration, as set forth herein, may be specifically enforced by Franchisor. Each party shall bear its own cost of mediation and arbitration, and Franchisor and Franchisee shall share mediation and arbitration costs equally except as provided in Section 22.8. This agreement to mediate and arbitrate shall survive any termination or expiration of this Agreement.

- 18.3.1 The parties shall not be required to first attempt to mediate or arbitrate a controversy, dispute or claim through mediation as set forth in this Section 18.3 if such controversy, dispute or claim concerns an allegation that a party has violated (or threatens to violate, or poses an imminent risk of violating):
- 18.3.1.1 Any federally protected intellectual property rights in the Proprietary Marks, the System, or in any Confidential Information;
 - 18.3.1.2 Any claims pertaining to or arising out of any warranty issue;
 - 18.3.1.3 Any of the restrictive covenants contained in this Agreement;
- 18.3.1.4 Any of Franchisee's payment obligations that are more than forty-five (45) days past due;
- 18.3.1.5 Any claims arising out of or related to fraud or misrepresentation by Franchisee, or Franchisee's insolvency;
- 18.3.1.6 Any claims relating to Franchisee's obligations on termination or expiration of this Agreement;
- 18.3.1.7 Any claims relating to any Transfer of an interest in Franchisee, the Franchised Business or Franchisee's assets; or
 - 18.3.1.8 Any matters involving danger, health or safety.
- 18.3.2 Interpretation. The provisions of this Agreement must be construed as independent of any other covenant or provision of this Agreement. However, if a court or arbitrator of competent jurisdiction determines that any provisions are unlawful, that court or arbitrator is to modify or interpret the provisions to the minimum extent necessary to have them comply with the law. Notwithstanding any provision relating to the state laws by and under which this Agreement must be governed and construed, all issues relating to arbitrability or the enforcement of the agreement to arbitrate in this Agreement must be governed by the United States Arbitration Act (9 U.S.C. § 1 et seq.) and the Federal common law of arbitration. The arbitrator, and not any federal, state or local court, shall have exclusive authority to resolve any dispute relating to the interpretation, applicability, unconscionability, arbitrability, enforceability or formation of this Agreement, including any claim of fraud in the inducement or that all or any part of the Agreement is void or voidable. However, the preceding sentence shall not apply to the clause entitled "Class Action Waiver."
- 18.3.3 **Judgment**. Judgment on an arbitration award may be entered in any court of competent jurisdiction. This judgment is binding, final and non-appealable.

- 18.3.4 **Failure to Appear**. The arbitration and mediation provisions in this Article are self-executing and remain in full force and effect after the expiration or sooner termination of this Agreement. If either party fails to appear at any properly noticed arbitration proceeding, notwithstanding failing to appear, an award may be entered against that party by default or otherwise.
- 18.3.5 Class Action Waiver. Any proceeding (whether mediation, arbitration, trial to a court or jury, appeal or otherwise) must be brought in the parties' individual capacity and not as a plaintiff or class member in any purported class, collective, representative, multiple plaintiff or similar proceeding ("Class Action"). Franchisee and Franchisor expressly waive any ability to maintain any Class Action in any forum. Further, an arbitration proceeding between Franchisor and Franchisee (or any of Franchisee's or Franchisor's affiliates and owners and guarantors) may not be consolidated with any other arbitration proceeding between them and any other franchisee, person or entity. Franchisee hereby agrees not to seek joinder of any of its claims with those of any other party. The arbitrator shall not have authority to combine or aggregate similar claims or conduct any Class Action, nor make an award to any person or entity not a party to the arbitration. Any claim that all or part of this Class Action Waiver is unenforceable, unconscionable, void or voidable may be determined only by a court of competent jurisdiction and not by an arbitrator. FRANCHISEE AND FRANCHISOR UNDERSTAND THAT WETHEY WOULD HAVE HAD A RIGHT TO LITIGATE THROUGH A COURT, TO HAVE A JUDGE OR JURY DECIDE OUR CASE, AND TO BE PARTY TO A CLASS OR REPRESENTATIVE ACTION. HOWEVER, FRANCHISEE AND FRANCHISOR UNDERSTAND AND CHOOSE TO WAIVE THAT RIGHT AND HAVE ANY CLAIMS DECIDED INDIVIDUALLY THROUGH ARBITRATION. It is Franchisee's and Franchisor's joint business judgment that the limitations of this subsection make good business sense, because:
- 18.3.5.1 the mediation and arbitration procedures contemplated by this Agreement (and which Franchisee and Franchisor agree are the core methods for resolving disputes) function most effectively on an individual case basis;
- 18.3.5.2 there are significant business and other factors present in each individual franchisee's situation which should be respected; and
- 18.3.5.3 the economic interests of lawyers on either side in class-wide or multiple plaintiff dispute, as well as the tendency to polarize positions, makes accommodation and compromise, as a practical business matter, less easily achieved, which would be a serious detriment to Franchisee's and Franchisor's business interests, as well as those of the entire Franchise System, in quickly, amicably and economically resolving any dispute.
- 18.4 **Selection of Venue**. Nothing contained in this Agreement shall prevent Franchisor from applying to and obtaining from any court having jurisdiction, a writ of attachment, a temporary injunction, preliminary injunction and/or other emergency relief available to safeguard and protect Franchisor's legitimate business interests. The parties expressly agree to the jurisdiction and venue of any court of general jurisdiction in Mecklenburg County, North Carolina and the jurisdiction and venue of the United States District Court for the Western District of North Carolina. Franchisee acknowledges that this Agreement has been entered into in the State of North Carolina, and that Franchisee is to receive valuable and continuing services emanating from Franchisor's headquarters in Huntersville, North Carolina, including, but not limited to, training, assistance, support, and the development of the System. In recognition of such services and their origin, Franchisee hereby irrevocably consents to the personal jurisdiction of the state and federal courts of North Carolina as set forth in this Section.
- 18.5 **Third Party Beneficiaries.** Franchisor's officers, directors, shareholders, members, agents and/or employees are express third-party beneficiaries of the provisions of this Agreement, including

the mediation and arbitration provisions set forth in this Section 18, each having authority to specifically enforce the right to mediate or arbitrate <u>any Dispute or claims</u> asserted <u>by Franchisee</u> against such person(s) by Franchisee.).

- 18.6 **Prior Notice of Claims**. As a condition precedent to commencing an action for damages or for violation or breach of this Agreement, Franchisee must notify Franchisor within thirty (30) days after the occurrence of the violation or breach, and failure to timely give such notice shall preclude any claim for damages.
- 18.7 **No Right to Offset.** Franchisee shall not withhold all or any part of any payment to Franchisor or any of its affiliates on the grounds of Franchisor's alleged nonperformance or as an offset against any amount Franchisor or any of Franchisor's affiliates allegedly may owe Franchisee under this Agreement or any related agreements.
- Injunctive Relief. The Franchised Business is intended to be one of a large number of businesses identified by the Proprietary Marks selling to the public the Approved Products and Approved Services associated with the Proprietary Marks. Consequently, a single franchisee's failure to comply with the terms of its franchise agreement is likely to cause irreparable damage to Franchisor, and damages at law would therefore be an inadequate remedy. Upon Franchisee's breach or threatened breach of any of the terms concerning any matters referenced in Section 18.3.1, Franchisor may seek an injunction restraining the breach and/or a decree of specific performance (with recovery of reasonable attorneys' attorneys' fees and costs incurred in obtaining equitable relief). Franchisor may do so without demonstrating or proving any actual damage. These equitable remedies are in addition to all other rights or remedies to which Franchisor may otherwise be entitled due to Franchisee's breach of this Agreement. Franchisor may seek this relief without posting any bond or security; however, if a court of competent jurisdiction requires a bond or security, a bond or security for \$1,000 is sufficient. Notwithstanding anything in this Agreement to the contrary, Franchisor may seek injunctive relief in any jurisdiction that has jurisdiction over Franchisee or any other party against whom the relief is sought. If injunctive relief is granted, Franchisee's only remedy will be the court's dissolution of the injunctive relief. If the injunctive relief was wrongfully issued, Franchisee expressly waives all claims for damages Franchisee incurred as a result of the wrongful issuance.
- 18.9 **Limitation of Action.** Franchisee further agrees that no cause of action arising out of or under this Agreement may be maintained by Franchisee against Franchisor unless brought before the expiration of one (1) year after the act, transaction or occurrence upon which such action is based or the expiration of one (1) year after the Franchisee becomes aware of facts or circumstances reasonably indicating that Franchisee may have a claim against Franchisor hereunder, whichever occurs sooner, and that any action not brought within this period shall be barred as a claim, counterclaim, defense, or setoff.
- 18.9.1 Franchisee hereby waives the right to obtain any remedy based on alleged fraud, misrepresentation or deceit by Franchisor, including, without limitation, rescission of this Agreement, in any mediation, arbitration, judicial, or other adjudicatory proceeding arising hereunder, except upon a ground expressly provided in this Agreement, or pursuant to any right expressly granted by any applicable statute expressly regulating the sale of franchises, or any regulation or rules promulgated thereunder.
- 18.9.2 Notwithstanding anything to the contrary contained herein, all actions shall be conducted on an individual, not a class-wide basis, and any proceeding between Franchisee, Franchisee's guarantors, and Franchisor or its affiliates or employees may not be consolidated with any other proceeding between Franchisor and any other person or entity.

18.10 Waiver of Punitive Damages. Franchisee The parties hereby waiveswaive to the fullest extent permitted by law, any right to or claim for any punitive, exemplary, incidental, indirect, special, or consequential damages (including, without limitation, lost profits) against Franchisor the other party arising out of any cause whatsoever (whether such cause be based in contract, negligence, strict liability, other tort or otherwise) and agrees that in the event of a dispute, that Franchisee's recovery is limited to actual damages. If any other term of this Agreement is found or determined to be unconscionable or unenforceable except for any reason, the foregoing provisions shall continue in full force and effect, including, without limitation, the waiver of any right to claim any consequential damages. Nothing in this Section shall be construed to prevent Franchisor from claiming and obtaining indemnified claims described in 13.2 and punitive or consequential exemplary damages, including lost future Royalty Fees for the balance of the term of this Agreement violation of the Lanham Act, Trademark or Copyright Infringement or Trademark Dilution, unauthorized dissemination of the confidential information or trade secrets.

18.11 THE PARTIES HEREBY AGREE TO WAIVE TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM, WHETHER AT LAW OR EQUITY, REGARDLESS OF WHICH PARTY BRINGS SUIT. THIS WAIVER SHALL APPLY TO ANY MATTER WHATSOEVER BETWEEN THE PARTIES HERETO WHICH ARISES OUT OF OR IS RELATED IN ANY WAY TO THIS AGREEMENT, THE PERFORMANCE OF EITHER PARTY, AND/OR FRANCHISEE'S PURCHASE FROM FRANCHISOR OF THE FRANCHISE AND/OR ANY PRODUCTS OR SERVICES.

19. REPRESENTATIONS

19.1 Call Center. If established, the The Call Center will beis provided on an "AS IS" and "AS AVAILABLE" basis. Franchisor does not covenant any level, quality, continuity or standard of operation for the Call Center, or covenant that the Call Center will be free from defaults, viruses or other harmful components, operate on a continuous or uninterrupted basis, or provide secure access to the Call Center or services provided thereby. FRANCHISOR MAKES NO REPRESENTATION OR WARRANTY OF ANY KIND, EITHER EXPRESS OR IMPLIED, INCLUDING, WITHOUT LIMITATION, WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, TITLE OR NON-INFRINGEMENT, OR WARRANTIES REGARDING THE QUALITY, ACCURACY, TIMELINESS, AVAILABILITY, SUITABILITY, OR RELIABILITY OF ANY SERVICE, OR SECURITY, USEFULNESS, LACK OF VIRUSES OR OTHER HARMFUL COMPONENTS, OR WARRANTIES REGARDING COMPLETENESS OF THE CONTENT OF THE CALL CENTER, OR WARRANTIES WITH RESPECT TO THE USE OR AVAILABILITY OF ANY INFORMATION, DATA, ITEM, APPARATUS, METHOD OR PROCESS INCLUDED IN THE CALL CENTER, OR THAT SUCH WILL MEET THE FRANCHISEE'S REQUIREMENTS, OR BE ERROR FREE OR NOT INFRINGE ON THE RIGHTS OF OTHERS, OR THAT DEFECTS WILL BE CORRECTED.

19.2 No Authority. NO SALESPERSON, REPRESENTATIVE OR OTHER PERSON HAS THE AUTHORITY TO BIND OR OBLIGATE FRANCHISOR EXCEPT FRANCHISOR'S AUTHORIZED OFFICER BY A WRITTEN DOCUMENT. FRANCHISEE ACKNOWLEDGES THAT NO REPRESENTATIONS, PROMISES, INDUCEMENTS, GUARANTEES OR WARRANTIES OF ANY KIND WERE MADE BY FRANCHISOR OR ON FRANCHISOR'S BEHALF WHICH HAVE LED FRANCHISEE TO ENTER INTO THIS AGREEMENT. FRANCHISEE UNDERSTANDS THAT WHETHER FRANCHISEE SUCCEEDS AS A FRANCHISEE IS DEPENDENT UPON FRANCHISEE'S EFFORTS, BUSINESS JUDGMENTS, THE PERFORMANCE OF FRANCHISEE'S EMPLOYEES, MARKET CONDITIONS AND VARIABLE FACTORS BEYOND FRANCHISOR'S CONTROL OR INFLUENCE. FRANCHISEE FURTHER UNDERSTANDS THAT SOME FRANCHISEES ARE MORE OR LESS SUCCESSFUL THAN OTHER FRANCHISEES AND THAT

FRANCHISOR HAS MADE NO REPRESENTATION THAT FRANCHISEE WILL DO AS WELL AS ANY OTHER FRANCHISEE.

19.3 Receipt. THE UNDERSIGNED ACKNOWLEDGES RECEIPT OF THIS AGREEMENT, WITH ALL BLANKS COMPLETED AND WITH ANY AMENDMENTS AND SCHEDULES, AT LEAST SEVEN (7) CALENDAR DAYS PRIOR TO EXECUTION OF THIS AGREEMENT. IN ADDITION, THE UNDERSIGNED ACKNOWLEDGES RECEIPT OF FRANCHISOR'S FRANCHISE DISCLOSURE DOCUMENT AT LEAST FOURTEEN (14) CALENDAR DAYS PRIOR TO THE EXECUTION OF THIS AGREEMENT OR FRANCHISEE'S PAYMENT OF ANY MONIES TO FRANCHISOR, REFUNDABLE OR OTHERWISE.

19.4 Opportunity for Review by Franchisee's Advisors. FRANCHISEE ACKNOWLEDGES THAT FRANCHISOR HAS RECOMMENDED, AND THAT FRANCHISEE HAS HAD THE OPPORTUNITY TO OBTAIN, REVIEW OF THIS AGREEMENT AND FRANCHISOR'S FRANCHISE DISCLOSURE DOCUMENT BY FRANCHISEE'S LAWYER, ACCOUNTANT OR OTHER BUSINESS ADVISOR PRIOR TO EXECUTION HEREOF.

19.5 Execution of Agreement. 3 Execution of Agreement. EACH OF THE UNDERSIGNED PARTIES WARRANTS THAT IT HAS THE FULL AUTHORITY TO SIGN AND EXECUTE THIS AGREEMENT. IF FRANCHISEE IS A PARTNERSHIP OR CORPORATION, THE PERSON EXECUTING THIS AGREEMENT ON BEHALF OF SUCH PARTNERSHIP OR CORPORATION WARRANTS TO FRANCHISOR, BOTH INDIVIDUALLY AND IN HIS/HER CAPACITY AS PARTNER OR OFFICER, THAT ALL OF THE PARTNERS OF THE PARTNERSHIP OR ALL OF THE SHAREHOLDERS OF THE CORPORATION, AS APPLICABLE, HAVE READ AND APPROVED THIS AGREEMENT, INCLUDING ANY RESTRICTIONS WHICH THIS AGREEMENT PLACES UPON RIGHTS TO TRANSFER THEIR INTEREST IN THE PARTNERSHIP OR CORPORATION.

20. GUARANTEE OF PRINCIPALS AND THEIR SPOUSES

- Quarantee. Franchisee's spouse and all partners in a limited partnership, shareholders in a corporate franchisee, or members of a limited liability company, as well as all general partners and managing members and their spouses hereby personally and unconditionally guarantee without notice, demand or presentment, the payment of all of Franchisee's monetary obligations under this Agreement and any other agreement between Franchisee and Franchisor and/or Franchisor's affiliates, as if each were an original party to this or any other agreement in his or her individual capacity. All such personal guarantors further agree to be bound by the restrictions upon Franchisee's activities upon transfer, termination or expiration, and nonrenewal of this Agreement as if each were an original party to this Agreement in his or her individual capacity. All such personal guarantors and their spouses must execute a continuing Personal Guaranty in the form attached as Exhibit A to this Agreement.
- 20.2 **Security Interest.** For valuable consideration, as security for the payment of/for: (a) all amounts owing or to be owed to usFranchisor and ourits affiliates by youFranchisee from time to time under this Agreement or any other agreement between youFranchisee and usFranchisor or ourFranchisor's affiliates, or otherwise; (b) yourFranchisee's performance of all obligations to usFranchisor and ourFranchisor's affiliates; and (c) the costs and expenses that weFranchisor and/or ourits affiliates incur to collect or attempt to collect amounts due from youFranchisee and to enforce this Section (together, the "Obligations"), youFranchisee hereby grant usgrants Franchisor a security interest in all of the assets of the Franchised Business or used by at, or in connection with yourFranchisee's Franchised Business, including but not limited to: (i) all equipment, furnishings, fixtures, motor vehicles, merchandise, inventory, goods and other tangible personal property; (ii) all accounts, accounts receivable, other receivables, contract

rights, leasehold interests, software, chattel paper and general intangibles; (iii) all instruments, documents of title, policies and certificates of insurance, securities, bank deposits, bank accounts and cash; (iv) all books, records and documents; (v) all permits and licenses for the operation of the Franchised Business; and (vi) all accessions, additions and improvements to, and all replacements, substitutions and parts for, and all proceeds and products of, the foregoing, including proceeds of insurance (collectively, the "Collateral"). Franchisee agrees to execute and deliver to Franchisor any other documents reasonably requested by Franchisor to create, maintain, perfect, or assure the priority of the security interest granted above. Franchisee hereby appoints Franchisor as its agent and attorney-in-fact to execute and deliver documents and to take all other actions (to the extent permitted by law) in Franchisee's name and on Franchisee's behalf that Franchisor may deem necessary or advisable to create, maintain, perfect, assure the priority of, or foreclose its security interest in and lien on the Collateral. YouFranchisee must execute and deliver to usFranchisor financing statements and/or such other documents as weFranchisor reasonably deemdeems necessary to perfect ourFranchisor's interest in the Collateral within 10 days of yourFranchisee's receipt of such documents from usFranchisor.

YouFranchisee will not remove the Collateral or any portion thereof or grant or permit any security interest in (or otherwise encumber) the Collateral, without ourFranchisor's prior written consent. You representFranchisee represents and warrantwarrants that the security interest granted is prior to all other security interests in the Collateral except for: (i) bona fide purchase money security interests; and (ii) if consented to by usFranchisor in writing, the security interest granted to a third party in connection with yourFranchisee's original financing for yourFranchisee's Franchisor Business, if any. In connection with any request for ourFranchisor's approval of a security interest, weFranchisor will exercise ourits Business Judgment, bearing in mind the interests of the borrower, lender, ourselvesFranchisor and the System. On the occurrence of any event entitling usFranchisor to terminate this Agreement or any other agreement between the parties, or if weFranchisor reasonably determined termines that we areFranchisor is not assured that yourFranchisee's (and/or any affiliates') obligations will be timely and fully paid and/or performed, weFranchisor will have all the rights and remedies of a secured party under the Uniform Commercial Code of the State in which yourFranchisee's Franchised Business is located, including, without limitation, the right to take possession of the Collateral. This appointment is coupled with an interest and is irrevocable as long as any of the Obligations remain outstanding.

21. NOTICES

All notices and requests to be given under this Agreement are to be in writing, and delivered by either hand delivery or overnight mail by a recognized carrier offering a delivery receipt, to the following addresses (which may be changed by written notice):

Franchisee Name/Address:

Franchisor: HFB FenceCo Franchising, LLC

107 Parr Drive

Huntersville, North Carolina 28078

Attn: Chief Legal Officer

Mailing any notice hereunder sent by U.S. certified mail, postage prepaid or when sent via Federal Express or a similar overnight courier shall be presumptive evidence of delivery of the notice or request.

22. MISCELLANEOUS

- 22.1 **Entire Agreement**. This Agreement contains the entire Agreement of the parties. There are no representations either oral or written, except those contained in this Agreement. This written Agreement includes all representations between the parties. This Agreement may not be modified except by a written document signed by both parties. Nothing in this Agreement or any related agreement is intended to disclaim the representations Franchisor made in the Franchise Disclosure Document furnished to Franchisee.
- 22.2 Construction of Language. The language of this Agreement shall be construed according to its fair meaning, and not strictly for or against either party. All words in this Agreement refer to whatever number or gender the context requires. If more than one party or person is referred to as Franchisee, their obligations and liabilities shall be joint and several. Headings are for reference purposes and do not control interpretation. Reference to Franchisee's "immediate family" means Franchisee's spouse, parents, children and siblings, and Franchisee's spouse's parents, children and siblings. Reference to Franchisee's "principals" means Franchisee's partners, officers, directors, shareholders, members and managers, as applicable. References to "Franchisor" and "Franchisee" include the party's successors, assigns or transferees. The parties have had a reasonable opportunity to review this Agreement. In the event of an ambiguity or if a question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by all of the parties, and no presumptions or burdens of proof shall arise in favor of any party by virtue of the authorship of any of the provisions of this Agreement.
- 22.3 **Severability**. If any provision of this Agreement is deemed invalid or inoperative for any reason, that provision shall be deemed modified to the extent necessary to make it valid and operative or, if it cannot be so modified, it shall then be severed, and the remainder of that provision shall continue in full force and effect as if this Agreement had been signed with the invalid portion so modified or eliminated; provided, however, that if any part of this Agreement relating to payments to Franchisor or any of its affiliates or protection of the Proprietary Marks or the Confidential Information, including the Operations Manual and Franchisor's other Trade Secrets, is declared invalid or unenforceable, then Franchisor, at Franchisor's option, may terminate this Agreement immediately upon written notice to Franchisee.
- 22.4 **State Law Applies**. If any provision of this Agreement, including, but not limited to, its provisions for transfer, renewal, termination, notice of termination, or cure rights, is inconsistent with any valid law or regulation of the state in which the Franchised Business is located, then the valid law or regulation of that state applicable to the Franchised Business shall supersede any provision of this Agreement that is less favorable to Franchisee.
- 22.5 Additional Documentation. Franchisee must, from time to time, subsequent to the date first set forth above, at Franchisor's request and without further consideration, execute and deliver such other documentation or agreements and take such other actions as Franchisor reasonably may require in order to effectuate the transactions contemplated herein. In the event that Franchisee fails to comply with the provisions of this Section, Franchisee hereby appoints Franchisor as Franchisee's attorney-in-fact to execute any and all documents on Franchisee's behalf reasonably necessary to effectuate the transactions contemplated herein.
- 22.6 **Force Majeure**. Neither Franchisee, Franchisor, nor Franchisor's affiliates will be liable for loss or damage or deemed to be in breach of this Agreement or any related agreement, if its failure to perform its obligations is not the fault nor within the reasonable control of the person due to perform, but results from, without limitation, acts of God or natural disaster (such as, but not limited to, violent storm, cyclone, hurricane, tornado, blizzard, earthquake, volcanic activity, landslide, tidal wave, tsunami, flood, damage or destruction by lightning, drought); accident, riots, war, terrorist act, epidemic, pandemic, quarantine, civil

commotion, breakdown of communication facilities, breakdown of web host, breakdown of Internet service provider, natural catastrophes, national or regional emergencies; governmental acts or omissions, changes in laws or regulations, national or local strikes, fire, explosion, generalized lack of availability of raw materials or energy; or any other cause, whether similar in kind to the foregoing or otherwise. Any delay resulting from any such cause will extend the time of performance for the period of such delay or for such other reasonable period of time as the parties agree in writing or will excuse performance, in whole or in part, as Franchisor deems reasonable. For the avoidance of doubt, this Force Majeure section shall not apply to a party's financial inability to perform its obligations hereunder, nor does this Section apply to negate or delay any payment obligation whether on account of a public health crisis such as COVID-19 or otherwise.

- Anti-Terrorist Activities. Franchisee certifies that neither Franchisee, nor Franchisee's owners, principals, employees, or anyone associated with Franchisee is listed in the Annex to Executive Order 13224. Franchisee agrees not to hire or have any dealings with a person listed in the Annex. Franchisee certifies that Franchisee has no knowledge or information that, if generally known, would result in Franchisee, Franchisee's owners, principals, employees, or anyone associated with Franchisee being listed in the Annex to Executive Order 13224. Franchisee agrees to comply with and/or assist Franchisor to the fullest extent possible in Franchisor's efforts to comply with the Anti-Terrorism Laws (as defined below). In connection with such compliance, Franchisee certifies, represents and warrants that none of Franchisee's property or interests are subject to being "blocked" under any of the Anti-Terrorism Laws and that Franchisee and Franchisee's owners or principals are not otherwise in violation of any of the Anti-Terrorism Laws. Franchisee is solely responsible for ascertaining what actions must be taken by Franchisee to comply with all such Anti-Terrorism Laws, and Franchisee specifically acknowledges and agrees that Franchisee's indemnification responsibilities as provided in Section 13.2 of this Agreement pertain to Franchisee's obligations under this Section 22.7. Any misrepresentation by Franchisee under this Section or any violation of the Anti-Terrorism Laws by Franchisee, Franchisee's owners, principals or employees shall constitute grounds for immediate termination of this Agreement and any other agreement Franchisee has entered into with Franchisor or one of Franchisor's affiliates in accordance with the terms of Section 15.2.1 of this Agreement. As used herein, "Anti-Terrorism Laws" means Executive Order 13224 issued by the President of the United States, the Terrorism Sanctions Regulations (Title 31, Part 595 of the U.S. Code of Federal Regulations), the Foreign Terrorist Organizations Sanctions Regulations (Title 31, Part 597 of the U.S. Code of Federal Regulations), the Cuban Assets Control Regulations (Title 31, Part 515 of the U.S. Code of Federal Regulations), the USA PATRIOT Act, and all other present and future federal, state and local laws, ordinances, regulations, policies, lists, and any other requirements of any Governmental Authority (including, without limitation, the United States Department of Treasury Office of Foreign Assets Control) addressing or in any way relating to terrorist acts and acts of war.
- Attorneys' Fees. If Franchisee is in breach or default of any monetary or non-monetary material obligation under this Agreement or any related agreement between Franchisee and Franchisor and/or Franchisor's affiliates, and Franchisor engages an attorney to enforce Franchisor's rights (whether or not formal judicial proceedings are initiated), Franchisee must pay all reasonable attorneys' fees, court costs and litigation expenses Franchisor incurs. If Franchisee institutes any legal action to interpret or enforce the terms of this Agreement, and Franchisee's claim in such action is denied or the action is dismissed, Franchisor is entitled to recover Franchisor's reasonable attorneys' fees, and all other reasonable costs and expenses incurred in defending against same, and to have such an amount awarded as part of the judgment in the proceeding.
- 22.9 **Amendment of Prior Agreements.** In order to enhance consistency and quality of operation, performance, dispute resolution and other matters, we amend our Franchisor amends its standard Franchise Agreement from time to time. As a result, this Agreement may be different from other Brand franchise agreements that Franchisee or Franchisee's affiliates may have signed in the past and may contain revised provisions regarding, among other things, modifications to the System, manner of payment of fees

and late fees, duties of franchisee, protection of trademarks, status and protection of Manuals and Confidential Information, technology requirements, advertising, insurance, accounting and records, transfers, default and termination, obligations on termination, franchisee covenants, taxes, indemnification, obligations to defend, approvals and waivers, notices, construction of agreement and applicable law. To cooperate with usFranchisor in the achievement of these goals and as a condition of the grant of an additional franchise, Franchisee agrees that all of Franchisee's or its affiliates' existing Brand franchise agreements with Franchisor or its affiliates are amended to match the provisions of this Agreement (if the existing franchise agreements do not already include these provisions), except with respect to the royalty feeRoyalty Fee rate, required marketing contributions and spending, other fees for which amounts are specified, territory description, Approved Location, contract term, renewal conditions, and transfer conditions set out in the prior agreements, which will remain unchanged. Provided, however, that any modifications to the Franchise Agreement that purports to change: (i) an investor protection under applicable state law; (ii) any applicable franchise relations law; (iii) the rules adopted under either; or (iv) a requirement that is a part of the state registration process is specifically excepted from this provision. FRANCHISEE ACKNOWLEDGES THAT THIS SECTION AMENDS ALL OF FRANCHISEE'S EXISTING FRANCHISE AGREEMENTS WITH FRANCHISOR AND THAT THE AMENDMENT WILL SURVIVE THE EXPIRATION OR TERMINATION OF THIS AGREEMENT.

22.10 **No Personal Liability**. Franchisee agrees that fulfillment of any and all of Franchisor's obligations written in this Agreement or based on any oral communications which may be ruled to be binding in a Court of Law shall be Franchisor's sole responsibility and none of Franchisor's agents, representatives, nor any individuals associated with Franchisor's franchise company shall be personally liable to Franchisee for any reason. This is an important part of this Agreement. Franchisee agrees that nothing that Franchisee has been told by Franchisor or Franchisor's representatives shall be binding unless it is written in this Agreement. This is an important part of this Agreement. Do not sign this Agreement if there is any question concerning its contents or any representations made.

22.11 You and we agreeFranchisee is advised that weFranchisor may conduct due diligence on the franchise sale process—and that we rely on your representations. The results of that experience as a material inducement to enter into the franchise relationship, without which we would not enter into this Franchise Agreement with you. Accordingly, you and we agree that any such diligence may be entered into evidence in any dispute resolution proceeding, and the limitations set forth in any state addenda or other amendment to this Agreement will not waiveact as a waiver any of our rights, to use evidence, pursue claims, or enjoy the protections from unlawful or unfair business practices/dishonesty under any applicable law or regulation to use such information so long as it is collected in accordance with applicable law.

(THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK. SIGNATURES APPEAR ON THE FOLLOWING PAGE.)

IN WITNESS WHEREOF, AND INTENDING TO BE LEGALLY BOUND HEREBY, THE PARTIES HERETO HAVE CAUSED THIS AGREEMENT TO BE EXECUTED EFFECTIVE THE DATE FIRST SET FORTH ABOVE.

FRANCHISOR:	FRANCHISEE:
HFB FENCECO FRANCHISING, LLC ₂ a North Carolina limited liability company	
	(Individual, Partnership or Corporation Name)
By:	Ву:
Name:	Name:
Title:	Title:
	Ву:
	Name:
	Title:

EXHIBIT Exhibit A

TO

To Franchise Agreement
HFB FENCECO FRANCHISINGFenceCo Franchising, LLC
FRANCHISE AGREEMENT

PERSONAL GUARANTY

NOTE: IF FRANCHISEE IS A CORPORATION, EACH OF FRANCHISEE'S SHAREHOLDERS AND THEIR SPOUSES MUST EXECUTE THE FOLLOWING UNDERTAKING. IF FRANCHISEE IS A PARTNERSHIP, EACH OF FRANCHISEE'S GENERAL PARTNERS AND THEIR SPOUSES MUST EXECUTE THE FOLLOWING UNDERTAKING. IF FRANCHISEE IS A LIMITED LIABILITY COMPANY, EACH OF FRANCHISEE'S MEMBERS AND MANAGERS AND THEIR SPOUSES MUST EXECUTE THE FOLLOWING UNDERTAKING.

Definition of "l	(persor	(s) or entity)			
	Sole Proprietorship Indi Corporation	vidual	_ 🗆	Limited Liability Company	
	General Partnership		Limited Partne	rship	

ARTICLE I PERSONAL GUARANTY

The undersigned persons (individually and collectively, "you" or "Personal Guarantors") hereby represent to HFB FenceCo Franchising, LLC ("Franchisor") that you are all of the owners, shareholders, general partners, or members and managers, or the spouse of any such owner, shareholder, general partner, member or manager of the Franchisee. In consideration of the grant by Franchisor to Franchisee, as provided in the foregoing franchise agreement (the "Franchise Agreement"), each of you hereby agree, in consideration of benefits received and to be received by each of you, jointly and severally, and for yourselves, your heirs, legal representatives and assigns, to be firmly bound by all of the terms, provisions and conditions of the foregoing Franchise Agreement, and any other agreement between Franchisee and Franchisor and/or its affiliates, and do hereby unconditionally guarantee the full and timely performance by Franchisee of each and every obligation of Franchisee under the aforesaid-Franchise Agreement or other agreement between Franchisor and Franchisee, including, without limitation, any indebtedness of Franchisee arising under or by virtue of the aforesaid Franchise Agreement and that you (jointly and individually) will not permit or cause any change in the percentage of Franchisee owned, directly or indirectly, by any person, without first obtaining the written consent of Franchisor prior to said proposed transfer, which consent must not be unreasonably withheld, and without first paying, or causing to be paid, to Franchisor the transfer fee provided for in said Franchise Agreement, if applicable, and without otherwise complying with the transfer provisions of the foregoing Franchise Agreement. You further agree to be bound by the in-term and post-term non-competition and non-solicitation covenants set forth in Section 17 of the Franchise Agreement, as well as all other covenants set forth in the Franchise Agreement, including, but not limited to, those concerning confidentiality (Section 5 of the Franchise Agreement) and indemnification (Section 13.2 of the Franchise Agreement). You agree that this personal guaranty (the

"Guaranty") will be governed by the dispute resolution procedures set forth in Section 18 of the Franchise Agreement.

ARTICLE II CONFIDENTIALITY

During the term of this Guaranty, you will receive information which Franchisor considers a trade secret and confidential information ("Confidential Information"). You shall not, during the term of this Guaranty or thereafter, communicate, divulge, or use for the benefit of any other person, partnership, association, corporation, or limited liability company, any Confidential Information, including, without limitation, Franchisor's Operations Manual and its contents; price lists and standards and specifications for the Approved Products and Approved Services; standards and specifications related to Franchisor's integrated bookkeeping system; and other methods, techniques, and know-how concerning the operation of a Franchised Business, of which you may be apprised by virtue of your role as a Personal Guarantor of Franchisee.

ARTICLE III NON-COMPETITION

- 1. **During the Term of the Franchise Agreement**. During the term of the Franchise Agreement, neither you, nor your owners, principals, officers, directors, nor any members of your family or the family of your owners, officers, directors, or principals may, directly or indirectly, for themselves, or through, on behalf of, or in conjunction with any other person or entity:
- a. Own, maintain, engage in, be employed as an officer, director, principal of, lend money to, extend credit to, or have any interest in any other business that, directly or indirectly, by itself or through licensees or franchisees, offers the Approved Products or Approved Services, or any other products and/or services authorized or offered for sale by System franchisees (a "Competitive Business") within the Designated Territory or the Designated Territory of any other System franchisee, provided that this Article III(1) does not apply to: (i) such person's ownership of a Franchised Business under a Franchise Agreement with Franchisor; or (ii) such person's ownership of a less than five percent (5%) legal or beneficial interest in any publicly traded company providing such services;
- b. Employ or seek to employ, any person who is at that time employed by Franchisor, Franchisor's affiliates, or otherwise directly or indirectly induce or seek to induce such person to leave his or her employment thereat; or
- c. Solicit any current, former, or prospective customer solicited by Franchisee's Franchised Business, or any other customer of whom Franchisee has become aware as a result of access to Franchisor's System, or other franchisees for any competitive purpose.
- 2. **After the Term of the Franchise Agreement**. For a period of two (2) years after the expiration and nonrenewal, transfer or termination of the Franchise Agreement, regardless of the cause, neither you, nor your owners, principals, officers, directors, nor any members of your family or the family of your owners, officers, directors, or principals may, directly or indirectly, for themselves or through, on behalf of, or in conjunction with any other person or entity:
- a. Own, maintain, engage in, be employed as an officer, director, principal of, lend money to, extend credit to, or have any interest in any Competitive Business: (i) within the Designated Territory; (ii) within a twenty five (25)—mile radius of the Designated Territory of any other Franchised Business; or (iii) within a twenty five (25)—mile radius of any System business operated by Franchisor or

its affiliate, provided that this Article III(2) does not apply to: (A) such person's ownership of a Franchised Business under a Franchise Agreement with Franchisor; or (B) such person's ownership of a less than five percent (5%) legal or beneficial interest in any publicly traded company providing services the same as or similar to a Competitive Business;

- b. Employ or seek to employ, any person who is at that time employed by Franchisor, Franchisor's affiliates, or otherwise directly or indirectly induce or seek to induce such person to leave his or her employment thereat; or
- c. Solicit any current, former, or prospective customer solicited by Franchisee's Franchised Business, or any other customer of whom Franchisee has become aware as a result of access to Franchisor's System, or other franchisees for any competitive purpose.
- 3. **Intent and Enforcement.** It is the parties' intent that the provisions of this Article III be judicially enforced to the fullest extent permissible under applicable law. Accordingly, the parties agree that any reduction in scope or modification of any part of the noncompetition provisions contained herein shall not render any other part unenforceable. In the event of the actual or threatened breach of this Article III by you, any of your principals, or any members of their immediate family, Franchisor shall be entitled to an injunction restraining such person from any such actual or threatened breach. You agree that in the event of the actual or threatened breach of this Article III, Franchisor's harm will be irreparable and that Franchisor has no adequate remedy at law to prevent such harm. You acknowledge and agree that each of you has previously worked or been gainfully employed in other careers and that the provisions of this Article III in no way prevent you from earning a living. You further acknowledge and agree that the time limitation of this Article III shall be tolled during any default under this Guaranty.

ARTICLE IV DISPUTE RESOLUTION

- 1. **Acknowledgment**. You acknowledge that this Guaranty is not a franchise agreement and does not confer upon you any rights to use the Franchisor's Proprietary Marks or its System.
- 2. **Governing Law**. This Guaranty shall be deemed to have been made in and governed by the laws of the State of North Carolina, without any reference to North Carolina conflict of laws principles.
- Internal Dispute Resolution. Franchisee and Franchisor believe that it is important to resolve any disputes amicably, quickly, cost-effectively, and professionally and to return to business as soon as possible. Franchisee and Franchisor have agreed that the provisions of Sections 3 and 4 of this Personal Guaranty support these mutual objectives and, therefore, agree as follows. and professionally and to return to business as soon as possible. Franchisee and Franchisor have agreed that the provisions of Sections 3 and 4 of this Personal Guaranty support these mutual objectives and, therefore, agree as follows. Any disagreement, litigation, claim, dispute, suit, action, controversy or proceeding of any type whatsoever, including any claim for equitable relief and/or where Franchisee is acting as a "private attorney general" suing pursuant to a statutory claim or otherwise between or involving a Guarantor, Franchisee and/or Franchisor on whatever theory and/or facts based, and whether or not arising out of this Guaranty (including any dispute or disagreement relating to arbitration, including the arbitrability of this Guaranty or any of its provisions), or offer, sale or negotiation of a Top Rail Franchise, or the relationship of the parties arising from the Franchise or from entering this Guaranty or the Franchise Agreement, or any claim that this Guaranty or the Franchise Agreement or any part thereof is invalid, illegal or otherwise voidable, void, or unenforceable ("Dispute") will be processed in the following manner, Franchisee and Franchisor each expressly waiving all rights to any court proceeding, except as expressly provided below in Section 4.1. Personal Guarantor must first bring any claim or Dispute between Franchisee and Franchisor or Personal

Guarantor and Franchisor to Franchisor's management, after providing notice as set forth in Section 4.8 below. Personal Guarantor must exhaust this internal Dispute resolution procedure before Personal Guarantor may bring Personal Guarantor's Dispute before a third party. This agreement to first attempt resolution of Disputes internally shall survive termination or expiration of this Guaranty. Personal Guarantor and Franchisor agree that the franchise relationship is unique and that as a result, it is important that anyone who serves as a mediator or arbitrator in a Dispute must have a minimum of seven (7) years' substantive experience in franchise law.

- Mediation/Arbitration. If Personal Guarantor and Franchisor are unable to resolve a Dispute using the internal Dispute resolution procedure, then Disputes which are not first resolved through the internal Dispute resolution procedure set forth in Section 3 above, must be submitted first to non-binding mediation in Mecklenburg County, North Carolina under the auspices of the American Arbitration Association ("AAA"), in accordance with AAA's Commercial Mediation Rules then in effect. A link to the AAA Commercial Arbitration Rules and Mediation Procedures can be found at https://www.adr.org/Rules. The mediation will be conducted by one mediator with at least seven years of substantive experience in franchise law who will be appointed under the American Arbitration Association's Commercial Mediation Rules and who will conduct the mediation in accordance with such rules. Mediation must be conducted on an individual, not a class-wide, basis; only Franchisor (and/or its affiliates and its and their respective owners, officers, directors, agents and employees) and Franchisee (and/or its Owners, guarantors, affiliates, officers, directors, agents and employees, if applicable) may be the parties to any mediation proceeding described in this Section, and no such mediation proceeding may be consolidated with any other mediation proceeding between Franchisor and any other person, corporation, limited liability company or partnership. Franchisor and Personal Guarantor agree that statements made by Franchisor, Personal Guarantor or any other party in any such mediation proceeding will not be admissible in any arbitration or other legal proceeding. If the mediation proceedings have been terminated either: (i) as the result of a written declaration of the mediator(s) that further mediation efforts are not worthwhile; or (ii) as a result of a if the mediation does not resolve the dispute by written declaration by Franchisoragreement within 14 days after the mediation, then, subject to the exclusions set forth in Section 4.1, all Disputes must be submitted to binding arbitration before one arbitrator of the AAA in accordance with its commercial arbitration rules. The arbitration will be conducted by one arbitrator with at least seven years of substantive experience in franchise law who will be appointed under the AAA's Commercial Arbitration Rules and who will conduct the arbitration in accordance with such rules. Except as provided in Section 4.1, no party may file litigation involving any Dispute unless it has complied with the mediation and arbitration provisions in this Article regarding that Dispute. "Filing litigation" includes asserting a counterclaim, third-party claim or other claim relating to a Dispute (irrespective of whether the party asserting the claim filed the legal action in which the claim is asserted). Franchisor's rights to mediation and arbitration, as set forth herein, may be specifically enforced by Franchisor. Each party shall bear its own cost of mediation and arbitration, and Franchisor and Personal Guarantor shall share mediation costs equally. This agreement to mediate and arbitrate shall survive any termination or expiration of this Guaranty.
- 4.1 The parties shall not be required to first attempt to mediate or arbitrate a controversy, Dispute or claim through mediation as set forth in this Section 4.1 if such controversy, Dispute or claim concerns an allegation that a party has violated (or threatens to violate, or poses an imminent risk of violating):
- 4.1.1 Any federally protected intellectual property rights in the Proprietary Marks, the System, or in any Confidential Information;
 - 4.1.2 Any claims pertaining to or arising out of any warranty issue;
 - 4.1.3 Any of the restrictive covenants contained in this Guaranty;

- 4.1.4 Any of Franchisee's payment obligations that are more than forty-five (45)
- 4.1.5 Any claims arising out of or related to fraud or misrepresentation by Personal Guarantor, Franchisee, or Franchisee's insolvency;
- 4.1.6 Any claims relating to Franchisee's obligations on termination or expiration of this Guaranty;
- 4.1.7 Any claims relating to any Transfer of an interest in Franchisee, the Franchised Business or Franchisee's assets; or
 - 4.1.8 Any matters involving danger, health or safety.
- 4.2 **Interpretation**. The provisions of this Guaranty must be construed as independent of any other covenant or provision of this Guaranty. However, if a court or arbitrator of competent jurisdiction determines that any provisions are unlawful, that court or arbitrator is to modify or interpret the provisions to the minimum extent necessary to have them comply with the law. Notwithstanding any provision relating to the state laws by and under which this Guaranty must be governed and construed, all issues relating to arbitrability or the enforcement of the agreement to arbitrate in this Guaranty must be governed by the United States Arbitration Act (9 U.S.C. § 1 et seq.) and the Federal common law of arbitration. The arbitrator, and not any federal, state or local court, shall have exclusive authority to resolve any Dispute relating to the interpretation, applicability, unconscionability, arbitrability, enforceability or formation of this Guaranty, including any claim of fraud in the inducement or that all or any part of the Guaranty is void or voidable. However, the preceding sentence shall not apply to the clause entitled "Class Action Waiver."
- 4.3 **Judgment**. Judgment on an arbitration award may be entered in any court of competent jurisdiction. This judgment is binding, final and non-appealable.
- 4.4 **Failure to Appear**. The arbitration and mediation provisions in this Article are self-executing and remain in full force and effect after the expiration or sooner termination of this Guaranty. If either party fails to appear at any properly noticed arbitration proceeding, notwithstanding failing to appear, an award may be entered against that party by default or otherwise.
- 4.5 Class Action Waiver. Any proceeding (whether mediation, arbitration, trial by a court or jury, appeal or otherwise) must be brought in the parties' individual capacity and not as a plaintiff or class member in any purported class, collective, representative, multiple plaintiff or similar proceeding ("Class Action"). Personal Guarantor and Franchisor expressly waive any ability to maintain any Class Action in any forum. Further, an arbitration proceeding between Franchisor and Personal Guarantor (or any of Franchisee's or Franchisor's affiliates and owners and guarantors) may not be consolidated with any other arbitration proceeding between them and any other franchisee, person or entity. Personal Guarantor hereby agrees not to seek joinder of any of its claims with those of any other party. The arbitrator shall not have authority to combine or aggregate similar claims or conduct any Class Action nor make an award to any person or entity not a party to the arbitration. Any claim that all or part of this Class Action Waiver is unenforceable, unconscionable, void or voidable may be determined only by a court of competent jurisdiction and not by an arbitrator. PERSONAL GUARANTOR AND FRANCHISOR UNDERSTAND THAT WETHEY WOULD HAVE HAD A RIGHT TO LITIGATE THROUGH A COURT, TO HAVE A JUDGE OR JURY DECIDE OUR CASE, AND TO BE PARTY TO A CLASS OR REPRESENTATIVE ACTION. HOWEVER, PERSONAL GUARANTOR AND FRANCHISOR UNDERSTAND AND

days past due;

CHOOSE TO WAIVE THAT RIGHT AND HAVE ANY CLAIMS DECIDED INDIVIDUALLY, THROUGH ARBITRATION. It's Personal Guarantor's and Franchisor's joint business judgment that the limitations of this subsection make good business sense, because:

- 4.5.1 the mediation and arbitration procedures contemplated by this Agreement (and which Personal Guarantor and Franchisor agree are the core methods for resolving Disputes) function most effectively on an individual case basis;
- 4.5.2 there are significant business and other factors present in each individual Personal Guarantor's situation which should be respected; and
- 4.5.3 the economic interests of lawyers on either side in class-wide or multiple plaintiff disputes, as well as the tendency to polarize positions, makes accommodation and compromise, as a practical business matter, less easily achieved, which would be a serious detriment to Personal Guarantor's and Franchisor's business interests, as well as those of the entire Franchise System, in quickly, amicably and economically resolving any Dispute.
- 4.6 **Selection of Venue**. Nothing contained in this Guaranty shall prevent Franchisor from applying to and obtaining from any court having jurisdiction, a writ of attachment, a temporary injunction, preliminary injunction and/or other emergency relief available to safeguard and protect Franchisor's interests. The parties expressly agree to the jurisdiction and venue of any court of general jurisdiction in Mecklenburg County, North Carolina and the jurisdiction and venue of the United States District Court for the Western District of North Carolina. Personal Guarantor acknowledges that this Guaranty has been entered into in the State of North Carolina, and that Franchisee, whose obligations Personal Guarantor is guaranteeing, is to receive valuable and continuing services emanating from Franchisor's headquarters in Huntersville, North Carolina, including, but not limited to, training, assistance, support and the development of the System. In recognition of such services and their origin, Personal Guarantor hereby irrevocably consents to the personal jurisdiction of the state and federal courts of North Carolina as set forth in this Section.
- 4.7 **Third Party Beneficiaries**. Franchisor's officers, directors, shareholders, members, agents and/or employees are express third party beneficiaries of the provisions of this Guaranty, including the mediation provision set forth in this Section 4, each having authority to specifically enforce the right to mediate claims asserted against such person(s) by Personal Guarantor.
- 4.8 **Prior Notice of Claims**. As a condition precedent to commencing an action for damages or for violation or breach of this Guaranty, Personal Guarantor must notify Franchisor within thirty (30) days after the occurrence of the violation or breach, and failure to timely give such notice shall preclude any claim for damages.
- 4.9 **No Right to Offset**. Personal Guarantor shall not withhold all or any part of any payment to Franchisor or any of its affiliates on the grounds of Franchisor's alleged nonperformance or as an offset against any amount Franchisor or any of Franchisor's affiliates allegedly may owe Franchisee under this Guaranty or any related agreements.
- 4.10 **Injunctive Relief.** The Franchised Business is intended to be one of a large number of businesses identified by the Proprietary Marks selling to the public the Approved Products and Approved Services associated with the Proprietary Marks. Consequently, a single Personal Guarantor's failure to comply with the terms of its franchise agreement is likely to cause irreparable damage to us, and damages at law would therefore be an inadequate remedy. On Personal Guarantor's breach or threatened breach of any of the terms concerning any matters referenced in Section 4.1, weFranchisor may seek an injunction

restraining the breach and/or a decree of specific performance (with recovery of reasonable attorneys' fees and costs incurred in obtaining equitable relief). WeFranchisor may do so without demonstrating or proving any actual damage. These equitable remedies are in addition to all other rights or remedies to which weFranchisor may otherwise be entitled because of Personal Guarantor's breach of this Guaranty. WeFranchisor may seek this relief without posting any bond or security. However, if a court of competent jurisdiction requires a bond or security, a bond or security for \$1,000 is sufficient. Notwithstanding anything in this Guaranty to the contrary, Franchisor may seek injunctive relief in any jurisdiction that has jurisdiction over Personal Guarantor or any other party against whom the relief is sought. If injunctive relief is granted, Personal Guarantor's only remedy will be the court's dissolution of the injunctive relief. If the injunctive relief was wrongfully issued, Personal Guarantor expressly waives all claims for damages Personal Guarantor incurred as a result of the wrongful issuance.

- 5. **Jurisdiction and Venue**. Nothing contained in this Guaranty shall prevent Franchisor from applying to and obtaining from any court having jurisdiction, a writ of attachment, a temporary injunction, preliminary injunction, and/or other emergency relief available to safeguard and protect Franchisor's interests. The parties expressly agree to the jurisdiction and venue of any court of general jurisdiction in Mecklenburg County, North Carolina and the jurisdiction and venue of the United States District Court for the Western District of North Carolina. The parties acknowledge and agree that this Guaranty has been entered into in the State of North Carolina, and that Franchisee, whose obligations Personal Guarantor is guaranteeing, is to receive valuable and continuing services emanating from Franchisor's headquarters in Huntersville, North Carolina, including, but not limited to, training, assistance, support and the development of the System. In recognition of such services and their origin, you hereby irrevocably consent to the personal jurisdiction of the state and federal courts of North Carolina as set forth herein.
- 6. Jury Trial Waiver. THE PARTIES HEREBY AGREE TO WAIVE TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM, WHETHER AT LAW OR EQUITY, REGARDLESS OF WHICH PARTY BRINGS SUIT. THIS WAIVER SHALL APPLY TO ANY MATTER WHATSOEVER BETWEEN THE PARTIES HERETO WHICH ARISES OUT OF OR IS RELATED IN ANY WAY TO THIS PERSONAL GUARANTY OR THE FRANCHISE AGREEMENT, THE PERFORMANCE OF EITHER PARTY, AND/OR YOUR PURCHASE FROM FRANCHISOR OF THE FRANCHISE, OPTION AND/OR ANY GOODS OR SERVICES.
- Waiver of Punitive Damages. YouThe parties hereby waive; to the fullest extent permitted by law, any right to or claim for any punitive, exemplary, incidental, indirect, special—or consequential damages (including, without limitation, lost profits) which you may have against Franchisor arising out of any cause whatsoever (whether such cause be based in contract, negligence, strict liability, other tort or otherwise) and agree that in the event of a Dispute, your recovery shall be limited to actual damages. If any other term of this Guaranty is found or determined to be unconscionable or unenforceable for any reason, the foregoing provisions shall continue in full force and effect, including, without limitation, the waiver of any right to claim any consequential damages.
- 7. Limitation on Action., or consequential damages (including, without limitation, lost profits) against the other party arising out of any cause whatsoever (whether such cause be based in contract, negligence, strict liability, other tort or otherwise) except for indemnified claims described in 13.2 of the Franchise Agreement and punitive or exemplary damages for violation of the Lanham Act, Trademark or Copyright Infringement or Trademark Dilution, unauthorized dissemination of the confidential information or trade secrets.
- 7-8. <u>Limitation on Action.</u> You agree that no cause of action arising out of or under this Guaranty or the Franchise Agreement may be maintained by you unless brought before the expiration of

one (1) year after the act, transaction or occurrence upon which such action is based or the expiration of one (1) year after you become aware of facts or circumstances reasonably indicating that you may have a claim against the Franchisor, whichever occurs sooner, and that any action not brought within this period shall be barred as a claim, counterclaim, defense or set-off. Notwithstanding anything to the contrary contained herein, all actions will be conducted on an individual, not a class wide, basis and any proceeding between you, Franchisor or its affiliates or employees may not be consolidated with any other proceeding between Franchisor and any other person or entity.

8-9. Attorneys' Fees. If the undersigned is in breach or default of any monetary or non-monetary material obligation under the Franchise Agreement or this Guaranty, or any related agreement between Franchisee and Franchisor and/or Franchisor's affiliates, and Franchisor engages an attorney to enforce Franchisor's rights (whether or not formal judicial proceedings are initiated), Franchisee must pay all reasonable attorneys' fees, court costs and litigation expenses Franchisor incurs. If the undersigned or Franchisee institutes any legal action to interpret or enforce the terms of this Guaranty or the Franchise Agreement, and the claim(s) in such action is denied or the action is dismissed, Franchisor is entitled to recover Franchisor's reasonable attorneys' fees, and all other reasonable costs and expenses incurred in defending against same, and to have such an amount awarded as part of the judgment in the proceeding.

9-10. Nonwaiver. Franchisor's failure to insist upon strict compliance with any provision of this Guaranty and the Franchise Agreement shall not be a waiver of Franchisor's right to do so, any law, custom, usage or rule to the contrary notwithstanding. Delay or omission by Franchisor respecting any breach or default shall not affect Franchisor's rights respecting any subsequent breaches or defaults. All rights and remedies granted in this Guaranty shall be cumulative. Franchisor's election to exercise any remedy available by law or contract shall not be deemed a waiver or preclude exercise of any other remedy.

10.11. Severability. The parties agree that if any provisions of this Guaranty may be construed in two ways, one of which would render the provision illegal or otherwise voidable or unenforceable and the other which would render it valid and enforceable, such provision shall have the meaning that renders it valid and enforceable. The language of all provisions of this Guaranty shall be construed according to fair meaning and not strictly construed for or against either party. The provisions of this Guaranty are severable, and this Guaranty shall be interpreted and enforced as if all completely invalid or unenforceable provisions were not contained herein, and partially valid and enforceable provisions shall be enforced to the extent that they are valid and enforceable. If any material provision of this Guaranty shall be stricken or declared invalid, the parties agree to negotiate mutually acceptable substitute provisions. In the event that the parties are unable to agree upon such provisions, Franchisor reserves the right to terminate this Guaranty.

41-12. Construction of Language. Any term defined in the Franchise Agreement that is not defined in this Guaranty will be ascribed the meaning given to it in the Franchise Agreement. The language of this Guaranty will be construed according to its fair meaning, and not strictly for or against either party. All words in this Guaranty refer to whatever number or gender the context requires. If more than one party or person is referred to as you, their obligations and liabilities must be joint and several. Headings are for reference purposes and do not control interpretation.

<u>12-13.</u> **Successors**. References to "Franchisor" or "the undersigned," or "you" include the respective parties' successors, assigns or transferees.

13.14. No Personal Liability. You agree that fulfillment of any and all of Franchisor's obligations written in this Guaranty or in the Franchise Agreement or based on any oral communications which may be ruled to be binding in a Court of Law shall be Franchisor's sole responsibility and none of

Franchisor's agents, representatives, nor any individuals associated with Franchisor's franchise company shall be personally liable to Franchisee or you for any reason.

 $(THE\ REMAINDER\ OF\ THIS\ PAGE\ IS\ INTENTIONALLY\ LEFT\ BLANK.\\ SIGNATURES\ APPEAR\ ON\ \underline{THE}\ FOLLOWING\ PAGE.)$

IN WITNESS WHEREOF, AND INTENDING TO BE LEGALLY BOUND HEREBY, THE PARTIES HERETO HAVE CAUSED THIS GUARANTY TO BE EXECUTED EFFECTIVE THE DATE FIRST SET FORTH ABOVE.

PERSONAL GUARANTORS:	SPOUSES:		
Ву:	By:		
Print Name:	Print Name:		
Address:	Address:		
Telephone:			
Ву:	By:		
Print Name:	Print Name:		
Address:	Address:		
Telephone:			

EXHIBIT B TO HFB FENCECO FRANCHISING, LLC FRANCHISE AGREEMENT

$\frac{\textbf{CONDITIONAL ASSIGNMENT OF FRANCHISEE'S TELEPHONE NUMBERS,}}{\textbf{FACSIMILE NUMBERS AND DOMAIN NAMES}}$

1	("Assignor"),
of which is hereby acknowledged, hereby condit facsimile numbers, domain names, as well as any lis	HFB FenceCo Franchising, LLC ("Assignee"), receipt ionally assigns to Assignee all telephone numbers, stings associated therewith, utilized by Assignor in the above referenced address (the "Assigned Property").
Telephone Number(s):	
Facsimile Number(s):	
Domain/Social Media Name(s) (as permitted by Fra	nchisor under the Franchise Agreement):
("Assignment") will become effective automatically Upon the occurrence of that condition, Assignor mand/or domain name registrar to assure the effective Assignee had been originally issued such Assigned 3. Assignor agrees to pay the telephon the effective date of assignment, all amounts owed to Assignment becomes effective. Assignor further agrees	e company and/or domain name registrar, on or before for the use of the Assigned Property up to the date this ees to indemnify Assignee for any sums Assignee must ment, and agrees to fully cooperate with the telephone
ASSIGNOR:	ASSIGNEE:
(Individual, Partnership or Corporation Name)	HFB FENCECO FRANCHISING, LLC
Ву:	Ву:
Name:	Name:
Title:	Title:
Date:	Date:

EXHIBIT C

TO HFB FENCECO FRANCHISING, LLC FRANCHISE AGREEMENTExhibit C

to FDD
HFB FenceCo Franchising, LLC

STATEMENT OF OWNERSHIP

Franchisee:							
Form of Ownership (Check One)							
☐ Individual ☐ Limited Liability Company ☐ Limited Partnership							
□ Corporation □	□ Corporation □ General Partnership						
If a Partnership, provide name and address of each partner showing percentage owned, whether active in management, and indicate the state in which the partnership was formed.							
If a Corporation, provide the state and date of incorporation, the names and addresses of each officer and director, and list the names and addresses of every shareholder showing what percentage of stock is owned by each.							
If a Limited Liability Company, provide the state and date of formation, the name and address of the manager(s), and list the names and addresses of every member and the percentage of membership interest held by each member.							
State and Date of Formation/Incorp	oration:						
Management (managers, officers, board of directors, etc.):							
<u>Name</u>			<u>Title</u>				
Members, Stockholders, Partners:							
<u>Name</u>	Add	ress	Percentage Owned				

This form is current and complete as of	, 20
	FRANCHISEE:
Date:	By:
240.	Printed Name: Title:

Exhibit D To FDD HFB FenceCo Franchising, LLC

CONFIDENTIALITY AND RESTRICTIVE COVENANT AGREEMENT

(for trained employees, shareholders, officers, directors, general partners, members and managers of Franchisee)

In consideration of my being a	of
(the "Franchisee"), and other good and valuable con	sideration, the receipt and sufficiency of which is hereby
acknowledged, I hereby acknowledge and agree th	at Franchisee has acquired the right and franchise from
HFB FenceCo Franchising, LLC (the "Company	") to establish and operate a franchised business (the
"Franchised Business") and the right to use, in the	e operation of the Franchised Business, the Company's
trade names, trademarks and service marks (the	"Proprietary Marks") and the Company's unique and
distinctive format and system relating to the establishment	plishment and operation of Franchised Businesses (the
"System"), as they may be changed, improved and	further developed from time to time in the Company's
sole discretion, only at the following authorized ar	d approved location:
	(the "Franchised Business Premises").

- 1. The Company possesses certain proprietary and confidential information and trade secrets relating to the operation of the System and the Franchised Business, including the Company's Manuals; price lists and standards and specifications for the Approved Products and Approved Services; standards and specifications related to the Company's integrated bookkeeping system; and other methods, techniques and know-how concerning the of operation of the Franchised Business which may be communicated to Franchisee or of which I may be apprised by virtue of my employment with Franchisee ("Confidential Information").
- 2. Any and all information, knowledge, know-how, and techniques which the Company specifically designates as confidential shall be deemed to be Confidential Information for purposes of this Confidentiality and Restrictive Covenant Agreement ("Agreement").
- 3. In my position with the Franchisee, the Company and Franchisee will disclose the Confidential Information to me in furnishing to me the training program and subsequent ongoing training, the Company's Operations Manual (the "Operations Manual") and other general assistance during the term of this Agreement.
- 4. I will not acquire any interest in the Confidential Information, other than the right to utilize it in the operation of the Franchised Business during the term hereof, and the use or duplication of the Confidential Information for any use outside the System would constitute an unfair method of competition.
- 5. The Confidential Information is proprietary, involves trade secrets of the Company, and is disclosed to me solely on the condition that I agree, and I do hereby agree, that I shall hold in strict confidence all Confidential Information and all other information designated by the Company as confidential. Unless the Company otherwise agrees in writing, I will disclose and/or use the Confidential Information only in connection with my duties with the Franchisee, and will continue to not disclose any such information even after I cease to be in that position and will not use any such information even after I cease to be in that position unless I can demonstrate that such information has become generally known or easily accessible other than by the breach of an obligation of Franchisee under the Franchise Agreement.

- 6. Except as otherwise approved in writing by the Company, I shall not, while in my position with the Franchisee, for myself, or through, on behalf of, or in conjunction with any person, persons, partnership, corporation or limited liability company, own, maintain, engage in, be employed by, or have any interest in any other business which operates or licenses any other competing business(es)—) within 25 miles of Franchisee's territory and the territory of any other franchisee in the system which information is available upon request, except for a Franchised Business operating under the System and Proprietary Marks.
- 7. I agree that each of the foregoing covenants shall be construed as independent of any other covenant or provision of this Agreement. If all or any portion of a covenant in this Agreement is held unreasonable or unenforceable by a court or agency having valid jurisdiction in an un-appealed final decision to which the Company is a party, I expressly agree to be bound by any lesser covenant subsumed within the terms of such covenant that imposes the maximum duty permitted by law, as if the resulting covenant were separately stated in and made a part of this Agreement.
- 8. I understand and acknowledge that the Company shall have the right, in its sole discretion, to reduce the scope of any covenant set forth in this Agreement, or any portion thereof, without my consent, effective immediately upon receipt by me of written notice thereof, and I agree to comply forthwith with any covenant as so modified.
- 9. The Company is a third-party beneficiary of this Agreement and may enforce it, solely and/or jointly with the Franchisee. I am aware that my violation of this Agreement will cause the Company and the Franchisee irreparable harm; therefore, I acknowledge and agree that the Franchisee and/or the Company may apply for the issuance of an injunction preventing me from violating this Agreement, and I agree to pay the Franchisee and the Company all the costs it/they incur(s), including, without limitation, legal fees and expenses, if this Agreement is enforced against me. Due to the importance of this Agreement to the Franchisee and the Company, any claim I have against the Franchisee or the Company is a separate matter and does not entitle me to violate, or justify any violation of, this Agreement.
- 10. This Agreement shall be construed under the laws of the State of North Carolina. The only way this Agreement can be changed is in writing signed by both the Franchisee and me.

D.,.

Бу
Name:
Title:
Date:
ACKNOWLEDGED BY FRANCHISEE:
Ву:
Name:

EXHIBIT D

TO

 $\underline{Exhibit\ E}$

to FDD

HFB FENCECO FRANCHISING FenceCo Franchising, LLC

ELECTRONIC FUNDS WITHDRAWAL AUTHORIZATION

Bank Name:
Routing #:
Acct. No.:
Entity Name on Acct.:
Effective as of the date of the signature below, ("Franchisee"), hereby authorizes HFB FenceCo Franchising, LLC ("Company"), or its designee, to withdraw funds from the above-referenced bank account, electronically or otherwise, and to make the following payments to Company under the Franchise Agreement for the Franchised Business located at:
(i) all Royalty Fees; (ii) all Brand Fund Contributions, Technology Fees, Call Center Fees or other recurring fees; and (iii) any other fees due and owing under the Franchise Agreement. Franchisee acknowledges that Royalty Fees and all other fees may be collected by Franchisor in the manner provided for in the Franchise Agreement. Such withdrawals shall occur on a monthly basis, or on such other schedule as Company shall specify in writing. Company is also authorized to deposit the Gross Revenue of Franchisee's Franchisee Business, less all amounts due under the Franchise Agreement, into the above-referenced account electronically or otherwise. Such deposits shall occur on a monthly basis, or on such other schedule as Company shall specify in writing. This authorization shall remain in full force and effect until terminated in writing by Company. Franchisee shall provide Company, in conjunction with this authorization, a voided check from the above-referenced account.
FRANCHISE AGREEMENT
ELECTRONIC FUNDS WITHDRAWAL AUTHORIZATION
Bank Name:
ABA#:
Acet. No.:
Acct. Name:

Effective as of the date of the signature below, _ ("Franchisee"), hereby authorizes HFB FenceCo Franchising, LLC ("Company"), or its designee, to withdraw funds from the above-referenced bank account, electronically or otherwise, and to make the following payments to Company under the Franchise Agreement for the Franchised Business located at: (i) all Royalty Fees; (ii) all Brand Fund Contributions, Technology Fees, Call Center Fees or other recurring fees; any (iii) and other fees due and owing under the Franchise Agreement. Franchisee acknowledges that Royalty Fees and all other fees may be collected by Franchisor in the manner provided for in the Franchise Agreement. Such withdrawals shall occur on a monthly basis, or on such other schedule as Company shall specify in writing. Company is also authorized to deposit the Gross Revenue of Franchisee's Franchised Business, less all amounts due under the Franchise Agreement, into the above referenced account, electronically or otherwise. Such deposits shall occur on a monthly basis, or on such other schedule as Company shall specify in writing. This authorization shall remain in full force and effect until terminated in writing by Company. Franchisee shall provide Company, in conjunction with this authorization, a voided check from the above referenced account. AGREED: FRANCHISOR: FRANCHISEE: HFB FENCECO FRANCHISING, LLC (Individual, Partnership or Corporation Name) Name: Date: Date: **FRANCHISEE:** (Individual, Partnership or Corporation Name) By: Name: Title:

Date:

Exhibit F to FDD HFB FenceCo Franchising, LLC

MULTI-UNIT ADDENDUM

MULTI-UNIT ADDENDUM

	This	Multi-Unit	Addendum	(the	"Addendum")	is	made	and	entered	into	01
		(t)	ne "Effective	Date")	, by and between	: (i)	HFB Fe	nceCo	Franchisi	ng, LL	С, а
North (Carolina	a limited liab	ility company	with a	a business addres	s at	107 Pari	Drive	, Huntersy	ille, N	ortl
Carolir	na 28078	3 ("Franchiso	r"); and (ii)								
a(n)			with a	busin	ess address at						
("Franc	hisee")										

BACKGROUND

EXHIBIT E TO HFB FENCECO FRANCHISING, LLC FRANCHISE AGREEMENT

SITE SELECTION ADDENDUM

HFB FenceCo Franchising, LLC, a North Carolina limited liability company with a principal
business address at 107 Parr Drive, Huntersville, North Carolina 28078 ("Franchisor") and ("Franchisee"), have on (the "Effective Date"),
entered into the foregoing Franchise Agreement for the operation of a franchised business using
Franchisor's Proprietary Marks and System (the "Franchised Business") and desire to supplement its terms,
as set forth below. The parties therefore agree as follows:
1. Within ninety (90) days after Franchisee receives notice of approval of the Franchise
Agreement, Franchisee must obtain a site, at Franchisee's expense, for the Franchised Business, which
Franchisor will approve as hereinafter provided. The site must be within the Designated Territory set forth
in the Data Sheet of the Franchise Agreement.
2. Franchisee's failure to obtain a site for the Franchised Business within the time required
in Paragraph 1 will constitute a default under the Franchise Agreement and this Site Selection Addendum ("Addendum"). Time is of the essence.
3. Prior to Franchisee's acquisition by lease or purchase of a site for the Franchised Business, Franchisee must submit to Franchisor in the form Franchisor specifies, a completed site review form, such other information or materials as Franchisor may reasonably require, and a letter of intent or other evidence satisfactory to Franchisor that confirms Franchisee's favorable prospects for obtaining the proposed site. Recognizing that time is of the essence, Franchisee must submit a proposed site, together with the information and materials required by this Paragraph 3, to Franchisor for Franchisor's approval within ninety (90) days after execution of this Addendum. Franchisor will have thirty (30) days after receipt of such information and materials from Franchisee to approve or disapprove, in Franchisor's sole discretion, the site as a location for the Franchised Business. No proposed site will be deemed approved unless Franchisor has expressly approved it in writing.
4. Franchisor will furnish to Franchisee such site selection guidelines, consultation and on- site evaluation as Franchisor deems advisable as part of Franchisor's evaluation of Franchisee's request for site approval. Franchisor will not, however, provide on site evaluation for any proposed site prior to Franchisor's receipt of the information and materials required by Paragraph 3 hereof. If Franchisor deems on-site evaluation necessary and appropriate, Franchisor will conduct one (1) on-site evaluation at

Franchisor's cost. For each additional on-site evaluation (if any), Franchisee will reimburse Franchisor for Franchisor's expenses, including, without limitation, the costs of travel, lodging and meals. 5. If Franchisee will be occupying the Franchised Business premises under a lease, Franchisee shall, upon Franchisor's request, prior to the execution of the lease, submit the lease to Franchisor for Franchisor's approval. Franchisor's approval of the lease may be conditioned upon Franchisee's execution of a Collateral Assignment of Lease in the form Franchisor prescribes, as well as the inclusion or exclusion of certain required provisions. Franchisee must furnish Franchisor with a copy of any executed lease within ten (10) days after execution thereof. After Franchisor has approved a site for the Franchised Business in writing and Franchisee acquired the site, the site will constitute the Approved Location referred to in Section 1.3 of the Franchise Agreement. Franchisee hereby acknowledges and agrees that Franchisor's approval of a site does not constitute an assurance, representation or warranty of any kind, express or implied, as to the suitability of the site for the Franchised Business or for any other purpose. Franchisor's approval of the site indicates only that Franchisor believes the site complies with acceptable minimum criteria established by Franchisor solely for Franchisor's purposes as of the time of the evaluation. Both parties to this Agreement acknowledge the application of criteria that have been effective with respect to other sites and premises may not be predictive of potential for all sites and that, subsequent to Franchisor's approval of a site, demographic and/or economic factors such as competition from other similar businesses included in or excluded from Franchisor's criteria could change, thereby altering the potential of a site. Such factors are unpredictable and are beyond Franchisor's control. Franchisor will not be responsible for the failure of a site approved by Franchisor to meet Franchisee's expectations as to revenue or operational criteria. Franchisee further acknowledges and agrees that Franchisee's acceptance of a franchise for the operation of the Franchised Business at the site is based on Franchisee's own independent investigation of the This Addendum constitutes an integral part of the Franchise Agreement between the parties hereto, and terms of this Addendum will be controlling with respect to the subject matter hereof. Except as modified or supplemented by this Addendum, the terms of the Franchise Agreement are hereby ratified and confirmed. IN WITNESS WHEREOF, the parties hereto have duly executed this Addendum on the day and vear first above written. **FRANCHISEE:** (Individual, Partnership or Corporation Name)

© 2023 HFB FenceCo Franchising, LLC

2023 FDD 1 v1F (MN)

FRANCHISOR:

HFB FENCECO FRANCHISING, LLC					
By:					
Name:					
Title:					

EXHIBIT F TO HFB FENCECO FRANCHISING, LLC FRANCHISE AGREEMENT

STATEMENT OF OWNERSHIP

Franchisect				
	Form of C	hwnorchin		
	(Chec)	_		
☐ Individual ☐	Sole Proprieto	rship	Limited Liability Company	
☐ Corporation ☐	General Partne	ership ===	Limited Partnership	
If a Partnership, provide ractive in management, and indicate	name and address	of each partne the partnersh	r showing percentage owned, wheth	
-		_	the names and addresses of each office	
and director, and list the names ar owned by each.	nd addresses of ev	ery shareholde	er showing what percentage of stock	
If a Limited Liability Con	pany, provide the	state and date	of formation, the name and address	
the manager(s), and list the name interest held by each member.	s and addresses o	f every memb	per and the percentage of membersh	
merest here by each member.				
State and Date of Formation/Incorp	poration:			
Management (managers, officers	, board of directo	ers, etc.):		
Name			Title	
M. 1. (4. 11.11. P. 4				
Members, Stockholders, Partner	91			
Name	Add	ress	Percentage Owned	
	I		I	

	er. Your managing owner as of the Effective Date is You may not change the managing owner.
without prior written approval.	
Identification of Designated N	nager. Your Designated Manager, if applicable, as of the Effective
Date is	. You may not change
the Designated Manager withou	rior written approval.
This form is current and	emplete as of .20 .
	FRANCHISEE:
Date:	
	Printed Name:
	Title:

EXHIBIT G TO HFB FENCECO FRANCHISING, LLC FRANCHISE AGREEMENT

COLLATERAL ASSIGNMENT OF LEASE

COLLATERAL ASSIGNMENT OF LEASE

Landlord:	Franchisor: HFB FenceCo Franchising, LLC
Notice Address:	Notice Address: 107 Parr Drive Huntersville, NC 28078
Telephone:	
Tenant:	
Leased Premises:	
1. <u>Use.</u> Tenant is a franchisee of F operation of a Top Rail business (or any name at	ranchisor. The Leased Premises shall be used only for the uthorized by Franchisor).
of any written notice of default ("Default") gi	ity to Cure. Landlord shall provide Franchisor with copies iven to Tenant under the Lease, and Landlord grants to cure any Default under the Lease (should Tenant fail to do no period in which Tenant may cure the Default.
Lease. If Landlord terminates the Lease for Tena a new Lease with Landlord on the same terms	I shall copy Franchisor on any notice of termination of the nnt's Default, Franchisor shall have the option to enter into and conditions as the terminated Lease. To exercise this 15 days after Franchisor receives notice of the termination
Tenant is terminated during the term of the Lease	ment. If the Franchise Agreement between Franchisor and e, then upon the written request of Franchisor, Tenant shall consents to the assignment of the Lease to Franchisor.
Tenant shall have the right to assign or sublet the sublease shall relieve Tenant or any guarantor of	otwithstanding any provision of the Lease to the contrary, a Lease to Franchisor, provided that no such assignment or cliability under the Lease. If Franchisor becomes the lessee are the right to assign or sublease its lease to a franchisee
6. <u>Authorization</u> . Tenant authorizeach other about Tenant and Tenant's business.	es Landlord and Franchisor to communicate directly with
or the Lease, or the termination of Tenant's right designee may, after giving reasonable prior notice and other material bearing Franchisor's brand n	the expiration or termination of the Franchise Agreement ht of possession of the Leased Premises, Franchisor or its ce to Landlord, enter the Leased Premises to remove signs ame, trademarks, and commercial symbols, provided that mage Franchisor or its designee causes by such removal.
	ris Collateral Assignment of Lease, Franchisor does not Premises or any obligation as Tenant under the Lease.

Executed by:

LANDLORD:
Ву:
Name:
Title:
Date:
TENANT:
By:
Name:
Title:
Date:
VD A NOW GOD
FRANCHISOR:
HFB FENCECO FRANCHISING, LLC
Ву:
Name:
Title:
Date:

EXHIBIT H TO HFB FENCECO FRANCHISING, LLC FRANCHISE AGREEMENT

FRANCHISOR DIRECTED ADVERTISING SERVICES AGREEMENT

This-Franchisor Directed Advertising Services Agreement ("FDASA") is entered into as of
, 20, by and between HFB FenceCo Franchising, LLC, a North Carolina limited liability company, with its principal place of business at 107 Parr Drive, Huntersville, North Carolina 28078 ("we," "us," "our" or "Franchisor") and
, a(n) with its principal place of business at ("you" or "Franchisee").
RECITALS
A. Franchisee and Franchisor are parties to that certain Franchise Agreement dated the date hereof (the "Franchise Agreement").
B. Pursuant to the terms of the Franchise Agreement, Franchisee is required to pay for Local Advertising Services (defined below) from Franchisor.
AGREEMENT
NOW THEREFORE, for good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties agree as follows:
1. Any capitalized terms used, but not defined herein, have the same meaning set forth in the Franchise Agreement.
2. <u>Term.</u> The FDASA will remain in effect indefinitely and will automatically terminate upon the expiration or termination of the Franchise Agreement.
3. <u>Default</u> . If you are in default of your obligations under the FDASA or the Franchise Agreement, we may suspend some or all of the Local Advertising Services without terminating either agreement. This remedy is in addition to any other remedies we may have at law or in equity.
4. <u>Termination</u> . We can terminate the FDASA prior to the end of the Term by written notice if: (i) you are in default under the Franchise Agreement or any other agreement with Franchisor or an affiliate of Franchisor; (ii) you are in default under the FDASA and the default continues for a period of ten (10) days after we deliver written notice of default; or (iii) we discontinue the Local Advertising Services under Section 8 of this FDASA.
5. Franchisor Directed Advertising Services. You and we agree that as Franchisor we have the right, in light of our marketing expertise and research regarding effective advertising of the System: (a) to formulate and design the Local Advertising Services (defined below), including the service options and vendors made available to franchisees from time to time; (b) to develop and control all content for the Local Advertising Services (the "Advertising"); and (c) to determine the uses and distribution of all such Advertising. We have the right to administer each service, as well as other services we elect to offer in the

future, which may include appointment generation platforms, billboards, radio ads, etc. We can change or end a service at any time and add new services at any time.

"Local Advertising Services" means our advertising of TOP RAILSM brand, the Services offered under the Proprietary Marks, and the System, as conducted in your Designated Territorry in coordination with our similar advertising service in the territories of other franchisees and nationally, and includes the several channels and forms of local marketing activities and services that we (or a designated vendor) offer from time to time and perform pursuant to this FDASA, which activities and services may include, but are not limited to: media management, advertising design, digital marketing for generation of customer leads, digital marketing for generation of customer leads, digital marketing for generation of customer reviews, tracking of leads, performance analytics, production and printing of marketing pieces, list management, mail management, market studies, strategy and analysis, research and development, shipping and delivery to the United States Postal Service or other carrier, other services we offer in our Business Judgment, and related administrative services.

Franchisor Directed Advertising Fee. For our Local Advertising Services, you agree to pay us or an affiliate we designate a non refundable fee (the "Franchisor Directed Advertising Fee") in an amount up to the Local Advertising Requirement, currently the greater of (\$3,000 per month or 3% of Gross Revenue per month). We reserve the right to designate the timing of payment and to set other policies relating to the Franchisor Directed Advertising Fee, but the fee will continue to be due monthly by the tenth (10^{th)} of the month. For example, although we have no obligation to do so, we may permit you to: (i) vary the monthly payment, subject to satisfaction of the annual amount required; and/or (ii) allocate the required amount among marketing channels that we offer. If not otherwise designated by us, you must pay the Franchisor Directed Advertising Fee at the same time you pay your Royalty Fee under the Franchise Agreement. You must pay the Franchisor Directed Advertising Fee by electronic funds transfer or such other method as we may designate from time to time. We and our affiliates may earn revenue and profits on products or services we provide and may receive relates licensing fees administrative fees commissions, or other payments on products and services that third party vendors provide. The Franchisor Directed Advertising Fee is credited to your required monthly Local Advertising Requirement. If local advertising directed by us is paid with a franchisee credit card (or other method of direct payment), all such amounts spent that are directed by us will be credited to the monthly Franchisor Directed Advertising Fee.

Changes to Franchisor Directed Advertising Fee. We may periodically increase or decrease the Franchisor Directed Advertising Fee in an amount we deem appropriate in our Business Judgment, but in no event will that increase cause the Franchisor Directed Advertising Fee to be more than the Local Advertising Requirement. Further, we reserve the right to change the Franchisor Directed Advertising Fee from a dollar amount to a percentage of sales or to any other formula we designate. We will give you thirty (30) days' advance notice by email or other means if we intend to change the Franchisor Directed Advertising Fee pursuant to this Section.

7. No Guaranteed Results. We do not represent or warrant that the Local Advertising Services or any individual service will reach a minimum number of persons in your Designated Territory, will reach any specific persons in the Designated Territory, or will produce any particular results for your Franchised Business. We obtain distribution lists and other data from third party data compilation and demographic information service provider(s) that we and/or our affiliates select. We make no representation or warranty regarding the accuracy of any such lists or other data we use to conduct the Local Advertising Services.

8. <u>Ownership.</u> You acknowledge and agree that, as between Franchisor and Franchisee, Franchisor owns all right, title and interest in and to any Advertising created in or for the Local Advertising Services, including all intellectual property rights therein or appurtenant thereto. Franchisee will not acquire any right, title or interest in the Advertising or any portion thereof pursuant to this FDASA or otherwise,

9. Discontinuation or Replacemen	t of Local Advertising Services. We have the right, in
our sole business judgment, to discontinue the Local	al Advertising Services, in whole or in part, and to require
you to pay all or a portion of the amount of the Fr	anchisor Directed Advertising Fee to one or more other
	ne place of the Local Advertising Services. In the event
that we discontinue the Local Advertising Services	s or replace it with an entirely new service, we will seek
feedback from franchisees, which feedback will be	e advisory only.
10. Assignment. The FDASA is full	ly assignable by Franchisor. Franchisee may not assign
the FDASA without the prior written consent of Fr	
IN WITNESS WHEREOF, the parties her representatives as indicated below. FRANCHISOR:	eto have executed this FDASA by their duly authorized FRANCHISEE:
HFB FENCECO FRANCHISING, LLC	
	(Individual, Partnership or Corporation Name)
By:	Ву:
Name:	Name:
Title:	Title:
Deter	Data

other than the right to access and use the Advertising as expressly granted in the Franchise Agreement,

subject to the terms and conditions of the Franchise Agreement and this FDASA.

EXHIBIT C TO HFB FENCECO FRANCHISING, LLC FRANCHISE DISCLOSURE DOCUMENT

MULTI UNIT ADDENDUM

MULTI-UNIT ADDENDUM

b. For purposes of this Addendum, if a capitalized term in this Addendum is not specifically defined herein, that term will be given the same definition that the term is afforded in the Applicable Franchise Agreements, as applicable. A. 2. Multi-Unit Fee in Lieu of Initial Franchise Fee. Notwithstanding anything contained in Section 3.1 of the Applicable Franchise Agreements, Franchisee will not be required to pay Franchisor an Initial Franchise Fee under any Applicable Franchise Agreement. Instead, Franchisee must pay Franchisor a lump-sum multi-unit fee amounting to \$ (the "Multi-Unit Fee") (collectively, the "Applicable Franchise Agreements") and, under each such Applicable Franchise	This Multi-Unit Addendum (the "Addendum") is made and entered into
Contemporaneous with the execution of this Addendum, Franchisee and Franchise entered into (
BACKGROUND A. Contemporaneous with the execution of this Addendum, Franchisee and Franchiso entered into() franchise agreements TR (collectively, the "Applicable Franchise Agreement, Franchisee obtained the right and under each such Applicable Franchise Agreement, Franchisee obtained the right and undertook the obligation to operate a franchised business under Franchisor's then-current proprietar mark(s) (each, a "Franchised Business"). B. Each Franchised Business has its own designated territory wherein Franchisee is required to actively promote and operate the Franchised Businesses (each, a "Designated Territory" and collectively the "Designated Territories"). C. Franchisor and Franchisee now wish to amend and otherwise clarify certain provisions in the Applicable Franchise Agreements, pursuant to the terms and conditions of this Addendum. AGREEMENT NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties agree as follows: 1. Background; Definitions. a. The parties acknowledge and agree that the Background portion of this Addendum including all definitions, representations and provisions set forth therein, is hereby incorporated by reference as if set forth in this Section. For purposes of this Addendum, if a capitalized term in this Addendum is no specifically defined herein, that term will be given the same definition that the term is afforded in the Applicable Franchise Agreements, as applicable. A. 2. Multi-Unit Fee in Lieu of Initial Franchise Fee. Notwithstanding anything contained in Section 3.1 of the Applicable Franchise Agreements, Franchisee will not be required to pay Franchisor an Initial Franchise Fee under any Applicable Franchise Agreement. Instead, Franchise agreement, Franchise education, the "Applicable Franchise Agreements" and under cook such Applicable Franchise development for the properties of the courted proprietary mark(c) (each, a "Franchised Business"). B. Cook Franchised Business has its own design	
AContemporaneous with the execution of this Addendum, Franchisee and Franchiso entered into (// //
AContemporaneous with the execution of this Addendum, Franchisee and Franchise entered into() franchise agreements TR (collectively, the "Applicable Franchise Agreements") and, under each such Applicable Franchise Agreement, Franchisee obtained the right and undertook the obligation to operate a franchised business under Franchiser's then-current proprietary mark(s) (each, a "Franchised Business"). B Each Franchised Business has its own designated territory wherein Franchisee is required to actively promote and operate the Franchised Businesses (each, a "Designated Territorise"). C Franchisor and Franchisee now wish to amend and otherwise clarify certain provisions in the Applicable Franchise Agreements, pursuant to the terms and conditions of this Addendum. AGREEMENT NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties agree as follows: 1 Background; Definitions. a The parties acknowledge and agree that the Background portion of this Addendum including all definitions, representations and provisions set forth therein, is hereby incorporated by reference as if set forth in this Section. b For purposes of this Addendum, if a capitalized term in this Addendum is no specifically defined herein, that term will be given the same definition that the term is afforded in the Applicable Franchise Agreements, as applicable, A 2 Multi-Unit Fee in Lieu of Initial Franchise Fee. Notwithstanding anything contained in Section 3.1 of the Applicable Franchises Agreements. Franchisee will not be required to pay Franchisor a lump-sum multi-unit fee amounting to \$ (the "Multi-Unit Fee") collectively, the "Applicable Franchise Agreements.") and, under each such Applicable Franchise afranchisee is required to approach the Pranchise obtained the right and undertook the obligation to operate a franchised business under Franchise and operate the Franchises Businesses (each, a "Designated Territory" and collectivel	"Franchisce").
entered into	BACKGROUND
to actively promote and operate the Franchised Businesses (each, a "Designated Territory" and collectively the "Designated Territories"). C. Franchisor and Franchisee now wish to amend and otherwise clarify certain provisions in the Applicable Franchise Agreements, pursuant to the terms and conditions of this Addendum. AGREEMENT NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties agree as follows: 1. Background: Definitions. a. The parties acknowledge and agree that the Background portion of this Addendum including all definitions, representations and provisions set forth therein, is hereby incorporated by reference as if set forth in this Section. b. For purposes of this Addendum, if a capitalized term in this Addendum is no specifically defined herein, that term will be given the same definition that the term is afforded in the Applicable Franchise Agreements, as applicable. A. 2. Multi-Unit Fee in Lieu of Initial Franchise Fee. Notwithstanding anything contained in Section 3.1 of the Applicable Franchise Agreements. Franchisee will not be required to parfranchisor an Initial Franchise Fee under any Applicable Franchise Agreement. Instead, Franchisee must pay Franchisor a lump-sum multi-unit fee amounting to \$ (the "Multi-Unit Fee") (collectively, the "Applicable Franchise Agreements") and, under each such Applicable Franchise Agreement, Franchisee obtained the right and undertook the obligation to operate a franchised business under Franchise obtained the right and undertook the obligation to operate a franchised business under Franchise Businesses (each, a "Franchised Businesses"). B.A. Each Franchised Businesse has its own designated territory wherein Franchise is required to actively promote and operate the Franchised Businesses (each, a "Designated Territory" and collectively.	entered into() franchise agreements TR(collectively, the "Applicable Franchis Agreements") and, under each such Applicable Franchise Agreement, Franchisee obtained the right and undertook the obligation to operate a franchised business under Franchisor's then-current proprietation.
AGREEMENT NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties agree as follows: 1. Background; Definitions. a. The parties acknowledge and agree that the Background portion of this Addendum including all definitions, representations and provisions set forth therein, is hereby incorporated by reference as if set forth in this Section. b. For purposes of this Addendum, if a capitalized term in this Addendum is no specifically defined herein, that term will be given the same definition that the term is afforded in the Applicable Franchise Agreements, as applicable. A. 2. Multi-Unit Fee in Lieu of Initial Franchise Fee. Notwithstanding anything contained in Section 3.1 of the Applicable Franchise Agreements, Franchisee will not be required to pay Franchisor an Initial Franchise Fee under any Applicable Franchise Agreement. Instead, Franchisee must pay Franchisor a lump-sum multi-unit fee amounting to \$ (the "Multi-Unit Fee") (collectively, the "Applicable Franchise Agreements") and, under each such Applicable Franchise Agreement, Franchisee obtained the right and undertook the obligation to operate a franchised decircle of the payments of the proprietary mark(s) (each, a "Franchised Business"). B.A. Each Franchised Business has its own designated territory wherein Franchises is required to actively premote and operate the Franchised Businesses (each, a "Designated Territory" and collectively	to actively promote and operate the Franchised Businesses (each, a "Designated Territory" and collectivel
NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties agree as follows: 1. Background: Definitions. a. The parties acknowledge and agree that the Background portion of this Addendum including all definitions, representations and provisions set forth therein, is hereby incorporated by reference as if set forth in this Section. b. For purposes of this Addendum, if a capitalized term in this Addendum is no specifically defined herein, that term will be given the same definition that the term is afforded in the Applicable Franchise Agreements, as applicable. A. 2. Multi-Unit Fee in Lieu of Initial Franchise Fee. Notwithstanding anything contained in Section 3.1 of the Applicable Franchise Agreements, Franchisee will not be required to pay Franchisor an Initial Franchise Fee under any Applicable Franchise Agreement. Instead, Franchisee mus pay Franchisor a lump-sum multi-unit fee amounting to \$ (the "Multi-Unit Fee") (collectively, the "Applicable Franchise Agreements") and, under each such Applicable Franchise Agreement, Franchisee obtained the right and undertook the obligation to operate a franchised business under Franchiser's then ourrent proprietary mark(s) (each, a "Franchised Business"). B.A. Each Franchised Business has its own designated territory wherein Franchisee is required to actively promote and operate the Franchised Businesses (each, a "Designated Territory" and collectively.	
A. Background: Definitions. a. The parties acknowledge and agree that the Background portion of this Addendum including all definitions, representations and provisions set forth therein, is hereby incorporated by reference as if set forth in this Section. b. For purposes of this Addendum, if a capitalized term in this Addendum is not specifically defined herein, that term will be given the same definition that the term is afforded in the Applicable Franchise Agreements, as applicable. A. 2. Multi-Unit Fee in Lieu of Initial Franchise Fee. Notwithstanding anything contained in Section 3.1 of the Applicable Franchise Agreements, Franchisee will not be required to pay Franchisor an Initial Franchise Fee under any Applicable Franchise Agreement. Instead, Franchisee must pay Franchisor a lump-sum multi-unit fee amounting to \$ (the "Multi-Unit Fee") (collectively, the "Applicable Franchise Agreements") and, under each such Applicable Franchise Agreement, Franchisee obtained the right and undertook the obligation to operate a franchised business under Franchiseor's then current proprietary mark(s) (each, a "Franchised Business"). B.A. Each Franchised Business has its own designated territory wherein Franchisee is required to actively promote and operate the Franchised Businesses (each, a "Designated Territory" and collectively.	<u>AGREEMENT</u>
a. The parties acknowledge and agree that the Background portion of this Addendum, including all definitions, representations and provisions set forth therein, is hereby incorporated by reference as if set forth in this Section. b. For purposes of this Addendum, if a capitalized term in this Addendum is not specifically defined herein, that term will be given the same definition that the term is afforded in the Applicable Franchise Agreements, as applicable. A. 2. Multi-Unit Fee in Lieu of Initial Franchise Fee. Notwithstanding anything contained in Section 3.1 of the Applicable Franchise Agreements, Franchisee will not be required to pay Franchisor an Initial Franchise Fee under any Applicable Franchise Agreement. Instead, Franchisee must pay Franchisor a lump-sum multi-unit fee amounting to \$ (the "Multi-Unit Fee"), (collectively, the "Applicable Franchise Agreements") and, under each such Applicable Franchise Agreement, Franchisee obtained the right and undertook the obligation to operate a franchised business under Franchise obtained the right and undertook the obligation to operate a franchised business under Franchised Business has its own designated territory wherein Franchisee is required to actively promote and operate the Franchised Businesses (each, a "Designated Territory" and collectively.	
b. For purposes of this Addendum, if a capitalized term in this Addendum is not specifically defined herein, that term will be given the same definition that the term is afforded in the Applicable Franchise Agreements, as applicable. A. 2. Multi-Unit Fee in Lieu of Initial Franchise Fee. Notwithstanding anything contained in Section 3.1 of the Applicable Franchise Agreements, Franchisee will not be required to pay Franchisor an Initial Franchise Fee under any Applicable Franchise Agreement. Instead, Franchisee must pay Franchisor a lump-sum multi-unit fee amounting to \$ (the "Multi-Unit Fee"), (coellectively, the "Applicable Franchise Agreement.") and, under each such Applicable Franchise Agreement, Franchisee obtained the right and undertook the obligation to operate a franchised business under Franchise's then current proprietary mark(s) (each, a "Franchised Business"). B.A. Each Franchised Business has its own designated territory wherein Franchisee is required to actively promote and operate the Franchised Businesses (each, a "Designated Territory" and collectively.	1. Background; Definitions.
A. — 2. Multi-Unit Fee in Lieu of Initial Franchise Fee. Notwithstanding anything contained in Section 3.1 of the Applicable Franchise Agreements, Franchise Fee in Lieu of Initial Franchise Will not be required to pay Franchisor an Initial Franchise Fee under any Applicable Franchise Agreement. Instead, Franchisee must pay Franchisor a lump-sum multi-unit fee amounting to \$ (the "Multi-Unit Fee"). (collectively, the "Applicable Franchise Agreements") and, under each such Applicable Franchise Agreement, Franchisee obtained the right and undertook the obligation to operate a franchised business under Franchises's then current proprietary mark(s) (each, a "Franchised Business"). B.A. Each Franchised Business has its own designated territory wherein Franchisee is required to actively promote and operate the Franchised Businesses (each, a "Designated Territory" and collectively.	including all definitions, representations and provisions set forth therein, is hereby incorporated by
Franchisor an Initial Franchise Fee under any Applicable Franchise Agreements, Franchisee will not be required to pay Franchisor an Initial Franchise Fee under any Applicable Franchise Agreement. Instead, Franchisee must pay Franchisor a lump-sum multi-unit fee amounting to \$	specifically defined herein, that term will be given the same definition that the term is afforded in the
to actively promote and operate the Franchised Businesses (each, a "Designated Territory" and collectively	Contained in Section 3.1 of the Applicable Franchise Agreements, Franchisee will not be required to particular franchise Fee under any Applicable Franchise Agreement. Instead, Franchisee must pay Franchisor a lump-sum multi-unit fee amounting to \$ (the "Multi-Unit Fee" (collectively, the "Applicable Franchise Agreements") and, under each such Applicable Franchise Agreement, Franchisee obtained the right and undertook the obligation to operate a franchised busine
to actively promote and operate the Franchised Businesses (each, a "Designated Territory" and collectively	B.A. Each Franchised Business has its own designated territory wherein Franchisee is require
	to actively promote and operate the Franchised Businesses (each, a "Designated Territory" and collectivel

C.A. Franchisor and Franchisee now wish to amend and otherwise clarify certain provisions in the Applicable Franchise Agreements, pursuant to the terms and conditions of this Addendum.

ACREEMENT

NOW, THEREFORE, for valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties agree as follows:

1. Background; Definitions.

- a. The parties acknowledge and agree that the Background portion of this Addendum, including all definitions, representations and provisions set forth therein, is hereby incorporated by
- b. For purposes of this Addendum, if a capitalized term in this Addendum is not specifically defined herein, that term will be given the same definition that the term is afforded in the Applicable Proposition Agreements, as applicable.
- 3. Minimum Monthly Royalty Fees. Notwithstanding anything contained in Section 3.2 of the Applicable Franchise Agreements, Franchisee's Minimum Royalty Fee is based upon the number of Franchised Businesses and the number of months the Franchised Business has been open and operating. The Minimum Royalty Fee is as follows:
- 3. Minimum Monthly Royalty Fees. Notwithstanding anything contained in Section 3.2 of the Applicable Franchise Agreements, Franchisec's Minimum Royalty Fee is based upon the number of Franchised Businesses and the number of months the Franchised Business has been open and operating. The Minimum Royalty Fee is as follows:

Time of Operation	4 Franchised Businesses	2 Franchised Businesses	3 Franchised Businesses	4 Franchised Businesses	5 Franchised Businesses	
Time of Operation	1 Franchised Business	2 Franchised Businesses	3 Franchised Businesses	4 Franchised Businesses	<u>5</u> Franchised Businesses	-
First 3 Months	7% - 5% of Gross Revenue	7% - 5% of Gi Revenue	ross			
4 to 12 Months	\$250	\$500	\$750	\$1,000	\$1,250	
Second Year	\$500	\$1,000	\$1,500	\$2,000	\$2,500	
Third Year	\$750	\$1,500	\$2,250	\$3,000	\$3,750	
Fourth Year +	\$1,000	\$2,000	\$3,000	\$4,000	\$5,000	
4 to 12 Months	\$300	\$600	\$900	\$1,200	\$1,500	
Second Year Third Year +	\$1,500 \$2,000	\$3,000 \$4,000	\$4,500 \$6,000	\$6,000 \$8,000	\$7,500 \$10,000	

For each additional Franchised Business above 5 units:

- the First 3 months is 7-5% of Gross Revenue
- 4-12 months is an additional \$300 per Franchised Business
- Second Year is an additional \$1,500 per Franchised Business
- Third Year + is an additional \$2,000 per Franchised Business
- 4. **Brand Fund, Local Advertising Requirement, and Call Center Fee.** Notwithstanding anything contained in Section 3.5, Section 3.6, and Section 3.611 of the Applicable Franchise Agreements, Franchisor and Franchisee acknowledge and agree that for purposes of determining Gross Revenue for the Brand Fund, Local Advertising Requirement, and Call Center Fee, Franchisee may aggregate Gross Revenue of each Franchised Business operating under this Addendum. For purposes of clarification, the monthly Local Advertising Requirement will be the greater of: (a) three thousand dollars (\$\$3,000) for the first Applicable Franchise Agreement and one thousand five hundred dollars (\$\$1,500) for each additional contiguous Applicable Franchise Agreement; andor (b) three percent (3%)% of aggregate Gross Revenue for each Applicable Franchise Agreement under this Addendum. The Initial Franchisor Directed Advertising Fee will be calculated as follows: an amount up to three thousand dollars (\$3,000) for the first Applicable Franchise Agreement, times two (2) months.
- 5. **Technology Fee:** Marketing Management Fee, and Online Local Presence Fee. Franchisor and Franchisee acknowledge and agree that Franchisee will utilize (a) the same technology package for all of its Franchised Businesses, (b) the same marketing materials for all of its Franchised Businesses, and (c) the same website for all of its Franchised Businesses, and Franchisor will only require Franchisee to pay the applicable technology fees Technology Fee, Marketing Management Fee, and Online Local Presence Fee as if Franchisee was only operating one Franchised Business.
- 6. **Approved Location**. Section 1.3 of the Applicable Franchise Agreements is hereby amended to clarify that if Franchisee's Designated Territories under the Applicable Franchise Agreements

are contiguous, then Franchisee will only be obligated to secure one (1) Approved Location that meets Franchisor's standards and specifications.

- 7. **Inventory.** Section 7.6.7 of the Applicable Franchise Agreements is hereby amended to clarify that Franchisee must maintain sufficient levels of inventory, vehicles, supplies and equipment in order to adequately meet consumer demand within the Designated Territories.
- 8. **Initial Training Program.** Section 8.1 of the Applicable Franchise Agreements is hereby amended to clarify that Franchisee is only obligated to attend the Initial Training Program once.
- 9. Grand Opening Marketing. Section 12.5.1 of the Applicable Franchise Agreements is hereby amended to clarify that if Franchisee operates the Franchised Business in Designated Territories that are contiguous to one another, then Franchisee is only obligated to spend the Grand Opening Marketing fee of \$15,000 (or \$22,000 if Franchisor increases such amount) once.
- 10. Transfer Fee. Section 14.3.2.8 of the Applicable Franchise Agreements is hereby amended to clarify that Franchisee is obligated to pay Franchisee a Transfer Fee of 7. Inventory. Section 7.6.7 of the Applicable Franchise Agreements is hereby amended to clarify that Franchisee must maintain sufficient levels of inventory, vehicles, supplies and equipment in order to adequately meet consumer demand within the Designated Territories.
- Initial Training Program. Section 8.1 of the Applicable Franchise Agreements is hereby amended to clarify that Franchisee is only obligated to attend the Initial Training Program once.
- 9. Grand Opening Marketing. Section 12.5.1 of the Applicable Franchise Agreements is hereby amended to clarify that if Franchisee operates the Franchised Business in Designated Territories that are contiguous to one another, then Franchisee is only obligated to spend the Grand Opening Marketing fee of \$10,000 (or \$15,000 if Franchisor increases such amount) once.
- 10. Transfer Fee. Section 14.3.2.8 of the Applicable Franchise Agreements is hereby amended to clarify that Franchisee is obligated to pay Franchisee a Transfer Fee of ten thousand dollars (\$10,000)\$10,000 in connection with each Applicable Franchise Agreement/Designated Territory that Franchisee wishes to transfer.
- 11. Renewal Fee. Section 2.2.9 of the Applicable Franchise Agreements is hereby amended to clarify that Franchisee is obligated to pay Franchisor a Renewal Fee of 11. Renewal Fee. Section 2.2.9 of the Applicable Franchise Agreements is hereby amended to clarify that Franchisee is obligated to pay Franchisor a Renewal Fee of five thousand dollars (\$\$5,000) in connection with the first Applicable Franchise Agreement that Franchisee wishes to renew. The Renewal Fee for an additional renewal of an Applicable Franchise Agreement under this Addendum that is renewed at the same time will be one thousand dollars (\$\$1,000).
- 12. **Transfer.** Franchisor has the right to sell, transfer, assign and/or encumber all or any part of Franchisor's assets and Franchisor's interest in, and rights and obligations under, this Addendum in Franchisor's sole discretion. Franchisee may not transfer its rights under this Addendum without Franchisor's prior written consent.
- 13. Default of Addendum Constitutes Default Under All Applicable Franchise Agreements. In the event Franchisee breaches any of the provisions of this Addendum, such breach will constitute a material default of all Applicable Franchise Agreements and must be cured within 30 days from Franchisee's receipt of Franchisor's written notice of such breach as set forth in Section 15.4 of the Applicable Franchise Agreements. If Franchisee fails to cure such breach(es) within the prescribed time

period, Franchisor may, at its option, terminate one or more of the Applicable Franchise Agreements immediately upon providing written notice. Notwithstanding the foregoing, Franchisor will not terminate any other Applicable Franchise Agreement under Franchisee's Multi-Unit Addendum solely based upon Franchisee's failure to operate a Franchised Business under one of the Applicable Franchise Agreements under Franchisee's Multi-Unit Addendum.

- 14. **Governing Law**. This Addendum shall be governed by and construed in accordance with the laws of the State of North Carolina, without reference to this state's conflict of laws principles.
- 15. Venue; Forum; Jurisdiction; Dispute Resolution. Franchisor and Franchisee agree and acknowledge that the venue, forum, jurisdiction, and dispute resolution provisions of the Applicable Franchise Agreements shall also apply to this Addendum.
- 16. Ratification of Applicable Franchise Agreements. Except as amended by this Addendum, any and all other terms and conditions set forth in the Applicable Franchise Agreements are hereby ratified and confirmed as if fully restated herein.
- 17. Entire Agreement. This Addendum constitutes the entire agreement between the parties and supersedes any and all prior negotiations, understandings, representations, and agreements. Nothing in this or in any related agreement, however, is intended to disclaim the representations Franchisor made in the Franchise Disclosure Document that was furnished to Franchisee.
- 18. No statement, questionnaire or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of: (i) waiving any claims under any applicable state franchise law, including fraud in the inducement; or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

<u>IN WITNESS WHEREOF</u>, the parties hereto, intending to be legally bound hereby, have duly executed and delivered this Addendum on the date and year first written above.

FRANCHISOR:	FRANCHISEE:
HFB FENCECO FRANCHISING, LLC, a North Carolina limited liability company	(Individual, Partnership or Corporation Name)
By:	By:
Name:	Name:
Title:	Title:
Date:	Date:

Exhibit G to FDD HFB FenceCo Franchising, LLC

SAMPLE ASSIGNMENT AND ASSUMPTION AGREEMENT TO A BUSINESS ENTITY

SAMPLE ASSIGNMENT AND ASSUMPTION AGREEMENT UPON TRANSFER TO A BUSINESS ENTITY

BACKGROUND

- B. Assignor, by transferring its ownership interest in Franchisee, desires to assign all Assignor's rights and obligations under the Franchise Agreement to Assignee. Assignee desires to obtain these rights and is willing to assume these obligations.
 - C. Franchisor is willing to permit the assignment and assumption.

Now therefore, in consideration of the mutual promises and consents provided in this Agreement, and other good and valuable consideration, the sufficiency and adequacy of which the parties hereby acknowledge, the parties agree as follows:

TERMS OF AGREEMENT

- 1. Assignment and Assumption. Subject to the terms of this Agreement, Assignor assigns to Assignee, by transfer of its ownership interest in Franchisee, all of Assignor's right, title and interest in the Franchise Agreements (including Amendments, ancillary agreements and Multi-Unit Addendum) and delegates to Assignee all of Assignor's obligations under the Franchise Agreements arising from and after the Effective Date. Assignee agrees to assume and perform all of Assignor's obligations under the Franchise Agreements arising from and after the Effective Date. Simultaneously with the Assignment contemplated herein, the assigned Franchise Agreements will be amended to specify _______ is the Franchisee.
- 2. No Discharge of Assignor. The assignment by Assignor of Assignor's rights and obligations under the Franchise Agreements will not relieve Assignor or Franchisee of obligations under the Franchise Agreements that accrue before the Effective Date or after the Effective Date that are related to events occurring before the effective date of the transfer. Notwithstanding the foregoing, Assignor must report all revenue collected prior to the Effective Date and pay all royalties due thereon, on or before the 10th of the month (or the next business day after the 10th if the 10th is not a business day) following the month in which the assignment transaction closes. Assignee will be responsible for reporting all amounts collected after the Effective Date and paying all royalties on amounts collected after the Effective Date in accordance with the Franchise Agreements between Assignee and Franchisor. In addition, if assignor collects any amounts after the Effective Date that relate to operation of the business prior to (or after) the Effective date, Assignor must report all such collections to Franchisor on a monthly basis consistent with terms of the Franchise Agreements in

effect as of the Effective Date and Assignor must pay royalties on those collections on a monthly basis consistent with terms of the Franchise Agreements in effect as of the Effective Date.

- 3. **Transfer Fees.** Because this is a transfer from an individual to a business entity, no transfer fee applies.
- 4. **Preconditions to Consent.** In order to cause Franchisor to consent to such sale, waive such right of first refusal, and to gain other benefits to the Assignees and the Assigners, the Assignees and Assignors have agreed to their respective commitments under this Consent.
- 5. Consent and Waiver of Right of First Refusal. As of the Effective Date, Franchisor consents to the transfer of the rights and obligations of the Assignor in and to the Franchisee, the Franchise Agreements, and the related franchised business to the Assignee. Assignor confirms that they have transferred all rights (whether held personally or otherwise) in and to the Franchisee, the Franchise Agreements, and the related franchise business to Assignee, together with all rights in and to all real and personal property used in the operations of the Franchise and the related Franchised Business. Franchisor hereby declines to exercise its right of first refusal with regard to the proposed transfer contemplated between the Assignee and Assignor.
- 6. Acknowledgment. Assignor, Assignee and Franchisor agree that the assignment of Assignor's Franchised Business and the Franchise Agreements (and the related agreements, if any) to Assignee were not affected by or through Franchisor, and that Franchisor's activities in connection therewith were limited to the exercise of its right of approval under the Franchise Agreements.
- Limited Release by Assignee. Assignee, for itself and its officers, directors, stockholders, members, managers, partners, agents, heirs, executors, administrators, legal successors and assigns, release and forever discharge HFB FenceCo Franchising, LLC, and DDL Investments, Inc., and each of their respective parents, subsidiaries and affiliates, and the respective officers, directors, stockholders, members, managers, partners, employees, agents, attorneys, contractors, legal successors and assigns of each of the forgoing entities in their corporate and individual capacities ("Franchisor-Related Persons/Entities"), of and from all manner of actions, causes of action, suits, debts, dues, sums of money, accounts, reckonings, bonds, bills, specialties, covenants, contracts, agreements, controversies, promises, variances, trespasses, damages, judgments, executions, claims and demands whatsoever, in law, admiralty or in equity, that Assignee and/or Assignee's officers, directors, stockholders, members, managers, partners, agents, heirs, executors, administrators, legal successors and assigns ever had, now has, or that they hereafter can or may have for, on or by reason of any matter, cause or thing whatsoever, from the beginning of the world to the Effective Date, including, without limitation, claims arising under federal, state and local laws, rules and ordinances that are related to the assignment of the Franchise Agreements ("Assignee Claims"). For the absence of confusion, the Assignee Claims do not relate to any claim of Assignee (or any person claiming through Assignee) arising under the Franchise Agreements following the Effective Date.

Being fully informed, Assignee(s) waive any rights on, and acknowledges that this release extends to all Assignee Claims Assignees have or might have against Franchisor-Related Persons/Entities, whether known or unknown.

8. General Release by Assignor/Franchisee and Guarantors. The Assignor, for themselves, Franchisee, and, as applicable, their officers, directors, stockholders, members, managers, partners, agents, heirs, executors, administrators, legal successors and assigns, release and forever discharge the Franchisor-Related Persons/Entities, of and from all manner of actions, causes of action, suits, debts,

dues, sums of money, accounts, reckonings, bonds, bills, specialties, covenants, contracts, agreements, controversies, promises, variances, trespasses, damages, judgments, executions, claims and demands whatsoever, in law, admiralty or in equity, that Assignor and/or Assignor's officers, directors, stockholders, members, managers, partners, agents, heirs, executors, administrators, legal successors and assigns, ever had, now has, or that they hereafter can or may have for, on or by reason of any matter, cause or thing whatsoever, from the beginning of the world to the day of this Agreement, including, without limitation, claims arising under federal, state and local laws, rules and ordinances ("Assignor Claims").

Being fully informed, Assignor(s) waive any rights on, and acknowledge(s) that this release extends to all Assignor Claims Assignors have or might have against Franchisor-Related Persons/Entities, whether known or unknown.

- 9. Assumption of Risk. The Guarantor and Assignor/Franchisee(s) expressly assume the risk of any mistake of fact or fact of which they may be unaware or that the true facts may be other than any facts now known or believed to exist by Guarantor or Assignor/Franchisee(s), and it is the Guarantor's, Assignor's, and Franchisee's intention to forever settle, adjust and compromise any and all present and/or future disputes with respect to all matters from the beginning of time to the date of this document finally and forever and without regard to who may or may not have been correct in their understanding of the facts, law or otherwise. All releases given by the Guarantor and Assignor/Franchisee are intended to constitute a full, complete, unconditional and immediate substitution for any and all rights, claims, demands and causes of action whatsoever which exist, or might have existed, on the date of this document. The Guarantor and Assignor/Franchisee represent and warrant that they have made such independent investigation of the facts, law and otherwise pertaining to all matters discussed, referred to or released in or by this document as the Guarantor and Assignor/Franchisee, in each of their independent judgment, believe necessary or appropriate. The Guarantor and Assignor/Franchisee have not relied on any statement, promise, representation or otherwise, whether of fact, law or otherwise, or lack of disclosure of any fact, law or otherwise, by the Franchisor-Related Persons/Entities or anyone else not expressly set forth herein, in executing this document and/or the related releases.
- 10. Amendment to Statement of Ownership. The Statement of Ownership to the Franchise Agreements is hereby amended by deleting it in its entirety and replacing it with the updated Statement of Ownership attached as Exhibit A to this Agreement.
- 11. **Personal Guaranty**. The Personal Guaranty to the Franchise Agreements is hereby amended by deleting it in its entirety and replacing it with the updated Personal Guaranty attached as Exhibit B to this Agreement.
- 12. Entire Agreement. This Agreement represents the entire understanding and agreement between the parties with respect to the subject matter of this Agreement and supersedes all other negotiations, understandings and representations, if any, made by the parties with respect to the subject of this Agreement.
- 13. Assignment; Parties In Interest. This Agreement is binding on and inures to the benefit of the parties and their respective successors and assigns, but this Agreement and the rights or obligations in this Agreement may not be assigned by any party without the other parties' consent, except that Franchisor has the right to assign this Agreement, to any person, firm, corporation or other entity, if the assignment is made with Franchisor's assignment of its rights under the Franchise Agreements. This Agreement is made for the benefit of the parties, and nothing in this Agreement is intended to or confers any benefit on any other person, association or entity. If the Assignors, Guarantors, Assignee or any third party, or anyone acting for or on behalf of such persons or any third party, or any of them, claim to have received, from Assignee, Assignor, or otherwise, any interest in any of the Assignee Claims or Assignor

Claims, commence, join in, or in any manner seek relief through any suit (or otherwise) arising out of, based upon or relating to any of the Assignee Claims or Assignor Claims released hereunder or in any manner asserts against all or any of the Franchisor-Related Persons/Entities relating to any of the Assignee Claims or Assignor Claims released hereunder, the Assignors (with regard to Assignor Claims), jointly and severally, and Assignees (with regard to Assignee Claims), jointly and severally, agree to pay all documented attorneys' fees and other documented costs incurred by any of the Franchisor-Related Persons/Entities in defending or otherwise responding to said suit or assertion directly to the Franchisor-Related Persons/Entities incurring such costs.

14. Notices. Notices required or permitted hereunder shall be given in writing and deemed given upon receipt when delivered personally or by overnight courier or by certified mail, return receipt requested, and addressed as follows (or as otherwise provided by a Party by written notice under this Section 14):

Notices to Assignee:

Notices to HFB FenceCo Franchising, LLC

<u>Franchisor:</u> <u>107 Parr Drive</u>

Huntersville, NC 28078 Attn: Legal Department

15. Authority/Agreement. The Parties each warrant and represent to the other that they have the power and authority to enter into this Consent and will be bound by the terms hereof. NOTWITHSTANDING ANY OTHER WRITING, CONVERSATION, OR OTHER COMMUNICATION OUTSIDE OF THIS CONSENT, THE PARTIES ACKNOWLEDGE AND AGREE THAT THERE ARE NO DISPUTES AS TO THE TERMS OF THIS CONSENT, INCLUDING BUT NOT LIMITED TO THE SCOPE OF THE RELEASES OUTLINED IN SECTIONS 7 AND 8 ABOVE, ALL SUCH OBJECTIONS, IF ANY, BEING EXTINGUISHED AND OF NO FURTHER FORCE OR EFFECT, BY SIGNING THIS CONSENT. ASSIGNORS REPRESENT, WARRANT, AGREE, AND ACKNOWLEDGE THAT THEY HAVE READ, UNDERSTAND, AND AGREE TO THE TERMS OF THIS SECTION 15 AND SECTION 8, AND THAT ASSIGNORS HAVE FURTHER CONSULTED WITH LEGAL COUNSEL WHO HAS READ THIS SECTION 15 AND SECTION 8, AND THAT ASSIGNORS UNDERSTAND AND FULLY AGREE TO THE TERMS OF SECTION 15 AND SECTION 8. ASSIGNORS FURTHER REPRESENT THAT THEY HAVE HAD SUFFICIENT TIME AND OPPORTUNITY TO REVIEW THE TERMS OF THIS SECTION 15 AND SECTION 8, SEEK ADVICE FROM THEIR COUNSEL OR RECORD AND THAT THEY MAKE THE REPRESENTATIONS AND AGREE TO THESE TERMS FREELY. ASSIGNEES REPRESENT, WARRANT, AGREE, AND ACKNOWLEDGE THAT THEY HAVE READ, UNDERSTAND, AND AGREE TO THE TERMS OF THIS SECTION 15 AND SECTION 7, AND THAT ASSIGNEES HAVE FURTHER CONSULTED WITH LEGAL COUNSEL WHO HAS READ THIS SECTION 15 AND SECTION 7, AND THAT ASSIGNEES UNDERSTAND AND FULLY AGREE TO THE TERMS OF SECTION 15 AND SECTION 7. ASSIGNEES FURTHER REPRESENT THAT THEY HAVE HAD SUFFICIENT TIME AND OPPORTUNITY TO REVIEW THE TERMS OF THIS SECTION 15 AND SECTION 7, SEEK ADVICE FROM THEIR COUNSEL OR RECORD AND THAT THEY MAKE THE REPRESENTATIONS AND AGREE TO THESE TERMS FREELY.

16. **Binding Effect**. This Consent shall be binding upon and inure to the benefit of the Parties and, their respective administrators, representatives, heirs, successors, officers, directors, affiliates, successors, insurers and assigns.

- 17. **Dispute Resolution**. In the event of a dispute between the Parties relating to the interpretation or enforcement of this Consent the Parties agree that the dispute resolution provisions of the Franchise Agreements requiring pre-arbitration mediation and arbitration of disputes will apply and that all court proceedings, arbitration proceedings, mediation proceedings, lawsuits, court hearings and other hearings initiated by Franchisor or Assignors must and will be venued exclusively in Huntersville, North Carolina. The Parties do hereby agree and submit to personal jurisdiction in the State of North Carolina for the purposes of any suit, proceeding or hearing brought to enforce or construe the terms of this Consent or to resolve any dispute or controversy arising under, as a result of, or in connection with the Consent or the parties' relationship. The parties do hereby agree and stipulate that any such suits, proceedings and hearings will be exclusively venued and held in Mecklenburg County, North Carolina. Assignors and Assignees waive any rights to contest such venue and jurisdiction and any claims that such venue and jurisdiction are invalid.
- 18. Attorney Fees. If any party retains the services of legal counsel to enforce the terms of this Agreement and prevails in such enforcement action, it shall be entitled to recover all reasonable and documented costs and expenses, including reasonable attorneys' fees incurred in enforcing the terms of this Agreement.
- 19. Venue and Jurisdiction. Assignor and Assignee agrees that they have and had a relationship with Franchisor at its offices in the State of North Carolina, and each Party agrees that, with the exception of Franchisor's right to seek injunctive relief in any appropriate jurisdiction, any action by or against Franchisor arising out of or relating to this Agreement shall be commenced and concluded in the State of North Carolina pursuant to the mediation, arbitration, and litigation venue and jurisdiction provisions of the Franchise Agreements. This Agreement shall be construed under the laws of the State of North Carolina, which laws shall control in the event of any conflict of law.
- 20. **Severability.** If any portion of this Consent is held invalid by a court or arbitrator, the remainder of this Consent shall be enforced in harmony with the purpose of this Consent and the intent of the Parties at the time of its making. If any provision of this Consent is held to be excessive as to time or scope, then that provision shall be interpreted as if its provisions reached only to the maximum extent enforceable under applicable law.
- 21. **Headings**. The headings used in this Consent are merely for identification of the substantive provisions; and to the extent they conflict with any substantive portion of this Consent, they are to be disregarded.
- 22. Construction. The Parties hereto hereby mutually acknowledge this Consent was mutually drafted and it shall not be construed for or against any Party on the basis of the drafting of any particular provision or the document of the whole, but rather shall be given a fair and reasonable interpretation, based on the plain language of this Consent in expressing the Parties' intent.
- 23. Counterparts. This Agreement may be executed in multiple counterparts by the various parties and the failure to have the signatures of all parties on a single Agreement shall not affect the validity or enforceability of any part of this Agreement against any party who executes any counterpart of the Agreement. Executed facsimile or other electronically delivered copies of this Agreement shall be deemed to be effective as original signatures.
- 24. Definitions. Unless otherwise stated, all terms used in this Agreement have the same meaning as indicated in the Franchise Agreements.

I HAVE READ THE ABOVE AGREEMENT AND UNDERSTAND ITS TERMS. I WOULD

NOT SIGN THIS AGREEMENT IF I DID NOT UNDERSTAND AND AGREE TO BE BOUND BY ITS TERMS.	

IN WITNESS WHEREOF, the parties have set their hands and seals.

FRANCHISOR:

HFB FENCECO FRANCHISING, LLC,

a North Carolina limited liability company

Jeffrey R. Dudan, President

ASSIGNOR:	
<u>, Individually</u>	
ASSIGNEE: [ENTITY] a	

$\frac{\text{EXHIBIT A TO ASSIGNMENT AND ASSUMPTION (TO BUSINESS ENTITY)}}{\text{STATEMENT OF OWNERSHIP}}$

FRANCHISEE:				
Form of Ownership (Check One)				
	Limited Liabil	ity Company	☐ Limited Partnership	
□ Corporation □ General Partnership				
If a Partnership , provide name and address of each partner showing percentage owned, whether active in management, and indicate the state in which the partnership was formed.				
If a Corporation , provide the state and date of incorporation, the names and addresses of each officer and director, and list the names and addresses of every shareholder showing what percentage of stock is owned by each.				
If a Limited Liability Company , provide the state and date of formation, the name and address of the manager(s), and list the names and addresses of every member and the percentage of membership interest held by each member.				
State and Date of Formation/Incorporation:				
Management (managers, officers, board of directors, etc.):				
<u>Name</u> <u>Title</u>				
Members, Stockholders, Partners:				
<u>Name</u>	Add	ress	Percentage Owned	

This form is current and com	plete as of	, 202_

FRANCHISEE:

[ENTITY]

EXHIBIT B TO ASSIGNMENT AND ASSUMPTION (TO BUSINESS ENTITY) HFB FENCECO FRANCHISING, LLC

- 12. Transfer. Franchisor has the right to sell, transfer, assign and/or encumber all or any part of Franchisor's assets and Franchisor's interest in, and rights and obligations under, this Addendum in Franchisor's sole discretion. Franchisee may not transfer its rights under this Addendum without Franchisor's prior written consent.
- 13. Default of Addendum Constitutes Default Under All Applicable Franchise Agreements. In the event Franchisee breaches any of the provisions of this Addendum, such breach will constitute a material default of all Applicable Franchise Agreements and must be cured within 30 days from Franchisee's receipt of Franchiser's written notice of such breach as set forth in Section 15.4 of the Applicable Franchise Agreements. If Franchisee fails to cure such breach(es) within the prescribed time period, Franchiser may, at its option, terminate one or more of the Applicable Franchise Agreements immediately upon providing written notice.
- 14. Governing Law. This Addendum shall be governed by and construed in accordance with the laws of the State of North Carolina, without reference to this state's conflict of laws principles.
- 15. Venue; Forum; Jurisdiction; Dispute Resolution. Franchisor and Franchisee agree and acknowledge that the venue, forum, jurisdiction, and dispute resolution provisions of the Applicable Franchise Agreements shall also apply to this Addendum.
- 16. Ratification of Applicable Franchise Agreements. Except as amended by this Addendum, any and all other terms and conditions set forth in the Applicable Franchise Agreements are hereby ratified and confirmed as if fully restated herein.
- 17. Entire Agreement. This Addendum constitutes the entire agreement between the parties and supersedes any and all prior negotiations, understandings, representations, and agreements. Nothing in this or in any related agreement, however, is intended to disclaim the representations Franchisor made in the Franchise Disclosure Document that was furnished to Franchisee. Franchisee acknowledges that it is entering into this Addendum as a result of its own independent investigation of the Franchised Businesses and not as a result of any representations about Franchisor made by its shareholders, officers, directors, employees, agents, representatives, independent contractors, or franchisees that are contrary to the terms set forth in this Addendum, or in any disclosure document, prospectus, or other similar document required or permitted to be given to Franchisee pursuant to applicable law.

IN WITNESS WHEREOF, the parties hereto, intending to be legally bound hereby, have duly executed and delivered this Addendum on the date and year first written above.

FRANCHISOR:	FRANCHISEE:
HFB FENCECO FRANCHISING, LLC	(Individual, Partnership or Corporation Name)
Ву:	Ву:
Name:	Name:
Title:	Title:
Date:	Date:

EXHIBIT D TO

HFB FENCECO FRANCHISING, LLC FRANCHISE DISCLOSURE DOCUMENTAGREEMENTS

PERSONAL GUARANTY

NOTE: IF FRANCHISEE IS A CORPORATION, EACH OF FRANCHISEE'S SHAREHOLDERS AND THEIR SPOUSES MUST EXECUTE THE FOLLOWING UNDERTAKING. IF FRANCHISEE IS A PARTNERSHIP, EACH OF FRANCHISEE'S GENERAL PARTNERS AND THEIR SPOUSES MUST EXECUTE THE FOLLOWING UNDERTAKING. IF FRANCHISEE IS A LIMITED LIABILITY COMPANY, EACH OF FRANCHISEE'S MEMBERS AND MANAGERS AND THEIR SPOUSES MUST EXECUTE THE FOLLOWING UNDERTAKING.

Definition of	"Franchisee:"		(r	person(s) or entity
	Individual	 Limited Liability Company		Corporation
	General Partnership	 Limited Partnership		

ARTICLE I PERSONAL GUARANTY

The undersigned persons (individually and collectively "you" or "Personal Guarantors") hereby represent to HFB FenceCo Franchising, LLC ("Franchisor") that you are all of the owners, shareholders, general partners, or members and managers, or the spouse of any such owner, shareholder, general partner, member or manager of the Franchisee. In consideration of the grant by Franchisor to Franchisee, as provided in the foregoing Franchise Agreements (the "Franchise Agreements"), each of you hereby agree, in consideration of benefits received and to be received by each of you, jointly and severally, and for your selves, your heirs, legal representatives and assigns, to be firmly bound by all of the terms, provisions and conditions of the foregoing Franchise Agreements, and any other agreement between Franchisee and Franchisor and/or its affiliates, and do hereby unconditionally guarantee the full and timely performance by Franchisee of each and every obligation of Franchisee under the aforesaid Franchise Agreements or other agreement between Franchisor and Franchisee, including, without limitation, any indebtedness of Franchisee arising under or by virtue of the aforesaid Franchise Agreements and that you (jointly and individually) will not permit or cause any change in the percentage of Franchisee owned, directly or indirectly, by any person, without first obtaining the written consent of Franchisor prior to said proposed transfer, which consent must not be unreasonably withheld, and without first paying or causing to be paid to Franchisor the transfer fee provided for in said Franchise Agreements, if applicable, and without otherwise complying with the transfer provisions of the foregoing Franchise Agreements. You further agree to be bound by the in-term and post-term non-competition and non-solicitation covenants set forth in Section 17 of the Franchise Agreements, as well as all other covenants set forth in the Franchise Agreements, including, but not limited to, those concerning confidentiality (Section 5 of the Franchise Agreements) and indemnification (Section 13.2 of the Franchise Agreements). You agree that this personal guaranty (the "Guaranty") will be governed by the dispute resolution procedures set forth in Section 18 of the Franchise Agreements.

ARTICLE II CONFIDENTIALITY

During the term of this Guaranty, you will receive information which Franchisor considers a trade

secret and confidential information ("Confidential Information"). You shall not, during the term of this Guaranty or thereafter, communicate, divulge, or use for the benefit of any other person, partnership, association, corporation, or limited liability company any Confidential Information, including, without limitation, Franchisor's Operations Manual and its contents; price lists and standards and specifications for the Approved Products and Approved Services; standards and specifications related to Franchisor's integrated bookkeeping system; and other methods, techniques and know-how concerning the operation of a Franchised Business of which you may be apprised by virtue of your role as a Personal Guarantor of Franchisee.

ARTICLE III NON-COMPETITION

During the Term of the Franchise Agreements. During the term of the Franchise Agreements, neither you nor your owners, principals, officers, directors, nor any members of your family or the family of your owners, officers, directors or principals may, directly or indirectly, for themselves or through, on behalf of, or in conjunction with any other person or entity: Own, maintain, engage in, be employed as an officer, director, principal of, lend money to, extend credit to, or have any interest in, any other business that, directly or indirectly, by itself or through licensees or franchisees, offers the Approved Products or Approved Services, or any other products and/or services authorized or offered for sale by System franchisees (a "Competitive Business") within the Designated Territory or the Designated Territory of any other System franchisee, provided that this Article III(1) does not apply to: (i) such person's ownership of a Franchised Business under a Franchise Agreements with Franchisor; or (ii) such person's ownership of a less than five percent (5%) legal or beneficial interest in any publicly-traded company providing such services; b. Employ or seek to employ any person who is at that time employed by Franchisor, Franchisor's affiliates, or any other System franchisee, or otherwise directly or indirectly induce or seek to induce such person to leave his or her employment thereat; or c. Solicit any current, former or prospective customer solicited by Franchisee's Franchised Business, or any other customer of whom Franchisee has become aware as a result of access to Franchisor's System, or other franchisees for any competitive purpose. After the Term of the Franchise Agreements. For a period of 2 years after the expiration and nonrenewal, transfer or termination of the Franchise Agreements, regardless of the cause, neither you, nor your owners, principals, officers, directors, nor any members of your family or the family of your owners, officers, directors, or principals may, directly or indirectly, for themselves or through, on behalf of, or in conjunction with any other person or entity: Own, maintain, engage in, be employed as an officer, director, principal of, lend money to, extend credit to, or have any interest in any Competitive Business: (i) within the Designated Territory; (ii) within a 25-mile radius of the Designated Territory of any other Franchised Business; or (iii) within a 25-mile radius of any System business operated by Franchisor or its affiliate, provided that this Article III(2) does not apply to: (A) such person's ownership of a Franchised Business under a Franchise Agreements with Franchisor; or (B) such person's ownership of a less than five percent (5%) legal or beneficial interest in any publicly-traded company providing services the same as or similar to a Competitive Business;

induce such person to leave his or her employment thereat; or

Franchisor's affiliates, or any other System franchisee, or otherwise directly or indirectly induce or seek to

Employ or seek to employ any person who is at that time employed by Franchisor,

- c. Solicit any current, former or prospective customer solicited by Franchisee's Franchised Business or any other customer of whom Franchisee has become aware as a result of access to Franchisor's System, or other franchisees for any competitive purpose.
- 3. Intent and Enforcement. It is the parties' intent that the provisions of this Article III be judicially enforced to the fullest extent permissible under applicable law. Accordingly, the parties agree that any reduction in scope or modification of any part of the noncompetition provisions contained herein shall not render any other part unenforceable. In the event of the actual or threatened breach of this Article III by you, any of your principals, or any members of their immediate family, Franchisor shall be entitled to an injunction restraining such person from any such actual or threatened breach. You agree that in the event of the actual or threatened breach of this Article III, Franchisor's harm will be irreparable, and that Franchisor has no adequate remedy at law to prevent such harm. You acknowledge and agree that each of you has previously worked or been gainfully employed in other careers and that the provisions of this Article III in no way prevent you from earning a living. You further acknowledge and agree that the time limitation of this Article III shall be tolled during any default under this Guaranty.

ARTICLE IV DISPUTE RESOLUTION

- 1. **Acknowledgment**. You acknowledge that this Guaranty is not a Franchise Agreements and does not confer upon you any rights to use the Franchisor's Proprietary Marks or its System.
- 2. **Governing Law.** This Guaranty shall be deemed to have been made in and governed by the laws of the State of North Carolina, without any reference to North Carolina conflict of laws principles.
- Internal Dispute Resolution. Franchisee and Franchisor believe that it is important to resolve any disputes amicably, quickly, cost-effectively and professionally and to return to business as soon as possible. Franchisee and Franchisor have agreed that the provisions of Sections 3 and 4 of this Personal Guaranty support these mutual objectives and, therefore, agree as follows. Any disagreement, litigation, claim, dispute, suit, action, controversy or proceeding of any type whatsoever, including any claim for equitable relief and/or where Franchisee is acting as a "private attorney general" suing pursuant to a statutory claim or otherwise between or involving a Guarantor, Franchisee and/or Franchisor, on whatever theory and/or facts based, and whether or not arising out of this Guaranty (including any dispute or disagreement relating to arbitration, including the arbitrability of this Guaranty or any of its provisions), or offer, sale or negotiation of a Top Rail Fence Franchise, or the relationship of the parties arising from the Franchise or from entering this Guaranty or the Franchise Agreements, or any claim that this Guaranty or the Franchise Agreements or any part thereof is invalid, illegal or otherwise voidable, void or unenforceable ("Dispute") will be processed in the following manner, Franchisee and Franchisor each expressly waiving all rights to any court proceeding, except as expressly provided below in Section 4.1. Personal Guarantor must first bring any claim or Dispute between Franchisee and Franchisor or Personal Guarantor and Franchisor to Franchisor's management after providing notice as set forth in Section 4.8 below. Personal Guarantor must exhaust this internal Dispute resolution procedure before Personal Guarantor may bring Personal Guarantor's Dispute before a third party. This agreement to first attempt resolution of Disputes internally shall survive termination or expiration of this Guaranty. Personal Guarantor and Franchisor agree that the franchise relationship is unique and that as a result, it is important that anyone who serves as a mediator or arbitrator in a Dispute must have a minimum of 7 years' substantive experience in franchise
- 4. Mediation/Arbitration. If Personal Guarantor and Franchisor are unable to resolve a Dispute using the internal Dispute resolution procedure, then Disputes, which are not first resolved through the internal Dispute resolution procedure set forth in Section 3 above must be submitted first to non-binding mediation in Mecklenburg County, North Carolina under the auspices of the American Arbitration Association ("AAA"), in accordance with AAA's Commercial Mediation Rules then in effect. The

mediation will be conducted by one mediator with at least seven years of substantive experience in franchise law who will be appointed under the AAA's Commercial Mediation Rules and who will conduct the mediation in accordance with such rules. Mediation must be conducted on an individual, not a class-wide, basis; only Franchisor (and/or its affiliates and its and their respective owners, officers, directors, agents and employees) and Franchisee (and/or its Owners, guarantors, affiliates, officers, directors, agents and employees, if applicable) may be the parties to any mediation proceeding described in this Section, and no such mediation proceeding may be consolidated with any other mediation proceeding between Franchisor and any other person, corporation, limited liability company or partnership. Franchisor and Personal Guarantor agree that statements made by Franchisor, Personal Guarantor or any other party in any such mediation proceeding will not be admissible in any arbitration or other legal proceeding. If the mediation proceedings have been terminated either: (i) as the result of a written declaration of the mediator(s) that further mediation efforts are not worthwhile; or (ii) if the mediation does not resolve the dispute by written agreement within 14 days after the mediation, then, subject to the exclusions set forth in Section 4.1, all Disputes must be submitted to binding arbitration before one arbitrator of the AAA in accordance with its commercial arbitration rules. The arbitration will be conducted by one arbitrator with at least seven years of substantive experience in franchise law who will be appointed under the AAA's Commercial Arbitration Rules and who will conduct the arbitration in accordance with such rules. Except as provided in Section 4.1, no party may file litigation involving any Dispute unless it has complied with the mediation and arbitration provisions in this Article regarding that Dispute. "Filing litigation" includes asserting a counterclaim, third-party claim or other claim relating to a Dispute (irrespective of whether the party asserting the claim filed the legal action in which the claim is asserted). Franchisor's rights to mediation and arbitration, as set forth herein may be specifically enforced by Franchisor. Each party shall bear its own cost of mediation and arbitration and Franchisor and Personal Guarantor shall share mediation costs equally. This agreement to mediate and arbitrate shall survive any termination or expiration of this Guaranty.

- 4.11 The parties shall not be required to first attempt to mediate or arbitrate a controversy, Dispute, or claim through mediation as set forth in this Section 4.1 if such controversy, Dispute, or claim concerns an allegation that a party has violated (or threatens to violate, or poses an imminent risk of violating):
- 4.11.1 Any federally protected intellectual property rights in the Proprietary Marks, the System, or in any Confidential Information;
 - 4.11.2 Any claims pertaining to or arising out of any warranty issue;
 - 4.11.3 Any of the restrictive covenants contained in this Guaranty;
 - 4.11.4 Any of Franchisee's payment obligations that are more than forty-five (45)

days past due;

- 4.11.5 Any claims arising out of or related to fraud or misrepresentation by Personal Guarantor, Franchisee, or Franchisee's insolvency;
- 4.11.6 Any claims relating to Franchisee's obligations on termination or expiration of this Guaranty;
- 4.11.7 Any claims relating to any Transfer of an interest in Franchisee, the Franchised Business or Franchisee's assets; or
 - 4.11.8 Any matters involving danger, health or safety.
- 4.12 **Interpretation.** The provisions of this Guaranty must be construed as independent of any other covenant or provision of this Guaranty. However, if a court or arbitrator of competent

jurisdiction determines that any provisions are unlawful, that court or arbitrator is to modify or interpret the provisions to the minimum extent necessary to have them comply with the law. Notwithstanding any provision relating to the state laws by and under which this Guaranty must be governed and construed, all issues relating to arbitrability or the enforcement of the agreement to arbitrate in this Guaranty must be governed by the United States Arbitration Act (9 U.S.C. § 1 et seq.) and the Federal common law of arbitration. The arbitrator, and not any federal, state or local court, shall have exclusive authority to resolve any Dispute relating to the interpretation, applicability, unconscionability, arbitrability, enforceability or formation of this Guaranty, including any claim of fraud in the inducement or that all or any part of the Guaranty is void or voidable. However, the preceding sentence shall not apply to the clause entitled "Class Action Waiver."

- 4.13 **Judgment**. Judgment on an arbitration award may be entered in any court of competent jurisdiction. This judgment is binding, final and non-appealable.
- 4.14 **Failure to Appear.** The arbitration and mediation provisions in this Article are self-executing and remain in full force and effect after the expiration or sooner termination of this Guaranty. If either party fails to appear at any properly noticed arbitration proceeding, notwithstanding failing to appear, an award may be entered against that party by default or otherwise.
- Class Action Waiver. Any proceeding (whether mediation, arbitration, trial by a court or jury, appeal or otherwise) must be brought in the parties' individual capacity and not as a plaintiff or class member in any purported class, collective, representative, multiple plaintiff or similar proceeding ("Class Action"). Personal Guarantor and Franchisor expressly waive any ability to maintain any Class Action in any forum. Further, an arbitration proceeding between Franchisor and Personal Guarantor (or any of Franchisee's or Franchisor's affiliates and owners and guarantors) may not be consolidated with any other arbitration proceeding between them and any other franchisee, person or entity. Personal Guarantor hereby agrees not to seek joinder of any of its claims with those of any other party. The arbitrator shall not have authority to combine or aggregate similar claims or conduct any Class Action nor make an award to any person or entity not a party to the arbitration. Any claim that all or part of this Class Action Waiver is unenforceable, unconscionable, void or voidable may be determined only by a court of competent jurisdiction and not by an arbitrator. PERSONAL GUARANTOR AND FRANCHISOR UNDERSTAND THAT FRANCHISOR WOULD HAVE HAD A RIGHT TO LITIGATE THROUGH A COURT, TO HAVE A JUDGE OR JURY DECIDE ITS CASE, AND TO BE PARTY TO A CLASS OR REPRESENTATIVE ACTION. HOWEVER, PERSONAL GUARANTOR AND FRANCHISOR UNDERSTAND AND CHOOSE TO WAIVE THAT RIGHT AND HAVE ANY CLAIMS DECIDED INDIVIDUALLY THROUGH ARBITRATION. It is Personal Guarantor's and Franchisor's joint business judgment that the limitations of this subsection make good business sense, because:
- 4.15.1 the mediation and arbitration procedures contemplated by this Guaranty (and which Personal Guarantor and Franchisor agree are the core methods for resolving Disputes) function most effectively on an individual case basis;
- 4.15.2 there are significant business and other factors present in each individual Personal Guarantor's situation which should be respected; and
- 4.15.3 the economic interests of lawyers on either side in class-wide or multiple plaintiff disputes, as well as the tendency to polarize positions makes accommodation and compromise, as a practical business matter, less easily achieved, which would be a serious detriment to Personal Guarantor's and Franchisor's business interests, as well as those of the entire Franchise System, in quickly, amicably and economically resolving any Dispute.
- 4.16 **Selection of Venue**. Nothing contained in this Guaranty shall prevent Franchisor from applying to and obtaining from any court having jurisdiction, a writ of attachment, a

temporary injunction, preliminary injunction and/or other emergency relief available to safeguard and protect Franchisor's interests. The parties expressly agree to the jurisdiction and venue of any court of general jurisdiction in Mecklenburg County, North Carolina and the jurisdiction and venue of the United States District Court for the Western District of North Carolina. Personal Guarantor acknowledges that this Guaranty has been entered into in the State of North Carolina, and that Franchisee, whose obligations Personal Guarantor is guaranteeing, is to receive valuable and continuing services emanating from Franchisor's headquarters in Huntersville, North Carolina, including, but not limited to, training, assistance, support and the development of the System. In recognition of such services and their origin, Personal Guarantor hereby irrevocably consents to the personal jurisdiction of the state and federal courts of North Carolina as set forth in this Section.

- 4.17 **Third Party Beneficiaries**. Franchisor's officers, directors, shareholders, members, agents and/or employees are express third party beneficiaries of the provisions of this Guaranty, including the mediation provision set forth in this Section 4, each having authority to specifically enforce the right to mediate claims asserted against such person(s) by Personal Guarantor.
- 4.18 Prior Notice of Claims. As a condition precedent to commencing an action for damages or for violation or breach of this Guaranty, Personal Guarantor must notify Franchisor within 30 days after the occurrence of the violation or breach, and failure to timely give such notice shall preclude any claim for damages.
- 4.19 **No Right to Offset**. Personal Guarantor shall not withhold all or any part of any payment to Franchisor or any of its affiliates on the grounds of Franchisor's alleged nonperformance or as an offset against any amount Franchisor or any of Franchisor's affiliates allegedly may owe Franchisee under this Guaranty or any related agreements.
- Injunctive Relief. The Franchised Business is intended to be one of a large number of businesses identified by the Proprietary Marks selling to the public the Approved Products and Approved Services associated with the Proprietary Marks. Consequently, a single Personal Guarantor's failure to comply with the terms of its Franchise Agreements is likely to cause irreparable damage to Franchisor, and damages at law would therefore be an inadequate remedy. On Personal Guarantor's breach or threatened breach of any of the terms concerning any matters referenced in Section 4.1, Franchisor may seek an injunction restraining the breach and/or a decree of specific performance (with recovery of reasonable attorneys' fees and costs incurred in obtaining equitable relief). Franchisor may do so without demonstrating or proving any actual damage. These equitable remedies are in addition to all other rights or remedies to which Franchisor may otherwise be entitled because of Personal Guarantor's breach of this Guaranty. Franchisor may seek this relief without posting any bond or security. However, if a court of competent jurisdiction requires a bond or security, a bond or security for \$1,000 is sufficient. Notwithstanding anything in this Guaranty to the contrary, Franchisor may seek injunctive relief in any jurisdiction that has jurisdiction over Personal Guarantor or any other party against whom the relief is sought. If injunctive relief is granted, Personal Guarantor's only remedy will be the court's dissolution of the injunctive relief. If the injunctive relief was wrongfully issued, Personal Guarantor expressly waives all claims for damages Personal Guarantor incurred as a result of the wrongful issuance.
- 5. **Jurisdiction and Venue**. Nothing contained in this Guaranty shall prevent Franchisor from applying to and obtaining from any court having jurisdiction, a writ of attachment, a temporary injunction, preliminary injunction, and/or other emergency relief available to safeguard and protect Franchisor's interests. The parties expressly agree to the jurisdiction and venue of any court of general jurisdiction in Mecklenburg County, North Carolina and the jurisdiction and venue of the United States District Court for the Western District of North Carolina. The parties acknowledge and agree that this Guaranty has been entered into in the State of North Carolina, and that Franchisee, whose obligations Personal Guarantor is guaranteeing, is to receive valuable and continuing services emanating from Franchisor's headquarters in Huntersville, North Carolina, including, but not limited to, training, assistance,

support and the development of the System. In recognition of such services and their origin, you hereby irrevocably consent to the personal jurisdiction of the state and federal courts of North Carolina as set forth herein.

- 6. Jury Trial Waiver. THE PARTIES HEREBY AGREE TO WAIVE TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM, WHETHER AT LAW OR EQUITY, REGARDLESS OF WHICH PARTY BRINGS SUIT. THIS WAIVER SHALL APPLY TO ANY MATTER WHATSOEVER BETWEEN THE PARTIES HERETO WHICH ARISES OUT OF OR IS RELATED IN ANY WAY TO THIS PERSONAL GUARANTY OR THE FRANCHISE AGREEMENTS, THE PERFORMANCE OF EITHER PARTY, AND/OR YOUR PURCHASE FROM FRANCHISOR OF THE FRANCHISE, OPTION AND/OR ANY PRODUCTS OR SERVICES.
- 7. Waiver of Punitive Damages. You waive, to the fullest extent permitted by law, any right to, or claim for, any punitive, exemplary, incidental, indirect, special or consequential damages (including, without limitation, lost profits) which you may have against Franchisor arising out of any cause whatsoever (whether such cause be based in contract, negligence, strict liability, other tort or otherwise) and agree that in the event of a Dispute, your recovery shall be limited to actual damages. If any other term of this Guaranty is found or determined to be unconscionable or unenforceable for any reason, the foregoing provisions shall continue in full force and effect, including, without limitation, the waiver of any right to claim any consequential damages.
- 8. <u>Limitation on Action.</u> You agree that no cause of action arising out of or under this Guaranty or the Franchise Agreements may be maintained by you unless brought before the expiration of 1 year after the act, transaction or occurrence upon which such action is based or the expiration of 1 year after you become aware of facts or circumstances reasonably indicating that you may have a claim against the Franchisor, whichever occurs sooner, and that any action not brought within this period shall be barred as a claim, counterclaim, defense or set-off. Notwithstanding anything to the contrary contained herein, all actions will be conducted on an individual, not a class-wide, basis and any proceeding between you, Franchisee, and Franchisor or its affiliates or employees may not be consolidated with any other proceeding between Franchisor and any other person or entity.
- 9. Attorneys' Fees. If the undersigned is in breach or default of any monetary or non-monetary material obligation under the Franchise Agreements or this Guaranty, or any related agreement between Franchisee and Franchisor and/or Franchisor's affiliates, and Franchisor engages an attorney to enforce Franchisor's rights (whether or not formal judicial proceedings are initiated), Franchisee must pay all reasonable attorneys' fees, court costs and litigation expenses Franchisor incurs. If the undersigned or Franchisee institutes any legal action to interpret or enforce the terms of this Guaranty or the Franchise Agreements, and the claim(s) in such action is denied or the action is dismissed, Franchisor is entitled to recover Franchisor's reasonable attorneys' fees, and all other reasonable costs and expenses incurred in defending against same, and to have such an amount awarded as part of the judgment in the proceeding.
- 10. Nonwaiver. Franchisor's failure to insist upon strict compliance with any provision of this Guaranty and the Franchise Agreements shall not be a waiver of Franchisor's right to do so, any law, custom, usage or rule to the contrary notwithstanding. Delay or omission by Franchisor respecting any breach or default shall not affect Franchisor's rights respecting any subsequent breaches or defaults. All rights and remedies granted in this Guaranty shall be cumulative. Franchisor's election to exercise any remedy available by law or contract shall not be deemed a waiver or preclude exercise of any other remedy.
- 11. **Severability.** The parties agree that if any provisions of this Guaranty may be construed in two ways, one of which would render the provision illegal or otherwise voidable or unenforceable and the other which would render it valid and enforceable, such provision shall have the meaning that renders it valid and enforceable. The language of all provisions of this Guaranty shall be construed according to

fair meaning and not strictly construed for or against either party. The provisions of this Guaranty are severable, and this Guaranty shall be interpreted and enforced as if all completely invalid or unenforceable provisions were not contained herein, and partially valid and enforceable provisions shall be enforced to the extent that they are valid and enforceable. If any material provision of this Guaranty shall be stricken or declared invalid, the parties agree to negotiate mutually acceptable substitute provisions. In the event that the parties are unable to agree upon such provisions, Franchisor reserves the right to terminate this Guaranty.

- 12. Construction of Language. Any term defined in the Franchise Agreements that is not defined in this Guaranty will be ascribed the meaning given to it in the Franchise Agreements. The language of this Guaranty will be construed according to its fair meaning, and not strictly for or against either party. All words in this Guaranty refer to whatever number or gender the context requires. If more than one party or person is referred to as you, their obligations and liabilities must be joint and several. Headings are for reference purposes and do not control interpretation.
- 13. Successors. References to "Franchisor" or "the undersigned," or "you" include the respective parties' successors, assigns or transferees.
- 14. No Personal Liability. You agree that fulfillment of any and all of Franchisor's obligations written in this Guaranty or in the Franchise Agreements or based on any oral communications which may be ruled to be binding in a Court of Law shall be Franchisor's sole responsibility and none of Franchisor's agents, representatives, nor any individuals associated with Franchisor's franchise company shall be personally liable to Franchisee or you for any reason.

IN WITNESS WHEREOF, AND INTENDING TO BE LEGALLY BOUND HEREBY, THE PARTIES HERETO HAVE CAUSED THIS GUARANTY TO BE EXECUTED EFFECTIVE THE DATE FIRST SET FORTH ABOVE.

PERSONAL GUARANTOR:

, Individually	, Individually

PERSONAL GUARANTOR:

Exhibit H to FDD HFB FenceCo Franchising, LLC

STATE SPECIFIC ADDENDA

CALIFORNIA ADDENDUM TO FRANCHISE DISCLOSURE DOCUMENT

To the extent the California Franchise Investment Law, Cal. Corp. Code §§ 31000-31516 or the California Franchise Relations Act, Cal. Bus. & Prof. Code §§20000-20043 applies, the terms of this Addendum apply.

THE CALIFORNIA FRANCHISE INVESTMENT LAW REQUIRES THAT A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE BE DELIVERED TOGETHER WITH THE DISCLOSURE DOCUMENT.

OUR WEBSITE, WWW.TOPRAILFENCE.COM, HAS NOT BEEN REVIEWED OR APPROVED BY THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION AND INNOVATION AND COMPLAINTS CONCERNING THE CONTENTS OF THIS WEBSITE MAY BE DIRECTED TO THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION AND INNOVATION AT WWW.DFPI.CA.GOV.

SECTION 31125 OF THE CALIFORNIA CORPORATIONS CODE REQUIRES US TO GIVE YOU A DISCLOSURE DOCUMENT, IN A FORM CONTAINING THE INFORMATION THAT THE COMMISSIONER MAY BY RULE OR ORDER REQUIRE, BEFORE A SOLICITATION OF A PROPOSED MATERIAL MODIFICATION OF AN EXISTING FRANCHISE.

The registration of this franchise offering by the California Department of Financial Protection and Innovation does not constitute approval, recommendation, or endorsement by the commissioner.

Item 3, Additional Disclosure:

Neither we nor any person described in Item 2 of the Disclosure Document is subject to any currently effective order of any National Securities Association or National Securities Exchange, as defined in the Securities Exchange Act of 1934, 15 U.S.C.A. 78a et seq. suspending or expelling such persons from membership in such association or exchange.

Item 6, Additional Disclosure:

The highest interest rate allowed by law in California is 10% annually or 5% annually plus the prevailing interest rate charged to banks by the Federal Reserve Bank of San Francisco, whichever is higher.

Item 17, Additional Disclosures:

The franchise agreement requires franchisee to execute a general release of claims upon renewal or transfer of the franchise agreement. California Corporations Code Section 31512 provides that any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with any provision of that law or any rule or order thereunder is void. Section 31512 voids a waiver of your rights under the Franchise Investment Law (California Corporations Code Section 20010 voids a waiver of your rights under the Franchise Relations Act (Business and Professions Code Sections 20000 – 20043)).

The franchise agreement requires application of the laws of the State of North Carolina. This provision may not be enforceable under California law.

<u>California Business and Professions Code Sections 20000 through 20043 provide rights to the franchisee concerning termination, transfer or non-renewal of a franchise. If the franchise agreement contains a provision that is inconsistent with the law, the law will control.</u>

The franchise agreement provides for termination upon bankruptcy. This provision may not be enforceable under federal bankruptcy law (11 U.S.C.A. §101 et seq.).

The franchise agreement requires binding arbitration. The arbitration will occur in Charlotte, North Carolina with the cost being borne by the parties as determined by the arbitrator. Prospective franchisees are encouraged to consult with private legal counsel to determine the applicability of California and federal laws (such as Business and Professions Code Section 20040.5, Code of Civil Procedure Section 1281, and the Federal Arbitration Act) to any provisions of a franchise agreement restricting venue to a forum outside the State of California.

The franchise agreement contains a covenant not to compete which extends beyond the termination of the franchise. This provision may not be enforceable under California law.

<u>California's Franchise Investment Law (Corporations Code sections 31512 and 31512.1)</u> states that any provision of a franchise agreement, franchise disclosure document, acknowledgment, questionnaire or other writing, including any exhibit thereto, disclaiming or denying any of the following shall be deemed contrary to public policy and shall be void and unenforceable. The law also prohibits a franchisor from disclaiming or denying:

- (a) Representations it, its employees, or its agents make to you,
- (b) your ability to rely on any representations it makes to you, or
- (c) any violations of the law.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

Fee Deferral

The Department of Financial Protection and Innovation requires that the franchisor defer the collection of all initial fees from California franchisees until the franchisor has completed all its preopening obligations and franchisee is open for business. For any development agreement, the payment of the development and initial fee attributable to a specific unit is deferred until that unit is open.

CALIFORNIA ADDENDUM TO FRANCHISE AGREEMENT

To the extent the California Franchise Investment Law, Cal. Corp. Code §§ 31000-31516 or the California Franchise Relations Act, Cal. Bus. & Prof. Code §§20000-20043 applies, the terms of this Addendum apply.

1. Notwithstanding anything to the contrary contained in the Franchise Agreement, to the extent that the Franchise Agreement contains provisions that are inconsistent with the following, such provisions are hereby amended:

The Franchise Agreement requires franchisee to execute a general release of claims upon renewal or transfer of the franchise agreement. California Corporations Code Section 31512 provides that any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with any provision of that law or any rule or order there under is void. Section 31512 voids a waiver of your rights under the Franchise Investment Law (California Corporations Code Section 20010 voids a waiver of your rights under the Franchise Relations Act (Business and Professions Code Sections 20000 – 20043)).

The Franchise Agreement requires application of the laws of the State of North Carolina. This provision may not be enforceable under California law.

California Business and Professions Code Sections 20000 through 20043 provide rights to the franchisee concerning termination or non-renewal of a franchise. The Federal Bankruptcy Code also provides rights to franchisee concerning termination of the Franchise Agreement upon certain bankruptcy-related events. If the Franchise Agreement is inconsistent with the law, the law will control.

The Franchise Agreement requires binding arbitration. The arbitration will occur in Charlotte, North Carolina with the cost being borne by the parties as determined by the arbitrator. Prospective franchisees are encouraged to consult with private legal counsel to determine the applicability of California and federal laws (such as Business and Professions Code Section 20040.5, Code of Civil Procedure Section 1281, and the Federal Arbitration Act) to any provisions of a franchise agreement restricting venue to a forum outside the State of California.

The Franchise Agreement contains a covenant not to compete which extends beyond the termination of the franchise. This provision may not be enforceable under California law.

California's Franchise Investment Law (Corporations Code sections 31512 and 31512.1) states that any provision of a franchise agreement, franchise disclosure document, acknowledgment, questionnaire or other writing, including any exhibit thereto, disclaiming or denying any of the following shall be deemed contrary to public policy and shall be void and unenforceable. The law also prohibits a franchisor from disclaiming or denying:

- (a) Representations it, its employees, or its agents make to you,
- (b) your ability to rely on any representations it makes to you, or
- (c) any violations of the law.
- Any capitalized terms that are not defined in this Addendum shall have the meaning given them in the Franchise Agreement.

- 3. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 4. **Fee Deferral.** The Department of Financial Protection and Innovation requires that the franchisor defer the collection of all initial fees from California franchisees until the franchisor has completed all its pre-opening obligations and franchisee is open for business. For any development agreement, the payment of the development and initial fee attributable to a specific unit is deferred until that unit is open.
- 5. Except as expressly modified by this Addendum, the Franchise Agreement remains unmodified and in full force and effect.

This Addendum is being entered into in connection with the Franchise Agreement. In the event of any conflict between this Addendum and the Franchise Agreement, the terms and conditions of this Addendum shall apply.

IN WITNESS WHEREOF, the undersigned does hereby acknowledge receipt of this addendum and have executed this Addendum as of the date Franchisor signs below.

FRANCHISOR:	FRANCHISEE:
HFB FENCECO FRANCHISING, LLC, a North Carolina limited liability company	
By:	By:
Name:	Name:
Title:	Title:
Date:	Date:
	By:
	Name:
	<u>Title:</u>
	Date:

HAWAII ADDENDUM

TO FRANCHISE DISCLOSURE DOCUMENT

To the extent the Hawaii Franchise Investment Law, Hawaii Rev. Stat. \$482E-1 - 482E-12 applies, the terms of this Addendum apply.

THESE FRANCHISES WILL BE/HAVE BEEN FILED UNDER THE FRANCHISE INVESTMENT LAW OF THE STATE OF HAWAII. FILING DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION OR ENDORSEMENT BY THE DIRECTOR OF COMMERCE—AND CONSUMER AFFAIRSREGULATORY AGENCIES OR A FINDING BY THE DIRECTOR OF COMMERCE AND CONSUMER AFFAIRSREGULATORY AGENCIES THAT THE INFORMATION PROVIDED HEREIN IS TRUE, COMPLETE AND NOT MISLEADING.

THE FRANCHISE INVESTMENT LAW MAKES IT UNLAWFUL TO OFFER OR SELL ANY FRANCHISE IN THIS STATE WITHOUT FIRST PROVIDING TO THE PROSPECTIVE FRANCHISEE, OR SUBFRANCHISOR, AT LEAST SEVEN DAYS PRIOR TO THE EXECUTION BY THE PROSPECTIVE FRANCHISEE OF ANY BINDING FRANCHISE OR OTHER AGREEMENT, OR AT LEAST SEVEN DAYS PRIOR TO THE PAYMENT OF ANY CONSIDERATION BY THE FRANCHISEE, OR SUBFRANCHISOR, WHICHEVER OCCURS FIRST, A COPY OF THE FRANCHISE DISCLOSURE DOCUMENT, TOGETHER WITH A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE.

THIS FRANCHISE DISCLOSURE DOCUMENT CONTAINS A SUMMARY ONLY OF CERTAIN MATERIAL PROVISIONS OF THE FRANCHISE AGREEMENT. THE CONTRACT OR AGREEMENT SHOULD BE REFERRED TO FOR A STATEMENT OF ALL RIGHTS, CONDITIONS, RESTRICTIONS AND OBLIGATIONS OF BOTH THE FRANCHISOR AND THE FRANCHISEE.

FRANCHISOR:	FRANCHISEE:
HFB FENCECO FRANCHISING, LLC, a North Carolina limited liability company	<u>a(n)</u>
By:	By:
Name:	Name:
Title:	Title:
Date:	Date:

HAWAII ADDENDUM TO FRANCHISE AGREEMENT

To the extent the Hawaii Franchise Investment Law, Hawaii Rev. Stat. §\$482E-1 – 482E-12 applies, the terms of this Addendum apply.

- 1. The franchisor having filed the Franchise Disclosure Document under the Franchise Investment Law of The State of Hawaii does not constitute approval, recommendation or endorsement by the Director of Commerce and Consumer Affairs or a finding by the Director of Commerce and Consumer Affairs that the information provided therein is true, complete and not misleading.
- 2. In accordance with the Hawaii Franchise Investment Law, we have provided to you, at least seven (7) days prior to your execution of any binding franchise or other agreement, or at least seven (7) days prior to your payment of any consideration, whichever occurs first, a copy of the Franchise Disclosure Document, together with a copy of all proposed agreements relating to the sale of the franchise.
- 3. The Franchise Disclosure Document contains only a summary of certain material provisions of the Franchise Agreement. You should refer to the Franchise Agreement for a statement of all rights, conditions, restrictions and obligations of both us and you.

IN WITNESS WHEREOF, the undersigned have executed this Addendum as of the date Franchisor signs below:

4. No statement, questionnaire, or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

<u>IN WITNESS WHEREOF, the undersigned have executed this Addendum as of the date Franchisor signs below.</u>

FRANCHISOR:	FRANCHISEE:
HFB FenceCo Franchising, LLC	
a North Carolina limited liability company	a(n)
By:	By:
Name:	Name:
Title:	Title:
Date:	Date:

	By:
	Name:
	Title:
	Date:
FRANCHISOR:	FRANCHISEE:
HFB FENCECO FRANCHISING, LLC, a North Carolina limited liability company	
a rotal caroma minea money company	<u>a(n)</u>
By:	Ву:
Name:	Name:
Title:	Title:
Date:	Date:

ILLINOIS ADDENDUM TO FRANCHISE DISCLOSURE DOCUMENT

To the extent the Illinois Franchise Disclosure Act, Ill. Comp. Stat. \$\$705/1 - 705/44 applies, the terms of this Addendum apply.

Item 17, Additional Disclosures:

The following statements are added to Item 17:

- 1. Illinois law governs the Franchise Agreement.
- 2. <u>In conformance with Section 4 of the Illinois Franchise Disclosure Act</u>, any provision in a franchise agreement that designates jurisdiction and venue in a forum outside of the State of Illinois is void. However, a franchise agreement may provide for arbitration to take place outside of Illinois.
- 3. Your rights upon termination and non-renewal of an agreement are set forth in Sections 19 and 20 of the Illinois Franchise Disclosure Act.
- 4. In conformance with Section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.
- 5. No statement, questionnaire or acknowledgement signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of: (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on behalf of the Franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 6. "NATIONAL ACCOUNTS" EXIST IN THIS FRANCHISE SYSTEM. FRANCHISOR RESERVES THE RIGHT TO ESTABLISH, IDENTIFY, NEGOTIATE THE TERMS FOR AND SERVICE "NATIONAL ACCOUNTS" WITHIN YOUR TERRITORY. FRANCHISOR OR A 3rd PARTY DEIGNATED BY THE FRANCHISOR MAY PROVIDE PRODUCTS & SERVICES TO A "NATIONAL ACCOUNT". IN YOUR TERRITORY WITH NO COMPENSATION PAID TO YOU.

ILLINOIS ADDENDUM TO FRANCHISE AGREEMENT

To the extent the Illinois Franchise Disclosure Act, Ill. Comp. Stat. \$\$705/1 - 705/44 applies, the terms of this Addendum apply.

- 1. Notwithstanding anything to the contrary contained in the Franchise Agreement, to the extent that the Franchise Agreement contains provisions that are inconsistent with the following, such provisions are hereby amended:
 - a. Illinois law governs the Franchise Agreement.
 - b. In conformance with—Section 4 of the Illinois Franchise Disclosure Act, any provision in a franchise agreement that designates jurisdiction or venue in a forum outside of the State of Illinois is void. However, a franchise agreement may provide for arbitration to take place outside of Illinois.
 - c. Your rights upon termination and non-renewal of a franchise agreement are set forth in Sections 19 and 20 of the Illinois Franchise Disclosure Act.
 - d. Section 41 of the Illinois Franchise Disclosure Act provides that any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.
 - e. In conformance with section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.
- 2. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of: (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 3. "National Accounts" exist in this franchise System. The Franchisor reserves the right to establish, identify, negotiate the terms for, and service "National Accounts" within your territory. The Franchisor or a third party designated by the Franchisor may provide products and services to a "National Account" within your territory with no compensation paid to you.
- 4. Except as expressly modified by this Addendum, the Franchise Agreement remains unmodified and in full force and effect.

This Addendum is being entered into in connection with the Franchise Agreement. In the event of any conflict between this Addendum and the Franchise Agreement, the terms and conditions of this Addendum shall apply.

IN WITNESS WHEREOF, the undersigned have executed this Addendum as of the date Franchisor signs below.

FRANCHISOR:	FRANCHISEE:
HFB FenceCo FranchisingFENCECO FRANCHISING, LLC a North Carolina limited liability company	a(n)
	Ву:
Ву:	Name:
Name:	Title:
Title:	Date:
Date:	
	By:
	Name:
	Title:
	Date:

MARYLAND ADDENDUM TO FRANCHISE DISCLOSURE DOCUMENT

To the extent the Maryland Franchise Registration and Disclosure Law, Md. Code Bus. Reg. §§14-201 – 14-233 applies, the terms of this Addendum apply.

Item 17, Additional Disclosures:

- 1. Our termination of the Franchise Agreement because of your bankruptcy may not be enforceable under applicable federal law (11 U.S.C.A. 101 et seq.).
- 2. You may bring a lawsuit in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law.
- 3. Any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.
- 4. The general release required as a condition of renewal, sale and/or assignment/transfer will not apply to any liability under the Maryland Franchise Registration and Disclosure Law.

Franchisee Acknowledgment / Compliance Certification:

The representations under this Franchise Acknowledgment/Compliance Certification are not intended, nor shall they act as a release, estoppel or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of: (i) waiving any claims under any applicable state franchise law, including fraud in the inducement; or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

Item 5, Additional Disclosures:

Based upon the franchisor's financial condition, the Maryland Securities Commissioner has required a financial assurance. Therefore, all initial fees and payments owed by franchisees shall be deferred until the franchisor completes its pre-opening obligations under the franchise agreement. In addition, all development fees and initial payments by area developers shall be deferred until the first franchise under the multi-unit addendum is open.

MARYLAND ADDENDUM TO FRANCHISE AGREEMENT

To the extent the Maryland Franchise Registration and Disclosure Law, Md. Code Bus. Reg. §§14-201 – 14-233 applies, the terms of this Addendum apply.

- 1. Notwithstanding anything to the contrary contained in the Franchise Agreement, to the extent that the Franchise Agreement contains provisions that are inconsistent with the following, such provisions are hereby amended:
 - a. All representations requiring prospective franchisees to assent to a release, estoppel or waiver of liability are not intended to nor shall they act as a release, estoppel or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law.
 - b. Nothing in the Franchise Agreement prevents the franchisee from bringing a lawsuit in Maryland for claims arising under the Maryland Franchise Registration and Disclosure Law.
 - c. Nothing in the Franchise Agreement operates to reduce the 3-year statute of limitations afforded to a franchisee for bringing a claim arising under the Maryland Franchise Registration and Disclosure Law. Further, any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.
 - d. The Federal Bankruptcy laws may not allow the enforcement of the provisions for termination upon bankruptcy of the franchisee.
- 2. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of: (i) waiving any claims under any applicable state franchise law, including fraud in the inducement; or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 3. Based upon the franchisor's financial condition, the Maryland Securities Commissioner has required a financial assurance. Therefore, all initial fees and payments owed by franchisees shall be deferred until the franchisor completes its pre-opening obligations under the franchise agreement. In addition, all development fees and initial payments by area developers shall be deferred until the first franchise under the multi-unit addendum is open.
- 4. Any capitalized terms that are not defined in this Addendum shall have the meaning given them in the Franchise Agreement.
- 5. Except as expressly modified by this Addendum, the Franchise Agreement remains unmodified and in full force and effect.

This Addendum is being entered into in connection with the Franchise Agreement. In the event of any conflict between this Addendum and the Franchise Agreement, the terms and conditions of this Addendum shall apply.

(Signature Page Follows)

IN WITNESS WHEREOF, the undersigned signs below.	d have executed this Addendum as of the date Franchiso
FRANCHISOR:	FRANCHISEE:
HFB FENCECO FRANCHISING, LLC, a North Carolina limited liability company	<u>a(n)</u>
Ву:	By:
Name:	Name:
Title:	Title:
Date:	Date:

MINNESOTA ADDENDUM TO FRANCHISE DISCLOSURE DOCUMENT

To the extent the Minnesota Franchise Act, Minn. Stat. \$\$80C.01 - 80C.22 applies, the terms of this Addendum apply.

State Cover Page and Item 17, Additional Disclosures:

Minn. Stat. Sec. 80C.21 and Minn. Rule 2860.4400J prohibit us from requiring litigation to be conducted outside of Minnesota, requiring waiver of a jury trial or requiring the franchisee to consent to liquidated damages, termination penalties or judgment notes. In addition, nothing in the Disclosure Document shall abrogate or reduce any of your rights as provided for in Minn. Stat. Sec. 80C, or your rights to any procedure, forum or remedies provided for by the laws of the jurisdiction.

Franchisee cannot consent to the franchisor obtaining injunctive relief. The franchisor may seek injunctive relief. A court will determine if a bond is required.

Supplier Control. You must purchase all or nearly all of the inventory or supplies that are necessary to operate your business from the franchisor, its affiliates, or suppliers that the franchisor designates, at prices that the franchisor or they set. These prices may be higher than prices you could obtain elsewhere for the same or similar goods. This may reduce the anticipated profit of your franchised business.

Item 6, Additional Disclosure:

NSF checks are governed by Minn. Stat. 604.113, which puts a cap of \$30 on service charges.

Item 13, Additional Disclosures:

The Minnesota Department of Commerce requires that a franchisor indemnify Minnesota franchisees against liability to third parties resulting from claims by third parties that the franchisee's use of the franchisor's trademark infringes upon the trademark rights of the third party. The franchisor does not indemnify against the consequences of a franchisee's use of a franchisor's trademark except in accordance with the requirements of the franchise agreement, and as the condition to an indemnification, the franchisee must provide notice to the franchisor of any such claim immediately and tender the defense of the claim to the franchisor accepts tender of defense, the franchisor has the right to manage the defense of the claim, including the right to compromise, settle or otherwise resolve the claim, or to determine whether to appeal a final determination of the claim.

As required by the Minnesota Franchise Act, Minn. Stat. Sec. 80C.12(g), we protect your right to use the Top Rail Proprietary Mark or indemnify you from any loss, costs or expenses arising out of any claim, suit or demand regarding the use of the Top Rail Proprietary Mark, if you were using it in the manner we authorized, and if we are timely notified of the claim and given the right to manage the defense of the claim, including the right to compromise, settle or otherwise resolve the claim, and to determine whether to appeal a final determination of the claim.

Item 17, Additional Disclosures:

Any condition, stipulation or provision, including any choice of law provision, purporting to bind any person who, at the time of acquiring a franchise is a resident of the State of Minnesota, or in the case of a partnership or corporation, organized or incorporated under the laws of the State of Minnesota, or purporting to bind a person acquiring any franchise to be operated in the State of Minnesota, to waive

compliance or which has the effect of waiving compliance with any provision of the Minnesota Franchise Law is void.

We will comply with Minn. Stat. Sec. 80C.14, subds. 3, 4 and 5, which requires, except in certain specified cases, that a franchisee be given 90-days' notice of termination (with 60 days to cure), 180-days' notice for nonrenewal of the Franchise Agreement, and that consent to the transfer of the franchise will not be unreasonably withheld.

Minnesota Rule 2860.4400D prohibits a franchisor from requiring a franchisee to assent to a general release, assignment, novation or waiver as a condition of purchasing the franchise that would relieve any person from liability imposed by Minnesota Statute \$80C.01 - 80C.22.

Minnesota Rule 2860.4400(G) prohibits a franchisor form imposing on a franchisee by contract or rule, whether written or oral, any standard of conduct that is unreasonable.

The limitations of claims section must comply with Minn. Stat. Sec. 80C.17, subd. 5.

Minn. Stat. §80C.21 and Minn. Rule 2860.4400J prohibit us from requiring litigation to be conducted outside Minnesota, requiring waiver of jury trial, or requiring you to consent to liquidated damages, termination penalties or judgment notes. In addition, nothing in the Franchise Disclosure Document or Franchise Agreement can abrogate or reduce any of your rights as provided for in Minnesota Statutes, Chapter 80C, or your rights to any procedure, forum or remedies provided for by the laws of Minnesota.

Fee Deferral

Item 5 and Item 7 of the Franchise Disclosure Document and Section 3.1 of the Franchise Agreement are amended to state that payment of initial franchise fees shall be deferred until Franchisor has satisfied its pre-opening obligations to Franchisee and Franchisee's Franchised Business is open for business.

MINNESOTA ADDENDUM TO FRANCHISE AGREEMENT

To the extent the Minnesota Franchise Act, Minn. Stat. \$\$80C.01 - 80C.22 applies, the terms of this Addendum apply.

- 1. Notwithstanding anything to the contrary contained in the Franchise Agreement, to the extent that the Franchise Agreement contains provisions that are inconsistent with the following, such provisions are hereby amended:
 - a. With respect to franchises governed by Minnesota Franchise Law, franchisor shall comply with Minn. Stat. Sec. 80C.14, subd. 4 which requires that except for certain specified cases, that franchise be given 180-days' notice for non-renewal of this Franchise Agreement.
 - b. The Minnesota Department of Commerce requires that franchisor indemnify franchisees whose franchise is located in Minnesota against liability to third parties resulting from claims by third parties that the franchisee's use of franchisor's Proprietary Marks infringe upon the trademark rights of the third party. Franchisor does not indemnify against the consequences of a franchisee's use of franchisor's trademark, but franchisor shall indemnify franchisee for claims against franchisee solely as it relates to franchisee's use of the Marks in accordance with the requirements of the Franchise Agreement and franchisor's standards. As a further condition to indemnification, the franchisee must provide notice to franchisor of any such claim immediately and tender the defense of the claim to franchisor. If franchisor accepts tender of defense, franchisor has the right to manage the defense of the claim, including the right to compromise, settle or otherwise resolve the claim, or to determine whether to appeal a final determination of the claim.
 - c. Franchisee will not be required to assent to a release, assignment, novation or waiver that would relieve any person from liability imposed by Minnesota Statute §§ 80C.01 – 80C.22.
 - d. With respect to franchises governed by Minnesota Franchise Law, franchisor shall comply with Minn. Stat. Sec. 80C.14, subd. 3 which requires that except for certain specified cases, a franchisee be given 90-days' notice of termination (with 60 days to cure). Termination of the franchise by the franchisor shall be effective immediately upon receipt by franchisee of the notice of termination where its grounds for termination or cancellation are: (1) voluntary abandonment of the franchise relationship by the franchisee; (2) the conviction of the franchisee of an offense directly related to the business conducted according to the Franchise Agreement; or (3) failure of the franchisee to cure a default under the Franchise Agreement which materially impairs the goodwill associated with the franchisor's trade name, trademark, service mark, logo type or other commercial symbol after the franchisee has received written notice to cure of at least twenty-four (24) hours in advance thereof.
 - e. According to Minn. Stat. Sec. 80C.21 in Minnesota Rules or 2860.4400J, the terms of the Franchise Agreement shall not in any way abrogate or reduce your rights as provided for in Minn. Stat. 1984, Chapter 80C, including the right to submit certain matters to the jurisdiction of the courts of Minnesota. In addition, nothing in this Franchise Agreement shall abrogate or reduce any of franchisee's rights as provided for in Minn. Stat. Sec. 80C, or your rights to any procedure, forum or remedy provided for by the laws of the State of Minnesota. In addition, Minnesota Rule 2860.4400(G) prohibits a franchisor from imposing on a franchisee by contract or rule, whether written or oral, any standard of conduct that is unreasonable.

- f. Any claims franchisee may have against the franchisor that have arisen under the Minnesota Franchise Laws shall be governed by the Minnesota Franchise Law.
- g. The Franchise Agreement contains a waiver of jury trial provision. This provision may not be enforceable under Minnesota law.
- h. Franchisee consents to the franchisor seeking injunctive relief without the necessity of showing actual or threatened harm. A court shall determine if a bond or other security is required.
- i. Any action pursuant to Minnesota Statutes, Section 80C.17, Subd. 5 must be commenced no more than 3 years after the cause of action accrues.
- 2. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of: (i) waiving any claims under any applicable state franchise law, including fraud in the inducement; or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 3. Any capitalized terms that are not defined in this Addendum shall have the meaning given them in the Franchise Agreement.
- 4. Except as expressly modified by this Addendum, the Franchise Agreement remains unmodified and in full force and effect.
- 5. <u>Fee Deferral</u>. Item 5 and Item 7 of the Franchise Disclosure Document and Section 3.1 of the Franchise Agreement are amended to state that payment of initial franchise fees shall be deferred until Franchisor has satisfied its pre-opening obligations to Franchisee and Franchisee's Franchised Business is open for business.

This Addendum is being entered into in connection with the Franchise Agreement. In the event of any conflict between this Addendum and the Franchise Agreement, the terms and conditions of this Addendum shall apply.

(Signature Page Follows)

IN WITNESS WHEREOF, the undersigned have executed this Addendum as of the date Franchisor signs below.

FRANCHISOR:	FRANCHISEE:
HFB FenceCo FranchisingFENCECO FRANCHISING, LLC. a North Carolina limited liability company	a(n)
	Ву:
Ву:	Name:
Name:	Title:
Title:	Date:
Date:	
	Ву:
	Name:
	Title:
	Date:

NEW YORK ADDENDUM TO FRANCHISE DISCLOSURE DOCUMENT

To the extent the New York General Business Law, Article 33, §§680 - 695 applies, the terms of this Addendum apply.

Cover Page, Additional Disclosure:

1. The following information is added to the cover page of the Franchise Disclosure Document.:

INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN EXHIBIT A OR YOUR PUBLIC LIBRARY FOR SOURCESSERVICES OF INFORMATION. REGISTRATION OF THIS FRANCHISE BY NEW YORK STATE DOES NOT MEAN THAT NEW YORK STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN THAT ANYTHING IN THETHIS FRANCHISE DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND NEW YORKTHE APPROPRIATE STATE DEPARTMENT OF LAW, BUREAU OF INVESTOR PROTECTION AND SECURITIES, 28 LIBERTY STREET, 21 FLOOR, NEW YORK, NEW YORK 10005 OR PROVINCIAL AUTHORITY. THE FRANCHISOR MAY, IF IT CHOOSES, NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE FRANCHISE DISCLOSURE DOCUMENT. HOWEVER, THE FRANCHISE TO ACCEPT TERMS WHICH THAT ARE LESS FAVORABLE THAN THOSE SET FORTH IN THIS FRANCHISE DISCLOSURE DOCUMENT.

Item 3, Additional Disclosure:

2. The last sentence in Item 3 is deleted and replaced with the following is to be added at the end of Item 3:

Neither we, nor any of our predecessors, nor any person identified in Item 2With the exception of what is stated above, nor anythe following applies to the franchisor, its predecessor, a person identified in Item 2, or an affiliate offering franchises under ourthe franchisor's principal trademark;

A. No such party has anyan administrative, criminal, or a material civil or arbitration action pending against himthat person alleging: a felony, a violation of anya franchise, antitrust, or securities law, fraud, embezzlement, fraudulent conversion, restraintmisappropriation of tradeproperty, unfair or deceptive practices, misappropriation of property, or comparable civil or misdemeanor allegations.

Neither we, nor any of our predecessors, nor any person identified in Item 2 above, nor any affiliate offering franchises under our trademark, B. No such party has pending actions, other than routine litigation incidental to the business, which are significant in the context of the number of franchisees and the size, nature, or financial condition of the franchise system or its business operations.

C. No such party has been convicted of a felony or pleaded <u>nolo</u> <u>contendere</u> to any othera felony charge or, <u>during within</u> the <u>ten10</u>-year period immediately preceding the application for registration, <u>has</u> been convicted of <u>a misdemeanor</u> or pleaded <u>nolo</u> <u>contendere</u> to <u>anya</u>

misdemeanor charge or been found liable in an arbitration proceeding or a civil action by final judgment, or has been the subject of any other material complaint or legal or arbitration proceeding if such misdemeanor conviction or charge, civil action, complaint, or other such proceeding involved a, a civil action alleging: violation of anya franchise—law, antifraud, or securities law, fraud, embezzlement, fraudulent conversion, restraint of trade, unfair or deceptive practices, or misappropriation of property, or unfair or deceptive practices, or comparable allegationallegations.

Neither we, nor any of our predecessors, nor any person identified in Item 2 above, nor any affiliate offering franchises under our trademark, D. No such party is subject to anya currently effective injunctive or restrictive order or decree relating to franchises the franchise, or under any federal, statea Federal, State or Canadian franchise, securities, antitrust, trade regulation or trade practice law as a result of, resulting from a concluded or pending action or proceeding brought by a public agency, or is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities and Exchange Act of 1934, suspending or expelling such person from membership in such association or exchange; or is subject to a currently effective injunctive or restrictive order relating to any other business activity as a result of an action brought by a public agency or department including, without limitation, actions affecting a license as a real estate broker or sales agent.

Item 4, Additional Disclosure:

Item 4 is deleted and replaced with the following:

Neither we nor any of our predecessors, affiliates, or officers, during the 10 year period immediately before the date of the Disclosure Document: (a) filed as debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code; (b) obtained a discharge of its debts under the Bankruptcy Code; or (c) was a principal officer of a company or a general partner in a partnership that either filed as a debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code or that obtained a discharge of its debts under the U.S. Bankruptcy Code during or within one year after the officer or general partner of the franchisor held this position in the company or partnership

3.

Item 17, Additional Disclosures:

The following is added to the end of the "Summary" sections of Item 17(c) and 17(), titled "Requirements for a franchisee to renew or extend," and Item 17(m);), entitled "Conditions for franchisor approval of transfer":

ToHowever, to the extent required by applicable law, all rights you enjoy and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force; it being the intent of this proviso that the non-waiver provisions of General Business Law Section 687-(4) and 687-(5) be satisfied.

4. The following language replaces the "Summary" section of Item 17(d) is deleted and replaced with the following language:, titled "Termination by franchisee":

You may terminate the agreement on any grounds available by law.

5. The following is added to the Summary section of Item 17(i): end

No assignment will be made except to an assignce who in good faith and judgment of the franchisor is willing and financially able to assume the franchisor's obligations under the Franchise Agreement.

The following is added to the "Summary" sections of ItemsItem 17(v), titled "Choice of forum," and Item 17(w);, titled "Choice of law":

The foregoing choice of law should not be considered a waiver of any right conferred upon the franchisor or upon the franchisee by Article 33 of the General Business Law of the State of New York.

- 6. Franchise Questionnaires and Acknowledgements--No statement, questionnaire, or acknowledgment signed or agreed to by a franchise in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 7. Receipts—Any sale made must be in compliance with § 683(8) of the Franchise Sale Act (N.Y. Gen. Bus. L. § 680 et seq.), which describes the time period a Franchise Disclosure Document (offering prospectus) must be provided to a prospective franchisee before a sale may be made. New York law requires a franchisor to provide the Franchise Disclosure Document at the earlier of the first personal meeting, 10 business days before the execution of the franchise or other agreement, or the payment of any consideration that relates to the franchise relationship.

NEW YORK ADDENDUM

TO FRANCHISE AGREEMENT

To the extent the New York General Business Law, Article 33, §§680 - 695 applies, the terms of this Addendum apply.

- 1. Notwithstanding anything to the contrary contained in the Franchise Agreement, to the extent that the Franchise Agreement contains provisions that are inconsistent with the following, such provisions are hereby amended:
 - a. Any provision in the Franchise Agreement that is inconsistent with the New York General Business Law, Article 33, Section 680 695 may not be enforceable.
 - b. Any provision in the Franchise Agreement requiring franchisee to sign a general release of claims against franchisor does not release any claim franchisee may have under New York General Business Law, Article 33, Sections 680-695.
 - c. The New York Franchise Law shall govern any claim arising under that law.
- 2. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of: (i) waiving any claims under any applicable state franchise law, including fraud in the inducement; or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 3. Any capitalized terms that are not defined in this Addendum shall have the meaning given them in the Franchise Agreement.
- 4. Except as expressly modified by this Addendum, the Franchise Agreement remains unmodified and in full force and effect.

This Addendum is being entered into in connection with the Franchise Agreement. In the event of any conflict between this Addendum and the Franchise Agreement, the terms and conditions of this Addendum shall apply.

(Signature Page Follows)

IN WITNESS WHEREOF, the undersigned have executed this Addendum as of the date Franchisor signs below.

FRANCHISOR:	FRANCHISEE:
HFB FenceCo FranchisingFENCECO FRANCHISING, LLC. a North Carolina limited liability company	a(n)
	Ву:
Ву:	Name:
Name:	Title:
Title:	Date:
Date:	
	By:
	Name:
	Title:
	Date:

NORTH CAROLINA ADDENDUM TO FRANCHISE DISCLOSURE DOCUMENT

This Addendum pertains to franchises/business opportunities sold in the State of North Carolina and will be for the purpose of complying with certain provisions of North Carolina Law:

Item 8. To the extent applicable, Item 8 of the disclosure document is amended to add the following:

If the seller (franchisor) fails to deliver the product(s), equipment or supplies necessary to begin substantial operation of the business within 45 days of the delivery date stated in your contract, you may notify the seller (franchisor) in writing and demand that the contract be cancelled.

NORTH CAROLINA ADDENDUM TO FRANCHISE AGREEMENT

This Addendum relates to franchises sold in North Carolina and is intended to comply with North Carolina statutes and regulations. In consideration of the execution of the Franchise Agreement, we and you agree to amend the Franchise Agreement as follows:

1. <u>Service of Process</u> . The name and receive service of process is:	d address of our agent in North Carolina authorized to
receive service or process is:	
Secretary of State of North Carolin	a
2 South Salisbury Street	
Raleigh, North Carolina 27601	
+. 2. Any capitalized terms the meaning given them in the Franchise Agreement.	at are not defined in this Addendum shall have the
2.1. 3. Except as expressly mod	lified by this Addendum, the Franchise Agreement
remains unmodified and in full force and effect.	
IN WITNESS WHEREOF, the undersigned	have executed this Addendum as of the date Franchisor
signs below.	
FRANCHISOR:	FRANCHISEE:
HFB FenceCo Franchising, LLC	
a North Carolina limited liability company	a(n)
By:	By:
· -	
Name:	Name:
Title:	Title:
Date:	Date:
	By:
	Name:
	Title:
	Date:

NORTH DAKOTA ADDENDUM TO FRANCHISE DISCLOSURE DOCUMENT

To the extent the North Dakota Franchise Investment Law, N.D. Cent. Code, §§51-19-01 – 51-19-17 applies, the terms of this Addendum apply.

1. Fee Deferral. Item 5 and Item 7 of the Franchise Disclosure Document and Section 3.1 of the Franchise Agreement regarding payment of the initial franchise fees are amended to state that payment of the initial franchise fees shall be deferred until all initial obligations owed to the Franchisee under the Franchise Agreement or other documents have been fulfilled by the Franchisor and the Franchisee has commenced doing business pursuant to the Franchise Agreement.

Item 17, Additional Disclosures:

4.2. The following statements are added to Item 17:

Any provision requiring franchisees to consent to the jurisdiction of courts outside North Dakota or to consent to the application of laws of a state other than North Dakota may be unenforceable under North Dakota law. Any mediation or arbitration will be held at a site agreeable to all parties. If the laws of a state other than North Dakota govern, to the extent that such law conflicts with North Dakota law, North Dakota law will control.

Any general release the franchisee is required to assent to as a condition of renewal is not intended to nor shall it act as a release, estoppel or waiver of any liability franchisor may have incurred under the North Dakota Franchise Investment Law.

Covenants not to compete during the term of and upon termination or expiration of the Franchise Agreement are enforceable only under certain conditions according to North Dakota law. If the Franchise Agreement contains a covenant not to compete that is inconsistent with North Dakota law, the covenant may be unenforceable.

The Franchise Agreement includes a waiver of exemplary and punitive damages. This waiver may not be enforceable under North Dakota law.

The Franchise Agreement stipulates that the franchisee shall pay all costs and expenses incurred by franchisor in enforcing the Franchise Agreement. For North Dakota franchisees, the prevailing party is entitled to recover all costs and expenses, including attorneys' fees.

The Franchise Agreement requires the franchisee to consent to a waiver of trial by jury. This waiver may not be enforceable under North Dakota law.

The Franchise Disclosure Document and Franchise Agreement state that franchisee must consent to the jurisdiction of courts outside that State of North Dakota. That requirement may not be enforceable under North Dakota law.

The Franchise Disclosure Document and Franchise Agreement may require franchisees to consent to termination. This requirement may not be enforceable under North Dakota law.

The Franchise Agreement requires the franchisee to consent to a limitation of claims within one year. To the extent this requirement conflicts with North Dakota law, North Dakota law will apply.

3. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of: (i) waiving any claims under any applicable state franchise law, including fraud in the inducement; or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

NORTH DAKOTA ADDENDUM TO FRANCHISE AGREEMENT

To the extent the North Dakota Franchise Investment Law, N.D. Cent. Code, \$\$51-19-01-51-19-17 applies, the terms of this Addendum apply.

- 1. Notwithstanding anything to the contrary contained in the Franchise Agreement, to the extent that the Franchise Agreement contains provisions that are inconsistent with the following, such provisions are hereby amended:
 - a. Any release executed in connection with a renewal shall not apply to any claims that may arise under the North Dakota Franchise Investment Law.
 - b. Covenants not to compete during the term of and upon termination or expiration of the Franchise Agreement are enforceable only under certain conditions according to North Dakota law. If the Franchise Agreement contains a covenant not to compete that is inconsistent with North Dakota law, the covenant may be unenforceable.
 - c. The choice of law other than the State of North Dakota may not be enforceable under the North Dakota Franchise Investment Law. If the laws of a state other than North Dakota govern, to the extent that such law conflicts with North Dakota law, North Dakota law will control.
 - d. The waiver of punitive or exemplary damages may not be enforceable under the North Dakota Franchise Investment Law.
 - e. The waiver of trial by jury may not be enforceable under the North Dakota Franchise Investment Law.
 - f. The requirement that arbitration be held outside the State of North Dakota may not be enforceable under the North Dakota Franchise Investment Law. Any mediation or arbitration will be held at a site agreeable to all parties.
 - g. The requirement that a franchisee consent to termination has been determined by the Commissioner to be unfair, unjust and inequitable within the intent of the North Dakota Franchise Investment Law. This requirement may not be enforceable under North Dakota law.
 - h. The Franchise Agreement states that franchisee must consent to the jurisdiction of courts located outside the State of North Dakota. This requirement may not be enforceable under North Dakota law.
 - i. The Franchise Agreement requires the franchisee to consent to a limitation of claims within one year. To the extent this requirement conflicts with North Dakota law, North Dakota law will apply.
 - j. The Franchise Agreement stipulates that the franchisee shall pay all costs and expenses incurred by the Franchisor in enforcing the Franchise Agreement. For North Dakota franchisees, the prevailing party is entitled to recover all costs and expenses, including attorneys' fees
- 2. Fee Deferral. Item 5 and Item 7 of the Franchise Disclosure Document and Section 3.1 of the Franchise Agreement regarding payment of the initial franchise fees are amended to state that payment

of the initial franchise fees shall be deferred until all initial obligations owed to the Franchisee under the Franchise Agreement or other documents have been fulfilled by the Franchisor and the Franchisee has commenced doing business pursuant to the Franchise Agreement.

- 3. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of: (i) waiving any claims under any applicable state franchise law, including fraud in the inducement; or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 2-4. Any capitalized terms that are not defined in this Addendum shall have the meaning given them in the Franchise Agreement.
- 3.5. Except as expressly modified by this Addendum, the Franchise Agreement remains unmodified and in full force and effect.

This Addendum is being entered into in connection with the Franchise Agreement. In the event of any conflict between this Addendum and the Franchise Agreement, the terms and conditions of this Addendum shall apply.

FRANCHISOR:	FRANCHISEE:
HFB FenceCo Franchising, LLC	
a North Carolina limited liability company	a(n)
By:	By:
Name:	Name:
Title:	Title:
Date:	Date:
	By:
	Name:
	Title:
	Date:
FRANCHISOR:	FRANCHISEE:
HER FENCECO FRANCHISING, LLC.	

a North Carolina limited liability company	<u>a(n)</u>
D	D.,,
By: Name:	By: Name:
Title:	Title:
Date:	Date:

RHODE ISLAND ADDENDUM TO FRANCHISE DISCLOSURE DOCUMENT

To the extent the Rhode Island Franchise Investment Act, R.I. Gen. Law ch. 395 \$\$19-28.1-1-19-28.1-34 applies, the terms of this Addendum apply.

Item 17, Additional Disclosure:

The following statement is added to Item 17:

Section 19-28.1-14 of the Rhode Island Franchise Investment Act provides that: -"A provision in a franchise agreement restricting jurisdiction or venue to a forum outside this state or requiring the application of the laws of another state is void with respect to a claim otherwise enforceable under this Act."

RHODE ISLAND ADDENDUM TO FRANCHISE AGREEMENT TO FRANCHISE AGREEMENT

To the extent the Rhode Island Franchise Investment Act, R.I. Gen. Law ch. 395 §§19-28.1-1 – 19-28.1-34 applies, the terms of this Addendum apply.

Notwithstanding anything to the contrary contained in the Franchise Agreement, to the extent that the Franchise Agreement contains provisions that are inconsistent with the following, such provisions are hereby amended:

Section 19-28.1-14 of the Rhode Island Franchise Investment Act provides that: "A provision in a franchise agreement restricting jurisdiction or venue to a forum outside this state or requiring the application of the laws of another state is void with respect to a claim otherwise enforceable under this Act."

- 2. Any capitalized terms that are not defined in this Addendum shall have the meaning given them in the Franchise Agreement.
- Except as expressly modified by this Addendum, the Franchise Agreement remains unmodified and in full force and effect.

This Addendum is being entered into in connection with the Franchise Agreement. In the event of any conflict between this Addendum and the Franchise Agreement, the terms and conditions of this Addendum shall apply.

IN WITNESS WHEREOF, the undersigned have executed this Addendum as of the date Franchisor

FRANCHISOR:	FRANCHISEE:
HFB FenceCo Franchising FENCECO FRANCHISING, LLC ₃ a North Carolina limited liability company	a(n)
	Ву:
Ву:	Name:
Name:	Title:
Title:	Date:
Date:	
	Ву:
	Name:
	Title:

Date:			

SOUTH CAROLINADAKOTA ADDENDUM

TO FRANCHISE DISCLOSURE DOCUMENT

This Addendum pertains to franchises business opportunities sold in the State of South Carolina Dakota and will be for the purpose of complying with certain provisions of South Carolina Dakota Law:

Item 8 of the franchise disclosure document is amended to add the following:

If the seller (franchisor) fails to deliver the product, equipment or supplies necessary to begin substantial operation of the business within 45 days of the delivery date stated in your contract, you may notify the seller (franchisor) in writing and demand that the contract be canceled.

1. Fee Deferral. Pursuant to Section 37-5B-5 of the South Dakota Codified Laws, Franchisor is required to defer its initial franchise fees until such time as the Franchisee has fulfilled all of its initial obligations under the Franchise Agreement and Franchisee is operational.

SOUTH-CAROLINA ADDENDUM TO FRANCHISE AGREEMENT

This Addendum pertains to franchises/business opportunities sold in the State of South Carolina and will be for the purpose of complying with certain provisions of South Carolina Law:

1. Cogency Global Inc. is hereby authorized to receive service of process. Cogency Global Inc. is located at 2 Office Park Court, Suite 103, Columbia, SC 29223.

1.2. 2.—Any capitalized terms that are not defined in this Addendum shall have the meaning given them in the Franchise Agreement.

3

3. Except as expressly modified by this Addendum, the Franchise Agreement remains unmodified and in full force and effect.

This Addendum is being entered into in connection with the Franchise Agreement. In the event of any conflict between this Addendum and the Franchise Agreement, the terms and conditions of this Addendum shall apply.

SOUTH DAKOTA ADDENDUM TO FRANCHISE AGREEMENT

This Addendum pertains to franchises sold in the State of South Dakota and will be for the purpose of complying with certain provisions of South Dakota Law:

- 1. Fee Deferral. Pursuant to Section 37-5B-5 of the South Dakota Codified Laws, Franchisor is required to defer its initial franchise fees until such time as the Franchisee has fulfilled all of its initial obligations under the Franchise Agreement and Franchisee is operational.
- 2. Any capitalized terms that are not defined in this Addendum shall have the meaning given them in the Franchise Agreement.
- 3. Except as expressly modified by this Addendum, the Franchise Agreement remains unmodified and in full force and effect.

This Addendum is being entered into in connection with the Franchise Agreement. In the event of any conflict between this Addendum and the Franchise Agreement, the terms and conditions of this Addendum shall apply.

IN WITNESS WHEREOF, the undersigned have executed this Addendum as of the date Franchisor signs below.

FRANCHISOR:	FRANCHISEE:
HFB FENCECO FRANCHISING, LLC, a North Carolina limited liability company	<u>a(n)</u>
By:	By:
Name:	Name:
Title:	Title:
Date:	Date:

VIRGINIA ADDENDUM TO FRANCHISE DISCLOSURE DOCUMENT

To the extent the Virginia Retail Franchising Act, Va. Code §§13.1-557 – 13.1-574 applies, the terms of this Addendum apply.

Item 17, Additional Disclosures:

Any provision in any of the contracts that you sign with us which provides for termination of the franchise upon the bankruptcy of the franchisee may not be enforceable under federal bankruptcy law (11 U.S.C. 101 et. seq.).

"According to Section 13.1 – 564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any grounds for default or termination stated in the franchise agreement does not constitute "reasonable cause," as that term may be defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable."

VIRGINIA ADDENDUM TO FRANCHISE AGREEMENT

To the extent the Virginia Retail Franchising Act, Va. Code §§13.1-557 – 13.1-574 applies, the terms of this Addendum apply.

- 1. Notwithstanding anything to the contrary contained in the Franchise Agreement, to the extent that the Franchise Agreement contains provisions that are inconsistent with the following, such provisions are hereby amended:
 - "According to Section 13.1-564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any ground for default or termination stated in the franchise agreement does not constitute "reasonable cause," as that term may be defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable."
- 2. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of: (i) waiving any claims under any applicable state franchise law, including fraud in the inducement; or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 3. Any capitalized terms that are not defined in this Addendum shall have the meaning given them in the Franchise Agreement.
- 4. Except as expressly modified by this Addendum, the Franchise Agreement remains unmodified and in full force and effect.

This Addendum is being entered into in connection with the Franchise Agreement. In the event of any conflict between this Addendum and the Franchise Agreement, the terms and conditions of this Addendum shall apply.

(Signature Page Follows)

 $\underline{\text{IN WITNESS WHEREOF, the undersigned have executed this Addendum as of the date Franchisor}} \\ \underline{\text{signs below.}}$

FRANCHISOR:	FRANCHISEE:
HFB FenceCo FranchisingFENCECO FRANCHISING, LLC ₂ a North Carolina limited liability company	a(n)
	Ву:
Ву:	Name:
Name:	Title:
Title:	Date:
Date:	
	D.
	By:
	Name:
	Title:
	Date:

WASHINGTON ADDENDUM TO FRANCHISE DISCLOSURE DOCUMENT

To the extent the Washington Franchise Investment Protection Act, Wash. Rev. Code §§19.100.010 – 19.100.940 applies, the terms of this Addendum apply.

Item 17, Additional Disclosure:

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW will prevail.

RCW 19.100.180 may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise.

In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.

A release or waiver of rights executed by a franchisee may not include rights under the Washington Franchise Investment Protection Act or any rule or order thereunder except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel.

Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.

Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained in the franchise agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington.

RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.

Personal Guarantees. If you are a partnership, corporation or limited liability company, all of your owners will be required to sign personal guarantees. This requirement places the personal assets of the franchise owner(s) at risk.

Diligence Review of Franchisee Sales Process Exhibit. The following is added to the Diligence Review of Franchisee Sales Process exhibit to the Franchise Disclosure Document:

The Due Diligence Review of Franchisee Sales Process does not waive any liability the franchisor may have under the Washington Franchise Investment Protection Act, RCW 19.100, and the rules adopted thereunder.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

Fee Deferral

Item 5 and Item 7 of the Franchise Disclosure Document, Section 3.1 of the Franchise Agreement, and Section 2 of the Multi-Unit Addendum are hereby revised to state that payment of all initial fees, including the Initial Franchise Fee, shall be deferred until after all of Franchisor's initial obligations are complete and the Franchised Business is open for business. In addition, the Multi-Unit Fee shall be prorated and collected as each unit under the Multi-Unit Addendum is opened.

WASHINGTON ADDENDUM TO FRANCHISE AGREEMENT AND MULTI-UNIT ADDENDUM

To the extent the Washington Franchise Investment Protection Act, Wash. Rev. Code §§19.100.010 – 19.100.940 applies, the terms of this Addendum apply.

Notwithstanding anything to the contrary contained in the Franchise Agreement, to the extent that the Franchise Agreement contains provisions that are inconsistent with the following, such provisions are hereby amended: In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW will prevail. RCW 19.100.180 may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington. A release or waiver of rights executed by a franchisee may not include rights under the Washington Franchise Investment Protection Act or any rule or order thereunder except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable. Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer. Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained in the franchise agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington. RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.

that section: "In some circumstances professional fees and expenses are only recoverable where franchisor is the substantially prevailing party." Section 22.9 of the Franchise Agreement is amended by adding the following to that section: "Provided, however, that any modifications to the Franchise Agreement that purport to change an investor protection included in Washington Franchise Investment Protection Act, RCW 19.100, the rules adopted thereunder, or required as part of the state registration process is specifically excepted from this provision." No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of: (i) waiving any claims under any applicable state franchise law, including fraud in the inducement; or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise. Any capitalized terms that are not defined in this Addendum shall have the meaning given them in the Franchise Agreement. Except as expressly modified by this Addendum, the Franchise Agreement remains unmodified and in full force and effect. Fee Deferral. Item 5 and Item 7 of the Franchise Disclosure Document, Section 3.1 of the Franchise Agreement and Section 2 of the Multi-Unit Addendum (if applicable) are hereby amended to state that all initial franchise fees described in Item 5 of the Franchise Disclosure Document will be deferred until franchisee has: (a) received all initial training that it is entitled to under the Franchise Agreement or Offering Circular; and (b) is open for business. If the Franchisor is selecting a Multi-Unit Offering, the Multi-Unit Fee will be prorated and collected as each unit is opened. This Addendum is being entered into in connection with the Franchise Agreement. In the event of any conflict between this Addendum and the Franchise Agreement, the terms and conditions of this Addendum shall apply. IN WITNESS WHEREOF, the undersigned does hereby acknowledge receipt of this addendum and has executed this Addendum as of the date Franchisor signs below. **FRANCHISOR: FRANCHISEE:** HFB FENCECO FRANCHISING, LLC a North Carolina limited liability company a(n) By: By: Name: Name:

Section 22.8 of the Franchise Agreement is amended by adding the following at the end of

Title:	Title:
Date:	Date:

WISCONSIN ADDENDUM

TO FRANCHISE DISCLOSURE DOCUMENT

To the extent the Wisconsin Franchise Investment Law, Wis. Stat. \$\$553.01 - 553.78 or Wisconsin Fair Dealership Law, Wis. Stat. \$\$135.01 - 135.07 applies, the terms of this Addendum apply.

Item 17, Additional Disclosures:

For all franchisees residing in the State of Wisconsin, we will provide you at least 90-days' prior written notice of termination, cancellation or substantial change in competitive circumstances. The notice will state all the reasons for termination, cancellation or substantial change in competitive circumstances and will provide that you have 60 days in which to cure any claimed deficiency. If this deficiency is cured within 60 days, the notice will be void. If the reason for termination, cancellation or substantial change in competitive circumstances is nonpayment of sums due under the franchise, you will have 10 days to cure the deficiency.

For Wisconsin franchisees, Ch. 135, Stats., the Wisconsin Fair Dealership Law, supersedes any provisions of the Franchise Agreement or a related contract which is inconsistent with the Law.

WISCONSIN ADDENDUM TO FRANCHISE AGREEMENT

To the extent the Wisconsin Franchise Investment Law, Wis. Stat. \$\$553.01 - 553.78 or Wisconsin Fair Dealership Law, Wis. Stat. \$\$135.01 - 135.07 applies, the terms of this Addendum apply.

1. Notwithstanding anything to the contrary contained in the Franchise Agreement, to the extent that the Franchise Agreement contains provisions that are inconsistent with the following, such provisions are hereby amended:

To the extent any of the provisions regarding notice of termination or change in dealership are in conflict with Section 135.04 of the Wisconsin Fair Dealership Law, the Wisconsin law shall apply.

- 2. Any capitalized terms that are not defined in this Addendum shall have the meaning given them in the Franchise Agreement.
- 3. Except as expressly modified by this Addendum, the Franchise Agreement remains unmodified and in full force and effect.

This Addendum is being entered into in connection with the Franchise Agreement. In the event of any conflict between this Addendum and the Franchise Agreement, the terms and conditions of this Addendum shall apply.

IN WITNESS WHEREOF, the undersigned have executed this Addendum as of the date Franchisor signs below.

ED ANGUEGE

FRANCHISUR:	FRANCHISEE:
HFB FenceCo FranchisingFENCECO FRANCHISING, LLC ₂ a North Carolina limited liability company	a(n)
	By:
Ву:	Name:
Name:	Title:
Title:	Date:
Date:	
	Ву:
	Name:
	Title:

ED ANGINGOD

Date:			
Date.			

EXHIBIT E

Exhibit I to FDD HFB FENCECO FRANCHISING FenceCo Franchising, LLC FRANCHISE DISCLOSURE DOCUMENT

FINANCIAL STATEMENTS

HFB FenceCo Franchising, LLC	
Audited Financial Statements	
For the Years Ended December 31, 2023 and 2022	



Audited Financial Statements

For the Year Ended December 31, 2022

Table of Contents

Independent Auditors' Report	1 - 2
Financial Statements:	_
Balance Sheet	3
Income Statement and Statement of Changes in Members' Equity	4
Statement of Cash Flows	5
Notes to the Financial Statements	6-7



Independent Auditors' Report

To the Management of HFB FenceCo Franchising, LLC Charlotte, North Carolina

Opinion

We have audited the accompanying financial statements of HFB FenceCo Franchising, LLC which comprise the balance sheet as of December 31, 2022 and the related statements of income, changes in members' equity and cash flows for the year then ended and the related notes to the financial statements.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of HFB FenceCo Franchising, LLC as of December 31, 2022 and the results of its operations and its cash flows for the year then ended in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audit in accordance with auditing standards generally accepted in the United States of America. Our responsibilities under those standards are further described in the Auditor's Responsibilities for the Audit of the Financial Statements section of our report. We are required to be independent of HFB FenceCo Franchising, LLC and to meet our other ethical responsibilities in accordance with the relevant ethical requirements relating to our audit. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Responsibilities of Management for the Financial Statements

Management is responsible for the preparation and fair presentation of the financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about HFB FenceCo Franchising, LLC's ability to continue as a going concern within one year after the date that the financial statements are available to be issued.

Auditor's Responsibilities for the Audit of the Financial Statements

Our objectives are to obtain reasonable assurance about whether the financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with generally accepted auditing standards will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the financial statements.

In performing an audit in accordance with generally accepted auditing standards, we:

· Exercise professional judgment and maintain professional skepticism throughout the audit.

- Identify and assess the risks of material misstatement of the financial statements, whether due to fraud
 or error, and design and perform audit procedures responsive to those risks. Such procedures include
 examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that
 are appropriate in the circumstances, but not for the purpose of expressing an opinion on the
 effectiveness of HFB FenceCo Franchising, LLC's internal control. Accordingly, no such opinion is
 expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant
 accounting estimates made by management, as well as evaluate the overall presentation of the financial
 statements
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that
 raise substantial doubt about HFB FenceCo Franchising, LLC's ability to continue as a going concern
 for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control related matters that we identified during the audit.

Janes, Goldein Associales, GRISPUC

Matthews, North Carolina March 23, 2023

HFB FenceCo Franchising, LLC Balance Sheet December 31, 2022

CURRENT ASSETS

Cash and Cash Equivalents	\$ 527,627
Initial Franchise Fees Receivable	70,206
Other Current Assets	36,588
Total Current Assets	634,421

TOTAL ASSETS	\$ 634,421

LIABILITIES & MEMBERS' EQUITY

TOTAL LIABILITIES & MEMBERS' EQUITY

CURRENT LIABILITIES

CURRENT LIABILITIES	
Accounts Payable	130,410
Accrued Expenses	82,920
Contract Liabilities	79,003
Line of Credit	165,000
Total Current Liabilities	457,333
Contract Liabilities, net of current portion	4,369
TOTAL LIABILITIES	461,702
MEMBERS' EQUITY	172,719

The accompanying notes are an integral part of the financial statements.

\$ 634,421

HFB FenceCo Franchising, LLC

Income Statement and Statement of Members' Equity For the Year Ended December 31, 2022

Revenue		
Initial Franchise Fees	\$	16,628
Less: Related Commissions and Fees		(15,000)
Net Initial Franchise Fees		1,628
Royalty and Other Fees	-	206
Total Revenue		1,834
Expenses		
Management and Support Services		253,977
General and Administrative		70,220
Marketing		45,495
Franchise Operations		8,117
Total Expenses	-	377,809
Operating Loss		(375,975)
Interest Expense		1,306
Net Loss	\$	(377,281)
Members' Equity, January 1st	\$	-
Members' Contributions, net		550,000
Net Loss		(377,281)
Members' Equity, December 31	\$	172,719

The accompanying notes are an integral part of the financial statements.

HFB FenceCo Franchising, LLC Statement of Cash Flows For the Year Ended December 31, 2022

Cash flows from operating activities:		
Net Loss	\$	(377,281)
Adjustments to reconcile net income		
to net cash used in operating activities:		
Changes in operating assets and liabilities:		
(Increase) in initial franchise fees receivable		(70,206)
(Increase) in other current assets		(36,588)
Increase in accounts payable		130,410
Increase in accrued expenses		82,920
Increase in contract liabilities		83,372
Net cash used in operating activities	_	(187,373)
Cash flows from financing activities:		
Proceeds from Line of Credit		165,000
Members' Contributions		550,000
Net cash provided by financing activities	-	715,000
Net change in cash and cash equivalents	_	527,627
Cash and Cash Equivalents, January 1st	_	
Cash and Cash Equivalents, December 31st	\$	527,627

The accompanying notes are an integral part of the financial statements.

HFB FenceCo Franchising, LLC Notes to the Financial Statements For the Year Ended December 31, 2022

Note 1 - Organization and Business

HFB FenceCo Franchising, LLC (the Company) is a North Carolina limited liability corporation formed in 2022 for the purpose of conducting franchise operations. The Company sells franchise rights to own and operate Top Rail franchises which sells, furnishes, and installs wood, steel, aluminum, vinyl and other types of fencing and related garden products for residential and commercial customers. In addition, the Company provides support services to franchisees. As of December 31, 2022, there were 27 franchise territories sold and three territories signed under licensing agreements, which permit the use of the Company's trademark in connection with the sale and distribution of products and services. There were 21 franchise territories operational at December 31, 2022.

Note 2 - Summary of Significant Accounting Policies

Basis of Accountin

The accompanying financial statements have been prepared using the accrual basis of accounting in accordance with accounting principles generally accepted in the United States of America.

Cash and Cash Equivalents

Cash and cash equivalents consist of cash in a bank deposit account. As of December 31, 2022, the Company's cash balance exceeded the Federal Deposit Insurance Corporation limits. Management deems the related risk of loss to be remote.

Revenue Recognition

The Company recognizes revenue in accordance with FASB Accounting Standards Codification Topic 606, Revenue from Contracts with Customers (ASC 606). ASC 606, as amended, is based on the principle that revenue is recognized to depict the contractual transfer of goods or services to customers in an amount that reflects the consideration to which the entity expects to be entitled in exchange for goods or services utilizing a five-step revenue recognition model, which includes (i) identify the contract with a customer; (ii) identify the performance obligations in the contract; (iii) determine the transaction price; (iv) allocate the transaction price to the performance obligations in the contract; and (v) recognize revenue when or as the entity satisfies the performance obligation.

The Company's revenues consist primarily of initial franchise fees charged to franchisees in exchange for the right to own and operate franchises within a certain territory. Once operational, the Company will also receive monthly royalties and other fees from franchisees. ASC 606 did not impact the timing of revenue recognition for the royalties and other fees which are recognized monthly as a percentage of revenue collected by the franchisees.

In applying ASC 606 to initial franchise fees, the Company adopted Accounting Standards Update 2021-02, Franchisors – Revenue from Contracts with Customers: Practical Expedient. The amendments allow a practical expedient that simplifies the application of the guidance regarding identification of performance obligations. Franchisors that are not public business entities may account for pre-opening services provided to a franchisee as distinct from the franchise license (if the services are consistent with those included in a predefined list within the guidance) and can be recognized as a single performance obligation.

Lump sum payments are typically due upon execution of a franchise agreement or renewal. The Company recognizes the revenue related to pre-opening services as a single performance obligation in accordance with the guidance. In addition, the Company identified the completion of training as a performance

HFB FenceCo Franchising, LLC Notes to the Financial Statements For the Year Ended December 31, 2022

obligation and recognizes revenue related to this upon completion. The remaining initial fee is related to the license to utilize the Company's brand for a specified period of time which is satisfied equally over the life of the agreement. This deferred revenue is recorded as a contract liability and amortized on a straightline basis over the life of the agreement.

Accounts Receivable

Accounts receivable at December 31, 2022 represent initial franchise fees due at the signing of the franchise agreement which have not yet been collected. Management reviewed all outstanding fees and determined an allowance for doubtful accounts was not necessary.

Use of Estimates

The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from these estimates.

Note 3 - Line of Credit

The Company has drawn \$165,000 on a line of credit of \$400,000 as of December 31, 2022. The line bears interest of 9% and expires on December 31, 2023 at which time all principal outstanding is due.

Note 4 - Related Party Transactions

The lender of the line of credit is a private trust which is a related party.

All management (and related payroll), research, supervision, financial, accounting, investment, procurement, human resources, legal services, information systems, communications and other required services are provided by DDL Investments, Inc., a related party. Expenses related to these services of \$253,997 are included in management and support services (\$94,717 included in accounts payable) as of December 31, 2022.

The operating territories belong to the founding members of the Company and were not subject to royalty fees during 2022.

Note 5 - Subsequent Events

In preparing these financial statements in accordance with Accounting Standards Codification No. 855 -Subsequent Events, the Company has evaluated events and transactions for potential recognition or disclosure through March 23, 2023, the date the financial statements were available to be issued.

THESE FINANCIAL STATEMENTS ARE PREPARED WITHOUT AN AUDIT. PROSPECTIVE FRANCHISES OR SELLERS OF FRANCHISES SHOULD BE ADVISED THAT NO CERTIFIED PUBLIC ACCOUNTANT HAD AUDITED THESE FIGURES OR EXPRESSED HIS/HER OPINION WITH REGARD TO THE CONTENT OR FORM.

HFB FENCECO FRANCHISING BALANCE SHEET AS OF JUNE 30, 2023	LLC	
		Inaudited) ne 30, 2023
ASSETS	•	
CURRENT ASSETS		
Cash	\$	136,116
Accounts receivable		319,476
Inventory		163
Other current assets		39,604
TOTAL CURRENT ASSETS	***	495,359
Fixed Assets, Net		
Woncurrent Assets, Wet		-
TOTAL ASSETS	<u> </u>	495,359
LIABILITIES AND MEMBERS' EQUITY		
CURRENT LIABILITIES		
Accounts payable	\$	318,942
Accrued expenses		137,188
Contract liabilities		85,003
Lines of credit		240,000
TOTAL CURRENT LIABILITIES	5 <u>24.</u>	781,133
Contract Liabilities (Net of Current Portion)		4,369
TOTAL LIABILITIES	\$	785,502
		(290,143)
MEMBERS' EQUITY		(290,145)

INCOME CTATEMENTS	LLC	
INCOME STATEMENTS		
FOR THE SIX MONTHS ENDED JUNE 30	, 2023	
		naudited) ne 30, 2023
Revenue		
Initial Franchise Fee Income	\$	1,291,190
Less: Related Commissions and Fees		(654,632)
Net Initial Franchise Fee Income		636,558
Royalties		19,899
Other Fee Income		9,819
Total Revenue	1.00	666,275
Expenses Call Center Expenses Franchise Development General and Administrative	s	2,344 198,914 375,958
Marketing Expenses		195,638
		192,079 5,788
Business Operations Expense Technology Fee Expenses		970,721
	8	
Technology Fee Expenses	\$	(304,445)
Technology Fee Expenses Total Expenses	\$	(304,445) 8,418

EXHIBIT F TO HFB FENCECO FRANCHISING, LLC FRANCHISE DISCLOSURE DOCUMENT

SAMPLE ASSIGNMENT AND ASSUMPTION AGREEMENT

$\frac{\text{SAMPLE ASSIGNMENT AND ASSUMPTION AGREEMENT}}{\text{UPON TRANSFER TO AN AUTHORIZED FRANCHISEE}}$

	This Assignmen	nt and As	ssumption Ag	greement Upon	Transfer to	an Authori	zed Franchis	ee (the
"Agr	eement") is made	this	day of	*	, 20	, by	and between	HFI
Fene	eement") is made eCo Franchising, l	LLC, a N	lorth Carolina	corporation, w	ith its princip	al place of	business at 1	07 Par
Drive	. Huntersvi	lle.	North	Carolina	28078	("Fran	chisor")	ane
				, a(n)			with its pr	incipa
place	of		business	at			_ · · · · · · · · · · · · · · · · · · ·	. 1
F				(referred to be	rein as "Fi	anchisee" "()wner
and	"Guarantor") (Fra	nchisee	Owner an	d Guarantor	iointly refer	red to as	"Assignor") and
	<u>outrained</u>) (116			, a(n)				
				, u(11)			the rissi	giice)
	1. Franchis	coo ont	ered into	a Franchice	Agreemer	t with	Franchicor	date
				ment") for a Te				
torrit	ory set forth in the l	Eronchice	Agreements	(the "Franchise	ed Business")	-	operated wit	iiii tiik
territ	ory set forth in the i	ranemse	rigicements	(the Tranchist	od Dusiness)	•		
	2 Assigno	n br: +n/	nofomina ita	orrmonshin in	tarast in Ero	nobicoo de	niman ta ana	ion ol
A anie	z. Assigno gnor's rights and ob			ownership in				
					nent to Assig	nee. Assigi	ce desires to	ootan
mese	rights and is willin	g to assu	me tnese obli	ganons.				
	2 E		11	4	1	•		
	3. Franchis	sor is Wil	ung to permit	tne assignmen	t and assumpt	ion.		
	NT 41 C '		6.4				1	
	Now therefore, i							
	other good and val			he sufficiency	and adequac	y of which	the parties	hereby
ackno	owledge, the parties	s agree as	: tollows:					
			B.4	CKCROUNI				
	A. Franchis	see ent	ered into	a Franchisc	- Agreemer	it with	Franchisor	-datec
				Agreement") fo			be operated	withir
t he te	erritory set forth in t	he Franc	hise Agreeme	ents (the "Franc	chised Busine	ss").		
	B.AAssigno							
	snor's rights and ob				nent to Assig	nee. Assign	ce desires to	obtair
these	rights and is willin	g to assu	me these obli	gations.				
	C.A. Franchis	or is wil	ling to permit	the assignmen	t and assumpt	ion.		
	New therefore, i							
	other good and val			he sufficiency	and adequae	y of which	the parties	hereby
aekn	owledge, the parties	agree as	follows:					
			TERMS	OF AGREEN	AENT			
	1. Assigna	nent and	 Assumption	-Subject to the	e terms of thi	s Agreeme	nt, Assignor	assigne
to As	signee, by transfer							
the F	Tranchise Agreeme	nt and de	elegates to A	ssignee all of	Assignor's ol	oligations 1	inder the Fra	anchise
Agre	ement arising on an	d after th	e Effective Da	ate. Assignee as	rees to assum	e and perfo	rm all of Ass	ignor'
	ations under the Fr							
	n a reasonable ame							

the approval of the transfer, Assignee will sign the then current form of franchise agreement, which agreement will supersede the assigned Franchise Agreement, subject to the addendum negotiated between Franchiser and Assignee. Simultaneously with the Assignment contemplated herein, the assigned Franchise Agreement will be amended to specify

[Assignee] as the Franchise.

2. No Discharge of Assignor. The assignment by Assignor of Assignor's rights and obligations under the Franchise Agreement will not relieve Assignor or Franchise of obligations under the Franchise Agreement that accrue before or after the Effective Date (that are related to events occurring before the effective date of the transfer). Notwithstanding the foregoing, Assignor must report all revenue collected prior to the Effective Date and pay all royalties due thereon, on or before the 10th of the month (or the next business day after the 10th if the 10th is not a business day) following the month in which the assignment transaction closes. Assignee will be responsible for reporting all amounts collected after the Effective Date and paying all royalties on amounts collected after the Effective Date in accordance with the Franchise Agreement between Assignee and Franchisor. In addition, if assignor collects any amounts after the Effective Date that relate to operation of the business prior to (or after) the Effective date, Assignor must report all such collections to Franchisor on a monthly basis consistent with terms of the Franchise Agreement in effect as of the Effective Date and Assignor must pay royalties on those collections on a monthly basis consistent with terms of the Franchise Agreement in effect as of the Effective Date.

3.1. Consent. Franchisor consents to Assignor's assignment of Assignor's rights and obligations under the Franchise Agreement to Assignee. Franchisor hereby declines to exercise its right of first refusal with regard to the proposed transfer contemplated between the Assignee and Assigner.

4.1. Acknowledgment. Assignor, Assignee and Franchisor agree that the assignment of Assignor's franchised business and the Franchise Agreement (and the related agreements, if any) to Assignee were not affected by or through Franchisor, and that Franchisor's activities in connection therewith were limited to the exercise of its right of approval under the Franchise Agreement.

5.1. Limited Release by Assignee. Assignee, for themselves and their officers, directors, stockholders, members, managers, partners, agents, heirs, executors, administrators, legal successors and assigns, release and forever discharge HFB FenceCo Franchising, LLC, its respective parents, subsidiaries and affiliates, and the respective officers, directors, stockholders, members, managers, partners, employees, agents, attorneys, contractors, legal successors and assigns of each of the forgoing entities in their corporate and individual capacities ("Franchisor Related Persons/Entities"), of and from all manner of actions, causes of action, suits, debts, dues, sums of money, accounts, reckonings, bonds, bills, specialties, covenants, contracts, agreements, controversies, promises, variances, trespasses, damages, judgments, executions, claims and demands whatsoever, in law, admiralty or in equity, that Assignee and/or Assignee's officers, directors, stockholders, members, managers, partners, agents, heirs, executors, administrators, legal successors and assigns ever had, now has, or that they hereafter can or may have for, on or by reason of any matter, cause or thing whatsoever, from the beginning of the world to the day of this Agreement, including, without limitation, claims arising under federal, state and local laws, rules and ordinances that are related to the assignment of the Franchise Agreements

Being fully informed, Assignee(s) waive any rights on, and acknowledges that this release extends to all Claims Assignees have or might have against Franchisor Related Persons/Entities, whether known or unknown.

6.1. General Release by Assignor/Franchisee and Guarantors. The Assignor, for themselves, Franchisee, and, as applicable, their officers, directors, stockholders, members, managers, partners, agents, heirs, executors, administrators, legal successors and assigns, release and forever discharge HFB FenceCo Franchising, LLC, its respective parents, subsidiaries, and affiliates and the respective

officers, directors, stockholders, members, managers, partners, employees, agents, attorneys, contractors, legal successors, and assigns of each of the forgoing entities (in their corporate and individual capacities), of and from all manner of actions, causes of action, suits, debts, dues, sums of money, accounts, reckonings, bonds, bills, specialties, covenants, contracts, agreements, controversies, promises, variances, trespasses, damages, judgments, executions, claims and demands whatsoever, in law, admiralty or in equity, that Assignor and/or Assignor's officers, directors, stockholders, members, managers, partners, agents, heirs, executors, administrators, legal successors and assigns, ever had, now has, or that they hereafter can or may have for, on or by reason of any matter, cause or thing whatsoever, from the beginning of the world to the day of this Agreement, including, without limitation, claims arising under federal, state and local laws, rules and ordinances.

Being fully informed, Assigner(s) waive any rights on, and asknowledge(s) that this release extends to all Claims Assignors have or might have against Franchisor-Related Persons/Entities, whether known or unknown.

7.1. The Guarantor and Assignor/Franchisee(s) expressly assume the risk of any mistake of fact or fact of which they may be unaware or that the true facts may be other than any facts now known or believed to exist by Guarantor or Franchisee(s), and it is the Guarantor's Franchisee's intention to forever settle, adjust and compromise any and all present and/or future disputes with respect to all matters from the beginning of time to the date of this document finally and forever and without regard to who may or may not have been correct in their understanding of the facts, law or otherwise. All releases given by the Guarantor and Franchisee are intended to constitute a full, complete, unconditional and immediate substitution for any and all rights, claims, demands and causes of action whatsoever which exist, or might have existed, on the date of this document. The Guarantor and Franchisee represent and warrant that they have made such independent investigation of the facts, law and otherwise pertaining to all matters discussed, referred to or released in or by this document as the Guarantor and Franchisee, in each of their independent judgment, believe necessary or appropriate. The Guarantor and Franchisee have not relied on any statement, premise, representation or otherwise, whether of fact, law or otherwise, or lack of disclosure of any fact, law or otherwise, by the Franchisor Related Persons/Entities or anyone else not expressly set forth herein, in executing this document and/or the related releases.

8.1. Entire Agreement. This Agreement represents the entire understanding and agreement between the parties with respect to the subject matter of this Agreement and supersedes all other negotiations, understandings and representations, if any, made by the parties with respect to the subject of this Agreement.

9. Assignment; Parties In Interest. This Agreement is binding on and inures to the benefit of the parties and their respective successors and assigns, but this Agreement and the rights or obligations in this Agreement may not be assigned by any party without the other parties' consent, except that Franchisor has the right to assign this Agreement, to any person, firm, corporation or other entity, if the assignment is made with Franchisor's assignment of its rights under the Franchise Agreement. This Agreement is made for the benefit of the parties, and nothing in this Agreement is intended to or confers any benefit on any other person, association or entity.

Notices to Assignee:		
Notices to Assignee.		

Notices to Franchisor: HFB FenceCo Franchising, LLC 107 Parr Drive

Huntersville, NC 28078 Attn: Logal Department

10.1. This Agreement shall be construed under the laws of the State of North Carolina, which laws shall control in the event of any conflict of law.

11. Each party acknowledges that the terms of this Agreement have been completely read and are fully understood and voluntarily accepted by each party, after having a reasonable opportunity to retain and confer with counsel. This Agreement is entered into after a full investigation by the parties, and the parties are not relying upon any statements or representations not embodied in this Agreement.

12.1.—In the event that Franchisor retains the services of legal counsel to enforce the terms of this Agreement, it shall be entitled to recover all costs and expenses, including reasonable attorneys' fees incurred in enforcing the terms of this Agreement.

13. Assignor agrees that Assignor has and had a relationship with Franchisor at its offices in the State of North Carolina and that, with the exception of Franchisor's right to seek injunctive relief in any appropriate jurisdiction, any action by or against Franchisor arising out of or relating to this Agreement shall be commenced and concluded in the State of North Carolina pursuant to the mediation, venue and jurisdiction provisions of the Franchise Agreement.

This Agreement may be executed in multiple counterparts by the various parties and the failure to have the signatures of all parties on a single Agreement shall not affect the validity or enforceability of any part of this Agreement against any party who executes any counterpart of the Agreement. Executed facsimile copies of this Agreement shall be deemed to be effective as original signatures.

14. Definitions. Unless otherwise stated, all terms used in this Agreement have the same meaning as indicated in the Franchise Agreement.

HHAVE READ THE ABOVE AGREEMENT AND UNDERSTAND ITS TERMS, I WOULD NOT SIGN THIS AGREEMENT IF I DID NOT UNDERSTAND AND AGREE TO BE BOUND BY ITS TERMS.

(THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK. SIGNATURES APPEAR ON FOLLOWING PAGE.)

FRANCHISOR:	
HFB FENCECO FRANCHISING, LLC	
B y :	=
Name:	=
Title:	=
Date:	=
TRANSFEROR:	
[NAME]	=
Ву:	=
Name:	=
Title:	=
Date:	=

EXHIBIT G TO HFB FENCECO FRANCHISING, LLC FRANCHISE DISCLOSURE DOCUMENT

Table of Contents

Independent Auditors' Report	1 - 2
Financial Statements:	
Balance Sheets	3
Income Statements and Statements of Changes in Members' Equity	4
Statements of Cash Flows	5
Notes to the Financial Statements	6 - 8



Independent Auditors' Report

To the Management of HFB FenceCo Franchising, LLC Charlotte, North Carolina

Opinion

We have audited the accompanying financial statements of HFB FenceCo Franchising, LLC which comprise the balance sheets as of December 31, 2023 and 2022 and the related statements of income, changes in members' equity and cash flows for the years then ended and the related notes to the financial statements.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of HFB FenceCo Franchising, LLC as of December 31, 2023 and 2022 and the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audit in accordance with auditing standards generally accepted in the United States of America. Our responsibilities under those standards are further described in the Auditor's Responsibilities for the Audit of the Financial Statements section of our report. We are required to be independent of HFB FenceCo Franchising, LLC and to meet our other ethical responsibilities in accordance with the relevant ethical requirements relating to our audit. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Responsibilities of Management for the Financial Statements

Management is responsible for the preparation and fair presentation of the financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about HFB FenceCo Franchising, LLC's ability to continue as a going concern within one year after the date that the financial statements are available to be issued.

Auditor's Responsibilities for the Audit of the Financial Statements

Our objectives are to obtain reasonable assurance about whether the financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an audito's report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with generally accepted auditing standards will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the financial statements.

In performing an audit in accordance with generally accepted auditing standards, we:

· Exercise professional judgment and maintain professional skepticism throughout the audit.

- Identify and assess the risks of material misstatement of the financial statements, whether due to fraud
 or error, and design and perform audit procedures responsive to those risks. Such procedures include
 examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that
 are appropriate in the circumstances, but not for the purpose of expressing an opinion on the
 effectiveness of HFB FenceCo Franchising, LLC's internal control. Accordingly, no such opinion is
 expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant
 accounting estimates made by management, as well as evaluate the overall presentation of the financial
 statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that
 raise substantial doubt about HFB FenceCo Franchising, LLC's ability to continue as a going concern
 for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control related matters that we identified during the audit.

Matthews, North Carolina April 17, 2024

Janes, Goldaling Associales, Ophis Pul

HFB FenceCo Franchising, LLC Balance Sheets December 31, 2023 and 2022

ASSETS		
Current Assets:	December 31, 2023	December 31, 2022
Cash and Cash Equivalents	\$ 144,359	\$ 527,627
Accounts Receivable, net	631,824	70,206
Other Current Assets	161,557	36,588
Total Current Assets	937,740	634,421
Long-Term Assets:		
Fixed Assets, net	83,301	-
Intangible Assets, net	112,745	
Total Long-Term Assets	196,046	-
TOTAL ASSETS	\$ 1,133,786	\$ 634,421
LIABILITIES & MEMBERS' EQUITY		
CURRENT LIABILITIES		
Accounts Payable	502,268	130,410
Accrued Expenses	1,304,416	82,920
Contract Liabilities	1,005,241	79,003
Line of Credit		165,000
Total Current Liabilities	2,811,925	457,333
Contract Liabilities, net of current portion	174,269	4,369
TOTAL LIABILITIES	2,986,194	461,702
MEMBERS' EQUITY	(1,852,408)	172,719
TOTAL LIABILITIES & MEMBERS' EQUITY	\$ 1,133,786	\$ 634,421

The accompanying notes are an integral part of the financial statements.

HFB FenceCo Franchising, LLC Income Statements and Statements of Members' Equity For the Years Ended December 31, 2023 and 2022

Initial Franchise Fees, net of deferrals (Note 2) \$ 3,969,274 \$ 16	,628
Less: Related Commissions and Fees (2,242,973) (15	,000)
Net Initial Franchise Fees 1,726,301 1	,628
Royalties and Other Fees 36,166	206
Total Revenue 1,762,467 1	,834
Expenses	
General & Administrative 1,159,484 324	,197
Franchise Development 802,801 29	,465
Marketing 899,548 16	,030
Operations 744,976 8	,117
Depreciation and Amortization 11,049	-
Total Expenses 3,617,858 377	,809
Operating Loss (1,855,391) (375	,975)
Interest Expense 19,736 1	,306
Net Loss \$ (1,875,127) \$ (377	,281)
Members' Equity, January 1st \$ 172,719 \$	_
Members' (Distributions) Contributions, net (150,000) 550	,000
Net Loss (1,875,127) (377	,281)
Members' Equity, December 31 \$ (1,852,408) \$ 172	,719

HFB FenceCo Franchising, LLC Statements of Cash Flows For the Years Ended December 31, 2023 and 2022

Cash flows from operating activities:	2023	2022
Net Loss	\$ (1,875,127)	\$ (377,281)
Depreciation and Amortization	11,049	-
Bad Debt Expense	19,423	-
Adjustments to reconcile net income		
to net cash provided by (used in) operating activities:		
Changes in operating assets and liabilities:		
(Increase) in accounts receivable	(581,041)	(70,206)
(Increase) in other current assets	(124,969)	(36,588)
Increase in accounts payable	371,858	130,410
Increase in accrued expenses	1,221,496	82,920
Increase in contract liabilities	1,096,138	83,372
Net cash provided by (used in) operating activities	138,827	(187,373)
Cash flows from investing activities:		
Purchase of fixed assets	(88,325)	-
Purchase of website and other intangibles	(118,770)	
Net cash used in investing activities	(207,095)	-
Cash flows from financing activities:		
(Payments on) Proceeds from Line of Credit	(165,000)	165,000
Members' (Distributions) Contributions	(150,000)	550,000
Net cash (used in) provided by financing activities	(315,000)	715,000
Net change in cash and cash equivalents	(383,268)	527,627
Cash and Cash Equivalents, January 1st	527,627	
Cash and Cash Equivalents, December 31st	\$ 144,359	\$ 527,627
Supplemental Cash Flow Information:		
Cash paid for Interest	\$ 10,000	\$ -

HFB FenceCo Franchising, LLC Notes to the Financial Statements For the Years Ended December 31, 2023 and 2022

Note 1 - Organization and Business

HFB FenceCo Franchising, LLC (the Company) is a North Carolina limited liability corporation formed in 2022 for the purpose of conducting franchise operations. The Company sells franchise rights to own and operate Top Rail franchises which sells, furnishes, and installs wood, steel, aluminum, vinyl and other types of fencing for residential and commercial customers. In addition, the Company provides support services to franchisees. There were 140 franchise territories sold as of December 31, 2023, 97 of which were operational.

Note 2 - Summary of Significant Accounting Policies

Basis of Accounting

The accompanying financial statements have been prepared using the accounting basis of accounting in accordance with accounting principles generally accepted in the United States of America.

Cash and Cash Equivalents

Cash and cash equivalents consist of cash in a bank deposit account. As of December 31, 2023, the Company's cash balance did not exceed the Federal Deposit Insurance Corporation limits.

Revenue Recognition

The Company recognizes revenue in accordance with FASB Accounting Standards Codification Topic 606, Revenue from Contracts with Customers (ASC 606). ASC 606, as amended, is based on the principle that revenue is recognized to depict the contractual transfer of goods or services to customers in an amount that reflects the consideration to which the entity expects to be entitled in exchange for goods or services utilizing a five-step revenue recognition model, which includes (i) identify the contract with a customer; (ii) identify the performance obligations in the contract; (iii) determine the transaction price; (iv) allocate the transaction price to the performance obligations in the contract; and (v) recognize revenue when or as the entity satisfies the performance obligation.

The Company's revenues consist primarily of initial franchise fees charged to franchisees in exchange for the right to own and operate franchises within a certain territory. Once operational, the Company will also receive monthly royalties and other fees from franchisees. ASC 606 did not impact the timing of revenue recognition for the royalties and other fees which are recognized monthly as a percentage of revenue collected by the franchisees.

In applying ASC 606 to initial franchise fees, the Company adopted Accounting Standards Update 2021-02, Franchisors – Revenue from Contracts with Customers: Practical Expedient. The amendments allow a practical expedient that simplifies the application of the guidance regarding identification of performance obligations. Franchisors that are not public business entities may account for pre-opening services provided to a franchisee as distinct from the franchise license (if the services are consistent with those included in a predefined list within the guidance) and can be recognized as a single performance obligation.

Lump sum payments are typically due upon execution of a franchise agreement or renewal. The Company recognizes the revenue related to pre-opening services as a single performance obligation in accordance with the guidance. In addition, the Company identified the completion of training as a performance obligation and recognizes revenue related to this upon completion. The remaining initial fee is related to the license to utilize the Company's brand for a specified period of time which is satisfied equally over the life of the agreement. This deferred revenue is recorded as a contract liability and amortized on a straight-line basis over the life of the agreement.

Revenue Recognition (continued)

HFB FenceCo Franchising, LLC Notes to the Financial Statements For the Years Ended December 31, 2023 and 2022

Franchisees are required to pay a royalty fee of 5-7% of the monthly gross receipts of the franchised business. Gross receipts are reported by the franchisee a month in arrears. The Company also charges a marketing fee or brand fund fee in addition to a technology fee for each franchisee. The payment of royalties and fees typically commences in the month following the completion of training.

Accounts Receivable

Accounts receivable at December 31, 2023 represent initial franchise fees due at the signing of the franchise agreement which have not yet been collected and royalties and fees owed by the franchisees not yet remitted. The Company recorded accounts receivable of \$651,247 and \$70,206 as of December 31, 2023 and 2022, respectively. Management reviewed all outstanding accounts and determined that an allowance for doubtful accounts of \$19,423 and \$0 was necessary as of December 31, 2023 and December 31, 2022, respectively.

Fixed Assets and Intangible Assets

In 2023, fixed assets were purchased with useful lives of 3 to 5 years consisting of software, website, and operation manuals. Depreciation and amortization expense of \$11,049 is included in the accompanying financial statements for the year ended December 31, 2023.

Use of Estimates

The preparation of financial statements in conformity with accounting principles generally accepted in the United States of America requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from these estimates.

Reclassifications

Certain prior year amounts have been reclassified for consistency with the current year presentation. These reclassifications had no effect on the reported results of operations.

Note 3 - Income Taxes

The Company is a limited liability company which is treated as a partnership for tax purposes. As such, the members report and pay income taxes through their personal tax returns. The Company generally does not pay income taxes. As a result, no provision for income taxes has been included in these financial statements.

The Company follows the guidance in ASC 740-10 as it relates to uncertain tax positions as of December 31, 2023 and has evaluated its tax positions taken for filings with the Internal Revenue Service and the state jurisdiction of North Carolina where it operates. The Company has not recorded any reserves, or related accruals for interest and penalties for uncertain income tax positions. The Company is subject to routine audits by taxing jurisdictions; however, there are currently no audits in progress.

Note 4 - Line of Credit

The Company had \$165,000 drawn on a line of credit with a private trust of \$800,000 in 2022 which was paid in full as of December 31, 2023.

In August 2023, the Company entered into a new line of credit with a private entity of \$1,000,000. Nothing was drawn as of December 31, 2023. The line bears interest of 9% and expires in December 2025.

HFB FenceCo Franchising, LLC Notes to the Financial Statements For the Years Ended December 31, 2023 and 2022

Note 5 - Related Party Transactions

The lenders of the line of credit are both private entities and related parties.

All management, research, supervision, financial, accounting, investment, procurement, human resources, legal services, information systems, communications and other required services are provided by DDL Investments, Inc. (DDL), a related party. As of December 31, 2023, there are no employees of the Company as all operational support is provided by DDL or their delegates. Substantially all of the operating expenses of the Company are paid through DDL as part of the management agreement. As of December 31, 2023, approximately \$1.47M (2022 - \$161K) was due to DDL included in accounts payable and accrued expenses in the accompanying balance sheet.

The operating locations in 2022 belong to the founding members of the Company and were not subject to royalty fees during 2022. There are 10 territories in 2023, operating as 2 locations, belonging to the founding members of the Company.

Note 6 – Fixed Assets and Intangible Assets

Fixed assets and intangible assets at December 31, 2023 and 2022 consist of the following:

	2023	2022
Software	\$88,325	\$ -
Less: Accumulated Amortization	(5,024)	-
	\$83,301	-
Website and Manuals	118,770	-
Less: Accumulated Depreciation	(6,025)	-
Fixed Assets, net	\$112,745	\$ -

Note 7 - Subsequent Events

In preparing these financial statements in accordance with Accounting Standards Codification No. 855 – Subsequent Events, the Company has evaluated events and transactions for potential recognition or disclosure April 17, 2024, the date the financial statements were available to be issued. The Company has no knowledge of significant subsequent events as of this date that would require adjustment to or disclosure in the financial statements.

Exhibit J to FDD HFB FenceCo Franchising, LLC

OPERATIONS MANUAL TABLE OF CONTENTS

HFB FENCECO FRANCHISING, LLC

OPERATIONS MANUAL TABLE OF CONTENTS

Sections	Number of Pages
Section A — Introduction to Top Rail FenceChapter 1: General Information	30 <u>6</u>
Chapter 2: The Brand	<u>4</u>
Section B Establishing a Top Rail FenceChapter 3: The Franchise Business Relationship	<u>384</u>
Section C — Top Rail Fence Personnel Chapter 4: Brand Operating Standards	34 <u>16</u>
Chapter 5: Staffing Best Practices	<u>15</u>
Chapter 6: Operations Best Practices	<u>12</u>
Section D—Chapter 7: Tech & Administrative Procedures Best Practices	25 <u>4</u>
Section E — Daily Procedures	25
Section F Selling and Marketing Top Rail Fence Services	17
Total Pages	169 61

EXHIBIT H

TO

Exhibit K

to FDD

HFB FENCECO FRANCHISINGFenceCo Franchising, LLC
FRANCHISE DISCLOSURE DOCUMENT

<u>LIST OF FRANCHISEES AND</u> <u>FRANCHISEES THAT HAVE LEFT THE SYSTEM IN THE PAST FISCAL YEAR</u>

	Name	Entity	Address	Ī	City		State	Terr. State	Zip Code	Phone	Email Address
	<mark>lunde</mark> Chris Ierman	1026 Westbrooke LaneTriangle Fence Builders, LLC	100 Floral Drive Easley		SCChapel 1	<u>Hill</u>	NC	AR	27516	(919) 260- 1480	chris.herman jay.sunde@toprailfence.com
Jar	Shane minChris Ierman	1026 Westbrooke LanePalm Beach Fence and Rail, LLC	13275 Machiave WayEasley	<u>elli</u>	SCPalm Beach Gardens	_	FL ₂₉	FL864 -630 -3099	33418	(404) 313- 9978	chris.herman shane.jarmin@toprailfence.com
Jar	Shane minChris Ierman	Palm Beach Fence and Rail, LLC1026 Westbrooke Lane	13275 Machiavo WayEasley	<u>elli</u>	SC <u>Palm</u> Beach Garden	_	FL ₂₉	FL864 -630 3099	33418	(404) 313- 9978	chris.herman shane.jarmin@toprailfence.com
Jar	Shane minChris Ierman	1026 Westbrooke LanePalm Beach Fence and Rail, LLC	13275 Machiave WayEasley	<u>elli</u>	SCPalm Beach Gardens	_	FL29 642	FL864 -630 3099	33418	(404) 313- 9978	chris.herman shane.jarmin@toprailfence.com
	Chris anMatthew and Alicia Heck	Matthew Heck, LLC1026 Westbrooke Lane	Easley 2920 W. I Salle Street	<u>La</u>	SC Tamp	<u>oa</u>	29642 FL	864- 630- 3099 <u>F</u> <u>L</u>	33607	(317) 607- 8313	chris.herman <u>matt.heck</u> @toprailfence.com
	Chris anMatthew and Alicia Heck	1026 Westbrooke LaneMatthew Heck, LLC	Easley 2920 W. I Salle Street	<u>_a</u>	SC Tamp	o <u>a</u>	29642 FL	864- 630- 3099 <u>F</u> <u>L</u>	33607	(317) 607- 8313	chris.herman matt.heck@toprailfence.com
	Herman Ryan an Epps	Van Epps Construction Services, LLC1026 Westbrooke Lane	Easley 1135 Richmond Gle Circle	<u>n</u>	SC <u>Milto</u>	<u>n</u>	29642 <u>GA</u>	864- 630- 3099 <u>G</u> <u>A</u>	30044	(678) 428- 3842	chris.herman @toprailfenceryan.vanepps@vanepp _s.com
Termar		1026 Westbrooke La	ne Easley	SC	29642	864- 30		chris.herman @toprailfenc			
Iermar		1026 Westbrooke La	ne Easley	SC	29642	864- 30		chris.herman @toprailfenc			
Termar		1026 Westbrooke La	ne Easley	SC	29642	864- 30		chris.herman @toprailfenc			
Iermar		1026 Westbrooke La	ne Easley	SC	29642	864- 30		chris.herman @toprailfenc			
	Ierman Ryan an Epps	1026 Westbrooke Lane Van Epps Construction Services, LLC	Easley 1135 Richmond Gle Circle	<u>n</u>	SC Milto	<u>n</u>	29642 <u>GA</u>	3099 <u>G</u> <u>A</u>	30044	(678) 428- 3842	ehris.herman @toprailfenceryan.vanepps@vanepp _s.com
	Herman Ryan an Epps	1026 Westbrooke Lane Van Epps Construction Services, LLC	Easley 1135 Richmond Gle Circle	<u>n</u>	SC <u>Milto</u>	<u>n</u>	29642 <u>GA</u>	864- 630- 3099 <u>G</u> <u>A</u>	30044	(678) 428- 3842	chris.herman @toprailfenceryan.vanepps@vanepp s.com

	Name	Entity	Address		City		State	Terr. State	Zip Code	Phone	Email Address
	<mark>Sunde</mark> Chris Ierman	1026 Westbrooke LaneTriangle Fence Builders, LLC	100 Floral Drive Easley		SC Chapel I	<u>Hill</u>	<u>NC</u>	AR	27516	(919) 260- 1480	chris.herman jay.sunde@toprailfence.com
Jar	Shane minChris Ierman	1026 Westbrooke LanePalm Beach Fence and Rail, LLC	13275 Machiave Way Easley	<u>lli</u>	SCPalm Beach Gardens		FL ₂₉ 642	6.211	33418	(404) 313- 9978	chris.herman shane.jarmin@toprailfence.com
Jar	Shane minChris Ierman	Palm Beach Fence and Rail, LLC1026 Westbrooke Lane	13275 Machiave Way Easley	<u>lli</u>	SCPalm Beach Gardens		FL29	630	33418	(404) 313- 9978	chris.herman <u>shane.jarmin</u> @toprailfence.com
Jar	Shane minChris Ierman	1026 Westbrooke LanePalm Beach Fence and Rail, LLC	13275 Machiave WayEasley	<u>lli</u>	SCPalm Beach Gardens		FL ₂₉	630	33418	(404) 313- 9978	chris.herman shane.jarmin@toprailfence.com
	Chris anMatthew and Alicia Heck	Matthew Heck, LLC1026 Westbrooke Lane	Easley 2920 W. I. Salle Street	<u>.a</u>	SC Tampa	<u>1</u>	29642 <u>FL</u>	864- 630- 3099 <u>F</u> <u>L</u>	33607	(317) 607- 8313	ehris.herman matt.heck@toprailfence.com
	Chris anMatthew and Alicia Heck	1026 Westbrooke Lane Matthew Heck, LLC	Easley 2920 W. L. Salle Street	<u>a</u>	SC <u>Tampa</u>	<u>1</u>	29642 <u>FL</u>	864- 630- 3099 <u>F</u> <u>L</u>	33607	(317) 607- 8313	chris.herman matt.heck@toprailfence.com
	Ilerman Ryan an Epps	Van Epps Construction Services, LLC1026 Westbrooke Lane	Easley 1135 Richmond Gler Circle	1	SC <u>Miltor</u>	<u>1</u>	29642 GA		30044	(678) 428- 3842	chris.herman @toprailfenceryan.vanepps@vanepp s.com
	Herman Ryan an Epps	1026 Westbrooke Lane Van Epps Construction Services, LLC	Easley 1135 Richmond Gler Circle	<u>1</u>	SC <u>Miltor</u>	<u>1</u>	29642 GA	3099 <u>G</u> <u>A</u>	30044	(678) 428- 3842	ehris.herman @toprailfenceryan.vanepps@vanepp s.com
	HermanRyan an Epps	1026 Westbrooke Lane Van Epps Construction Services, LLC	Easley 1135 Richmond Gler Circle	<u>1</u>	<u>SCMiltor</u>	<u>1</u>	29642 <u>GA</u>		30044	(678) 428- 3842	chris.herman <u>@toprailfenceryan.vanepps@vanepp</u> <u>s</u> .com
Chr	is Herman	Westbrooke Fencing, Inc.	1026 Westbrook Lane	xe	Easley		SC	GA	29642	864-630- 3099	chris.herman @toprailfence.com
r Berry		104 Wellington Avenu	ue Greenville	SC	29609	719 05		tucker.berry @toprailfence	e.com		
r Berry		104 Wellington Avenu	ue Greenville	SC	29609	719-		tucker.berry @toprailfence	e.com		
r Berry		104 Wellington Avenu	ue Greenville	SC	29609	719-		tucker.berry @toprailfence	2.com		

104 Wellington Avenue

Berry

29609

SC

Greenville

719-651-0523 @toprailfence.com

tucker.berry
@toprailfence.com

1	Name	Entity	Address	City	State	Terr. State	Zip Code	Phone	Email Address		
	undeChris Ierman	1026 Westbrooke Lane Triangle Fence Builders, LLC	100 Floral Drive Easley	SCChapel Hill	<u>NC</u>	AR	27516	(919) 260- 1480	chris.herman jay.sunde@toprailfence.com		
Jarr	Shane minChris Ierman	1026 Westbrooke LanePalm Beach Fence and Rail, LLC	13275 Machiavelli WayEasley	SC <u>Palm</u> Beach Gardens	<u>FL29</u> 642	FL864 -630- 3099	33418	(404) 313- 9978	ehris.herman shane.jarmin@toprailfence.com		
Jarr	Shane ninChris Ierman	Palm Beach Fence and Rail, LLC1026 Westbrooke Lane	13275 Machiavelli WayEasley	SCPalm Beach Gardens	FL ₂₉ 642	FL864 -630- 3099	33418	(404) 313- 9978	chris.herman <u>shane.jarmin</u> @toprailfence.com		
Jarr	Shane ninChris Ierman	1026 Westbrooke LanePalm Beach Fence and Rail, LLC	13275 Machiavelli WayEasley	SC <u>Palm</u> Beach Gardens	FL ₂₉	FL864 -630- 3099	33418	(404) 313- 9978	chris.herman shane.jarmin@toprailfence.com		
Herma Heck	Chris mMatthew and Alicia Heck	Matthew Heck, LLC 1026 Westbrooke Lane	Easley2920 W. La Salle Street	S€ Tampa	29642 FL	864- 630- 3099 <u>F</u> <u>L</u>	33607	(317) 607- 8313	ehris.herman matt.heck@toprailfence.com		
Herma Heck	Chris mMatthew and Alicia Heck	1026 Westbrooke LaneMatthew Heck, LLC	Easley2920 W. La Salle Street	SC Tampa	29642 FL	864- 630- 3099 <u>F</u> <u>L</u>	33607	(317) 607- 8313	chris.herman matt.heck@toprailfence.com		
	Ierman Ryan an Epps	Van Epps Construction Services, LLC1026 Westbrooke Lane	Easley 1135 Richmond Glen Circle	SC <u>Milton</u>	29642 <u>GA</u>	864- 630- 3099 <u>G</u> <u>A</u>	30044	(678) 428- 3842	chris.herman @toprailfenceryan.vanepps@vanepp s.com		
r Berry		104 Wellington Aven	ue Greenville So			icker.berry toprailfence	e.com				

HFB FenceCo Franchising, LLC
Current Franchisees with Unopened Outlets as of December 31, 2023:2022

]	Name	Entity	Address	City	State	Terr. State	Zip Code	Phone	Email Address
C	<u>Herman</u> Brent hapman e Vandall	729 E. Mountain StreetWestbrook e Fencing, Inc.	1026 Westbrooke <u>Lane</u> Kernersville	<u>Easley</u> NC	27284 SC	704- 477- 7963 <u>G</u> <u>A</u>	29642	864-630- 3099	brent.chapman @windowhero.com VandallLJ@gmail.comchris.herman @toprailfence.com
Chri	s Herman	Westbrooke Fencing, Inc.	1026 Westbrooke <u>Lane</u>	<u>Easley</u>	<u>SC</u>	<u>GA</u>	<u>29642</u>	864-630- 3099	chris.herman@toprailfence.com
Chri	s Herman	Westbrooke Fencing, Inc.	1026 Westbrooke Lane	Easley	<u>SC</u>	GA	<u>29642</u>	864-630- 3099	chris.herman@toprailfence.com
Chri	s Herman	Westbrooke Fencing, Inc.	1026 Westbrooke Lane	Easley	<u>SC</u>	<u>GA</u>	<u>29642</u>	864-630- 3099	chris.herman@toprailfence.com
Luke '	t Chapman VandallBret ory Staples	Fortes In Fide Fencing, LLC	729 E. Mountain 1054 19th Street West	KernersvilleDes Moines	NC <u>IA</u>	27284 <u>I</u> <u>A</u>	704-477- 7963 <u>502</u> 6 <u>5</u>	(515) 778- 3363	brent.chapman @windowhero.com VandallLJ@gmail.combret.staples@t oprailfence.com
Luke '	t-Chapman VandallBret ory Staples	Fortes In Fide Fencing, LLC	729 E. Mountain 1054 19th Street West	KernersvilleDes Moines	NC <u>IA</u>	27284 <u>I</u> <u>A</u>	704-477- 7963 <u>502</u> 6 <u>5</u>	(515) 778- 3363	brent.chapman @windowhero.com VandallLJ@gmail.combret.staples@t oprailfence.com
	Mark Johns Lisa Johns	24651 SE 21* PlaceBailey Johns Family Ventures, Inc.	Sammamish2794 Charter Oak Drive	WA <u>Aurora</u>	98075 <u>IL</u>	206- 696- 1151 <u>IL</u>	60502	(720) 530- 1461	ohmjhyun @gmailmark.johns@toprailfence.c om
	Mark Johns Lisa Johns	24651 SE 21** PlaceBailey Johns Family Ventures, Inc.	Sammamish2794 Charter Oak Drive	WA <u>Aurora</u>	98075 <u>IL</u>	206- 696- 1151 <u>IL</u>	60502	(720) 530- 1461	ohmjhyun @gmailmark.johns@toprailfence.c om
~	Mark Johns Lisa Johns	24651 SE 21** PlaceBailey Johns Family Ventures, Inc.	Sammamish2794 Charter Oak Drive	WA <u>Aurora</u>	98075 <u>IL</u>	206- 696- 1151 <u>IL</u>	60502	(720) 530- 1461	ohmjhyun @gmailmark.johns@toprailfence.c om
lark Jo	hns and Lisa Johns	Bailey Johns Family Ventures, Inc.	2794 Charter Oak <u>Drive</u>	<u>Aurora</u>	<u>IL</u>	<u>IL</u>	60502	(720) 530- 1461	mark.johns@toprailfence.com
And	rew Deck	ECAC, LLC	12247 Old Stone Drive	Indianapolis	<u>IN</u>	<u>IN</u>	<u>46236</u>	(317) 441- 5712	andy.deck@toprailfence.com
And	rew Deck	ECAC, LLC	12247 Old Stone Drive	Indianapolis	<u>IN</u>	<u>IN</u>	<u>46236</u>	(317) 441- 5712	andy.deck@toprailfence.com
And	rew Deck	ECAC, LLC	12247 Old Stone Drive	Indianapolis	<u>IN</u>	<u>IN</u>	<u>46236</u>	(317) 441- 5712	andy.deck@toprailfence.com

N	<u>ame</u>	<u>Entity</u>	Address	City	Stat e	Terr. State	Zip Code	<u>Phone</u>	Email Address
Jay	<u>Sunde</u>	Triangle Fence Builders, LLC	100 Floral Drive	<u>Chapel</u> <u>Hill</u>	<u>NC</u>	<u>KS</u>	<u>27516</u>	(919) 260- 1480	jay.sunde@toprailfence.com
Jay	<u>Sunde</u>	Triangle Fence Builders, LLC	100 Floral Drive	<u>Chapel</u> <u>Hill</u>	<u>NC</u>	<u>KS</u>	<u>27516</u>	(919) 260- 1480	jay.sunde@toprailfence.com
Jay	<u>Sunde</u>	Triangle Fence Builders, LLC	100 Floral Drive	<u>Chapel</u> <u>Hill</u>	<u>NC</u>	<u>KS</u>	<u>27516</u>	(919) 260- 1480	jay.sunde@toprailfence.com
Jay :	<u>Sunde</u>	Triangle Fence Builders, LLC	100 Floral Drive	<u>Chapel</u> <u>Hill</u>	<u>NC</u>	<u>KS</u>	<u>27516</u>	(919) 260- 1480	jay.sunde@toprailfence.com
Cheek	Michael & Sarah Cheek	BS Enterprise, Inc.	276 Hub Lake Road	<u>Trout</u>	<u>LA</u>	<u>LA</u>	<u>71371</u>	(318) 794- 6483	billy.cheek@toprailfence.com
Cheek	Michael & Sarah Cheek	BS Enterprise, Inc.	276 Hub Lake Road	Trout	<u>LA</u>	<u>LA</u>	71371	(318) 794- 6483	billy.cheek@toprailfence.com
Cheek	Michael & Sarah Cheek	BS Enterprise, Inc.	276 Hub Lake Road	<u>Trout</u>	<u>LA</u>	<u>LA</u>	71371	(318) 794- 6483	billy.cheek@toprailfence.com
onatha	n Arnold	Altera Holdings, LLC	1301 Orchard Lakes Drive	St. Louis	<u>MO</u>	MO	63146	(314) 323- 4879	jonathan.arnold@toprailfence.com
Jay	<u>Sunde</u>	Triangle Fence Builders, LLC	100 Floral Drive	<u>Chapel</u> Hill	<u>NC</u>	<u>NC</u>	27516	(919) 260- 1480	jay.sunde@toprailfence.com
Jay	<u>Sunde</u>	Triangle Fence Builders, LLC	100 Floral Drive	<u>Chapel</u> <u>Hill</u>	<u>NC</u>	<u>NC</u>	<u>27516</u>	(919) 260- 1480	jay.sunde@toprailfence.com
Taylo	r Moss	T Moss Enterprises, LLC.	1000 East Woodlawn Road Unit 306	Charlotte	<u>NC</u>	<u>NC</u>	<u>28209</u>	(980) 339- 6891	taylor.moss@toprailfence.com
Taylo	r Moss	T Moss Enterprises, LLC.	1000 East Woodlawn Road Unit 306	Charlotte	<u>NC</u>	<u>NC</u>	<u>28209</u>	(980) 339- 6891	taylor.moss@toprailfence.com
Tucke	r Berry	Brookview Renovations, Inc.	104 Wellington Ave.	Greenville	<u>SC</u>	<u>NC</u>	<u>29609</u>	(719) 651- 0523	tucker.berry@toprailfence.com
Tucke	r Berry	Brookview Renovations, Inc.	104 Wellington Ave.	Greenville	<u>SC</u>	<u>NC</u>	<u>29609</u>	(719) 651- 0523	tucker.berry@toprailfence.com
Tucke	r Berry	Brookview Renovations, Inc.	104 Wellington Ave.	Greenville	<u>SC</u>	<u>NC</u>	<u>29609</u>	(719) 651- 0523	tucker.berry@toprailfence.com

]	<u>Name</u>	<u>Entity</u>	Address	<u>City</u>	<u>State</u>	Terr. State	Zip Code	<u>Phone</u>	Email Address
Tuc	ker Berry	Brookview Renovations, Inc.	104 Wellington Ave.	Greenville	<u>SC</u>	<u>NC</u>	<u>29609</u>	(719) 651-0523	tucker.berry@toprailfence.com
Tuc	ker Berry	Brookview Renovations, Inc.	104 Wellington Ave.	Greenville	<u>SC</u>	<u>NC</u>	<u>29609</u>	<u>(719)</u> <u>651-0523</u>	tucker.berry@toprailfence.com
	hapman and Vandall	<u>Chapman</u> <u>Vandall Inc.</u>	729 E. Mountain Street	Kernersville	<u>NC</u>	<u>NC</u>	<u>27284</u>	704-477- 7963	brent.chapman@toprailfence.com
	hapman and Vandall	<u>Chapman</u> <u>Vandall Inc.</u>	729 E. Mountain Street	Kernersville	<u>NC</u>	<u>NC</u>	<u>27284</u>	704-477- 7963	brent.chapman@toprailfence.com
	hapman and Vandall	<u>Chapman</u> Vandall Inc.	729 E. Mountain Street	Kernersville	<u>NC</u>	<u>NC</u>	<u>27284</u>	<u>704-477-</u> <u>7963</u>	brent.chapman@toprailfence.com
Chri	s Herman	Westbrooke Fencing, Inc.	1026 Westbrooke Lane	Easley	<u>SC</u>	<u>NC</u>	29642	864-630- 3099	chris.herman@toprailfence.com
Tuc	<u>ker Berry</u>	Brookview Renovations, Inc.	104 Wellington Avenue	Greenville	<u>SC</u>	<u>NC</u>	<u>29609</u>	719-651- 0523	tucker.berry@toprailfence.com
Kris	and Carrie Orth	Lightway Ventures, LLC	920 Countrywood Way	<u>Sapulpa</u>	<u>OK</u>	<u>OK</u>	<u>74066</u>	(302) 331-0961	kris.orth@toprailfence.com
Tay	lor Moss	T Moss Enterprises, LLC	1000 East Woodlawn Road Unit 306	Charlotte	<u>NC</u>	<u>SC</u>	<u>28209</u>	(980) 339-6891	taylor.moss@toprailfence.com
Chri	s Herman	Westbrooke Fencing, Inc.	1026 Westbrooke Lane	Easley	<u>SC</u>	<u>SC</u>	<u>29642</u>	864-630- 3099	chris.herman@toprailfence.com
Chri	s Herman	Westbrooke Fencing, Inc.	1026 Westbrooke Lane	Easley	<u>SC</u>	<u>SC</u>	<u>29642</u>	864-630- 3099	chris.herman@toprailfence.com
Chri	s Herman	Westbrooke Fencing, Inc.	1026 Westbrooke Lane	Easley	<u>SC</u>	<u>SC</u>	29642	864-630- 3099	chris.herman@toprailfence.com
Chri	s Herman	Westbrooke Fencing, Inc.	1026 Westbrooke Lane	Easley	<u>SC</u>	<u>SC</u>	<u>29642</u>	864-630- 3099	chris.herman@toprailfence.com
Chri	s Herman	Westbrooke Fencing, Inc.	1026 Westbrooke Lane	Easley	<u>SC</u>	<u>SC</u>	<u>29642</u>	864-630- 3099	chris.herman@toprailfence.com

1	<u>Vame</u>	<u>Entity</u>	Address	<u>City</u>	<u>State</u>	Terr. State	Zip Code	<u>Phone</u>	Email Address
Tuc	ker Berry	Brookview Renovations, Inc.	104 Wellington Avenue	Greenville	<u>SC</u>	<u>SC</u>	29609	719-651- 0523	tucker.berry@toprailfence.com
Tuc	ker Berry	Brookview Renovations, Inc.	104 Wellington Avenue	Greenville	<u>SC</u>	<u>SC</u>	<u>29609</u>	719-651- 0523	tucker.berry@toprailfence.com
Tuc	ker Berry	Brookview Renovations, Inc.	104 Wellington Avenue	Greenville	<u>SC</u>	<u>SC</u>	<u>29609</u>	719-651- 0523	tucker.berry@toprailfence.com
Tuc	ker Berry	Brookview Renovations, Inc.	104 Wellington Avenue	Greenville	<u>SC</u>	<u>SC</u>	<u>29609</u>	719-651- 0523	tucker.berry@toprailfence.com
Der	rek Long	GTTA, LLC.	1701 Eagle Drive	<u>Hixson</u>	<u>TN</u>	<u>TN</u>	<u>37343</u>	(435) 764- 8012	derrek.long@toprailfence.com
<u>Der</u>	rek Long	GTTA, LLC.	1701 Eagle Drive	<u>Hixson</u>	<u>TN</u>	<u>TN</u>	<u>37343</u>	(435) 764- 8012	derrek.long@toprailfence.com
	na & Jack nannon	Helack, LLC	4885 Gwynne Road	Memphis	<u>TN</u>	<u>TN</u>	<u>38117</u>	<u>(901) 237-</u> <u>3637</u>	jack.shannon@toprailfence.com
	na & Jack nannon	Helack, LLC	4885 Gwynne Road	Memphis	<u>TN</u>	<u>TN</u>	<u>38117</u>	(901) 237- 3637	jack.shannon@toprailfence.com
ellon	& Robert & Catherine ward Wile	CCLR, LLC	11 Double Palm Way	Ormond Beach	<u>FL</u>	<u>TN</u>	<u>32174</u>	(770) 402- 3604	laura.wellon@toprailfence.com
ellon	& Robert & Catherine ward Wile	CCLR, LLC	11 Double Palm Way	Ormond Beach	<u>FL</u>	<u>TN</u>	<u>32174</u>	(770) 402- 3604	laura.wellon@toprailfence.com
ellon	& Robert & Catherine ward Wile	CCLR, LLC	11 Double Palm Way	Ormond Beach	<u>FL</u>	<u>TN</u>	<u>32174</u>	(770) 402- 3604	laura.wellon@toprailfence.com
ellon	& Robert & Catherine ward Wile	CCLR, LLC	11 Double Palm Way	Ormond Beach	<u>FL</u>	TN	32174	(770) 402- 3604	laura.wellon@toprailfence.com
ellon	& Robert & Catherine ward Wile	CCLR, LLC	11 Double Palm Way	Ormond Beach	<u>FL</u>	<u>TN</u>	<u>32174</u>	(770) 402- 3604	laura.wellon@toprailfence.com

Na	<u>me</u>	<u>Entity</u>	<u>Address</u>	City	<u>State</u>	Terr. State	Zip Code	<u>Phone</u>	Email Address
Chris l	<u>Herman</u>	Westbrooke Fencing, Inc.	1026 Westbrooke Lane	Easley	<u>SC</u>	TN	29642	864-630- 3099	chris.herman@toprailfence.com
Chris l	<u>Herman</u>	Westbrooke Fencing, Inc.	1026 Westbrooke Lane	Easley	<u>SC</u>	<u>TN</u>	<u>29642</u>	864-630- 3099	chris.herman@toprailfence.com
Chris l	<u>Herman</u>	Westbrooke Fencing, Inc.	1026 Westbrooke Lane	Easley	<u>SC</u>	<u>TN</u>	<u>29642</u>	864-630- 3099	chris.herman@toprailfence.com
Chris l	<u>Herman</u>	Westbrooke Fencing, Inc.	1026 Westbrooke Lane	<u>Easley</u>	<u>SC</u>	<u>TN</u>	<u>29642</u>	864-630- 3099	chris.herman@toprailfence.com
Chris l	<u>Herman</u>	Westbrooke Fencing, Inc.	1026 Westbrooke Lane	Easley	<u>SC</u>	<u>TN</u>	<u>29642</u>	864-630- 3099	chris.herman@toprailfence.com
ames	<u>Graass</u>	Rankin Capital, LLC	871 Lake Carolyn Parkway Unit 163	Irving	<u>TX</u>	<u>TX</u>	<u>75039</u>	(214) 502- 9131	james.graass@toprailfence.com
	<u>ı Mark</u> ill	<u>Team H&H</u> Investments, LLC.	1405 Woodlands Drive	<u>Tyler</u>	<u>TX</u>	<u>TX</u>	<u>75703</u>	<u>(903) 920-</u> <u>7979</u>	josh.hill@toprailfence.com
ames	<u>Graass</u>	Rankin Capital, LLC	871 Lake Carolyn Parkway Unit 163	Irving	<u>TX</u>	<u>TX</u>	<u>75039</u>	(214) 502- 9131	james.graass@toprailfence.com
ames	<u>Graass</u>	Rankin Capital, LLC	871 Lake Carolyn Parkway Unit 163	Irving	<u>TX</u>	<u>TX</u>	<u>75039</u>	(214) 502- 9131	james.graass@toprailfence.com
ames	<u>Graass</u>	Rankin Capital, LLC	871 Lake Carolyn Parkway Unit 163	Irving	<u>TX</u>	<u>TX</u>	<u>75039</u>	<u>(214) 502-</u> <u>9131</u>	james.graass@toprailfence.com
ames	<u>Graass</u>	Rankin Capital, LLC	871 Lake Carolyn Parkway Unit 163	Irving	<u>TX</u>	<u>TX</u>	<u>75039</u>	(214) 502- 9131	james.graass@toprailfence.com
Mendo	ando oza and r Ruiz	RIME Residential & Commercial Services, LLC	4800 Steiner Ranch Blvd. Unit #15-202	<u>Austin</u>	TX	<u>TX</u>	<u>78732</u>	(305) 343- 5238	rudy.bennett@toprailfence.com
Mendo	ando oza and r Ruiz	RIME Residential & Commercial Services, LLC	4800 Steiner Ranch Blvd. Unit #15-202	Austin	TX	<u>TX</u>	<u>78732</u>	(305) 343- 5238	rudy.bennett@toprailfence.com
Mendo	ando oza and r Ruiz	RIME Residential & Commercial Services, LLC	4800 Steiner Ranch Blvd. Unit #15-202	Austin	<u>TX</u>	TX	<u>78732</u>	(305) 343- 5238	rudy.bennett@toprailfence.com

<u>N</u>	<u>lame</u>	<u>Entity</u>	Address	<u>City</u>	<u>State</u>	Terr. State	Zip Code	<u>Phone</u>	Email Address
	lo Mendoza vier Ruiz	RIME Residential & Commercial Services, LLC	4800 Steiner Ranch Blvd. Unit #15-202	<u>Austin</u>	<u>TX</u>	<u>TX</u>	<u>78732</u>	(305) 343- 5238	rudy.bennett@toprailfence.com
	and Tracy zeman	Advanced Home Enterprises, LLC	29010 Stratwood Bend Lake	Katy	<u>TX</u>	<u>TX</u>	77494	<u>(713) 628-</u> <u>0170</u>	brett.bozeman@toprailfence.com
	and Tracy zeman	Advanced Home Enterprises, LLC	29010 Stratwood Bend Lake	Katy	<u>TX</u>	TX	77494	(713) 628- 0170	brett.bozeman@toprailfence.com
	and Tracy zeman	Advanced Home Enterprises, LLC	29010 Stratwood Bend Lake	Katy	<u>TX</u>	<u>TX</u>	77494	(713) 628- 0170	brett.bozeman@toprailfence.com
Rudy	Bennett	JJLB, LLC	14591 South Chaumont Court	Draper	<u>UT</u>	<u>UT</u>	84020	(801) 837- 3582	rudy.bennett@toprailfence.com
Rudy	Bennett	JJLB, LLC	14591 South Chaumont Court	<u>Draper</u>	<u>UT</u>	<u>UT</u>	84020	(801) 837- 3582	rudy.bennett@toprailfence.com
Rudy	Bennett	JJLB, LLC	14591 South Chaumont Court	<u>Draper</u>	<u>UT</u>	<u>UT</u>	<u>84020</u>	(801) 837- 3582	rudy.bennett@toprailfence.com
Lace	Stewart & y Lavette ewart	Stewart Fencing, LLC	11175 Lakeshore Court	Glen Allen	<u>VA</u>	<u>VA</u>	23059	(804) 539- 1340	bob.stewart@toprailfence.com
Lace	Stewart & v Lavette ewart	Stewart Fencing, LLC	11175 Lakeshore Court	Glen Allen	<u>VA</u>	<u>VA</u>	23059	(804) 539- 1340	bob.stewart@toprailfence.com
Ja	/ Ohm	Sonamu, LLC.	24651 SE 21st Place	Sammamish	WA	WA	98075	<u>206-696-</u> <u>1151</u>	jay.ohm@toprailfence.com
Jay	/ Ohm	Sonamu, LLC.	24651 SE 21st Place	Sammamish	<u>WA</u>	<u>WA</u>	98075	206-696- 1151	jay.ohm@toprailfence.com
<u>Ja</u>	/ Ohm	Sonamu, LLC.	24651 SE 21st Place	Sammamish	<u>WA</u>	<u>WA</u>	<u>98075</u>	<u>206-696-</u> <u>1151</u>	jay.ohm@toprailfence.com

HFB FenceCo Franchising, LLC Franchisees with Unopened Outlets as of December 31, 2023

-	<u>Name</u>	<u>Entity</u>	Address	City	State	Terr. State	Zip Code	Phone	Email Address
Ster	ve Dravis	Synergy Fence, LLC	1871 Winterberry Street	Mobile	AL	<u>AL</u>	<u>36695</u>	(251) 751- 8059	steve.dravis@toprailfence.com
	Raferty and n Ryan West	REST Holdings, Inc.	6505 E Dorado Avenue	Greenwood Village	<u>CO</u>	<u>CO</u>	80111	<u>(773) 485-</u> <u>1780</u>	sean.raftery@toprailfence.com
	Raferty and n Ryan West	REST Holdings, Inc.	6505 E Dorado Avenue	Greenwood Village	<u>CO</u>	<u>CO</u>	80111	(773) 485- 1780	sean.raftery@toprailfence.com
	Raferty and n Ryan West	REST Holdings, Inc.	6505 E Dorado Avenue	Greenwood Village	<u>CO</u>	<u>CO</u>	80111	(773) 485- 1780	sean.raftery@toprailfence.com
	Raferty and n Ryan West	REST Holdings, Inc.	6505 E Dorado Avenue	Greenwood Village	<u>CO</u>	<u>CO</u>	80111	(773) 485- 1780	sean.raftery@toprailfence.com
	Raferty and n Ryan West	REST Holdings, Inc.	6505 E Dorado Avenue	Greenwood Village	<u>CO</u>	<u>CO</u>	80111	(773) 485- 1780	sean.raftery@toprailfence.com
Tar	ner Tom	Cavaco Fencing, LLC	1930 Kendall Street	Lakewood	<u>CO</u>	<u>CO</u>	80214	(720) 207- 1835	tanner.tom@toprailfence.com
Tar	ner Tom	<u>Cavaco</u> Fencing, LLC	1930 Kendall Street	Lakewood	<u>CO</u>	<u>CO</u>	80214	(720) 207- 1835	tanner.tom@toprailfence.com
Christo	pher Gragg	High Plains Patriot, Inc.	5746 Regal Oak Lane	Parker	<u>CO</u>	<u>CO</u>	80134	(720) 839- 5428	chris.gragg@toprailfence.com
Christo	pher Gragg	High Plains Patriot, Inc.	5746 Regal Oak Lane	Parker	<u>CO</u>	<u>CO</u>	80134	(720) 839- 5428	chris.gragg@toprailfence.com
Christo	pher Gragg	High Plains Patriot, Inc.	5746 Regal Oak Lane	Parker	<u>CO</u>	<u>CO</u>	80134	(720) 839- 5428	chris.gragg@toprailfence.com
Bryce	Humprey	BDH Holdings, Inc.	98 Riverside Drive Apt. 5F	New York	NY	CT	10024	(214) 794- 8789	bryce.humphrey@gmail.com
dam J Kellin	. Krasne and i A. Krasne	Curious Craft Ventures, LLC	6597 Willow Lake Circle	Fort Myers	<u>FL</u>	FL	33966	(402) 909- 2592	adam.krasne@toprailfence.com
dam J Kellin	. Krasne and i A. Krasne	Currious Craft Ventures, LLC	6597 Willow Lake Circle	Fort Myers	FL	<u>FL</u>	33966	(402) 909- 2592	adam.krasne@toprailfence.com

HFB FenceCo Franchising, LLC Franchisees with Unopened Outlets as of December 31, 2023

<u>Name</u>	Entity	Address	City	State	Terr. State	Zip Code	Phone	Email Address
dam J Krasne and Kellini A. Krasne	Currious Craft Ventures, LLC	6597 Willow Lake Circle	Fort Myers	FL	<u>FL</u>	33966	(402) 909- 2592	adam.krasne@toprailfence.com
uclides Jose "EJ" Marin	Marin Gallucci Group, Inc.	822 Azalea Place	Weston	FL	<u>FL</u>	<u>33327</u>	<u>(954) 309-</u> <u>3009</u>	ej.marin@toprailfence.com
uclides Jose "EJ" Marin	Marin Gallucci Group, Inc.	822 Azalea Place	Weston	FL	<u>FL</u>	33327	<u>(954) 309-</u> <u>3009</u>	ej.marin@toprailfence.com
uclides Jose "EJ" Marin	Marin Gallucci Group, Inc.	822 Azalea Place	Weston	FL	<u>FL</u>	33327	<u>(954) 309-</u> <u>3009</u>	ej.marin@toprailfence.com
ison Ecechi Okoro Orji	Elo Fencing Group, LLC	1244 SW 123rd Street	<u>Miami</u>	FL	<u>FL</u>	<u>33186</u>	(305) 281- 0694	edison.okoro@toprailfence.com
ison Ecechi Okoro <u>Orji</u>	Elo Fencing Group, LLC	1244 SW 123rd Street	Miami	<u>FL</u>	<u>FL</u>	<u>33186</u>	(305) 281- 0694	edison.okoro@toprailfence.com
Justin Green	Next Chapter Investments. LLC	511 Newhall Lane	<u>Debary</u>	<u>FL</u>	<u>FL</u>	<u>32713</u>	(407) 416- 2573	justin.green@toprailfence.com
Justin Green	Next Chapter Investments. LLC	511 Newhall Lane	Debary	FL	FL	32713	(407) 416- 2573	justin.green@toprailfence.com
ven E. McFate and Kristine McFate	No Boundaries, LLC	758 E Huckleberry Court	<u>Kuna</u>	<u>ID</u>	<u>ID</u>	<u>83634</u>	(208) 863- 4337	kris.mcfate@toprailfence.com
ven E. McFate and Kristine McFate	No Boundaries, LLC	758 E Huckleberry Court	Kuna	<u>ID</u>	<u>ID</u>	<u>83634</u>	(208) 863- 4337	kris.mcfate@toprailfence.com
Frank Martin	Fence King, LLC	1197 Washington Blvd.	<u>Birmingham</u>	MI	<u>MI</u>	<u>48009</u>	(586) 801- 5500	frank.martin@toprailfence.com
Frank Martin	Fence King, LLC	1197 Washington Blvd.	Birmingham	MI	<u>MI</u>	<u>48009</u>	<u>(586) 801-</u> <u>5500</u>	frank.martin@toprailfence.com
Frank Martin	Fence King, LLC	1197 Washington Blvd.	Birmingham	<u>MI</u>	<u>MI</u>	<u>48009</u>	(586) 801- 5500	frank.martin@toprailfence.com
Frank Martin	Fence King, LLC	1197 Washington Blvd.	Birmingham	MI	<u>MI</u>	<u>48009</u>	(586) 801- 5500	frank.martin@toprailfence.com
	•			· · · · · · · · · · · · · · · · · · ·				

HFB FenceCo Franchising, LLC Franchisees with Unopened Outlets as of December 31, 2023

<u>N</u>	ame_	<u>Entity</u>	Address	<u>City</u>	State	Terr. State	Zip Code	<u>Phone</u>	Email Address
	Harris and Nehring	Fortress 6, LLC	11239 Parleys Cone Court	Las Vegas	NV	MN	<u>89135</u>	<u>(702) 328-</u> <u>2236</u>	chade.nehring@toprailfence.com
	Harris and Nehring	Fortress 6, LLC	11239 Parleys Cone Court	Las Vegas	NV	MN	<u>89135</u>	<u>(702) 328-</u> <u>2236</u>	chade.nehring@toprailfence.com
	Harris and Nehring	Fortress 6, LLC	11239 Parleys Cone Court	Las Vegas	NV	MN	<u>89135</u>	(702) 328- 2236	chade.nehring@toprailfence.com
Chade	Harris and Nehring	Fortress 6, LLC	11239 Parleys Cone Court	Las Vegas	NV	MN	<u>89135</u>	<u>(702) 328-</u> <u>2236</u>	chade.nehring@toprailfence.com
and	Williamson Kathy iamson	Marilee Ventures, LLC	105 Mellomeade Pkwy.	Brandon	<u>MS</u>	<u>MS</u>	<u>39047</u>	(601) 862- 7558	richard.williamson@toprailfence.c om
Seth	Schwed	Schwed Fence Company, LLC.	10568 Churchhill Drive	Powell	<u>OH</u>	<u>OH</u>	43065	(614) 715- 0196	seth.schwed@toprailfence.com
Seth	Schwed	Schwed Fence Company, LLC.	10568 Churchhill Drive	Powell	<u>OH</u>	<u>OH</u>	<u>43065</u>	(614) 715- 0196	seth.schwed@toprailfence.com
Seth	Schwed	Schwed Fence Company, LLC.	10568 Churchhill Drive	Powell	<u>OH</u>	<u>OH</u>	<u>43065</u>	<u>(614) 715-</u> <u>0196</u>	seth.schwed@toprailfence.com
	Finn and la Finn	FinnWin Corp.	2109 Bellanca Court	<u>Flower</u> <u>Mound</u>	<u>TX</u>	<u>TX</u>	<u>75028</u>	(972) 746- 7374	jerry.finn@toprailfence.com
	Finn and la Finn	FinnWin Corp.	2109 Bellanca Court	<u>Flower</u> <u>Mound</u>	<u>TX</u>	<u>TX</u>	<u>75028</u>	<u>(972) 746-</u> <u>7374</u>	jerry.finn@toprailfence.com
	Finn and da Finn	FinnWin Corp.	2109 Bellanca Court	Flower Mound	<u>TX</u>	<u>TX</u>	<u>75028</u>	<u>(972) 746-</u> <u>7374</u>	jerry.finn@toprailfence.com
	Finn and da Finn	FinnWin Corp.	2109 Bellanca Court	<u>Flower</u> <u>Mound</u>	<u>TX</u>	<u>TX</u>	<u>75028</u>	(972) 746- 7374	jerry.finn@toprailfence.com
	v Frye and new Frye	AMAF Services, LLC.	5960 W. Parker Road #278	Plano	<u>TX</u>	<u>TX</u>	<u>75093</u>	(949) 538- 7993	andy.frye@toprailfence.com
	v Frye and new Frye	AMAF Services, LLC.	5960 W. Parker Road #278	<u>Plano</u>	<u>TX</u>	<u>TX</u>	<u>75093</u>	<u>(949) 538-</u> <u>7993</u>	andy.frye@toprailfence.com
	v Frye and new Frye	AMAF Services, LLC.	5960 W. Parker Road #278	Plano	<u>TX</u>	<u>TX</u>	<u>75093</u>	(949) 538- 7993	andy.frye@toprailfence.com

<u>HFB FenceCo Franchising, LLC</u> <u>Franchisees with Unopened Outlets as of December 31, 2023</u>

<u>N</u>	<u>ame</u>	<u>Entity</u>	Address	City	State	Terr. State	Zip Code	<u>Phone</u>	Email Address
	Frye and ew Frye	AMAF Services, LLC.	5960 W. Parker Road #278	<u>Plano</u>	<u>TX</u>	<u>TX</u>	<u>75093</u>	<u>(949) 538-</u> <u>7993</u>	andy.frye@toprailfence.com
	Frye and ew Frye	AMAF Services, LLC.	5960 W. Parker Road #278	Plano	TX	<u>TX</u>	<u>75093</u>	<u>(949) 538-</u> <u>7993</u>	andy.frye@toprailfence.com
	Frye and ew Frye	AMAF Services, LLC.	5960 W. Parker Road #278	Plano	TX	<u>TX</u>	<u>75093</u>	(949) 538- 7993	andy.frye@toprailfence.com
	Frye and ew Frye	AMAF Services, LLC.	5960 W. Parker Road #278	Plano	TX	<u>TX</u>	<u>75093</u>	(949) 538- 7993	andy.frye@toprailfence.com
John S	Schussler	<u>Crow Line</u> Enterprises, LLC.	1627 NE 77th Street	Seattle	WA	<u>WA</u>	<u>98115</u>	(206) 331- 1811	John.schussler@toprailfence.com
John S	Schussler	Crow Line Enterprises, LLC.	1627 NE 77th Street	<u>Seattle</u>	WA	<u>WA</u>	98115	(206) 331- 1811	John.schussler@toprailfence.com

HFB FenceCo Franchising, LLC

Former Franchisees

The name and last known address of every Franchisee who had a Top Rail Franchise transferred, terminated, cancelled, not renewed, or otherwise voluntarily or involuntarily ceased to do business under our Franchise Agreement during our last fiscal year, or who has not communicated with us within ten weeks of the Issuance Date of this Franchise Disclosure DocumentApril 19, 2024, are listed below. If you buy this Franchise, your contact information may be disclosed to other buyers when you leave the Franchise System.

None

EXHIBIT I TO

Exhibit L to FDD HFB FENCECO FRANCHISINGFenceCo Franchising, LLC FRANCHISE DISCLOSURE DOCUMENT

 $\frac{\text{DILIGENCE REVIEW OF FRANCHISEE}}{\text{ACKNOWLEDGEMENTS}} \\ \frac{\text{ACKNOWLEDGEMENTS}}{\text{ACKNOWLEDGEMENTS}} \\ \\ \frac{\text{ACKNOWLEDGEMENTS}}{\text{ACKNOWLEDGEMENTS}} \\ \frac{\text{$

DILIGENCE REVIEW OF FRANCHISEE AFFIRMATIONS AND ACKNOWLEDGEMENTS SALES PROCESS

This document is not applicable to and shall not be used as to any franchise offer and/or sale involving any California resident and/or franchisee as the Franchise Acknowledgement violates California Corporations Code sections 31512 and 31512.1.

As you know, HFB FenceCo Franchising, LLC ("we" or "us"), and you are preparing to enter into a Franchise Agreement and Multi-Unit Addendum (as applicable) for the operation of a franchise (a "Franchised Business"). The purposes of this Questionnaire are to determine whether any statements or promises were made to you that we have not authorized or that may be untrue, inaccurate or misleading, to be certain that you have been properly represented in this transaction, and to be certain that you understand the limitations on claims you may make by reason of the purchase and operation of your franchise. You cannot sign or date this Questionnaire the same day as the Receipt for the Franchise Disclosure Document, but you must sign and date it the same day you sign the Franchise Agreement and Multi-Unit Addendum (as applicable) and pay your franchise fee. Please review each of the following questions carefully and provide honest responses to each question. If you answer "No" to any of the questions below, please explain your answer on the back of this sheet.

Yes	No	1.	Have you received and personally reviewed the Franchise Agreement and Multi- Unit Addendum (as applicable), as well as each exhibit or schedule attached to the agreements you intend to enter into with us?
Yes	No	2.	Have you received and personally reviewed the Franchise Disclosure Document we provided?
Yes	No	3.	Did you sign a receipt for the Franchise Disclosure Document indicating the date you received it?
Yes	No	4.	Do you understand all the information contained in the Franchise Disclosure Document, Franchise Agreement and Multi-Unit Addendum (as applicable) you intend to enter into with us?
Yes	No	5.	Have you reviewed the Franchise Disclosure Document, Franchise Agreement and Multi-Unit Addendum (as applicable) with a lawyer, accountant or other professional advisor and discussed the benefits and risks of operating a Franchised Business with these professional advisor(s)?
Yes	No	6.	Do you understand the success or failure of your Franchised Business will depend in large part upon your skills, abilities and efforts and those of the persons you employ, as well as many factors beyond your control such as weather, competition, interest rates, the economy, inflation, labor and supply costs, lease terms and the marketplace?
Yes	No	7.	Do you understand that you must devote your personal full-time attention and best efforts to the management and operation of your Franchised Business and will not own, maintain, engage in, be employed by or have any interest in any other business other than the Franchised Business?
Yes	No		If you do not intend to devote your personal full-time attention and best efforts,

		did you receive permission from the franchisor to own, maintain, engage in, be employed by or have any interest in any other business?
Yes No	8.	Do you understand we have only granted you certain exclusive territorial rights under the Franchise Agreement and Multi-Unit Addendum (as applicable) and that we have reserved certain rights under the Franchise Agreement and/or Multi-Unit Addendum (as applicable)?
Yes No	9.	Do you understand we and our affiliates retain the exclusive unrestricted right to engage, directly or through others, in the providing of services under the Top Rail mark or other mark, at any location outside your Designated Territory, without regard to the proximity of these activities to the premises of your Franchised Business?
Yes No	10.	Do you understand all disputes or claims you may have arising out of or relating to the Franchise Agreement must be mediated, at our option, in Mecklenburg County, North Carolina?
Yes No	11.	Do you understand the Franchise Agreement provides you can only collect compensatory damages on any claim under or relating to the Franchise Agreement and you are not entitled to any punitive, consequential, or other special damages?
Yes No	12.	Do you understand the sole entity or person against whom you may bring a claim under the Franchise Agreement is HFB FenceCo Franchising, LLC?
Yes No	13.	Do you understand all persons whose names appear on the Franchise Agreement must successfully complete the appropriate initial training program(s) before we will allow the Franchised Business to open or consent to a transfer of that Franchised Business?
Yes No	14.	Do you understand that we require you to successfully complete certain initial training program(s) and if you do not successfully complete the applicable training program(s) to our satisfaction, we may terminate your Franchise Agreement?
Yes No	15.	Do you understand we do not have to sell you a franchise or additional franchises or consent to your purchase of existing franchises?
Yes No	16.	Do you understand that we will send written notices, as required by your Franchise Agreement, to either your Franchised Business or home address until you designate a different address by sending written notice to us?
Yes No	17.	Do you understand that we will not approve your purchase of a franchise, or we may immediately terminate your Franchise Agreement and/or Multi-Unit Addendum (as applicable), if we are prohibited from doing business with you under any anti-terrorism law enacted by the United States Government?
Yes No	18.	Is it true that no broker, employee or other person speaking on our behalf made any statement or promise regarding the costs involved in operating a Franchised Business that is not contained in the Franchise Disclosure Document or that is

	contrary to, or different Disclosure Document?	from, the information contained in the	Franchise
Yes No 19.	any statement or promise earnings, the likelihood of total amount of revenue contained in the Franchi	regarding the actual, average or projected f success, the amount of money you may ea a Franchised Business will generate, the Disclosure Document or that is contrapromation contained in the Franchise I	profits or rn, or the at is not ary to, or
Yes No 20.	any statement or promise your Franchise Agreeme concerning advertising, training, support service of	mployee or other person speaking on our bel or agreement, other than those matters addent and/or Multi-Unit Addendum (as a marketing, media support, marketing pe or assistance that is contrary to, or different the Franchise Disclosure Document?	lressed in oplicable) netration,
Yes No 21.	our behalf has solicited or payment in money, pro	mployee, or other person providing services r accepted any loan, gratuity, bribe, gift or perty or services from you in connectio chase with exception of those payments Disclosure Document?	any other n with a
Yes No 22.	Did you receive the Franc completed and signed this	hise Disclosure Document at least 14 days b Questionnaire?	efore you
Yes No 23.	Did you receive the Francand signed this Questionn	hise Agreement at least 7 days before you caire?	ompleted
WILL RELY ON THE THAT YOU HAVE TRUTHFULLY TO THE Do not sign this Question [Do not sign this Question]	M. BY SIGNING THIS CONSIDERED EACH OF ABOVE QUESTIONS. The property of the consideration of	S ARE IMPORTANT TO US AND TEQUESTIONNAIRE, YOU ARE REPRESEQUESTION CAREFULLY AND RESERVESTION. OF the franchise is to be located in the resident, or the franchise is to be located in the inigton resident, or the franchise is to be located in the inigton resident, or the franchise is to be located in the inigton resident, or the franchise is to be located in the inigton resident, or the franchise is to be located in the inigton resident, or the franchise is to be located in the inigton resident, or the franchise is to be located in the inigton resident.	ENTING ONDED California.] Maryland.]
Signature of Franchise A	pplicant	Signature of Franchise Applicant	
Signature of Franchise A Name (please print)	pplicant	Signature of Franchise Applicant Name (please print)	

EXHIBIT J TO

Exhibit M to FDD HFB FENCECO FRANCHISINGFenceCo Franchising, LLC FRANCHISE DISCLOSURE DOCUMENT

STATE EFFECTIVE DATES

State Effective Dates

The following states have franchise laws that require that the Franchise Disclosure Document be registered or filed with the state, or be exempt from registration: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin.

This document is effective and may be used in the following states, where the document is filed, registered or exempt from registration, as of the Effective Date stated below:

State	Effective Date		
California	Pending December 11, 2023		
Hawaii	Not Registered Pending		
Illinois	May 2, 2023 Pending		
Indiana	May 10, 2023 Pending		
Maryland	October 24, 2023 Pending		
Michigan	January 8, 2024		
Minnesota	August 28, 2023 Pending		
New York	September 11, 2023 Pending		
North Dakota	September 12, 2023 Pending		
Rhode Island	May 15, 2023 Pending		
South Dakota	November 17, 2023 Pending		
Virginia	July 11, 2023 Pending		
Washington	October 12, 2023 Pending		
Wisconsin	May 10, 2023 Pending		

Other states may require registration, filing, or exemption of a franchise under other laws, such as those that regulate the offer and sale of business opportunities or seller-assisted marketing plans.

EXHIBIT K

 \mathbf{TO}

Exhibit N
to FDD

HFB FENCECO FRANCHISING FenceCo Franchising, LLC
FRANCHISE DISCLOSURE DOCUMENT

ITEM 23 RECEIPTS

RECEIPTS (OUR COPY)

This disclosure document summarizes certain provisions of the Franchise Agreement and other information in plain language English. Read this disclosure document and all agreements carefully.

If HFB FenceCo Franchising, LLC offers you a franchise, it must provide this disclosure document to you:

a) 14 calendar-days before you sign a binding agreement with, or make a payment to, the franchisorus or an affiliate in connection with the proposed franchise sale-, or

(b) Under New York and Rhode Island require that we give you this disclosure document law, if applicable, at the earlier of the(i) your first personal meeting or to discuss the franchise, or (ii) 10 business days before the execution of the franchise or other agreements you sign a binding agreement with, or make payment of any consideration that relates to us or an affiliate in connection with the proposed franchise relationship.sale, or

(c) Michigan requires that we give you this disclosure document at least 10 business days before the execution of any binding franchise agreement or other agreement, or the payment of any consideration, whichever occurs first,

(d) Under Iowa law, if applicable, at the earlier of the first personal meeting or 14 calendar days before the execution of the franchise agreement or other agreement or the payment of any consideration that relates to the franchise relationship.

If HFB FenceCo Franchising, LLC does not deliver this disclosure document on time or if it contains a false or misleading statement or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, DC 20580, and the state administrator identified in Exhibit A of this Franchise Disclosure Document. A list of franchisor's agents registered to receive service of process is listed as Exhibit A of this Franchise Disclosure Document.

The name, principal business address, and telephone number of each franchise seller offering the franchise is:

Zack Dudan, Homefront Brands, 107 Parr Drive, Huntersville, NC 28078; (980) 441-1121

Jason Jobe, Homefront Brands, 107 Parr Drive, Huntersville, NC 28078; (980) 441-1121

Brian Dungan, Homefront Brands, 107 Parr Drive, Huntersville, NC 28078; (980) 441-1121

Michael D. Aiello, Homefront Brands, 107 Parr Drive, Huntersville, NC 28078; (980) 441-1121

Todd Bingham, Homefront Brands, 107 Parr Drive, Huntersville, NC 28078; (980) 441-1121

Issuance Date: April 19, 2024

I received a disclosure document dated April 19, 2024, that included the following Exhibits:

Exhibit B Franchise Agreement

Exhibit C Statement of Ownership

Exhibit D Confidentiality and Restrictive Covenant Exhibit E Electronic Funds Withdrawal Authorization

Exhibit F Multi-Unit Addendum

Exhibit G Sample Assignment and Assumption Agreement

Exhibit H State Specific Addenda Exhibit I Financial Statements

Exhibit J Operations Manual Table of Contents

List of Franchisees and Franchisees That Have Left the System in the Past Fiscal Year Exhibit K

Diligence Review of Franchisee Sales Process Exhibit L

Exhibit M State Effective Dates

Exhibit N Receipts

If an Individual:	If a Business Entity:	
Signature:	Signature:	
Printed Name:	Printed Name:	
Telephone No.:	Entity:	
Date:	Title:	
	Date:	
	igning, dating, and mailing it to Zack Dudan at 107 Parr Drive, ack Dudan at development@toprailfence.com. You may keep the	Field Code Changed

RECEIPTS (YOUR COPY)

This disclosure document summarizes certain provisions of the Franchise Agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If HFB FenceCo Franchising, LLC offers you a franchise, it must provide this disclosure document to you:

(a) 14 calendar days before you sign a binding agreement with, or make a payment to, us or an affiliate in connection with the proposed franchise sale, or

(b) Under New York law, if applicable, at the earlier of (i) your first personal meeting to discuss the franchise, or (ii) 10 business days before you sign a binding agreement with, or make payment to us or an affiliate in connection with the proposed franchise sale, or

(c) Michigan requires that we give you this disclosure document at least 10 business days before the execution of any binding franchise agreement or other agreement or the payment of any consideration, whichever occurs first, or

(d) Under Iowa law, if applicable, at the earlier of the first personal meeting or 14 calendar days before the execution of the franchise agreement or other agreement or the payment of any consideration that relates to the franchise relationship.

If HFB FenceCo Franchising, LLC does not deliver this disclosure document on time or if it contains a false or misleading statement or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, DC 20580, and the state administrator identified in Exhibit A of this Franchise Disclosure Document. A list of franchisor's agents registered to receive service of process is listed as Exhibit A of this Franchise Disclosure Document.

The name, principal business address, and telephone number of each franchise seller offering the franchise is:

Zack Dudan, Homefront Brands, 107 Parr Drive, Huntersville, NC 28078; (980) 441-1121

Jason Jobe, Homefront Brands, 107 Parr Drive, Huntersville, NC 28078; (980) 441-1121

Brian Dungan, Homefront Brands, 107 Parr Drive, Huntersville, NC 28078; (980) 441-1121

Michael D. Aiello, Homefront Brands, 107 Parr Drive, Huntersville, NC 28078; (980) 441-1121

Todd Bingham, Homefront Brands, 107 Parr Drive, Huntersville, NC 28078; (980) 441-1121

Issuance Date: April 29, 2023 19, 2024

I-have received a disclosure document issueddated April 29, 2023, which 19, 2024, that included the following Exhibits:

Exhibit A List of State Administrators and Agents for Service of Process

Exhibit B Franchise Agreement

Exhibit C Statement of Ownership

Exhibit D Confidentiality and Restrictive Covenant
Exhibit E Electronic Funds Withdrawal Authorization

Exhibit CF Multi-Unit Addendum

Exhibit G Sample Assignment and Assumption Agreement

Exhibit <u>DH</u> State Specific Addenda Exhibit <u>EI</u> Financial Statements

Exhibit F Sample Assignment and Assumption Agreement

Exhibit GJ Operations Manual Table of Contents

Exhibit <u>HK</u> List of Franchisees and Franchisees That Have Left the System in the Past Fiscal Year Exhibit <u>HL</u> <u>Diligence Review of Franchisee Affirmation and Acknowledgments Sales Process</u>

Exhibit JM State Effective Dates

Exhibit KN Receipts

If an Individual:	If a Business Entity:	
Signature:	Signature:	
Printed Name:	Printed Name:	
Telephone No.:	Entity:	
Date:	Title:	
	Date:	
	er, personal delivery or mailing it to Zack Dudan at 107 Parr Drive,	
Huntersville, NC 28078; or emailing (as an attachment) to Za	ck Dudan at development@toprailfence.com. You may keep the	Field Code Changed

RECEIPTS (VOUR COPY)

This disclosure document summarizes certain provisions of the Franchise Agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If HFB FenceCo Franchising, LLC offers you a franchise, it must provide this disclosure document to you 14 calendar days before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sale.

New York and Rhode Island require that we give you this disclosure document at the earlier of the first personal meeting or 10 business days before the execution of the franchise or other agreements or payment of any consideration that relates to the franchise relationship. Michigan requires that we give you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement, or the payment of any consideration, whichever occurs first.

If HFB FenceCo Franchising, LLC does not deliver this disclosure document on time or if it contains a false or misleading statement or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, DC 20580, and the state administrator identified in Exhibit A of this Franchise Disclosure Document. A list of franchisor's agents registered to receive service of process is listed as Exhibit A of this Franchise Disclosure Document.

The name, principal business address, and telephone number of each franchise seller offering the franchise is: Zack Dudan, Homefront Brands, 107 Parr Drive, Huntersville, NC 28078; (980) 441-1121 Jason Jobe, Homefront Brands, 107 Parr Drive, Huntersville, NC 28078; (980) 441 1121 Brian Dungan, Homefront Brands, 107 Parr Drive, Huntersville, NC 28078; (980) 441-1121

Issuance Date: April 29, 2023

disclosure document issued April 29, 2023, which included the following Exhibits:			
List of State Administrators and Agents for Service of Process			
Franchise Agreement			
Multi Unit Addendum			
State Specific Addenda			
Financial Statements			
Sample Assignment and Assumption Agreement			
Operations Manual Table of Contents			
List of Franchisees and Franchisees That Have Left the System in the Past Fiscal Year			
Franchisee Affirmation and Acknowledgments			
State Effective Dates			
Receipts			
If a Business Entity:			
Signature:			
Printed Name:			
Entity:			
Title:			

Date: ___